File No. 150689

Committee Item No. _____ Board Item No. _____

COMMITTEE/BOARD OF SUPERVISORS

AGENDA PACKET CONTENTS LIST

Committee: Budget & Finance Committee

Date July 8, 2015

Board of Supervisors Meeting

Date _____

Cmte Board

	Motion
X	Resolution
	Ordinance
	Legislative Digest
\square	Budget and Legislative Analyst Report
	Youth Commission Report
\boxtimes	Introduction Form
	Department/Agency Cover Letter and/or Report
	MOU
	Grant Information Form
	Grant Budget
	Subcontract Budget
\mathbf{X}	Contract/Agreement
\mathbf{X}	Form 126 – Ethics Commission
	Award Letter
	Application
	Public Correspondence

OTHER (Use back side if additional space is needed)

· 🔲	

Completed by:_	Linda Wong	Date_	July 2, 2015
Completed by:		Date_	

FILE NO. 150689

RESOLUTION NO.

[Agreement Extension - TEGSCO, LLC, dba San Francisco Auto Return - Towing and Storage of Abandoned and Illegally-Parked Vehicles]

Resolution approving the Second Amendment to the Amended and Restated Agreement with TEGSCO, LLC, dba San Francisco AutoReturn, for towing, storage, and disposal of abandoned and illegally-parked vehicles to extend the Agreement and the License for eight months, from August 1, 2015, through March 31, 2016.

WHEREAS, On July 31, 2005, the San Francisco Municipal Transportation Agency (SFMTA) executed an agreement with San Francisco AutoReturn to manage a network of ten local and Bay Area towing firms that provide towing support for the San Francisco Police Department (SFPD) and the SFMTA ("the Towing Agreement") and Appendix C to that agreement, a revocable license for a long-term storage facility for towed vehicles at Pier 70 ("the License"); and

WHEREAS, In addition, San Francisco AutoReturn secured a separate license in 2005 with Caltrans for property on 7th Street (across from the Hall of Justice) to be used as the primary storage lot for towed vehicles and the customer service center; and

WHEREAS, In 2010 the SFMTA Board and this Board approved an extension of the Towing Agreement and the License for five years until July 31, 2015; and

WHEREAS, In May 2014 both Boards approved transferring the long-term storage facility from Pier 70 to property at 2650 Bayshore Blvd. and replacing Appendix C to the Towing Agreement (the License) with Appendix H, the license for the Bayshore site ("the Bayshore License"); and

Supervisor Wiener BOARD OF SUPERVISORS WHEREAS, The SFMTA requires an additional eight months after expiration of the Towing Agreement and the Bayshore License to release a new RFP for towing services, negotiate a new towing agreement; and

WHEREAS, In order to reimburse San Francisco AutoReturn for the rent increase of \$53,621 per month that it must pay to Caltrans during the last eight months of the existing lease for the 7th Street site, the SFMTA will allow San Francisco AutoReturn to offset its rent of \$147,985.76 per month at the Bayshore facility by \$53,000 per month, for a total of \$424,000 over the eight-month period; and

WHEREAS, On June 16, 2015, the SFMTA Board of Directors approved Resolution No. 15-091, authorizing the Second Amendment to the Towing Agreement extending the Agreement for eight months, and authorizing the First Amendment to Appendix H of the Towing Agreement, the Bayshore License; and

WHEREAS, In November 2014, the Board of Supervisors passed Resolution No. 432-14 urging the SFMTA to evaluate the feasibility of waiving or otherwise mitigating towing and storage fees for individual members of the public whose vehicles were towed and identified by the SFPD as stolen; and

WHEREAS, In addition to approving extension of the Towing Agreement and the License, the SFMTA Board on June 16, 2015, approved a waiver policy for owners of stolen vehicles; and

WHEREAS, This policy will go into effect on December 1, 2015; and

WHEREAS, The Towing Agreement and Bayshore License amendment are on file with the Clerk of the Board of Supervisors in File No. <u>150689</u>; and

WHEREAS, The Planning Department has determined that the actions contemplated in this Resolution comply with the California Environmental Quality Act (California Public

Supervisor Wiener BOARD OF SUPERVISORS Resources Code, Sections 21000, et seq.); said determination is on file with the Clerk of the Board of Supervisors in File No. <u>150689</u> and is incorporated herein by reference; and

WHEREAS, The Board affirms this determination; now, therefore, be it

RESOLVED, That the San Francisco Board of Supervisors authorizes execution of the Second Amendment to the Amended and Restated Agreement and Property Use License for towing, storage, and disposal of abandoned and illegally-parked vehicles with TEGSCO, LLC, dba San Francisco AutoReturn, and the First Amendment to Appendix H to that Agreement, the Amended and Restated Revocable License to Enter and Use Property, to extend the Agreement and the License for eight months, from August 1, 2015, to March 31, 2016; and, be it

FURTHER RESOLVED, That within 30 days of the document being fully executed by all parties, the final document shall be provided to the Clerk of the Board for inclusion in the official file.

JULY 8, 2015

lte	m 9 Department:			
File	e 15-0689 San Francisco Municipal Transportation Agency (MTA)			
EX	ECUTIVE SUMMARY			
	Legislative Objectives			
•	The proposed resolution would approve the second amendment to the existing towing service agreement and property use license between the San Francisco Municipal Transportation Agency (MTA) and San Francisco AutoReturn to (i) extend the service agreement by eight months from August 1, 2015 through March 31, 2016; and (ii) authorize a rent credit from MTA to AutoReturn for \$53,000 per month, totaling \$424,000 over the proposed eight-month extension period.			
	Key Points			
•	On July 31, 2005, MTA entered into a towing service agreement and property use license with Tegsco, LLC d.b.a. San Francisco AutoReturn (AutoReturn) following a competitive Request for Proposals (RFP) process.			
•	Under the existing agreement, the revenues collected by AutoReturn are used to pay for operating expenses. AutoReturn pays MTA one percent of all gross revenues and a per vehicle referral fee.			
•	AutoReturn uses two privately-owned facilities to perform the vehicle storage services under this agreement. The monthly rental rate for the 7th and Harrison Street short-term storage facility was increased by \$53,621 from \$36,379 to \$90,000 as of March 1, 2015. This increase was an adjustment to the market rental rate following the expiration of the long-term lease held by AutoReturn at the end of February 2015.			
•	A new fee waiver policy, for towed vehicles that are determined to have been stolen, will go into effect on December 1, 2015. The new fee waiver policy will require AutoReturn to waive or reimburse fees paid by San Francisco residents as well as non-residents whose vehicles were stolen and then subsequently towed.			
	Fiscal Impact			
•	The total estimated net revenues payable by AutoReturn to the MTA under the proposed second amendment are \$181,794 for the proposed eight-month extension period of the existing agreement from August 1, 2015 through March 31, 2016.			
•	According to Mr. Steven Lee, MTA Senior Manager of Financial Services and Contracts, the proposed \$424,000 rent credit over the proposed eight-month extension period is reasonable as it provides time for MTA to award a new towing agreement and is needed to offset AutoReturn's rental costs during the eight-month extension period.			
	Recommendation			
	Approve the proposed resolution.			

MANDATE STATEMENT

Mandate Statement

Charter Section 9.118(a) requires that any contract entered into by a Department, which has anticipated revenues of one million dollars or more, or the modification, amendment or termination of any contract which when entered into had anticipated revenue of one million dollars or more shall be subject to approval of the Board of Supervisors by resolution.

BACKGROUND

On July 31, 2005, the San Francisco Municipal Transportation Agency (MTA) entered into a towing service agreement and property use license with Tegsco, LLC d.b.a. San Francisco AutoReturn (AutoReturn) following a competitive Request for Proposals (RFP) process. Under this existing service agreement, AutoReturn provides services consisting of towing, storage, and disposal of abandoned and illegally parked vehicles on behalf of the San Francisco Police Department (SFPD) and the MTA. Under the existing agreement, the revenues collected by AutoReturn are used to pay for operating expenses. Further, under the existing agreement, AutoReturn pays MTA one percent of all gross revenues and a per vehicle referral fee.¹

The original agreement, which was previously approved by the Board of Supervisors, was for five years from August 1, 2005 through July 31, 2010. Although the agreement provided for an option to extend, at the sole discretion of the MTA, the agreement did not specify the term of the extension.

First Amendment and Revision to the Original Service Agreement

In June 2007, the Board of Supervisors approved the first amendment to the agreement to increase the towing and storage fee rate, payable by the owners of towed vehicles to AutoReturn, due to a significant drop in towing volume. The increase to the towing and storage fee was to offset decreases in AutoReturn's gross revenue from declining tow volumes and fixed contract costs.

In July 2010, the Board of Supervisors approved a five-year extension of the existing service agreement and property use license between the MTA and AutoReturn from August 1, 2010 through July 31, 2015.

DETAILS OF PROPOSED LEGISLATION

The proposed resolution would approve the second amendment to the existing towing service agreement and property use license between MTA and AutoReturn. Under the proposed second amendment, MTA would:

BUDGET AND LEGISLATIVE ANALYST

¹ The per vehicle referral fee is a fee paid by AutoReturn to the MTA for each tow referred by the San Francisco Police Department and MTA Parking Enforcement Control Officers.

- Extend the agreement for an eight-month period from August 1, 2015 through March 31, 2016, while the MTA completes a competitive Request For Proposals (RFP) process for awarding a new towing service agreement; and
- Authorize a rent credit to AutoReturn of \$53,000 per month, totaling \$424,000 over the eight-month extension period, to offset the increased monthly rent being paid by AutoReturn for the rental of a short-term storage facility located at 7th and Harrison Streets.

Increase in Rental Rate for Short-Term Storage Facility located at 7th and Harrison Streets

AutoReturn uses two privately-owned facilities to perform the vehicle storage services under this agreement, including (i) the long-term storage facility located at 2650 Bayshore Boulevard,² and (ii) the short-term storage facility located at 7th & Harrison Streets. AutoReturn's monthly rent for the 7th and Harrison Street location was increased by \$53,621 from \$36,379 to \$90,000 as of March 1, 2015. This increase was an adjustment to the market rental rate following the expiration of the long-term lease held by AutoReturn at the end of February 2015.

Fee Waiver for Stolen Vehicles

A new fee waiver policy, for towed vehicles that are determined to have been stolen, will go into effect on December 1, 2015. The new fee waiver policy will require AutoReturn to waive or reimburse fees paid by San Francisco residents as well as non-residents whose vehicles were stolen and then subsequently towed. This policy was adopted by the MTA in response to the November 2014 Board of Supervisors request made to MTA to evaluate the feasibility of waiving or mitigating towing and storage fees for owners of vehicles that were towed and then identified by the San Francisco Police Department as having been stolen.

FISCAL IMPACT

Under the existing agreement between the MTA and AutoReturn, AutoReturn is required to pay the MTA annual payments consisting of one percent of all gross revenues realized by AutoReturn and a per vehicle referral fee of \$22.50 per vehicle towed, if the towed vehicles were referred by the San Francisco Police Department or MTA Parking Enforcement Control Officers. Table 1 below shows the payments made by AutoReturn to MTA, totaling \$13,709,492 over the ten-year term from August 1, 2005 through July 31, 2015.

BUDGET AND LEGISLATIVE ANALYST

² In May 2014, the Board of Supervisors approved the transfer of the long-term storage facility from Pier 70 to 2650 Bayshore Boulevard.

Fiscal Year	Referral Fees	1% of Gross	Total Revenues
FISCAL TEAL		Revenue	Paid to MTA
FY 2005-06	\$1,261,905	\$153,739	\$1,415,644
FY 2006-07	1,332,164	166,903	1,499,067
FY 2007-08	1,373,579	208,090	1,581,669
FY 2008-09	1,409,100	216,911	1,626,011
FY 2009-10	1,227,336	187,087	1,414,423
FY 2010-11	1,146,828	186,268	1,333,096
FY 2011-12	1,123,498	186,491	1,309,989
FY 2012-13	. 979,811	163,580	1,143,391
FY 2013-14	1,043,354	160,935	1,204,289
FY 2014-15 (est.)	1,019,655	162,258	1,181,913
Total	\$11,917,230	\$1,792,262	\$13,709,492

Table 1. Annual Payments Paid by AutoReturn to MTA

Source: Municipal Transportation Agency (MTA) staff.

The total estimated net revenues payable by AutoReturn to the MTA under the proposed second amendment are \$181,794 for the proposed eight-month extension period of the existing agreement from August 1, 2015 through March 31, 2016, as shown in Table 2 below.

Table 2. Estimated Net Revenues to MTA during ProposedEight-Month Extension of Existing Agreement

Estimated Revenues Payable by AutoReturn to MTA during Eight-Month Extension	\$831,794
Rent Credit Paid by MTA to AutoReturn	(424,000)
Reimbursement by MTA to AutoReturn for Waiver of Fees for Towed Stolen Vehicles	(226,000)
Net Estimated Revenues Paid by AutoReturn to MTA	\$181,794

Source: Municipal Transportation Agency (MTA) staff.

*Projected revenues paid to MTA by AutoReturn calculated as the average annual payment received by MTA over four fiscal years from FY 2010-11 through FY 2013-14.

According to Mr. Steven Lee, MTA Senior Manager of Financial Services and Contracts, at the request of AutoReturn, the proposed \$424,000 rent credit is needed to offset AutoReturn's increased rental costs during the proposed eight-month extension period due to the decline in tow volumes. According to Mr. Lee, MTA considers this rent credit to be reasonable over the eight-month extension period, providing time for the MTA to award a new towing agreement. MTA issued the new RFP on June 26, 2015 and anticipates that a new towing service agreement, which would be subject to Board of Supervisors approval, should be awarded by March of 2016.

RECOMMENDATION

Approve the proposed resolution.



SEMTA Municipal Transportation Agency

Edwin M. Lee, Mayor

Tom Nolan, Chairman Cheryl Brinkman, Vice-Chairman Joél Ramos, Director Gwyneth Borden, Director

Malcolm Heinicke, Director Cristina Rubke, Director

Edward D. Reiskin, Director of Transportation

June 23, 2015

The Honorable Members of the Board of Supervisors **City and County of San Francisco** 1 Dr. Carlton B. Goodlett Place, Room 244 San Francisco, CA 94102

Subject: Extension of Towing Agreement with San Francisco AutoReturn

Honorable Members of the Board of Supervisors:

The SFMTA requests your approval to enter into an eight-month extension of the current towing agreement from August 1, 2015 to March 31, 2016. The purpose of this briefing is to provide information to support the extension and to describe two changes in the extension. No additional funding is required for this extension but the two changes have a financial impact to the agency totaling \$650,000 for the extension term.

Background and Extension Summary

The current towing agreement with San Francisco AutoReturn (AutoReturn), which has been in place since August 2005, will expire July 31, 2015. The SFMTA Board of Directors approved the release of a Request for Proposals (RFP) on June 16, 2015 and it is expected that the proposal evaluation, contract negotiations and (if necessary) transition to a new vendor will take approximately eight months beyond the current expiration date. The release of the RFP took longer than expected because SFMTA conducted a best practices review of towing contracts to determine the best way to approach the scope for these services going forward. This review resulted in a major change to the service model in the new contract.

In addition to allowing for the competitive bidding process during the eight-month extension, the proposed extension will accomplish the following:

1. Establish a stolen vehicle policy for waiver or reimbursement of towing and storage fees for all individuals that will go into effect on December 1, 2015. A summary of the current towing and storage fees, and the current and new waiver/reimbursement policies are below:

Type of Fee (as of July 1, 2015)	Current Waiver Policy (Established by the BOS in 2005)	New Waiver Policy
SFMTA Administrative Tow Fee (\$266.00)	Waived only for SF Residents whose stolen vehicles were towed in San Francisco.	 Waived for SF residents whose stolen vehicles were recovered in San Francisco; 50% waiver for non-SF residents whose stolen vehicles are recovered in San Francisco.
SFMTA Administrative Storage Fee (\$3.00)/day	Waived only for SF Residents whose stolen vehicles were towed in San Francisco.	Waived for all individuals whose stolen vehicles were recovered in San Francisco

1 South Van Ness Avenue 7th Floor, San Francisco, CA 94103

www.sfmta.com

Memo to the Board of Supervisors Agreement with AutoReturn June 23, 2015 Page 2 of 3

Type of Fee (as of July 1, 2015)	Current Waiver Policy (Established by the BOS in 2005)	New Waiver Policy
Contractor's Towing Fee (\$225.75)	All are required to pay.	Waived for all individuals whose stolen vehicles were recovered in San Francisco
Contractor's Storage Fees (\$68.25)/day	4-hour grace period for all before accrual begins.	• 48-hour grace period before accrual begins for SF residents;
		• 24-hour grace period before accrual begins for non-SF residents.

stolen vehicles for the extension term will result in a loss of revenue of approximately \$226,000 from December 1 through March 31, 2016.

The SFMTA is in the process of updating the Transportation Code, Division II to legislate waiver and reimbursement conditions for towing and storage fees, including the new stolen vehicle waiver policy. These conditions currently reside in the Administrative Code, Section 10C, but must be transferred to reflect the SFMTA Board's oversight responsibility for all fare, fines and fees. The code modifications and the required public hearing for the fee waivers will be on August 18, 2015.

Following the approval of the extension agreement, outdated sections of the Administrative Code need to be amended by the Board of Supervisors to reflect jurisdiction with regard to waivers, as the Board of Supervisors still has authority over the SFPD's Traffic Offender Fee. This action does not preclude the waiver policy from being implemented.

- 2. Extends the lease the SFMTA has with AutoReturn for the long-term storage and auction facility to run concurrently with extension term.
- 3. Allows the towing contractor to offset \$53,000 per month of the 7th Street rent increase against the monthly rent for the long-term storage lot during the extension period to ensure continuing customer service support for short-term storage at the 7th Street location. In March 2015, Caltrans regained leasing authority for the 7th Street location. They then increased the rent from \$36,379 per month to \$90,000 per month, citing current market rates. AutoReturn agreed to pay the increased rent through July 2015, at an additional cost of \$268,105, but has determined they do not have the resources to continue to assume the rent increase cost. This action will result in increased costs to the Agency of \$424,000 in rent payments over the eightmonth extension term.
- 4. Provides the SFMTA with sufficient time to negotiate with Caltrans for transfer of the 450-7th Street customer service center and short-term vehicle storage parking lot from AutoReturn to the SFMTA. The Agency seeks to retain the 7th Street location because of its proximity to the SFPD Traffic Division at 850 Bryant Street, where the majority of police hold waivers are issued, and because it is readily accessible to public transportation.

The new contract will attempt to address the two biggest challenges under the current agreement related to paying the contractor based on tow volume and the amount of cost recovery fees charged to the public. While fee waivers for stolen vehicles are part of the existing contract extension, additional ways to reduce operating costs will be evaluated for the new contract to support cost recovery fee reductions (e.g. replacing the current tow volume dependent methodology to a

Memo to the Board of Supervisors Agreement with AutoReturn June 23, 2015 Page 3 of 3

management fee methodology for contractor payments, implementing technology that can reduce program costs).

Recommendation

The SFMTA recommends that the Board of Supervisors authorize the Agency to extend the agreement for Towing and Storage of Abandoned and Illegally-Parked vehicles with TEGGSCO, dba San Francisco AutoReturn, for eight months, to March 31, 2016 and to support the waiver of certain fees for individuals with stolen vehicles and to allow the towing contractor to offset the rent increase related to the 7th Street short-term storage location

Thank you for your consideration of the proposed extension. Should you have any questions or require more information, please do not hesitate to contact me any time.

Sincerely,

Edward D. Reiskin Director of Transportation

Towing Agreement

Current Towing Agreement Summary:

- In place since 2005; expires July 31, 2015.
- No provision for waiver of towing and storage fees for individuals with stolen vehicles.
- Tow RFP to be released after June 16, 2015 SFMTA Board approval.
- Current contractor leases 7th Street customer service center and short term storage lot from Caltrans on a month-to-month basis.
- Current contractor leases Bayshore long-term storage lot from SFMTA.

Extension Agreement Summary (approved by the SFMTA Board on June 16, 2015, pending Board of Supervisors approval):

- Extends agreement by eight months, to March 31, 2016.
- Will establish a stolen vehicle waiver fee policy to be approved by the SFMTA Board on August 18, 2015, to go into effect on December 1, 2015, as follows:

<u>Type of Fee</u>	<u>SF Residents</u>	Non-SF Residents
SFMTA Administrative Tow Fee (\$266.00)	Waived	50% waived
SFMTA Administrative Storage Fee (\$3.00)	Waived	Waived
Contractor's Tow Fee (S225.75)	Waived	Waived
Contractor's Storage Fee (\$68.25)	48-hour grace period before storage fees begin to accrue	24-hour grace period before storage fees begin to accrue

- Cost to SFMTA: ~\$226,000 for the extension term.
- Allows SFMTA time to negotiate transfer of 7th Street Customer Service Center and short-term storage lot with Caltrans from current contractor to SFMTA.
- Allows time for proposal evaluation and new contract award, and transition between contractors if needed.
- Allows current contractor, during the extension period, to offset \$53,000 of \$147,985.76 rent for Bayshore to pay for Caltrans rent increase for the short term storage site from \$36,379 to \$90,000. Estimated cost to the SFMTA for the 8 months extension is \$424K.

New Agreement

- Completed best practices review to serve as guidance for new contract.
- Will continue waivers of fees for individuals with stolen vehicles.
- Will move from a tow-volume dependent business model where the Contractor is paid directly by the public, to a flat management fee and variable per tow fee hybrid, paid to the Contractor by the SFMTA.
- Will have one towing and one storage fee, set by the SFMTA, instead of two types of each fee; and will reflect the agency's commitment to fee reduction. Have both tow storage lots directly under SFMTA control.
- Will update technology to maximize customer service support during the tow and vehicle recovery process.
- Will attempt to reduce fees towing fees to the public to the extent possible.
- Will be before the SF MTA Board of Directors and Board of Supervisors for approval in the 1st quarter of 2016.



City Hall 1 Dr. Carlton B. Goodlett Place, Room 244 San Francisco 94102-4689 Tel. No. 554-5184 Fax No. 554-5163 TDD/TTY No. 554-5227

June 29, 2015

File No. 150689

Sarah Jones Environmental Review Officer Planning Department 1650 Mission Street, 4th Floor San Francisco, CA 94103

BOARD of SUPERVISORS

Dear Ms. Jones:

On June 23, 2015, Supervisor Scott Wiener introduced the following legislation:

File No. 150689

Resolution approving the Second Amendment to the Amended and Restated Agreement with TEGSCO, LLC, dba San Francisco AutoReturn, for towing, storage, and disposal of abandoned and illegally-parked vehicles to extend the Agreement and the License for eight months, from August 1, 2015, through March 31, 2016.

This legislation is being transmitted to you for environmental review.

Angela Calvillo, Clerk of the Board By: Linda Wong, Assistant Clerk

Attachment

c: Joy Navarrete, Environmental Planning Jeanie Poling, Environmental Planning

Not defined as a project under CEQA Sections 15378 and 15060(c)(2) because it does not result in a physical change in the environment.

- Jeanie Polity 6/29/15

City and County of San Francisco Municipal Transportation Agency One South Van Ness Ave. 7th floor San Francisco, California 94103

Second Amendment to the Restated and Amended Agreement for Towing and Storage of Abandoned and Illegally-Parked Vehicles

THIS AMENDMENT (this "Amendment") is made as of August 1, 2015, in San Francisco, California, by and between TEGSCO, LLC. d.b.a. San Francisco AutoReturn ("Contractor"), and the City and County of San Francisco, a municipal corporation ("City"), acting by and through its Municipal Transportation Agency ("SFMTA").

RECITALS

A. City and Contractor have entered into the Agreement (as defined below).

B. City and Contractor desire to modify the Agreement on the terms and conditions set forth herein to extend the Agreement by eight months and waive the Contractor's towing fee for individuals with stolen vehicles and allow a grace period of up to 72 hours for the Contractor's storage fees for stolen vehicles.

NOW, THEREFORE, Contractor and the City agree as follows:

1. **Definitions.** The following definitions shall apply to this Amendment:

1a. Agreement. The term "Agreement" shall mean the Agreement dated July 31, 2015 between Contractor and City, as amended by the:

First Amendment, dated May 1, 2014

1b. Contract Monitoring Division. Contract Monitoring Division. Effective July 28, 2012, with the exception of Sections 14B.9(D) and 14B.17(F), all of the duties and functions of the Human Rights Commission under Chapter 14B of the Administrative Code (LBE Ordinance) were transferred to the City Administrator, Contract Monitoring Division ("CMD"). Wherever "Human Rights Commission" or "HRC" appears in the Agreement in reference to Chapter 14B of the Administrative Code or its implementing Rules and Regulations, it shall be construed to mean "Contract Monitoring Division" or "CMD" respectively.

1c. Other Terms. Terms used and not defined in this Amendment shall have the meanings assigned to such terms in the Agreement.

2. Modifications to the Agreement. The Agreement is modified as follows:

2a. Section 2 (Term of the Agreement) is replaced in its entirety to read as follows:

SFMTA P-550 (9-14)

1 of 3

The Term of this Agreement shall be from July 31, 2010 through March 31, 2016.

2b. Subsection (a) (City Waivers) of Section 12.2 of Appendix A – Scope of Work (Credits Due to Contractor) is replaced in its entirety to read as follows:

(a) City Waivers

In the event that (i) SFMTA, DPH or the SFPD determines pursuant to a post-storage hearing as required by Vehicle Code § 22852 that the towing, storage, transfer, lien and/or other fees shall be waived for a vehicle, or (ii) SFMTA, DPH or the SFPD waives the fees for the towing, storage, transfer and/or lien of a vehicle, or (iii) SFMTA or (iv) the SFPD waives the fees for the towing and storage of a vehicle for one of the reasons enumerated in §§ 10C.1 or 10C.8 of the San Francisco Administrative Code, or for towed vehicles identified by the SFPD as stolen then no such fees shall be charged by Contractor to the owner or operator of such vehicle. In the event that the owner or operator of a vehicle has paid Contractor for towing, storage, transfer, lien and/or other fees and the City subsequently waives the tow, storage, transfer, lien and/or other fees for that owner or operator, then Contractor shall directly reimburse the owner or operator in full the amounts previously paid to Contractor for such vehicle. When the City waives towing, storage, transfer and/or lien fees as provided for in this section, the City shall pay Contractor only the towing, storage, transfer and/or lien fees that would have been owed by the vehicle owner or operator, and such fees shall not include any SFMTA Administrative Fees as Contractor might otherwise charge. When the City waives storage fees as provided in this Section, the City shall pay Contractor storage fees for each such vehicle as set forth above for the first three (3) days of storage. There shall be no storage charge for the fourth (4th) through the ninth (9th) days of storage. For the tenth (10th) storage day and all days thereafter, City shall pay Contractor ten dollars (\$10.00) per day for the storage of such vehicles. Adjustments and credits and payments due to Contractor as a result of City waivers shall be calculated and submitted to the City through the City's claims process and are to be paid within one (1) month from the date of submission by the Contractor.

There shall be no late payment charges or interest assessed against City for late payment.

When SFMTA or the SFPD orders Contractor to release a vehicle pursuant to Vehicle Code § 22654(e) relating to authorization for moving a vehicle otherwise lawfully parked, City shall pay the cost of the tow and storage charges for a period not to exceed seventy-two (72) hours."

3. Effective Date. Each of the modifications set forth in Section 2 shall be effective on and after August 1, 2015.

4. Legal Effect. Except as expressly modified by this Amendment, all of the terms and conditions of the Agreement shall remain unchanged and in full force and effect.

IN WITNESS WHEREOF, Contractor and City have executed this Amendment as of the date first referenced above.

CITY

San Francisco Municipal Transportation Agency

Edward D. Reiskin Director of Transportation

Approved as to Form: Dennis J. Herrera City Attorney

By:

Mariam Morley Deputy City Attorney

San Francisco Municipal Transportation Agency Board of Directors Resolution No: ______ Adopted: ______ Attest:

Secretary, SFMTA Board of Directors

Board of Supervisors

Resolution No:	
Adopted:	

Attest:

Clerk of the Board

CONTRACTOR

TESGCO, LLC, d.b.a. San Franciseo AutoReturn

John Wicker

President and CEO

City vendor number: 01-0688299

SFMTA P-550 (9-14)

LEASE AMENDMENT

THIS FIRST AMENDMENT (this "Amendment") is made as of _August 1, 2015, by and between CITY AND COUNTY OF SAN FRANCISCO, a municipal corporation ("City") acting by and through its Municipal Transportation Agency ("SFMTA"), and TEGSCO, LLC., a California limited liability corporation, d.b.a San Francisco AutoReturn ("Licensee").

RECITALS

A. City and Licensee have previously entered into an existing license agreement, dated as of May 1, 2014 (the "Original License"), for the Premises (as defined in the Original License) located at 2650 Bayshore Boulevard, Daly City, California. The Original License, as amended by this Amendment, shall be referred to herein as the "License Agreement".

B. The term of the Original License expires on July 31, 2015, and the parties now desire to extend such term on the terms and conditions as set forth herein.

AGREEMENT

ACCORDINGLY, in consideration of the matters described in the foregoing Recitals, and for other good and valuable consideration, the receipt and sufficiency of which are mutually acknowledged, the City and Licensee agree as follows:

1. <u>Extension of License Term</u>. Section 4.1 of the Original License is hereby amended and restated to read in its entirety as follows:

"4.1 Original Term. The privilege given to Licensee pursuant to this License is temporary only and shall commence upon the Commencement Date (as described in the Basic License Information) and shall terminate on March 31, 2016, or the date of earlier termination of this License pursuant to the terms of this License or the Towing Agreement (the "Expiration Date"). Without limiting any of its rights hereunder, City may at its sole option freely revoke this License at any time, without cause and without any obligation to pay any consideration to Licensee. Licensee acknowledges its receipt and acceptance of the Premises on May 1, 2013."

2. <u>7th Street Lease Rent Credit</u>. Pursuant to a SF Parking Airspace Lease between Licensee and the State of California (SF-80-11), dated as of January 6, 2015 (the "7th Street Lease"), Licensee leases certain real property at 450 7th Street in San Francisco, California, from the State of California, for the customer service center and short-term towing and storage facility Licensee must provide under the Towing Agreement. Notwithstanding anything to the contrary in Article 5 of the Original License, if Licensee makes a rent payment of \$90,000 or more under the 7th Street Lease to the State of California for any month between August 1, 2015 through March 31, 2016, Tenant shall receive a rent credit of \$53,000 against the Base Fee owed by Licensee for such month under the License Agreement.

3. <u>Criminal History Inquiries for Employment.</u>

1

(a) Unless exempt, Licensee agrees to comply with and be bound by all of the provisions of San Francisco Administrative Code Chapter 12T (Criminal History in Hiring and Employment Decisions; "Chapter 12 T"), which are hereby incorporated as may be amended from time to time, with respect to applicants and employees of Licensee who would be or are performing work at the Premises.

(b) Licensee shall incorporate by reference the provisions of Chapter 12T in all sublicenses of some or all of the Premises, and shall require all sublicensees to comply with such provisions. Licensee's failure to comply with the obligations in this subsection shall constitute a material breach of the License Agreement.

(c) Licensee and sublicensees shall not inquire about, require disclosure of, or if such information is received base an Adverse Action on an applicant's or potential applicant for employment, or employee's: (1) Arrest not leading to a Conviction, unless the Arrest is undergoing an active pending criminal investigation or trial that has not yet been resolved; (2) participation in or completion of a diversion or a deferral of judgment program; (3) a Conviction that has been judicially dismissed, expunged, voided, invalidated, or otherwise rendered inoperative; (4) a Conviction or any other adjudication in the juvenile justice system; (5) a Conviction that is more than seven years old, from the date of sentencing; or (6) information pertaining to an offense other than a felony or misdemeanor, such as an infraction.

(d) Licensee and sublicensees shall not inquire about or require applicants, potential applicants for employment, or employees to disclose on any employment application the facts or details of any conviction history, unresolved arrest, or any matter identified in subsection (c) above. Licensee and sublicensees shall not require such disclosure or make such inquiry until either after the first live interview with the person, or after a conditional offer of employment.

(e) Licensee and sublicensees shall state in all solicitations or advertisements for employees that are reasonably likely to reach persons who are reasonably likely to seek employment with Tenant or subtenant at the Premises that the Licensee and sublicensees will consider for employment qualified applicants with criminal histories in a manner consistent with the requirements of Chapter 12T.

(f) Licensee and sublicensees shall post the notice prepared by the Office of Labor Standards Enforcement ("OLSE"), available on OLSE's website, in a conspicuous place at the Premises and at other workplaces within San Francisco where interviews for job opportunities at the Premises occur. The notice shall be posted in English, Spanish, Chinese, and any language spoken by at least 5% of the employees at the Premises or other workplace at which it is posted.

(g) Licensee and sublicensees understand and agree that upon any failure to comply with the requirements of Chapter 12T, the City shall have the right to pursue any rights or remedies available under Chapter 12T or this Permit, including but not limited to a penalty of \$50 for a second violation and \$100 for a subsequent violation for each employee, applicant or other person as to whom a violation occurred or continued, termination or suspension in whole or in part of the License Agreement.

(h) If Licensee has any questions about the applicability of Chapter 12T, it may contact the SFMTA for additional information. SFMTA may consult with the Director of the City's Office of Contract Administration who may also grant a waiver, as set forth in Section 12T.8.

2

4. <u>Bottled Drinking Water</u>. Unless exempt, Licensee agrees to comply fully with and be bound by all of the provisions of the San Francisco Bottled Water Ordinance, as set forth in San Francisco Environment Code Chapter 24, including the administrative fines, remedies, and implementing regulations provided therein, as the same may be amended from time to time. The provisions of Chapter 24 are incorporated herein by reference and made a part of the License Agreement as though fully set forth.

5. <u>No Joint Venture</u>. This Amendment or any activity by the City hereunder does not create a partnership or joint venture between the City and Licensee relating to the License Agreement or otherwise. This Amendment does not constitute authorization or approval by the City of any activity conducted by Licensee, and the City shall in no way be responsible for the acts or omissions of Licensee on the Premises or otherwise.

6. <u>Attorneys Fees</u>. In the event a dispute arises concerning this Amendment, the party not prevailing in such dispute shall pay any and all costs and expenses incurred by the other party in enforcing or establishing its rights hereunder, including, without limitation, court costs and reasonable attorneys' fees. For purposes of this Amendment, reasonable fees of attorneys of City's Office of the City Attorney shall be based on the fees regularly charged by private attorneys with the equivalent number of years of experience in the subject matter area of the law for which the City Attorney's services were rendered who practice in the City of San Francisco in law firms with approximately the same number of attorneys as employed by the Office of the City Attorney.

7. <u>References</u>. No reference to this Amendment is necessary in any instrument or document at any time referring to the Lease. Any future reference to the Lease shall be deemed a reference to such document as amended hereby.

8. <u>Applicable Law</u>. This Amendment shall be governed by, construed and enforced in accordance with the laws of the State of California.

9. <u>Further Instruments</u>. The parties hereto agree to execute such further instruments and to take such further actions as may be reasonably required to carry out the intent of this Amendment.

10. <u>Effective Date</u>. The date of which this Amendment shall become effective as of the date this Amendment is duly executed and exchanged by the parties hereto.

11. <u>Miscellaneous</u>. Except as expressly modified herein, the terms, covenants and conditions of the Original License shall remain unmodified and in full force and effect. The Original License as amended by this Amendment constitutes the entire agreement of the parties concerning the subject matter hereof, and supersedes and conceals any and all previous negotiations, agreements, or understandings, if any, regarding the matters contained herein. The execution of this Amendment shall not constitute a waiver of relinquishment of any rights that City may have relating to the Original License. Licensee and City hereby ratify and confirm all of the provisions of the Original License as amended by this Amendment.

In witness whereof, the parties hereto have executed this Amendment as of the date written above.

[SIGNATURES ON FOLLOWING PAGE]

4

LICENSEE:

TEGSCO, LLC, a California limited liability company d.b.a. San Francisco AutoReturn

By: John Wicker, President and CEO

CITY AND COUNTY OF SAN FRANCISCO, a municipal corporation, acting by and through its Municipal Transportation Agency

By:

Edward D. Reiskin Director of Transportation

APPROVED AS TO FORM: DENNIS J. HERRERA, City Attorney

By:

Mariam Morley, Deputy City Attorney

San Francisco Municipal Transportation Agency Board of Directors Resolution No: ______ Adopted: ______

Attest:

Secretary, SFMTA Board of Directors

Board of Supervisors

Resolution No: ______Adopted: ______

Attest:

Clerk of the Board

CITY:

SAN FRANCISCO MUNICIPAL TRANSPORTATION AGENCY BOARD OF DIRECTORS

RESOLUTION No. 15-091

WHEREAS, The City, through the SFMTA, is responsible for efficiently controlling and removing illegally parked and abandoned vehicles, in order to maximize the amount of on-street parking, manage traffic congestion and maintain clean, safe and functional streets; and,

WHEREAS, On July 31, 2005, the SFMTA executed SFMTA Contract #2013-45 with San Francisco AutoReturn to manage a network of ten local and Bay Area towing firms that provide towing support for the SFMTA and the San Francisco Police Department (SFPD) and Appendix C, a license agreement for a long-term storage facility at Pier 70; and,

WHEREAS, In addition, the Contractor secured a separate license with Caltrans for the primary storage lot that contains the customer service center where the majority of the public retrieves their vehicles; and,

WHEREAS, The agreement was extended for five years in 2010; and

WHEREAS, In May 2014 Appendix C was replaced with Appendix H when both Boards approved transferring the long-term storage facility from Pier 70 to 2650 Bayshore Blvd.; and,

WHEREAS, The agreement will expire on July 31, 2015; and,

WHEREAS, The SFMTA requires an additional eight months to release the RFP, negotiate the contract and transfer leasing of the primary storage lot from the Contractor to the SFMTA; and,

WHEREAS, In order to pay for an increase to the primary storage facility lease amount from \$36,379 per month to \$90,000 per month the SFMTA will allow the Contractor to offset its rent of \$147,985.76 per month at the Bayshore facility by \$53,000 per month, for a total of \$424,000 during the eight-month extension period; and

WHEREAS, In November 2014, the Board of Supervisors passed Resolution No. 432-14, urging the SFMTA to evaluate the feasibility of waiving or otherwise mitigating towing and storage fees for individual members of the public whose vehicles were towed and identified by the SFPD as stolen; and,

WHEREAS, Waiving the contractor's towing and storage fees for individuals whose vehicles have been stolen will result in the SFMTA having to reimburse AutoReturn; and the SFMTA would no longer receive administrative fees currently paid by non-San Francisco residents whose vehicles were stolen, at a cost of approximately \$1 million annually; and,

WHEREAS, The proposed approval actions are categorically exempt from Environmental Review (Class 1 Operation, repair, maintenance, permitting, leasing, licensing, or minor alteration of existing public or private structures, facilities, mechanical equipment or topographical features involving negligible or no expansion of use beyond that existing at the time of the lead agency's determination); now, therefore, be it

RESOLVED, That the San Francisco Municipal Transportation Agency Board of Directors authorizes the Director or his designee to execute the Second Amendment to San Francisco Municipal Transportation Agency Contract No. 2013-45, the Amended and Restated Towing Agreement with TEGSCO, LLC, dba San Francisco AutoReturn for towing and storage of illegally-parked and abandoned vehicles, to extend the Agreement by eight months to March 31, 2016 at no additional cost to the City, pending Board of Supervisors approval; and be it further

RESOLVED, That the San Francisco Municipal Transportation Agency Board of Directors approves the First Amendment to Appendix H of San Francisco Municipal Transportation Agency Contract No. 2013-45, the Property Use License for Towing, Storage and Disposal of Abandoned and Illegally Parked Vehicles, at 2650 Bayshore Blvd, Daly City, CA for long-term storage and lien sales of towed vehicles, to extend the License eight months to March 31, 2016, and to allow San Francisco AutoReturn to deduct \$53,000 from the monthly rent of the Licensed facility to cover the increased monthly rent assessed by Caltrans for the customer service center and primary vehicle storage facility at 7th Street; and be it further

RESOLVED, That the SFMTA Board of Directors approves a policy for stolen vehicles of waiving SFMTA administrative towing and storage fees and the contractor's towing fee, and granting a 72-hour grace period before the contractor's storage fees begin to accrue.

I certify that the foregoing resolution was adopted by the San Francisco Municipal Transportation Agency Board of Directors at its meeting of June 16, 2015.

R. Boomer

Secretary to the Board of Directors San Francisco Municipal Transportation Agency



Edwin M. Lee, Mayor

Tom Nolan, *Chairman* Gwyneth Borden, *Director* Jerry Lee, *Director* Cristina Rubke, *Director* Cheryl Brinkman, Vice-Chairman Malcolm Heinicke, Director Joél Ramos, Director

Edward D. Reiskin, Director of Transportation

Transportation Code Modification: Waiver for Towing and Storage Fees

The SFMTA proposes to amend the Transportation Code to formalize the Agency's current practice of offering waivers for towing and administrative fees paid by the public in cases such as car theft.

Statutorily Exempt from Environmental Review—CEQA Guidelines Section 15273 Rates, Tolls, Fares and Charges: CEQA does not apply to the establishment, modification, structuring, restructuring or approval of rates, tolls, fares or other charges by public agencies.

62

5/18/2015

Erik Jaszewski, SFMTA San Francisco Municipal Transportation Agency Date

SAN FRANCISCO MUNICIPAL TRANSPORTATION AGENCY BOARD OF DIRECTORS

RESOLUTION No. 15-091

WHEREAS, The City, through the SFMTA, is responsible for efficiently controlling and removing illegally parked and abandoned vehicles, in order to maximize the amount of on-street parking, manage traffic congestion and maintain clean, safe and functional streets; and,

WHEREAS, On July 31, 2005, the SFMTA executed SFMTA Contract #2013-45 with San Francisco AutoReturn to manage a network of ten local and Bay Area towing firms that provide towing support for the SFMTA and the San Francisco Police Department (SFPD) and Appendix C, a license agreement for a long-term storage facility at Pier 70; and,

WHEREAS, In addition, the Contractor secured a separate license with Caltrans for the primary storage lot that contains the customer service center where the majority of the public retrieves their vehicles; and,

WHEREAS, The agreement was extended for five years in 2010; and

WHEREAS, In May 2014 Appendix C was replaced with Appendix H when both Boards approved transferring the long-term storage facility from Pier 70 to 2650 Bayshore Blvd.; and,

WHEREAS, The agreement will expire on July 31, 2015; and,

WHEREAS, The SFMTA requires an additional eight months to release the RFP, negotiate the contract and transfer leasing of the primary storage lot from the Contractor to the SFMTA; and,

WHEREAS, In order to pay for an increase to the primary storage facility lease amount from \$36,379 per month to \$90,000 per month the SFMTA will allow the Contractor to offset its rent of \$147,985.76 per month at the Bayshore facility by \$53,000 per month, for a total of \$424,000 during the eight-month extension period; and

WHEREAS, In November 2014, the Board of Supervisors passed Resolution No. 432-14, urging the SFMTA to evaluate the feasibility of waiving or otherwise mitigating towing and storage fees for individual members of the public whose vehicles were towed and identified by the SFPD as stolen; and,

WHEREAS, Waiving the contractor's towing and storage fees for individuals whose vehicles have been stolen will result in the SFMTA having to reimburse AutoReturn; and the SFMTA would no longer receive administrative fees currently paid by non-San Francisco residents whose vehicles were stolen, at a cost of approximately \$1 million annually; and,

WHEREAS, The proposed approval actions are categorically exempt from Environmental Review (Class 1 Operation, repair, maintenance, permitting, leasing, licensing, or minor alteration of existing public or private structures, facilities, mechanical equipment or topographical features involving negligible or no expansion of use beyond that existing at the time of the lead agency's determination); now, therefore, be it

RESOLVED, That the San Francisco Municipal Transportation Agency Board of Directors authorizes the Director or his designee to execute the Second Amendment to San Francisco Municipal Transportation Agency Contract No. 2013-45, the Amended and Restated Towing Agreement with TEGSCO, LLC, dba San Francisco AutoReturn for towing and storage of illegally-parked and abandoned vehicles, to extend the Agreement by eight months to March 31, 2016 at no additional cost to the City, pending Board of Supervisors approval; and be it further

RESOLVED, That the San Francisco Municipal Transportation Agency Board of Directors approves the First Amendment to Appendix H of San Francisco Municipal Transportation Agency Contract No. 2013-45, the Property Use License for Towing, Storage and Disposal of Abandoned and Illegally Parked Vehicles, at 2650 Bayshore Blvd, Daly City, CA for long-term storage and lien sales of towed vehicles, to extend the License eight months to March 31, 2016, and to allow San Francisco AutoReturn to deduct \$53,000 from the monthly rent of the Licensed facility to cover the increased monthly rent assessed by Caltrans for the customer service center and primary vehicle storage facility at 7th Street; and be it further

RESOLVED, That the SFMTA Board of Directors approves a policy for stolen vehicles of waiving SFMTA administrative towing and storage fees and the contractor's towing fee, and granting a 72-hour grace period before the contractor's storage fees begin to accrue.

I certify that the foregoing resolution was adopted by the San Francisco Municipal Transportation Agency Board of Directors at its meeting of June 16, 2015.

R. Boomer_

Secretary to the Board of Directors San Francisco Municipal Transportation Agency



SFMTA Municipal Transportation Agency Edwin M. Lee, Mayor

Tom Nolan, *Chairman* Gwyneth Borden, *Director* Jerry Lee, *Director* Cristina Rubke, *Director* Cheryl Brinkman, Vice-Chairman Malcolm Heinicke, Director Joél Ramos, Director

Edward D. Reiskin, Director of Transportation

Transportation Code Modification: Waiver for Towing and Storage Fees

The SFMTA proposes to amend the Transportation Code to formalize the Agency's current practice of offering waivers for towing and administrative fees paid by the public in cases such as car theft.

Statutorily Exempt from Environmental Review—CEQA Guidelines Section 15273 Rates, Tolls, Fares and Charges: CEQA does not apply to the establishment, modification, structuring, restructuring or approval of rates, tolls, fares or other charges by public agencies.

4

5/18/2015 Date

Erik Jaszewski, SFMTA San Francisco Municipal Transportation Agency

1 South Van Ness Avenue 7th Floor, San Francisco, CA 94103

415.701.4500

City and County of San Francisco Municipal Transportation Agency One South Van Ness Ave. 7th floor San Francisco, California 94103

First Amendment to Amended and Restated Service Agreement and Property Use License for Towing, Storage and Disposal of Abandoned and Illegally Parked Vehicles

THIS AMENDMENT (this "Amendment") is made as of May 1, 2014 in San Francisco, California, by and between TEGSCO, LLC, a California limited liability company doing business as San Francisco AutoReturn ("Contractor"), and the City and County of San Francisco, a municipal corporation ("City"), acting by and through its Municipal Transportation Agency ("SFMTA").

RECITALS

A. City and Contractor have entered into the Agreement (as defined below).

B. City and Contractor desire to modify the Agreement on the terms and conditions set forth herein to reflect the relocation of the Secondary Storage Facility and one of the Designated Facilities from property owned by the Port of San Francisco at Pier 70 in San Francisco to a portion of the premises at 2650 Bayshore Boulevard in Daly City that City leases from Prologis, L.P.

NOW, THEREFORE, Contractor and the City agree as follows:

1. **Definitions.** The following definitions shall apply to this Amendment:

- a. Agreement. The term "Agreement" shall mean the Amended and Restated Service Agreement and Property Use License for Towing, Storage and Disposal of Abandoned and Illegally Parked Vehicles dated July 31, 2010 between Contractor and City.
- **b. Other Terms.** Terms used and not defined in this Amendment shall have the meanings assigned to such terms in the Agreement.

2. Modifications to the Agreement. The Agreement is hereby modified as follows:

a. Section 1.29

Section 1.29 of the Agreement is deleted in its entirety:

b. Section 1.42

Section 1.42 of the Agreement is amended in its entirety to read as follows:

1.42 <u>Performance Bond</u>: The Performance Bond which Contractor is required to maintain to guarantee the performance of Contractor's obligations under this Agreement, as further

May 1, 2014

described in Section 12.1 and 12.2 of this Agreement. The Performance Bond and all replacement Performance Bonds provided by Contractor during the Term of this Agreement shall be attached hereto as Appendix F and incorporated by reference as though fully set forth herein.

c. Section 1.44

Section 1.44 of the Agreement is deleted in its entirety.

d. Section 1.45

Section 1.45 of the Agreement is deleted in its entirety.

e. Section 1.48

Section 1.48 of the Agreement is amended in its entirety to read as follows:

1.48 Property, Properties: Real property owned or leased by the City and licensed to Contractor for the purpose of this Agreement, including a portion of the leased premises at 2650 Bayshore Boulevard in Daly City, and any other properties that may be licensed to Contractor by City for the purpose of this Agreement during its Term.

f. Section 1.57

Section 1.57 of the Agreement is amended in its entirety to read as follows:

1.57 <u>Secondary Storage Facility</u>: The facility primarily used for long term storage of impounded vehicles, located at 2650 Bayshore Boulevard in Daly City, California, as further described in Appendix A, Section 11.1(c).

g. Section 1.62

Section 1.62 of the Agreement is amended in its entirety to read as follows:

1.62 <u>Term</u>: The duration of this Agreement as established in Section 2 herein, and any additional period during which Contractor completes repair, remediation or other work required for the termination of the License agreement for 2650 Bayshore Boulevard in Daly City, as set forth in Appendix H of this Amendment ("the Bayshore License"), and the tasks listed in Section 54 of this Agreement for transition to a successor contractor or successor agreement.

h. Section 2

Section 2 of the Agreement is amended in its entirety to read as follows:

2. Term of the Agreement

The Term of this Agreement shall be five years, from the July 31, 2010 through July 31, 2015.

i. Section 4

Section 4 of the Agreement is amended in its entirety to read as follows:

May 1, 2014

4. Services Contractor Agrees to Perform

The Contractor shall tow and store any vehicles that SFMTA, DEPTH or the SPED orders removed from any public street or highway or from private property within the City, in accordance with the requirements of the San Francisco Transportation Code and the Vehicle Code, and shall perform such other related services as are described in this Agreement, in accordance with Appendix A, "Scope of Work," and Appendix B "Operations Plan". Appendix A, Appendix B, and the Operations Plan to be adopted and amended as provided in Section 14 of Appendix A are all attached hereto and incorporated by reference as though fully set forth herein. Rates and charges to the public for services under this Agreement shall be as set forth in Appendix F, attached hereto and incorporated by reference as though fully set forth herein.

All vehicle handling and storage required by this Agreement shall be conducted at one of the Designated Facilities. As of the Effective Date of the First Amendment to this Agreement and pursuant to the Bayshore License, City hereby licenses to Contractor the use of one of the real properties on which the Designated Facilities are currently located -- a portion of the premises at 2650 Bayshore Boulevard in Daly City. The Bayshore License is attached to the First Amendment to this Agreement as Appendix H and is incorporated by reference as though fully set forth herein. The parties acknowledge that Contractor's Customer Service Center and Primary Storage Facility operations are conducted at a property owned by Caltrans and leased by Contractor -- 450 7th Street. Should City cease to use Contractor for towing services, the terms for the City's continued use of the 450 7th Street property are attached hereto as Appendix G and are incorporated by reference as though fully set forth herein.

City hereby consents to Contractor's towing and storage of vehicles that are not towed or stored pursuant to a Tow Request with the prior written approval of City and subject to any conditions imposed in such written approval; provided however, that (I) Contractor shall at all times conduct itself in accordance with the Customer service standards of this Agreement so as not create any negative effect on Contractor's public image and reputation as the City's towing contractor, and (ii) Contractor's operations with respect to such vehicles shall not create any adverse impact on its performance of all requirements of this Agreement. City may revoke its consent at any time without cause by written notice to Contractor. The following sections of this Agreement shall apply to any towing or storage of vehicles by Contractor within the City and County of San Francisco that are not towed or stored pursuant to a Tow Request: Amended and Restated Service Agreement Sections 8, 9, 11, 12, 13, 14, 15, 16, 17, 20, 22, 26, 27, 28, 29, 31, 32, 35, 37, 38, 40, 41, 42, 50; Appendix A Sections 1, 3, 4, 5, 6, 7, 9, 11, 12.1(b), 12.3, 12.5, 12.6, 13, 14; Appendix B, Appendices E and F, and the Bayshore License during its term, if such towed or stored vehicle is taken to the Secondary Storage Facility.

j. Section 11

Section 11 of the Agreement is amended in its entirety to read as follows:

11. Required Insurance

Subject to approval by the City's Risk Manager of the insurers and policy forms, Contractor shall place and maintain throughout the Term of this Agreement, and pay the cost thereof, the following insurance policies:

11.1. Comprehensive general liability insurance with limits not less than \$2,000,000 (such

May 1, 2014

limit may be provided through a primary and excess policy) each Occurrence, combined single limit for bodily injury and property damage, or in such greater amount and limits as SFMTA may reasonably require from time to time, including coverage for contractual liability, personal injury, broad form property damage, products and completed operations, independent contractors (excluding towing and dismantling subcontractors), and mobile equipment.

11.2. Sudden and accidental pollution insurance with limits not less than \$1,000,000 for each occurrence.

11.3. Comprehensive business/commercial automobile liability insurance with limits not less than \$2,000,000 for each Occurrence combined single limit for bodily injury and property damage, including coverage for owned, non-owned and hired automobiles. If Contractor does not own or lease company vehicles that are subject to motor vehicle registration, then only non-owned and hired coverage is required.

11.4. Garage-keeper's legal liability insurance with limits not less than \$5,000,000 (such limit may be provided through a primary and excess policy) for each Occurrence combined single limit for loss and damage to vehicles in Contractor's care, custody or control caused by fire, explosion, theft, riot, civil commotion, malicious mischief, vandalism or collision, with any deductible not to exceed \$25,000 for each Occurrence. Contractor may insure or self-insure loss of non-automobile property in the care, custody, or control of the garage keeper with a limit of \$5000.

11.5. Workers' Compensation Insurance, including Employers' Liability, with limits not less than \$1,000,000 for each accident, covering all Employees employed by Contractor in the performance of this Agreement to provide statutory benefits as required by the laws of the State of California. Said policy shall be endorsed to provide that the insurer waives all rights of subrogation against the City.

11.6. Environmental impairment liability insurance with limits not less than \$1,000,000 each occurrence, covering the sudden and accidental release of hazardous materials and the resulting costs of clean up.

Except as set forth above, any deductibles in the policies listed above shall not exceed \$25,000 each occurrence. The insurance policies shall be endorsed to name as an additional insured the City and County of San Francisco and its respective departments, commissioners, officers, agents and employees.

The agreements between the Contractor and its towing and dismantling subcontractors (as applicable) shall require that reasonable insurance is maintained and that the City will be named as an additional insured on the general liability policy.

11.7. Commercial General Liability and Commercial Automobile Liability Insurance policies must be endorsed to provide:

Name as Additional Insured the City and County of San Francisco, its Officers, Agents, and Employees.

That such policies are primary insurance to any other insurance available to the Additional Insureds, with respect to any claims arising out of this Agreement, and that insurance applies separately to each insured against whom claim is made or suit is brought, but the inclusion of more than on insured shall not operate to increase the insurer's limits of liability.

Should any of the required insurance be provided under a form of coverage that includes a general annual aggregate limit or provides that claims investigation or legal defense costs be

included in such general annual aggregate limit, such general annual aggregate limit shall be double the occurrence or claims limits specified above.

11.8. Regarding Workers' Compensation, Contractor hereby agrees to waive subrogation which any insurer of Contractor may acquire from Contractor by virtue of the payment of any loss. Contractor agrees to obtain any endorsement that may be necessary to effect this waiver of subrogation. The Workers' Compensation policy shall be endorsed with a waiver of subrogation in favor of the City for all work performed by the Contractor, its employees, agents and subcontractors.

11.9. Should any of the required insurance be provided under a claims-made form, Contractor shall maintain such coverage continuously throughout the term of this Agreement and, without lapse, for a period of three years beyond the expiration of this Agreement, to the effect that, should occurrences during the contract term give rise to claims made after expiration of the Agreement, such claims shall be covered by such claims-made policies.

11.10. Subject to the provisions of Section 18.1, should any required insurance lapse during the term of this Agreement, requests for payments originating after such lapse shall not be processed until the City receives satisfactory evidence of reinstated coverage as required by this Agreement, effective as of the lapse date. If insurance is not reinstated, the City may, at its sole option, terminate this Agreement effective on the date of such lapse of insurance.

11.11. Before commencing any operations under this Agreement, Contractor shall furnish to City certificates of insurance and additional insured policy endorsements with insurers with ratings comparable to A-, VIII or higher, that are authorized to do business in the State of California, and that are satisfactory to City, in form evidencing all coverages, including additional insured endorsements and the policy declaration page for any umbrella policies, set forth above. Failure to maintain insurance shall constitute a material breach of this Agreement.

11.12. Approval of the insurance by City shall not relieve or decrease the liability of Contractor hereunder.

k. Section 12.2

Section 12.2 of the Agreement is amended in its entirety to read as follows:

12.2 Performance Bond

12.2.1 Performance Guarantee. Upon the Effective Date of this Agreement, Contractor shall provide to the City, and shall maintain throughout the Term of this Agreement and for a period of at least ninety (90) days after expiration or termination of this Agreement, or until all of Contractor's obligations have been performed under this Agreement, whichever date is later, a performance guarantee of two million dollars (\$2,000,000), which shall consist of a Performance Surety Bond of two million dollars (\$2,000,000) in favor of the City and County of San Francisco, a municipal corporation, acting by and through its Municipal Transportation Agency, guarantying the faithful performance by Contractor of this Agreement and of the covenants, terms and conditions of this Agreement, including all monetary obligations set forth herein, and including liquidated damages and any dishonesty on the part of Contractor.

The City may draw upon such Performance Surety Bond in circumstances which include, but are not limited to:

May 1, 2014

n:\ptc\as2013\1000468\00891055.doc

a. To ensure regulatory compliance in the event that Contractor receives a notice of violation or other regulatory order from a governmental or regulatory agency with jurisdiction over Contractor's operations or the Properties used by Contractor for the performance of this Agreement and Contractor does not achieve compliance with the notice of violation or order to the satisfaction of the issuing agency within the time specified by the agency or by the City if the agency does not specify a timeframe.

b. To reimburse the City for any fine or other charge assessed against the City related to any notice of violation or other regulatory order issued to Contractor.

c. To reimburse the City for costs associated with City's environmental assessments or corrective action related to Contractor's violation of any of the requirements of Appendix H, which may be performed at the City's sole discretion.

d. To satisfy any overdue payment obligations owed by Contractor to City pursuant to the Bayshore.

e. To satisfy fines assessed by City against Contractor pursuant to the Bayshore License.

f. To compensate City for losses or damage to property caused

by Contractor.

12.2.2 Performance Surety Bond Requirements. The Performance Surety Bond required by this Section 12 shall be issued on a form and issued by a financial institution acceptable to the City in its sole discretion, which financial institution shall (a) be a bank, insurance or trust company doing business and having an office in the State of California, (b) have a combined capital and surplus of at least \$25,000,000, and (c) be subject to supervision or examination by federal or state authority. If Contractor defaults with respect to any provision of this Agreement, City may, but shall not be required to, make its demand under said Performance Surety Bond for all or any portion thereof, to compensate City for any loss or damage which City may have incurred by reason of Contractor's default or dishonesty, including (but not limited to) any claim for fines or liquidated damages; provided, however, that City shall present its written demand to said bank, insurance or trust company for payment under said Performance Surety Bond only after City first shall have made its demand for payment directly to Contractor, and five (5) full days have elapsed without Contractor having made payment to City.

12.2.3 Expiration or Termination of Performance Surety Bond. The term of the Performance Surety Bond shall apply for individual one-year periods, and may be extended by the insurance, bank or trust company by Continuation Certificate. The insurance, bank or trust company herein may, if it so elects, terminate its obligation under this bond by serving at least forty five (45) days written notice of its intention to do so upon the SFMTA. In the event the City receives notice from the issuer of the Performance Surety Bond that the Performance Surety Bond will be terminated, not renewed or will otherwise be allowed to expire for any reason during the period from the commencement of the Term of this Agreement to ninety (90) days after the expiration or termination of this Agreement or the conclusion of all of Contractor's obligations under the Agreement, whichever occurs last, and Contractor fails to provide the City with a replacement Performance Surety Bond (in a form and issued by a financial institution acceptable to the City) within ten (10) days following the City's receipt of such notice, such occurrence shall be

6 of 18

an Event of Default as defined in Section 18 of this Agreement. However, neither nonrenewal by the insurance, bank or trust company, nor the failure or inability of the Contractor to file a replacement bond in the event of nonrenewal, shall itself constitute a loss to the City recoverable under the Performance Surety Bond or any renewal or continuation thereof. Insurance, bank or trust company's liability under the Performance Surety Bond and all continuation certificates issued in connection therewith shall not be cumulative and shall in no event exceed the amount as set forth in the Performance Surety Bond or in any additions, riders, or endorsements properly issued by the insurance, bank or trust company as supplements thereto.

12.2.5 Demands Upon Performance Surety Bond. City may use all or any portion of the Performance Surety Bond to compensate City for any loss or damage that it may have incurred by reason of Contractor's negligence or breach. Such loss or damage may include without limitation any damage to or restoration of the Properties for which Contractor is responsible, and claims for fines and/or liquidated damages. Should the City terminate this Agreement due to a breach by Contractor, the City shall have the right to draw from the Performance Surety Bond those amounts necessary to pay any fees or other financial obligations under the Agreement or the Bayshore License, and perform the towing and storage services described in this Agreement until such time as the City procures another contractor and the agreement between the City and that contractor becomes effective.

12.2.6 Depletion of Performance Surety Bond. If any portion of the Performance Surety Bond is used by City, Contractor shall provide written proof that the Performance Surety Bond has been restored to its initial value, which shall require a replacement Performance Surety Bond in the face amount of the required Performance Surety Bond. Contractor's failure to do so within the time limits specified in Section 18.1.1(f) shall constitute an Event of Default as defined in Section 18 of this Agreement.

12.2.7 Dispute Resolution. In the event that a dispute arises between the City and Contractor concerning this Agreement or the use or maintenance of the Performance Surety Bond, Contractor may appeal to the Director of Transportation within fourteen (14) days of demand on the Performance Surety Bond with evidence supporting Contractor's claim for relief from the demand on the Performance Surety Bond. The Director of Transportation will respond within fourteen (14) days. Any failure of the Director of Transportation to respond within fourteen (14) days shall be deemed a rejection of Contractor's claim for relief from the demand on the Performance Surety Bond. Contractor's claim for relief from demands on the Performance Surety Bond and the Director of Transportation's response to such demand shall constitute the administrative remedy for Agreement interpretation described in Section 46 herein. Each party reserves its remedies in equity and law. No decision by the City concerning the Performance Surety Bond shall prevent Contractor from seeking restoration of the funds by appropriate legal action.

I. Section 12.5

Section 12.5 of the Agreement is amended in its entirety to read as follows:

12.5 Environmental Oversight Deposit. Upon the Effective Date of this Agreement, Contractor shall provide to the City, and shall maintain and replenish throughout the term of the Bayshore License and for a period of at least ninety (90) days after termination or expiration of the Bayshore License, an Environmental Oversight Deposit in the amount of ten thousand dollars (\$10,000), which shall be deposited in an account specified by City. If Contractor receives a notice of violation or other regulatory order from a governmental or regulatory agency with jurisdiction over 2650 Bayshore Boulevard, Daly City, California and or its operations), and such notice is not cured within fourteen (14) days, the City may draw from this deposit to reimburse the City for staff costs incurred by the City while inspecting site conditions and enforcing and administering the Hazardous Materials provisions of the Bayshore License. If Contractor receives a notice of violation or other regulatory order from a governmental or regulatory agency with jurisdiction over the site and or its operations, and such notice is cured within fourteen (14) days, the City may draw from this deposit in an amount not to exceed \$500 to reimburse the City for staff costs incurred by the City. The City will submit an invoice to Contractor for any such costs, and Contractor will pay such invoiced amounts within thirty (30) days to replenish the Environmental Oversight Deposit. Contractor's failure to pay such costs within thirty (30) days, or to replenish the Environmental Oversight Deposit if drawn upon, will constitute an Event of Default.

m. Section 13

Section 13 of the Agreement is amended in its entirety to read as follows:

13. Insurance and Performance Guarantee Requirements

All insurance policies and Performance Surety Bonds obtained pursuant to this Agreement shall be endorsed to provide that thirty (30) days prior written notice of cancellation, non-renewal or reduction in coverage or limits shall be given to SFMTA in the manner and at the addresses specified below.

Two copies of any Performance Surety Bond, and two copies of each original policy or policy endorsement of insurance shall be provided to SFMTA upon the Effective Date of this Agreement, and complete copies of any insurance policies obtained pursuant to this Agreement shall be provided to SFMTA if requested at any time.

13.1 Upon City's request, Contractor shall provide satisfactory evidence that Contractor has adequately provided for Social Security and Unemployment Compensation benefits for Contractor's Employees.

Contractor shall comply with the provisions of any insurance policy covering Contractor or the City, and with any notices, recommendations or directions issued by any insurer under such insurance policies so as not to adversely affect the insurance coverage.

n. Section 15.

Section 15 of the Agreement is amended in its entirety to read as follows:

15. Indemnification

Contractor shall indemnify and save harmless City and its officers, agents and employees from, and, if requested, shall defend them against any and all loss, cost, damage, injury, liability, and claims thereof for injury to or death of a person, including Employees of Contractor or loss of or damage to property, resulting directly or indirectly from Contractor's performance of this Agreement, including, but not limited to, the use of Contractor's facilities or equipment provided by City or others, regardless of the negligence of, and regardless of whether liability without fault is imposed or sought to be imposed on City, except to the extent that such indemnity is void or otherwise unenforceable under applicable law in effect on or validly retroactive to the date of this Agreement, and except where such loss, damage, injury, liability or claim is the result of the active negligence or willful misconduct of City and is not contributed to by any act of, or by any omission to perform some duty imposed by law or agreement on Contractor, its subcontractors or either's agent or employee. The foregoing indemnity shall include, without limitation, reasonable fees of attorneys, consultants and experts and related costs and City's costs of investigating any claims against the City. The foregoing indemnification does not include the limitations on Contractor's liability described in Sections 24.3 and 24.8 of the Bayshore License.

In addition to Contractor's obligation to indemnify City, Contractor specifically acknowledges and agrees that it has an immediate and independent obligation to defend City from any claim which actually or potentially falls within this indemnification provision, even if the allegations are or may be groundless, false or fraudulent, which obligation arises at the time such claim is tendered to Contractor by City and continues at all times thereafter.

As to any intellectual property that Contractor provides to the City in the performance of this Agreement, Contractor shall indemnify and hold City harmless from all loss and liability, including attorneys' fees, court costs and all other litigation expenses for any infringement of the patent rights, copyright, trade secret or any other proprietary right or trademark, and all other intellectual property claims of any person or persons, arising as a consequence of the use by City of the intellectual property supplied by the Contractor, or any of its officers or agents.

o. Section 18.1.6

Section 18.1.6 of the Agreement is amended in its entirety to read as follows:

18.1.6 Failure of Contractor to timely perform its obligations under the Bayshore License.

p. Section 19.3

Section 19.3 of the Agreement is amended in its entirety to read as follows:

19.3 Duties Upon Termination. Upon termination of this Agreement, the City and Contractor shall promptly pay to the other, as soon as is determinable after the effective date of termination, all amounts due each other under the terms of this Agreement, and upon such payment neither shall have any further claim or right against the other, except as expressly provided herein. Upon the effective date of termination, Contractor shall deliver to the SFMTA the originals of all
books, permits, plans, Records, licenses, contracts and other documents pertaining to Contractor's operation under this Agreement, any insurance policies, bills of sale or other documents evidencing title or rights of the City, and any and all other Records or documents pertaining to Contractor's operation under this Agreement, any insurance policies, bills of sale or other documents evidencing title or rights of the City, and any and all other Records or documents pertaining to the Designated Facilities, whether or not enumerated herein, which are requested by the City or necessary or desirable for the ownership, leasing and operation of the Designated Facilities, which are then in possession of Contractor. Contractor further agrees to do all other things reasonably necessary to cause an orderly transition of the management and operation of the Services provided by Contractor under this Agreement without detriment to the rights of the City or to the continued operation of such services.

q. Section 54.2

Section 54.2 of the Agreement is amended in its entirety to read as follows:

54.2 Contractor agrees to take all actions as may be necessary or as the City may direct for the protection and preservation of any property related to this Agreement that is in the possession of Contractor and in which City has or may acquire an interest, including any environmental remediation required under Appendix H. Contractor acknowledges that City has a vested interest for payment of fees for vehicles that Contractor has towed and/or is storing, and Contractor shall not impair said interest and shall take all actions reasonably necessary to safeguard City's interest in said towed and stored vehicles.

r. Section 4.5

Section 4.5 of Appendix A to the Agreement is amended in its entirety to read as follows:

4.5

No ID Vehicles

(a) **NO ID Designation**. All vehicles with no visible VIN shall be impounded under a "NO ID" number and shall be designated as a vehicle subject to Investigative Police Hold and held for inspection by the SFPD Auto Detail regardless of which City agency initiated the Tow Request. NO ID vehicles shall be included in regular reports to the SFPD of Police Hold vehicles as specified in Sections 4.6(d) and 13 of this Appendix A. Contractor shall keep NO ID vehicles within the NO ID area (excluding oversized vehicles), with the exception that Contractor shall move a NO ID vehicle out of the NO ID area within twenty-four (24) hours of receiving a request to do so by the SFPD.

(b) NO ID Procedure. If a VIN is found following inspection by the SPED, Contractor shall follow applicable lien sale provisions of the Vehicle Code for processing that vehicle. Otherwise, Contractor shall designate the vehicle as an "Unable to Identify" or "UNTIDY" vehicle and, after receipt of a written release by the SPED (DMV Form 462, "Public Agency Authorization of Disposal of Vehicle" or successor form), such UTID vehicle shall be disposed of as required by the Vehicle Code and in accordance with instructions on DMV Form 462 or successor form. (c) **SFPD Access to NO ID Vehicles.** Contractor shall allow SFPD personnel with written authorization from the Chief of Police or an officer designated by the Chief of Police to remove parts from any NO ID vehicle, except as prohibited by Appendix H.

s. Section 10.2

Section 10.2 of Appendix A to the Agreement is amended in its entirety to read as follows:

10.2 Network Connections. Within thirty (30) days of the Effective Date of the Agreement, or other date approved in writing by City, the Contractor shall provide data lines to connect the following locations to the Contractor's TVMS:

Contractor Division	Location
Customer Service Center	450 7th Street
Primary Storage Facility	450 7th Street
Secondary Storage Facility	2650 Bayshore Boulevard, Daly City

Note: Alternate sites may be approved by the City.

Contractor shall provide T1 Internet data lines that connect the Customer Service Center and Primary Storage Facility to the Contractor's TVMS. Contractor shall provide data connections to the Primary and Secondary Storage Facilities using a T1 data line. Contractor shall install a firewall at all locations where there is a direct connect to the Internet to ensure the security of the data.

Within thirty (30) days of the Effective Date of this Agreement, or other date approved in writing by City, Contractor shall provide a T1 data line that connects the Customer Service Center with the SFMTA server headquarters at 1 South Van Ness Avenue. This connection will be used to provide a connection to the Contractor's TVMS for City staff and create a pathway for the Contractor's staff and systems to access the City's CMS. At any time during the Term of this Agreement, with the City's prior, written approval, and subject to all the conditions contained in such approval, the parties may agree to eliminate the requirement for the T1 data line that connects the networks of the City and Contractor. Because both the CMS and TVMS systems can now be accessed directly via the Internet, the parties may agree to eliminate the T1 data line to simplify the systems administration for the City and Contractor, and to eliminate the monthly costs of both the data connection and the associated networking equipment (see Section 10.1).

Contractor shall provide the following City locations with real-time access to the Contractor's TVMS within thirty (30) days of the Effective Date of this Agreement:

City Division	Location		
SFMTA Enforcement	505 7th Street		
SFMTA Citation Division	11 South Van Ness Avenue		
SFMTA Hearing Division	11 South Van Ness Avenue		
SFMTA Administration	1 South Van Ness Avenue, 8th Floor		
SFPD NOID Office	850 Bryant Street		
SFPD STOP Program	850 Bryant Street		

For users located at the City locations listed above who cannot connect to the Contractor's system using either a direct connection to the Internet or the data line connecting the Customer Service Center to 1 South Van Ness Avenue, Contractor shall configure a single Virtual Private Network (VPN) utilizing 3DES encryption per location. Remote clients shall be able to connect to Contractor's network through remote VPN client software and DSL Internet connections. Contractor shall bear only the cost of the DSL Internet services and the corresponding DSL modems May 1, 2014 11 of 18 n;http://da2013/1000468/00891055.doc

and/or routers. City shall identify and provide a computer (Windows XP, or later version) in each location on which the VPN client software will be installed, for which the DSL Internet service will be established, and for which the access to the Contractor's TVMS will be provided. The City shall also be responsible for any telecommunications cabling that is required for the DSL connections to be established in each location.

t. Section 11.1

Section 11.1 of Appendix A to the Agreement is amended in its entirety to read as follows:

11.1 Designated Facilities. Contractor shall use the Designated Facilities for all service requirements of this Agreement, except as otherwise approved in advance in writing by City. The location or relocation of any Designated Facility shall be subject to prior written approval by City. City hereby approves the current location of all Designated Facilities. Contractor may use Designated Facilities for towing and storing vehicles that are not towed or stored at the request of the City with the prior written approval of City and subject to any conditions imposed in such approval. Contractor may allow Employee and vendor parking at Designated Facilities, subject to any limitations set forth in the Bayshore License, so long as it does not interfere with Contractor's performance of towing and impound services to all standards and requirements of this Agreement.

(a) Customer Service Center. The Customer Service Center shall provide a location for Customers recovering vehicles in person to pay for towing and storage charges, Citation fees and penalties, and other applicable fees, and/or to process any documentation required for vehicle release. The Customer Service Center shall be open to the public twenty-four (24) hours per day, 365 days per year. If the Customer Service Center is relocated outside of the Hall of Justice at 850 Bryant Street, Contractor shall provide a security guard in any area open to the public at all times that such facility is open to the public at its sole expense. The Customer Service Center must be located at or near the Primary Storage Facility.

(b) **Primary Storage Facility**

(i) Authorized Facility. City hereby approves the lot at 450 7th Street as a Primary Storage Facility.

(ii) Vehicle Storage and Retrieval. Contractor must provide a covered area at the Primary Storage Facility where Customers can wait while their vehicle is being retrieved. The Primary Storage Facility shall be open twenty-four (24) hours per day, 365 days per year.

(iii) Tows to Primary Storage. Contractor shall take all towed vehicles to the Primary Storage Facility for short-term storage if they are not subject to a Police Hold or are not taken directly to the Secondary Storage Facility. Contractor shall store all towed vehicles at the Primary Storage Facility for twenty-four (24) hours after being towed, unless vehicles are required by this Agreement to be directly towed to the Secondary Storage Facility. Vehicles stored at the Primary Storage Facility may be moved to the Secondary Storage Facility after the first twenty-four (24) hours. Contractor shall ensure that it transfers vehicles from Primary to Secondary Storage frequently enough to ensure that there is, at all times, sufficient space for newly-towed vehicles to be received at the Primary Storage Facility. Contractor shall not conduct any vehicle maintenance or vehicle parts sales at the Primary Storage Facility, except for maintenance of forklifts or other lot operations equipment.

(c) Secondary Storage Facility

(i) Authorized Facility. City hereby grants Contractor a license to occupy and use the Property at 2650 Bayshore Boulevard, Daly City as a Secondary Storage Facility as of the Effective Date of the Agreement during the Term of and so long as the Contractor complies with the terms and conditions of Appendices C and D.

Vehicle Storage and Retrieval. Contractor shall use the Secondary (ii) Storage Facility to store vehicles, including vehicles towed directly to the Secondary Storage Facility and most vehicles which are not picked up by the public within twenty-four (24) hours. Contractor shall conduct vehicle lien sales at this location. This facility shall be open to the public from 8:00 a.m. to 6:00 p.m. Monday through Friday. Outside of operating hours, Contractor shall secure the Secondary Storage Facility using security personnel. All vehicles must be available for retrieval from the Secondary Storage Facility by Contractor's staff twenty-four (24) hours per day, 365 days per year, with the exception of the following categories of vehicles: (1) type II vehicles (more than 8,500 lbs. but less than 26,500 lbs., trucks, buses, and unattached trailers); (2) type V vehicles (more than 26,500 lbs., trucks, buses, and unattached trailers; (3) stolen recovery vehicles (vehicles stolen and recovered by SFMTA or SFPD); (4) hit and run vehicles (vehicles involved in hit and run accidents); (5) accident vehicles (vehicles involved in accidents and towed by the SFMTA or the SFPD); and (6) abandoned/missing parts vehicles (vehicles abandoned by owner/driver and missing parts).

(iii) Tows to Secondary Storage. Contractor shall tow all Scofflaw, arrest, accident, Abandoned, recovered stolen vehicles, oversized vehicles, SFPD STOP Administrative Police Hold vehicles, disabled vehicles, dilapidated vehicles and other vehicles as directed by the City directly to the Secondary Storage Facility. The specific policies for whether vehicles are towed to the Primary Storage Facility or directly to the Secondary Storage Facility may be changed at any time subject to City's prior written agreement.

(iv) Facility Management. Contractor agrees to assume all responsibilities for use of storage facilities at 2650 Bayshore Boulevard, Daly City in accordance with the Bayshore License attached as Appendix H to the First Amendment to this Agreement and incorporated by reference herein.

Contractor shall manage the Secondary Storage Facilities, in compliance with the Bayshore License, to meet the following guidelines:

(1) Contractor shall remove vehicles that have been legally cleared for disposal on a weekly basis.

(2) Contractor shall, at a minimum, hold a vehicle lien sale auction once a week.

(3) Vehicles shall be placed in such a way that no more than four (4) vehicles shall need to be moved to clear a passage for any vehicle.

(4) Two (2) feet of clearance space shall be maintained between the sides of all vehicles.

(5) The Secondary Storage Facility personnel shall comply with all municipal, state, and federal codes and safety regulations at all times.

(6) The Secondary Storage Facility shall be clean and maintained at all times.

(7) Facilities shall be screened from public view except for necessary gates.

(8) Gates shall be at least eight (8) feet high and maintained in good condition.

(9) The parking and storage surface shall be maintained in accordance with all requirements of Appendices B and the Bayshore License Agreement.

(10) Security systems, including ample lighting and a surveillance system, shall be in place and operational at all times for the entire area occupied by Contractor under the Bayshore.

(11) Contractor shall not permit the public to walk through the lot unescorted by an employee of Contractor.

(d) Central Dispatch. Contractor's Central Dispatch facility shall operate twenty-four (24) hours per day, 365 days per year. City hereby approves any location of Central Dispatch at the Primary Storage Facility or at Contractor's Headquarters Office, so long as Central Dispatch is located within the City and County of San Francisco.

(e) Changes in Facilities

(i) Approval. City may approve relocation of Designated Facilities, including shifting Contractor's operations between existing Designated Facilities, terminating the use of one or more Designated Facilities, or adding new Designated Facilities. Any such relocation or change to Designated Facilities shall require prior written approval of City.

(ii) Service Standards. In the event that City approves the relocation of any Designated Facility, the parties acknowledge that certain response times and maximum charges contemplated by this Agreement may require modification to take into account the changed geographic circumstances of Contractor's operations. Any written approval of a change to the Designated Facilities listed in this Agreement shall include a revised schedule of fees and/or response times to which the parties have agreed as part of the relocation, if necessary.

(iii) Consolidation. Contractor has relocated the Customer Service Center and Primary Storage Facility to a single location at the 450 7th Street site. Should City cease to use Contractor for towing services prior to February 28, 2015, the terms for City's continued use of the 450 7th Street property are set forth in Appendix G.

u. Section 11.2

Section 11.2 of Appendix A to the Agreement is amended in its entirety to read as follows:

11.2 **Property Maintenance Requirements.** All costs associated with maintenance of Designated Facilities shall be the sole responsibility of the Contractor.

All open areas within the Designated Facilities used for vehicle storage shall be maintained in a clean, secure, neat, and visually presentable manner. Contractor shall not dismantle or crush

May 1, 2014

14 of 18

vehicles or remove vehicle fluids on the any of the facilities to be used in the performance of this Agreement except in compliance with environmental regulations and the applicable requirements of Appendices B and the Bayshore License Agreement. Any removal of fluids from vehicles shall be conducted in a manner that complies with all requirements of this Agreement, and may only be performed by a licensed contractor, and into portable containers that are immediately removed from the facility.

v. Section 12.1

Section 12.1 of Appendix A to the Agreement is amended in its entirety to read as follows:

12.1 Payments Due to City

(a) Referral Fee

Contractor shall submit to the City a Referral Fee of \$20 per tow, excluding dropped tows as described in Appendix A, Section 4.2. The Referral Fee shall be the same for every type of vehicle, and shall increase each twelve (12) month period on each July 1st by the Consumer Price Index for the San Francisco Region as published by the United States Department of Labor, Bureau of Labor Statistics on January 1. Adjustments will be rounded to the nearest twenty-five cents (\$0.25). No Referral Fee shall be paid for:

- (1) Vehicles owned by the City under the jurisdiction of the SFMTA or the SFPD, or any other Courtesy Tow performed pursuant to Section 2.4 of this Appendix A, and
- (2) Vehicles for which a waiver of towing, storage, transfer and/or lien fees is issued by SFMTA, DPH or SFPD.

(b) Percentage Fee

Contractor shall submit to the City a percentage fee of one percent (1%) on annual Gross Revenues from all money collected during the term of this Agreement. This fee shall be initially paid in the fifteenth (15th) month after the Agreement is signed, in the thirteenth (13th) month thereafter for the previous twelve (12) month term, and yearly thereafter.

(c) SFMTA Administrative Fees

Prior to releasing the vehicle, Contractor shall collect a pass-through SFMTA Administrative Towing Fee for all vehicles recovered by the vehicle owner. The amount of the SFMTA Administrative Towing Fee is subject to change in accordance with the provisions of San Francisco Transportation Code § 305. In addition, prior to releasing the vehicle, Contractor shall collect a pass-through daily SFMTA Administrative Storage Fee for stored vehicles based upon the number of days the vehicle has been stored prior to recovery by the vehicle owner. No Administrative Fee shall be collected for:

- (1) Any vehicle owned by the City under the jurisdiction of the SFMTA or the SFPD, or any other Courtesy Tow performed pursuant to Section 2.4 of this Appendix A.
- (2) Any vehicle for which the Customer produces a written waiver of the SFMTA Administrative Fee issued by SFMTA, DPH or the SFPD.
- SFPD Traffic Offender Fee 15 of 18

May 1, 2014

(**d**)

If applicable, when a vehicle is sold at a lien sale and there are funds to satisfy all other fees as defined in Section 12.3(c) of this Appendix A, then Contractor shall pay to the SFPD or into an account designated by the SFPD an SFPD Traffic Offender Fee in an amount set by the San Francisco Police Commission. City and Contractor may agree in writing to utilize a mechanism other than the process described in Section 12.1(g) for collection of the SFPD Traffic Offender Fee.

(e) Citation Fees

Contractor shall collect payments of Citation fees from Customers with towed vehicles and from members of the public whose vehicles have not been towed, in accordance with all requirements set forth in this Agreement.

(f) Liquidated Damages and Fines

Contractor shall pay to City the amounts of any liquidated damages or fines assessed pursuant to this Appendix A, Section 15 and, Section 6.6 of the Bayshore License Agreement.

(g) Deposit of Fees Due to City

Except as otherwise specified herein, Contractor shall deposit all funds collected under this Section 12 within twenty-four (24) hours of receipt into an account specified by the City, Monday through Friday, not including weekends and holidays. Any funds with a deadline for deposit which falls on a weekend or a holiday shall be deposited no later than the next business day. All funds due to City under this Section shall be paid by Contractor without prior demand by the City and without any deduction, setoff, or counterclaim whatsoever, except as expressly provided herein. The parties may agree upon alternative procedures for Contractor's payment to City, but any such change must be approved in advance, by City, in writing.

(h) Payment Shortages

If Contractor fails to collect all amounts due from a Customer, Contractor shall be responsible to reimburse the City for any amounts not collected as required herein, unless the failure is caused solely by the negligence of City or a failure of the CMS. Contractor shall follow any procedures required by the City to report overages or shortages.

w. Section 14.2

Section 14.2 of Appendix A to the Agreement is amended in its entirety to read as follows:

14.2 Approval Process. All elements of the Operations Plan shall be subject to City review and approval. All Operations Plan elements must be initially submitted no later than the deadlines set forth in Appendix B for each Operations Plan element. The deadline for any Operations Plan element described in the Agreement may be extended by written approval of SFMTA upon the request of and a showing of good cause by Contractor. City shall have sixty (60) days to review each element submitted, and either approve it as submitted or request revisions. Contractor shall respond to a request for revisions within twenty (20) days. City will have fifteen (15) days to either approve the revised Plan element or request further revisions. Contractor and City shall from this point on have five (5) days to either approve the revised Plan element as submitted, submit further requests for revisions or to respond to requests for revisions. Each revision must reflect tracking of document versions, including date and source of revisions, and each exchange of versions between the parties shall be accompanied by an executed document substantially in the form of Appendix B.

n:\ptc\as2013\1000468\00891055.doc

- x. Appendix C to the Agreement is deleted in its entirety.
- y. Appendix D to the Agreement is deleted in its entirety and is replaced by Appendix H, which is attached to this Amendment.

- **3.** Effective Date. Each of the modifications set forth in Section 2 shall be effective on and after the date of this Amendment.
- 4. Legal Effect. Except as expressly modified by this Amendment, all of the terms and conditions of the Agreement shall remain unchanged and in full force and effect.

IN WITNESS WHEREOF, Contractor and City have executed this Amendment as of the date first referenced above.

CITY	CONTRACTOR
San Francisco Municipal Transportation Agency	TEGSCO, LLC, d.b.a.\San Francisco AutoReturn
Edward D. Reiskin Director of Transportation	JOHN WICKER President and CEO 945 Bryant Street, Suite 350 San Francisco, CA 94103
Authorized by:	
Municipal Transportation Agency Board of Directors Resolution No: 14-050 Adopted: 412014 Attest: 2014 Roberta Boomer, Secretary SFMTA Board of Directors	
Approved as to Form:	
Dennis J. Herrera City Attorney	
By: Mariam M. Morley Deputy City Attorney	

APPENDIX A: SCOPE OF WORK

TABLE OF CONTENTS

1.	точ	EQUIPMENT DISPATCH REQUIREMENTS	1		
	1.1	HOURS OF SERVICE			
	1.2	CENTRAL DISPATCH			
	1.3	COMMUNICATIONS EQUIPMENT			
		(a) Equipment Requirements	1		
		(b) Audio Recordings	1		
		(c) Dispatch Tow Car Operators	1		
2.	TOW REQUEST RESPONSE TIME REQUIREMENTS1				
	2.1	ROUTINE REQUESTS	1		
		(a) Scheduled and Dispatch Tows	1		
		(b) Abandoned Vehicle Tows	2		
	2.2	EXPEDITED TOW RESPONSE TIME REQUIREMENT	2		
	2.3	REGIONAL SWEEPS	2		
	2.4	COURTESY TOWS AND ROADSIDE SERVICE	2		
	2.5	FAILURE TO RESPOND	2		
	2.6	TOW SERVICE PLAN			
3.	VEH	VEHICLE INTAKE REQUIREMENTS			
	3.1	-			
	3.2	VALUATION			
	3.3				
		(a) VIN	3		
		(b) Personal Property	3		
	3.4	DIGITAL PHOTO RECORDING	3		
4.	VEHICLE HANDLING REQUIREMENTS 4				
	4.1	IMPROPER DISPOSAL OF VEHICLES			
	4.2	VEHICLE DROP			
	4.3	PERSONAL PROPERTY RELEASES			
		(a) Standard for Release	4		
		(b) Vehicles Subject to Hold	4		
		(c) Personal Property Release	4		
	4.4	RETRIEVAL REQUIREMENTS FROM SECONDARY TO PRIMARY STORAGE FACILITY	5		
	4.5	No ID Vehicles	5		
		(a) NO ID Designation	5		
		(b) NO ID Procedure	3		
		(c) SFPD Access to NO ID Vehicles	5		
		APPENDIX A: SCOPE OF WORK n:\ptc1\as2010\0900371\00630742.doc			

	4.6	PROG	CEDURES FOR VEHICLES IMPOUNDED BY THE SFPD
		(a)	Investigative Police Hold Vehicles
		(b)	Release
		(c)	Recovery of Stolen Vehicles
		(d)	Reporting
			(i) Reports Required
			(ii) Police Hold Storage Charges
	4.7	ADM	INISTRATIVE HOLD PROCEDURES
		(a)	Administrative Hold
		(b)	Release Restrictions
		(c)	Administrative Hold Reports
5.	CUS	гоме	R SERVICE
	5.1	CUST	TOMER INTAKE AND PROCESSING
		(a)	Telephone System 7
		(b)	Live Operators
		(c)	In-Person Customer Service
			(i) Facility
			(ii) Staffing
			(iii) Customer Service Representative
			(iv) Self-Service
			(v) Reports
		(d)	Customer Service via Internet
		(e)	Customer Service Manual
	5.2	Сом	PLAINTS AND CLAIMS
		(a)	Claims Procedure
		(b)	Complaint Procedures
		(c)	Claims/Complaint Status Reports
		(d)	Claims/Complaint Processing Plan
	5.3	DISSI	EMINATION OF INFORMATION TO THE PUBLIC10
•		(a)	Posted Information
		(b)	Translation of Posted Information
6.	PRO	CEDU	RES FOR VEHICLE RECOVERY 11
	6.1	FORM	M OF PAYMENT
		(a)	In-Person Payments
		(b)	Phone and Internet Payments
	6.2	Coli	LECTION OF PARKING CITATION PAYMENTS AND RELATED FEES; DEPOSIT
		(a)	Citation Payments
		(b)	Outstanding Fees

		(c)	Collection Fees for Non-Towed Vehicle Citations		
	6.3	INTE			
	6.4	INTERNET SERVICES			
	6.5	VEH	ICLE RECOVERY		
		(a)	Vehicle Location	13	
		<i>(b)</i>	Vehicle Release Procedure		
		(c)	Vehicles Not Subject to Release	14	
	6.6	FORM	M OF RECEIPT	14	
	6.7	DISC	LOSURE OF DEFICIENCY CLAIM AT TIME OF VEHICLE ABANDONMENT		
	6.8	INVALID PAYMENTS			
7.	PRO	PROCEDURES FOR LIEN SALES14			
	7.1	NOT	IFICATION TO REGISTERED OWNER	14	
	7.2	LIEN	SALES		
		(a)	Lien Sale Procedures		
		(b)	Pre-Registration		
	7.3	Post	Γ-LIEN SALE PROCEDURES	16	
		(a)	Driver's License and Vehicle Registration		
		<i>(b)</i>	Deficiency Claims		
	7.4	DISP	OSAL OF UNSOLD VEHICLES	16	
8.	STAFFING REQUIREMENTS1				
	8.1	ADEQUATE STAFFING			
	8.2	SUBC	CONTRACTORS	17	
		(a)	Subcontractor Designation		
		(b)	Subcontractor Compensation		
		(c)	Subcontractor Performance Standards		
		(d)	Subcontractor Compliance with Licenses and Permits		
		(e)	Subcontractor Identifying Equipment and Uniforms		
	8.3	TRAINING			
	8.4	POLICY AND PROCEDURE MANUAL			
	8.5	STAFFING PLAN			
	8.6	MANAGEMENT CHANGES			
	8.7	ON-CALL MANAGER			
	8.8	CON	TRACT MONITOR		
		(a)	Contract Monitor Payment Fund		
		<i>(b)</i>	Duties of Contract Monitor		
	8.9	VEHI	ICLE AUCTIONEER		
		(a) ·	Auctioneer Staffing Plan		
			ADDENIDIY AS SCODE OF WORK		

• •

	11.3	FACILITY SECURITY	
	11.2	PROPERTY MAINTENANCE REQUIREMENTS	
		(iii) Consolidation	
		(ii) Service Standards	
		(i) Approval	
		(e) Changes in Facilities	
		(d) Central Dispatch	
		(iv) Facility Management	
		(iii) Tows to Secondary Storage	
		(i) Vehicle Storage and Retrieval	
		(c) Secondary Storage Facility (i) Authorized Facility	
		(ii) Vehicle Storage and Retrieval	
		(i) Authorized Facility	
		(b) Primary Storage Facility	
	٠	(a) Customer Service Center	30
	11.1	DESIGNATED FACILITIES	
11.	FAC]	ILITIES REQUIREMENTS	
	10.5	HANDHELD DEVICES	
	10.4	TVMS FUNCTIONALITY	
	10.3	CITY ACCESS TO TVMS	
	10.2	NETWORK CONNECTIONS	
		(b) Router Configuration	
		(a) Firewall Configuration	
	10.1	COMPUTER HARDWARE AND SOFTWARE	
10.	той	ED VEHICLE MANAGEMENT SYSTEM	
	9.6	ELECTRONIC TOW INVENTORY SLIPS	
	9.5	VEHICLE TOW RECORDS AND CMS	
	9.4	TOW CAR DATA TERMINAL	
	9.3	TWO-WAY RADIO COMMUNICATIONS EQUIPMENT	
		(b) GPS Software	
		(a) GPS Equipment	
	9.2	GPS TRACKING SYSTEMS FOR TOW CARS	21
	9.1	Tow Cars	
9.	EQU	IPMENT AND INFORMATION SERVICES REQUIREMENTS	
		(c) Vehicle Sales to Certain Individuals Prohibited	20
	•	(b) Information Provided to Auctioneer	

	11.4	PRO	TECTION OF VEHICLE AND CONTENTS	
	11.5	LIEN	VEHICLE STORAGE	
	11.6	RIGI	HT TO INSPECT	
12.	FEES,	PAY	MENTS AND CREDITS	
	12.1	PAY	MENTS DUE TO CITY	
		(a)	Referral Fee	
		(b)	Percentage Fee	
		(c)	SFMTA Administrative Fees	
		(d)	SFPD Traffic Offender Fee	
		(e)	Citation Fees	
		(f)	Liquidated Damages and Fines	
		(g)	Deposit of Fees Due to City	
		(h)	Payment Shortages	
	12.2	CREI	DITS DUE TO CONTRACTOR	
		(a)	City Waivers	
		(b)	Police Investigative Hold Storage Fee	
		(c)	Non-Towed Vehicle Citation Collection Fees	
		(d)	Other Offset Allowances	
	12.3	CHARGES TO CUSTOMERS		
		(a)	Maximum Towing and Storage Charges	
		(b)	Vehicle Transfer Fee	
		(c)	Application of Funds Collected	
	12.4	ADD	ITIONAL FEES PROPOSED BY CONTRACTOR	
	12.5	ACCOUNT RECONCILIATION		
	12.6	CON	TRACTOR COMPLIANCE AUDIT FUND	
		(a)	Contractor Contribution	
		(b)	Contribution Adjustment	
13.	REPO	RŤIN	NG AND RECORDS REQUIREMENTS	
	13.1		18 Records	
		(a)	Records of Transaction	
		(b)	Weekly Management Report	
		(c)	Monthly Management Reports	
		(d)	Auction Report	
	13.2		IMUM REQUIRED REPORTING	
		(a)	Daily Reports	
	•	<i>(b)</i>	Weekly Reports	
		(c)	Monthly Reports	

		(d) Quarterly Reports
		(e) Additional Reports
	13.3	RECORDS MAINTENANCE
14.	OPER	ATIONS PLAN
	14.1	GENERAL PROVISIONS
	14.2	APPROVAL PROCESS
	14.3	LINE ITEM APPROVAL
	14.4	FINAL OPERATIONS PLAN
	14.5	SUBSEQUENT MODIFICATIONS TO OPERATIONS PLAN
15.	LIQU	IDATED DAMAGES
	15.1	ASSESSMENT OF LIQUIDATED DAMAGES
	15.2	DAMAGES CALCULATION
	15.3	STAFFING
	15.4	SUBCONTRACTING
	15.5	EQUIPMENT
	15.6	RESPONSE TIMES
	15.7	RECORD KEEPING AND REPORTING REQUIREMENT'S
	15.8	PLAN SUBMITTALS
	15.9	CUSTOMER SERVICE STANDARDS
	15.10	VEHICLE HANDLING REQUIREMENTS
	15.11	FINANCIAL OBLIGATIONS

APPENDIX A

SCOPE OF WORK

1. Tow Equipment Dispatch Requirements

1.1 Hours of Service

Contractor shall respond to all Tow Requests and shall intake and release towed vehicles twenty-four (24) hours per day, 365 days per year in accordance with the standards specified in this Agreement.

1.2 Central Dispatch

Contractor shall dispatch Tow Equipment from its Central Dispatch facility. Contractor shall provide dispatch staff to receive Tow Requests twenty-four (24) hours per day, 365 days per year. During Peak Towing Hours, a dispatch supervisor must be on duty to direct staff and address any issues or escalations raised by the City that relate to a Tow Request. At all times, Contractor must ensure that:

- (1) Tow Request phone calls must be answered within thirty seconds or less with no busy signal for 95% of the calls and within 90 seconds for the remaining 5% of the calls during a given calendar month; and
- (2) Tow Requests must be assigned to a subcontractor towing firm within five (5) minutes or directly to a Tow Car operator within ten (10) minutes.

1.3 Communications Equipment

(a) Equipment Requirements

Within thirty (30) days of the Effective Date of this Agreement, Contractor shall provide a dedicated telephone line to be used exclusively for communication between Central Dispatch and the City, and a minimum of two (2) two-way radios or equivalent equipment to be used as an alternative communications method.

(b) Audio Recordings

Contractor shall provide equipment that will create an audio recording with a time/date stamp of all Tow Requests communications by telephone between SFMTA and the Tow Desk and Contractor's Central Dispatch Facility. Contractor shall maintain such audio recordings for a minimum of 120 days. Upon City's request, Contractor shall provide City access to the audio recordings within twenty-four (24) hours for the purpose of reviewing or copying the recordings.

(c) Dispatch Tow Car Operators

Communication between Central Dispatch and towing subcontractors shall be by means of landline communications, radios, mobile telephones, or electronic two-way data terminals. Any other method of communication to be used between Central Dispatch and Tow Car operators must be approved by the City in writing.

2. Tow Request Response Time Requirements

2.1 Routine Requests

(a) Scheduled and Dispatch Tows

Upon receiving a Tow Request for a Scheduled Tow or Dispatch Tow Contractor shall be on site with the appropriate Tow Equipment for the vehicle type to be towed, or with Tow Equipment specifically requested by City in the Tow Request, within thirty-five (35) minutes of a Dispatch Tow Request during Peak Towing Hours and within twenty-five (25) minutes at all other times, or within ten (10) minutes of the time designated for initiation of a Scheduled Tow. Contractor may notify City that it is requesting an extension of time of up to fifteen (15) minutes to process a Scheduled or Dispatch Tow request for reasons beyond the Contractor's control, which are defined as traffic congestion or travel distance, and the City shall inform Contractor if the request for extension of time is denied. Contractor shall respond within the applicable time limit for at least 94% of the Dispatch Tows performed in a given calendar month.

(b) Abandoned Vehicle Tows

Contractor must respond to each Abandoned Vehicle Tow Request within twenty-four (24) hours unless otherwise specified in the Tow Request.

2.2 Expedited Tow Response Time Requirement

City may specify that an Expedited Tow is required in the Tow Request. During Peak Towing Hours, Contractor shall respond to a Tow Request for an Expedited Tow within thirty (30) minutes of the Tow Request. Contractor may notify City that it is requesting an extension of time of up to fifteen (15) minutes to process an Expedited Tow request for reasons beyond the Contractor's control, which are defined as traffic congestion or travel distance, and the City shall inform Contractor if the request for extension of time shall be denied . Contractor shall respond within the applicable time limit for 100% of the Expedited Tows performed in a given calendar month.

2.3 Regional Sweeps

Contractor shall participate in Regional Sweeps requested by SFMTA or SFPD as a part of its regular towing services. SFMTA or SFPD shall notify Contractor at least forty-eight (48) hours in advance (with seventy-two (72) hours advance written notice provided when possible) of the date for a Regional Sweep, and shall inform Contractor of the number of Tow Cars required, the location, and the time that tows are to begin. City reserves the right to modify and/or expand Regional Sweep programs, and shall notify Contractor of these modifications and/or expansions no later than forty-eight (48) hours in advance of the date of a Regional Sweep. Regional Sweeps are subject to the performance standards of Scheduled Tows under this Agreement.

2.4 Courtesy Tows and Roadside Service

At the request of City, Contractor shall remove any disabled SFMTA or SFPD vehicle or render timely roadside assistance, including but not limited to starting vehicles, delivering gasoline and changing flat tires. These requests for towing or roadside assistance shall be provided within thirty-five (35) minutes of City's request. Contractor may notify City that it is requesting an extension of time of up to fifteen (15) minutes to process a Courtesy Tow request for reasons beyond the Contractor's control, which are defined as traffic congestion or travel distance, and the City shall inform Contractor if the request for extension of time shall be denied. The costs for these services shall not be charged to City, nor may Contractor pass the cost of these services to its subcontractors. Should Contractor fail to respond to a request in accordance with this Section, City may elect to acquire the services from another source and Contractor shall be responsible for City's direct and indirect costs in acquiring the requested services from another source.

2.5 Failure to Respond

In the event that Contractor fails to respond and/or fails to furnish necessary Tow Equipment at the designated point of tow within the time periods specified in this Agreement, the City shall have the right, by whatever means appropriate, to remove or cause the removal of the vehicle which was the subject of the Tow Request and transport it to the Contractor's storage facility, whereupon Contractor shall process and handle such vehicle in accordance with all requirements of this Agreement. In such event, Contractor shall be responsible for reimbursing City's direct and indirect costs for removing the vehicle, excluding subsequent damage caused by the alternate towing service provider.

2.6 Tow Service Plan

Contractor shall submit a Tow Service Plan to the City describing how it will deploy its subcontractors in sufficient detail to allow City to determine that Contractor will meet service requirements specified in this Agreement. This plan shall be approved and adopted as part of the Operations Plan, as provided in Section 14 of this Appendix A.

3. Vehicle Intake Requirements

3.1 Tow Data

Contractor shall ensure that a record of each Dispatch Tow is created in ADS by means of a Tow Car data terminal or by computer at Central Dispatch within five (5) minutes of receiving the tow request and the record must be created in TVMS within fifteen (15) minutes of completion of the tow. A record of each Scheduled Tow shall be created in TVMS within fifteen (15) minutes of completion of the tow. In addition, within fifteen (15) minutes of a towed vehicle at a Designated Facility, Contractor shall record, at a minimum, the following data:

- (1) the time the initial call for service was requested
- (2) the time the Tow Car arrived on the scene of the tow
- (3) the time the Tow Car arrived at the vehicle intake facility.
- (4) Contractor shall meet these timing requirements in ninety percent (90%) of all tows within any reasonable audit period identified by City.

3.2 Valuation

Contractor shall have each impounded vehicle assigned a Lien Category (Lien 1, Lien 2 or Lien 3 Vehicle) by trained personnel as soon as possible during vehicle intake, but in no event later than seventy-two (72) hours after the vehicle's initial arrival at a Designated Facility. Contractor shall assign a value to each vehicle under penalty of perjury as required by California Vehicle Code Section 22670(b). After Contractor initiates the lien process for a vehicle with the DMV, Contractor's valuation and classification of the vehicle under this Section 3.2 shall be subject to later adjustment only as directed by, or with the approval of, the City's designated Contract Monitor in compliance with the Vehicle Valuation Standards Plan as defined in Section 8.8(b). In no event shall any impounded vehicle be pulled from a pending auction for the purpose of reclassifying the vehicle after the auction has begun.

3.3 Inspection

(a) VIN

Contractor shall visually inspect any vehicle for which a lien is requested to confirm and record the VIN of each vehicle, and include the VIN in its lien request to the DMV within seven (7) days. Contractor shall notify the SFPD within twenty-four (24) hours of becoming aware of any vehicle in its possession for which the license plate and the VIN do not match. Exceptions to the deadline for collecting the VIN include:

- (1) Vehicles subject to SFPD investigative holds.
- (2) Vehicles identified as NO ID vehicles pursuant to Section 4.5 of this Appendix A which require SFPD inspection.
- (3) Other cases of extenuating circumstances as approved by the City.

(b) Personal Property

During intake inspection, personal property in the vehicle of more than nominal value that is visible from the exterior without opening any locked compartment shall be inventoried, and as part of the inventory Contractor shall record whether or not the vehicle has a locked storage compartment. This information shall be recorded in the TVMS system and/or the vehicle inventory forms that are kept on file at the Primary Storage Facility. Contractor, SFMTA, DPH and SFPD shall endeavor to keep the vehicle locked to the maximum extent possible during the towing and storage process. Contractor may remove and separately store personal property from the vehicle for security or other reasons as necessary.

3.4 Digital Photo Recording

Contractor shall have cameras at all vehicle intake facilities and shall take photos of all four (4) sides of the exterior of each vehicle the first time that it is brought into any Designated Facility, except for over-sized

vehicles that cannot be brought into the Secondary Storage Facility through the main entrance and must enter the facility from the secondary entrance. These images shall be stored electronically and in a manner that allows prompt retrieval within one (1) working day of any City request. City agrees that the requirements of this Section 3.4 may be satisfied by extracting still images from continuous video footage.

4. Vehicle Handling Requirements

4.1 Improper Disposal of Vehicles

If, in violation of applicable law or this Agreement, Contractor releases, sells, disposes of, or otherwise loses possession of or is unable to locate within its possession any vehicle that it has towed or impounded under this Agreement, notwithstanding any other criminal or civil penalties levied by a court of law, Contractor shall have sixty (60) days to resolve any Claim filed by the vehicle owner, and must submit any proposed Claim settlement to City for approval prior to finalizing such Claim. If Contractor fails to satisfactorily resolve such Claim within sixty (60) days, City may require Contractor to pay to the vehicle owner the blue book value of the vehicle at the time of sale, destruction or loss of the vehicle as determined by the City. If City directs Contractor to resolve a Claim after sixty (60) days by paying the blue book value of a vehicle to the owner, City may assess liquidated damages against Contractor in accordance with Section 15 of this Appendix A. This contractual remedy shall not preclude the vehicle owner from taking any other appropriate legal action against the Contractor.

4.2 Vehicle Drop

The Tow Car operator shall release a vehicle without assessing any towing charges upon request of the SFMTA or SFPD officer or DPH employee present at the site of the tow. The City officer or employee may require release of the vehicle pursuant to this Section 4.2 when the owner or operator of the vehicle is present to claim the car before all of the following three (3) events occur:

- (1) The towing apparatus is completely attached to the vehicle in a manner consistent with the proper use of the Tow Car equipment and in accordance with industry standards, and
- (2) All required data has been collected, and
- (3) The Tow Car has been put in gear by the operator and has started to drive away.

The City may assess liquidated damages against Contractor pursuant to Section 15 of this Appendix A for failure to release a vehicle as required by this Section 4.2. In addition, the Contractor shall reimburse the owner or operator of the vehicle towed in violation of this Section all expenses incurred by the owner or operator to recover the vehicle plus \$100 per occurrence.

4.3 Personal Property Releases

(a) Standard for Release

Contractor shall only release personal property found within any vehicle in its custody when it is satisfied that the claimant is entitled to access the vehicle and its contents and the vehicle is not subject to a Police or Administrative Hold.

(b) Vehicles Subject to Hold

No person shall be allowed to retrieve personal property from a vehicle subject to an Investigative or Administrative Police Hold without the prior written authorization of the SFPD.

(c) Personal Property Release

A personal property release allows the claimant to enter the vehicle, with the supervision of Contractor, to obtain property from the towed vehicle. Contractor's supervision shall include preparing a written inventory of the items removed by the claimant, but Contractor shall have no responsibility for assisting the claimant to remove personal property from the vehicle. Contractor shall require the claimant to sign the inventory statement listing the item(s) they removed from the vehicle, and shall file the personal property release data collected in the TVMS system or in a paper file with a file name cross referencing the Tow Request ID in the TVMS.

4.4 Retrieval Requirements from Secondary to Primary Storage Facility

If the Customer's vehicle is stored at the Secondary Storage Facility, Contractor shall provide either a free shuttle service enabling a Customer who appears at the Primary Storage Facility to retrieve a vehicle directly from the Secondary Storage Facility, one-way taxi fare for the Customer between the Primary Storage Facility and the Secondary Storage Facility, or free Tow Car service to tow the vehicle back to the Primary Storage Facility. In all cases, the vehicle must be released to the Customer within one (1) hour of the Customer's payment of fees at the Customer Service Center, with the exception of the categories of vehicles listed in Section 11.1(c)(ii).

4.5 No ID Vehicles

(a) NO ID Designation

All vehicles with no visible VIN shall be impounded under a "NO ID" number and shall be designated as a vehicle subject to Investigative Police Hold and held for inspection by the SFPD Auto Detail regardless of which City agency initiated the Tow Request. NO ID vehicles shall be included in regular reports to the SFPD of Police Hold vehicles as specified in Sections 4.6(d) and 13 of this Appendix A. Contractor shall keep NO ID vehicles within the NO ID area (excluding oversized vehicles), with the exception that Contractor shall move a NO ID vehicle out of the NO ID area within twenty-four (24) hours of receiving a request to do so by the SFPD.

(b) NO ID Procedure

If a VIN is found following inspection by the SFPD, Contractor shall follow applicable lien sale provisions of the Vehicle Code for processing that vehicle. Otherwise, Contractor shall designate the vehicle as an "Unable to Identify" or "UTID" vehicle and, after receipt of a written release by the SFPD (DMV Form 462, "Public Agency Authorization of Disposal of Vehicle" or successor form), such UTID vehicle shall be disposed of as required by the Vehicle Code and in accordance with instructions on DMV Form 462 or successor form.

(c) SFPD Access to NO ID Vehicles

Contractor shall allow SFPD personnel with written authorization from the Chief of Police or an officer designated by the Chief of Police to remove parts from any NO ID vehicle, except as prohibited by Appendices B, C and D.

4.6 **Procedures for Vehicles Impounded by the SFPD**

(a) Investigative Police Hold Vehicles

The SFPD may designate any vehicle for which it has made a Tow Request as an Investigative Police Hold vehicle. Investigative Police Hold vehicles shall be stored in a segregated, secure area, which may be located in a Designated Facility. Within ninety (90) days of the Effective Date of this Agreement, Contractor shall provide secured and fenced Investigative Police Hold storage locations at the Designated Facilities which shall include the following features:

- (1) A vehicle inspection area with six (6) indoor bays that contain car lifts, air compressors, and access to 220V electricity at the Primary Storage Facility.
- (2) An office area with two (2) phone lines dedicated for use by SFPD personnel (upon request of the SFPD) at the Primary Storage Facility.
- (3) Indoor space for at least twenty-five (25) NO ID vehicles; and
- (4) Secured indoor space for at least 100 Investigative Police Hold vehicles; and
- (5) Secured outdoor space for at least 175 Investigative Police Hold vehicles; and
- (6) Contractor shall provide a covered storage space that holds at least fifty (50) vehicles secured for SFPD investigations at the Primary Storage Facility that meets with the approval of the SFPD. The SFPD has approved the following arrangement: Storage space for twelve (12) vehicles is located behind a secure fence under the freeway between columns R58 and R56. Additional space that can accommodate at least 38

vehicles is provided by a secure fence that will run from in front of the 6 bays to the carwash building, and from the carwash building to the fence line across from R56. Any change to this configuration must be approved, in advance and in writing, by the SFPD.

No person shall be allowed access to an Investigative Police Hold vehicle or retrieve personal property from such vehicle without written authorization from the SFPD. If the SFPD designates an Investigative Police Hold vehicle as an evidentiary vehicle at the time of the Tow Request, Contractor shall ensure that the towing and storage of the vehicle is conducted in accordance with any standards for handling and preservation of evidence provided to Contractor by the SFPD in writing. Contractor shall maintain the Investigative Police Hold areas in a manner which ensures its ability to locate vehicles requested by SFPD within one (1) hour of SFPD's request. Contractor may, from time to time, request training for Tow Car operators and Employees for the handling of evidentiary vehicles from the SFPD. Contractor shall submit Police Department Procedures to the City describing in detail how it will process Investigative Police Hold vehicles to meet service requirements specified in this Agreement. These procedures shall be approved and adopted as part of the Operations Plan, as provided in Section 14 of this Appendix A.

(b) Release

Contractor shall not release or allow parts to be removed from any vehicle towed by the SFPD without a written release authorization from the SFPD. The SFPD will provide to Contractor a standard form to be used for all written release authorizations and a list of individuals authorized to sign vehicle and personal property releases, including exemplars of those individuals' signatures. Contractor may not release an Investigative Police Hold vehicle without a release form signed by an authorized individual as designated by the SFPD. Contractor shall inform the Customer that release of a vehicle subject to Police Hold may only be obtained by going to a SFPD station to request that the vehicle be released.

(c) Recovery of Stolen Vehicles

Recovered stolen vehicles may only be released upon presentation of SFPD Form 425 or as otherwise specified by SFPD in writing. Contractor shall cooperate with City in the coordination of electronic information between DMV and City, between City agencies, and between Contractor and City for the purpose of early identification of stolen vehicles and prompt notification of the owner. Procedures for waiver and reimbursement of towing and storage charges for stolen vehicles are described in Section 12.2(a) of this Appendix A.

(d) Reporting

(i) Reports Required

Contractor shall submit to the SFPD a weekly report listing all Investigative Police Hold vehicles that are currently being stored by Contractor in accordance with the requirements of Section 13 of this Appendix A, delivered to the person designated by the SFPD as the inspector in charge of auto investigative holds. Contractor shall issue reports to SFPD personnel designated by the Chief of Police in writing, the Director of SFMTA or her or his designee, and the SFMTA Contract Administrator upon occurrence of the following events:

- (1) A "300 vehicle warning" notice on each day that the number of Investigative Police Hold vehicles stored by Contractor exceeds 300 vehicles; and
- (2) An inventory report of Investigative Police Hold vehicles on each day that the number of Investigative Police Hold vehicles stored by Contractor exceeds 325 vehicles.

(ii) Police Hold Storage Charges

On each day or portion of a day that the Contractor has more than 350 Investigative Police Hold vehicles in storage, Contractor may charge the SFPD a daily Investigative Police Hold storage fee per vehicle, as further described in Section 12.2(b) of this Appendix A.

4.7 Administrative Hold Procedures

Page 6 of 45

(a) Administrative Hold

SFMTA or the SFPD may designate a vehicle as an Administrative Hold vehicle. Contractor shall identify and track Administrative Police Holds as either "STOP" holds or "Traffic Administration" holds. Vehicles subject to Administrative Hold by the SFPD or the SFMTA shall not be processed or otherwise treated as Investigative Police Hold vehicles.

(b) Release Restrictions

Administrative Hold vehicles that are impounded in accordance with the provisions of the Vehicle Code, including Administrative Holds resulting from the SFPD's STOP Program, shall not be released until Contractor receives written authorization for the release by the SFPD. Contractor may proceed with the lien sale of the vehicle in accordance with all applicable lien sale requirements, without written SFPD release authorization. SFPD STOP Administrative Hold vehicles must be held for at least thirty (30) days prior to lien sale.

(c) Administrative Hold Reports

Within twenty-four (24) hours of a request from SFMTA or SFPD, Contractor shall produce a report of SFMTA and SFPD Administrative Holds that lists the City employee and department requesting the hold for each vehicle subject to Administrative Hold.

5. Customer Service

Contractor shall interact with Customers who contact Contractor for the purpose of retrieving towed vehicles in person, by phone using an interactive telephone answering system ("IVR") and live operators, and by internet. When a Customer makes an inquiry of Contractor by any means of communication, Contractor shall provide the Customer with accurate and timely information regarding their rights with respect to the towed vehicle under this Agreement and all applicable federal, state and local laws and regulations. Contractor shall at a minimum implement the Customer service activities described in this Agreement. All materials created by Contractor which are intended for use by Customers, whether in written, electronic or audio format, shall be made available in Spanish, spoken Cantonese and written Chinese. Contractor shall also make best efforts to provide bilingual staff to assist Customers in the Mandarin, Russian and Vietnamese languages.

5.1 Customer Intake and Processing

(a) Telephone System

Contractor shall establish and maintain one (1) customer service phone line that the City or the public may call for information regarding towed vehicles. This phone line shall be independent of the phone line used for requests for Dispatch Tows. Contractor shall make information available to City and Customers through both an automated IVR, as further described in Section 6.3 of this Appendix A, and through live operators as described in Section 5.1(b) below. The automated IVR and call distribution system to live operators must have sufficient lines, instruments, hardware, software, and overflow safeguards to meet the service requirements of this Agreement.

(b) Live Operators

Contractor's telephone operators must be available to respond to calls from the public twenty-four (24) hours per day, 365 days per year. During each monthly reporting period, at least ninety-five percent (95%) of all customer service calls received during Peak Service Hours must reach a live operator within three (3) minutes of request; of the remaining five percent (5%) of Peak Service Hour customer service calls, at least four percent (4%) must reach a live operator within ten (10) minutes of request. At all other times, at least eighty percent (80%) of all customer service calls must reach a live operator within three (3) minutes of request; of the remaining twenty percent (20%) of customer service calls, at least nineteen percent (19%) must reach a live operator within ten (10) minutes of request (19%) must reach a live operator within ten (10) minutes of request (19%) must reach a live operator within ten (10) minutes of request (19%) must reach a live operator within ten (10) minutes of request (19%) must reach a live operator within ten (10) minutes of request (19%) must reach a live operator within ten (10) minutes of request (19%) must reach a live operator within ten (10) minutes of request (19%) must reach a live operator within ten (10) minutes of request during the monthly reporting period.

(c) In-Person Customer Service

(i) Facility

Contractor shall operate a Customer Service Center that is open to the public twenty-four (24) hours a day, 365 day per year. The Customer Service Center shall have equipment that records the time that each Customer enters the waiting area and how long they wait for service at the service window. Contractor shall store this data and summarize it in the Customer Service reports required by subsection 5.1(c)(v). In all cases, a vehicle must be released to a Customer within one (1) hour of the Customer's completion of all payment and documentation requirements for vehicle release.

(ii) Staffing

During Peak Service Hours, Contractor shall maintain adequate and sufficiently trained staff to simultaneously serve six (6) Customers, including "quick service" window(s), in accordance with the standards set forth in this Agreement. Wait times for ninety-five percent (95%) of all customers served in person shall be no longer than ten (10) minutes.). Contractor shall cross-train all personnel so that window staff are trained to answer telephone calls if they are not assisting a Customer in person and call volumes require additional telephone operators. Telephone operators shall be trained to assist window staff when walk-in wait times exceed ten (10) minutes and windows are available. All Employees of Contractor who have regular, continuous contact with members of the public shall be neat in appearance and courteous to the public.

(iii) Customer Service Representative

Contractor shall provide a Customer Service Representative in addition to the regularly scheduled service window staff and telephone operators. An appropriately trained Customer Service Representative shall be on duty during the hours of 7:00 a.m. through 7:00 p.m., Monday through Friday, excluding City holidays, to assist Customers apart from normal window service, including but not limited to escalated service inquiries, suggestions, Complaints, assistance to disabled Customers and other out-of-the-ordinary Customer needs. The designated Customer Service Representative may assist Customers at a window or assist telephone callers while not occupied with escalated service inquiries, so long as service level requirements for phone and walk-in service as defined in this Section 5 are met.

(iv) Self-Service

Contractor shall provide Customers who use the self-service phone system and the self-service web site access to a special "quick service" window(s) or self-service kiosk(s) for expedited service.

(v) Reports

Contractor shall submit a monthly Customer Service Report that shows window staffing patterns, average wait times for Customers, and number of Customers served by hour.

(d) Customer Service via Internet

Contractor shall create and maintain an internet site that provides general information to the Customer, including the ability to ascertain the status of a specific towed vehicle and to make payments of Mandatory Fees with a credit card. Contractor's internet site shall generally be available twenty-four (24) hours a day, 365 days per year. Internet site maintenance and down-time should be scheduled between the hours of 12:00 a.m. and 6:00 a.m., except as otherwise necessary. City may assess liquidated damages pursuant to Section 15 of this Appendix A for failure to meet these standards; however Contractor may request a waiver of liquidated damages in advance of maintenance that must be performed outside the hours of 12:00 a.m. to 6:00 a.m.. Detailed specifications and requirements for the internet site are set forth in Section 6.4 of this Appendix A.

(e) Customer Service Manual

Within forty-five (45) days of the Effective Date of the Agreement, Contractor shall create a Customer Service Manual for Employees describing the policies and procedures for assisting Customers with vehicles towed under this Agreement. This manual shall be a reference for Contractor staff and subcontractors. The Customer Service Manual shall include, but is not limited to Employee training, guidelines for dissemination of information to the public, specifications for the Customer Service Center, cleanliness and safety standards for all facilities, and procedures for solicitation of feedback from Customers served. The Customer Service Manual shall be approved and adopted as part of the Operations Plan as provided in Section 14 of this Appendix A.

5.2 Complaints and Claims

(a) Claims Procedure

Contractor shall establish a procedure by which persons whose vehicles have been towed and/or stored ("Claimants") may file a Claim against Contractor. Contractor shall respond to all Claims within fourteen (14) days of receipt of Claim, either to accept, deny or request further information for investigation. Contractor shall endeavor to resolve each Claim within ninety (90) days of receipt from Claimant, and shall resolve all Claims within six (6) months unless (i) such Claim is abandoned by the Claimant's failure to respond to Contractor's communication for a period of thirty (30) days, or (ii) the Claim is filed in court. Contractor shall in all cases endeavor to resolve Claims fairly and expeditiously. Contractor shall designate a Claims Manager who shall supervise Contractor's Claims procedures and shall be available during regular business hours to discuss Claims with Claimants in person or by telephone. Contractor shall maintain records of all Claims filed and of all correspondence with Claimants, denials of Claims, settlement offers and amounts paid on Claims.

(b) Complaint Procedures

Contractor shall establish a procedure by which Customers ("Complainants") may submit Complaints about Contractor's performance of the services under this Agreement. Contractor shall make available to Customers a Complaint Form, which shall include a self-addressed pre-paid postage envelope. Contractor's Complaint procedure shall allow Complaints to be submitted by mail, fax or internet. Customers shall be able to request a Complaint Form by telephone, fax, in person or by email. Contractor shall record the name, telephone number, and address of each Complainant and the details of each Complaint. Contractor shall respond to all Complaints, regardless of origin of request for service (by mail, phone, in person or by internet) within seven (7) days of receipt of Complaint.

(c) Claims/Complaint Status Reports

Contractor shall submit a monthly Claims/Complaint Status Report that contains the following information:

- (1) Claim/Complaint tracking number
- (2) Name of Claimant/Complainant
- (3) Date Claim/Complaint received
- (4) Name of Employee who processed Claim/Complaint
- (5) Brief description of Claim/Complaint
- (6) Estimated value of Claim, when available
- (7) Verified amount of Claim
- (8) Status of Claim/Complaint
- (9) Average Claim/Complaint resolution time
- (10) Brief description of Claim/Complaint resolution
- (11) Date of resolution of Claim/Complaint

Contractor shall deliver reports regarding Claims and Complaints to the City in accordance with the reporting requirements of Section 13 of this Appendix A. Contractor also shall retain any supporting documents submitted with a Claim or Complaint in accordance with Record retention requirements of this Agreement. Contractor shall respond to City requests to review Records related to Claims and Complaints within seven (7) days. Contractor shall track Complaints and Claims using both paper forms and electronic records. Using electronic records, Contractor shall provide the City with supplemental, specialized reports regarding any Complaint or Claim upon request within seven (7) days.

(d) Claims/Complaint Processing Plan

APPENDIX A: SCOPE OF WORK n:\ptc1\as2010\0900371\00630742.doc

Page 9 of 45.

Contractor shall provide the City with its Claims/Complaint Processing Plan, along with the form of Complaint and Claim forms to be used under this Agreement, within forty-five (45) days of the Effective Date of this Agreement. This plan shall be approved and adopted as part of the Operations Plan as provided in Section 14 of this Appendix A.

5.3 Dissemination of Information to the Public

(a) Posted Information

The Contractor shall distribute the following information to its towing subcontractors and make available on its internet site and in the Customer Service Center, or in a different location as specified below, that are accessible to the public in a conspicuous location:

- (1) Statement that a complete copy of this "Tow Agreement with the City and County of San Francisco" is available for review on the internet (including the URL for the internet site) or may be obtained at the same maximum per-page costs for copies which the City makes available to the public under the Sunshine Ordinance, San Francisco Administrative Code Chapter 67.
- (2) A statement of the Customer's rights and obligations pursuant to Vehicle Code § 22852.
- (3) Schedules of all current towing, storage and additional charges as established pursuant to this Agreement.
- (4) Notice required by Vehicle Code § 22850.3 that any vehicle impounded pursuant to Vehicle Code § 22850 may only be released upon proof of current registration or upon issuance of a notice to appear with proof of correction of the registration violation, at the discretion of the impounding agency.
- (5) Notice explaining the right to a post-storage hearing and the procedure by which a post-storage hearing may be requested from SFMTA, DPH or the SFPD.
- (6) Procedures for filing a Claim for damages incurred to the vehicle or contents thereof as a result of the tow or while in storage, and associated Claim forms.
- (7) Procedure by which all unclaimed vehicles are sold at public auction, including the location of such auctions, and a statement indicating that all in attendance at such auction shall have an equal opportunity to bid.
- (8) List of publications in which such auctions are advertised.
- (9) Instructions for bidders interested in attending the public vehicle auctions.
- (10) At the Secondary Storage Facility, a preliminary list of all vehicles to be auctioned seven (7) days in advanced of the auction date.
- (11) At the Secondary Storage Facility, results of the previous vehicle auction, which must include the list of vehicles by make, model and year and the final sale price.
- (12) A list of Tow Car firms that Contractor has evaluated for compliance with industry standards based on maintenance of insurance and permits which are not affiliated with Contractor, along with their contact information, that Customers may contact for towing services.

(b) Translation of Posted Information

Except for items 5.3(a)(10) and 5.3(a)(11) above, information provided at the Customer Service Center shall be available to the public in English, Spanish, and Chinese and up to three other languages that may be designated by the City. For items 5.3(a)(10) and 5.3(a)(11) above, Contractor shall provide a summary sheet in Spanish and Chinese explaining how to read the lists of auctioned vehicles. Specifications for signage or documentation

APPENDIX A: SCOPE OF WORK n:\ptc1\as2010\0900371\00630742.doc

Page 10 of 45

produced, including but not limited to multiple languages required, wording, size of letters, font used and methods of display shall be approved by the City in advance of posting or publication. Contractor shall publish all information listed above on its internet site and distribute it to all of its towing subcontractors.

6. Procedures for Vehicle Recovery

6.1 Form of Payment

(a) In-Person Payments

At the time that a Customer contacts Contractor by telephone, internet or in person, Contractor shall communicate the amount of all Mandatory Fees and any other fees owing as of the date of the contact, and shall collect all Mandatory Fees owing prior to releasing the vehicle to the Customer. Customer Service Center staff shall be trained in procedures to accept payments when electronic cash registers or Contractor's computer systems are not operational. The following forms of payment shall be valid for Customers making payment in person at the Customer Service Center:

- (1) Cash;
- (2) Personal checks with valid ID card as proof of identity if verified through check verification service provided by Contractor; and
- (3) Credit card (MasterCard, VISA, American Express, and Novus/Discover Card) with a valid ID card as proof of identity; and
- (4) Debit/ATM card with a valid ID card as proof of identity.

(b) Phone and Internet Payments

For payment over the phone or the internet, Contractor shall accept at a minimum:

- (1) Credit cards (Visa, Mastercard); and
- (2) Debit/ATM cards with credit card company affiliations suitable for internet payments.

6.2 Collection of Parking Citation Payments and Related Fees; Deposit

(a) Citation Payments

Contractor shall accept payment of Citations from all Customers at its Customer Service Center, regardless of whether or not the Customer has had a vehicle towed. In order to allow Contractor to accept Citation payments, City shall give Contractor access to its CMS through the City's WAN and/or via the internet. If a Customer wishes to dispute any Citation, Contractor shall refer the Customer to the SFMTA Citation Division.

(b) Outstanding Fees

At the time that a Customer contacts Contractor regarding payment of fees Contractor shall determine Customer's outstanding Citations and the amount due on each, using SFMTA's CMS. If Customer has five (5) or more Delinquent Citations, Customer must pay all Delinquent Citations before Contractor may release the vehicle. If Customer has fewer than five (5) Delinquent Citations, Contractor must inform Customer of the amount of fees due for outstanding Citations and the option to pay outstanding Citation fees through Contractor. If any person wishes to pay outstanding Citation fees that are not yet Delinquent, the Contractor shall accept payment at its Customer Service Center, or by internet or telephone credit card payment. At the time of payment, Contractor shall record Citation payments online in real-time to the CMS twenty-four (24) hours per day, 365 days per year. Notwithstanding the requirements of this Section 6, Contractor shall not be required to provide information over the internet on outstanding Citation fees, nor shall Contractor be required to accept credit card payments over the internet until such time as Phase 3 of Contractor's internet site is implemented, as described in Section 6.4 herein. Contractor shall not accept payments for Citations that are marked in the CMS as being under Administrative Review, Hearing, or Project 20. Contractor shall also require payments for boot

fees that may have been charged prior to the vehicle being towed and any insufficient funds fees that may have been assessed for prior Citation payments.

(c) Collection Fees for Non-Towed Vehicle Citations

The threshold amount of funds collected daily for Citations associated with vehicles that have not been towed, under which amount Contractor is required to collect Citation fees for non-towed vehicles without any reimbursement credit from City (the "Daily Collection Limit"), is currently set at \$5,952 per day, based on the current average parking fine payment of \$48.29 per Citation (the "Average Citation Amount"). Should the amount collected by Contractor exceed the Daily Collection Limit amount more than twice within a given calendar month, Contractor shall be entitled to a processing fee in the form of a credit in an amount equivalent to seven and a half percent (7.5%) of the amount received above the Daily Collection Limit, beginning on the third (3rd) day and continuing on each day within the calendar month that Citation fee collections exceed the Daily Collection Limit. Should the Average Citation Amount increase or decrease by more than one percent (1%), the Daily Collection Limit shall increase or decrease by the same percentage. The Average Citation Amount and the Daily Collection Limit amount shall be reviewed by City and recalculated as necessary at the beginning of each contract year.

6.3 Interactive Voice Response System

Within sixty (60) days of the Effective Date of the Agreement, Contractor shall provide an interactive voice response system ("IVR") that will answer incoming calls and offer callers a choice of menus with information on towing services. Contractor's proposed telephone answering system, its equipment, functionality and its message content shall be subject to review and approval by the City. At a minimum, the IVR must contain the following features:

- (1) Capability for users to obtain general information about the status and location of a vehicle, including Investigative and Administrative Police Hold information
- (2) Ability to hear a detailed listing and summarized total of all Mandatory Fees and other fees that have been applied to the vehicle (excluding Delinquent Citations)
- (3) Option to pay Mandatory Fees and other fees by credit card and receive instructions for the expedited release of the vehicle from a live operator (excluding Delinquent Citations)
- (4) Options for information on Contractor's locations and operating hours
- (5) Ability to listen to general information regarding Contractor and City policies and procedures
- (6) Information on the weekly lien sale auctions
- (7) Option to transfer to a live operator
- (8) The IVR must be functional ninety-seven percent (97%) of the time during the hours of 6:00 a.m. to 12:00 a.m. during a monthly reporting period.

Within 180 days of the date that the City provides an Application Programming Interface ("API") that Contractor can use to integrate its TVMS with the City's CMS, the Contractor's IVR shall provide the following additional capabilities: Contractor's IVR shall communicate in real-time with the TVMS to accept payments for any outstanding fees and optional services available to the public and with the City's Citation processing system to pay any outstanding Citations. The IVR shall advise the Customer of the date range for which the stated fees are applicable, as well as a time limit within which the vehicle must be retrieved without incurring additional storage charges once the fees have been paid. The IVR shall ensure that all fees for five (5) or more Delinquent Citations, boot fees and insufficient funds fees shall be included in the Mandatory Fees required for release of the vehicle presented to the Customer for payment on the IVR. In addition, Contractor must advise the Customer of any other outstanding amount listed in the Citation processing system. Contractor's IVR shall inform Customers of the payment of Mandatory Fees required for release of the vehicle, and shall separately advise of the amounts of any other outstanding fees that may be paid simultaneously through the IVR system. Following

integration of the TVMS and CMS, Contractor shall provide Customers with the ability to make credit card payments for Mandatory Fees through its IVR system.

6.4 Internet Services

Within ninety (90) days of the Effective Date of this Agreement, Contractor shall design and implement an internet site that allows the Customer to access information on towed vehicles, not including outstanding Citation fees or the ability to make credit card payments over the internet ("Phase 2"). In Phase 2, a Customer shall be able to look up vehicle data on the internet site using a minimum of two (2) of the following data look-up fields:

- (1) License number
- (2) Make
- (3) Towed from location (by street name)
- (4) Tow date

Within 180 days of the date that the City provides an Application Programming Interface ("API") that Contractor can use to integrate its TVMS with the CMS, Contractor shall provide the ability for the public to access all information regarding a towed vehicle and all related outstanding fees and charges ("Phase 3"). In Phase 3, Contractor shall provide the capacity on its internet site to make on-line, real-time payments of Mandatory Fees and any other fees for Citations, using a credit card with appropriate security restrictions for payment types accepted. The internet site shall advise the Customer of the date and time range for which the stated fees are applicable, as well as a time limit within which the vehicle must be retrieved once the fees have been paid without incurring additional storage charges. The internet site shall ensure that all fees for five (5) or more Delinquent Citations, boot fees, and insufficient funds fees shall be included in the Mandatory Fees required for release of the vehicle that are presented to the Customer for payment on the internet site. Contractor's internet site shall inform Customers of the payment amount of Mandatory Fees required, and shall separately advise of the amounts of any other outstanding fees that may be voluntarily paid through the internet site. The internet site created to satisfy the requirements of this Section shall meet Department of Telecommunications and Information Services standards for accessibility.

6.5 Vehicle Recovery

(a) Vehicle Location

Contractor's staff in the Customer Service Center and vehicle storage facilities shall be able to identify and locate a towed vehicle in Contractor's possession by using the license number, the VIN, or three (3) or more of the following identifiers: vehicle make and model, vehicle color, date of tow, and location of tow. Contractor's Customer Service Center and vehicle storage facility staff shall use the TVMS to provide information regarding the tow, and shall also be trained in procedures for manual processing of vehicle pick-up requests during times that the TVMS is not operational.

(b) Vehicle Release Procedure

When appearing to recover a towed vehicle, the Customer shall be required to provide evidence that she/he is a person entitled to possession of the vehicle. This shall include, but is not limited to, a key to the vehicle, a letter authorizing the person to pick up the vehicle signed by the registered owner, or a valid photo identification establishing a person's right to claim the vehicle. When Contractor is satisfied that the Customer is entitled to possession of the vehicle, Contractor shall record the identity of the Customer and the number of the photo identification provided by the Customer into the TVMS. After obtaining payment of all fees owing, the Customer shall be issued a tow receipt, with a copy to be signed by the Customer and retained by Contractor, and shall be directed to the Designated Facility where the vehicle is located to meet a Customer service agent. This Customer service agent shall either assist the Customer with the physical retrieval of their vehicle, direct them to the Contractor-provided shuttle, or arrange a pre-paid, one-way taxi ride to the Secondary Storage Facility in order to retrieve their vehicle. Contractor shall require any person who claims the right to possession of a vehicle to show a valid driver's license or to identify a licensed driver before allowing the vehicle to be

driven off of the Contractor's facilities and onto the public streets, except that Contractor may release a vehicle towed by SFPD to the person listed on the release issued by SFPD without requiring that person to show a valid driver's license.

(c) Vehicles Not Subject to Release

Contractor shall not release a vehicle that is subject to Police or Administrative Hold without a written authorization for release from the agency that impounded the vehicle. Contractor shall not release a vehicle unless Mandatory Fees have been paid or have been waived as provided in this Agreement. Contractor may not release any vehicle without proof of compliance with vehicle registration laws, which at the time of vehicle intake may be ascertained by visual inspection of registration tags on license plates. Contractor shall be liable for and hold City harmless from all claims arising out of the improper release of a vehicle, unless such release is caused solely by the negligence of SFMTA, SFPD, DPH or the DMV.

6.6 Form of Receipt

Contractor shall provide all Customers with a receipt for services rendered. The form of the receipt shall be subject to approval by City. All receipts shall include the following information:

A clear and succinct statement, in a legible text of at least 10-point type, informing the Customer that by law they may protest a vehicle tow, that a hearing to protest any tow must be requested within thirty (30) days from the date of tow, and setting forth current telephone numbers for SFMTA, DPH and the SFPD for the purpose of requesting a hearing. SFMTA may require changes to the language or form of such statement printed on Contractor's receipt. In addition, each Customer's receipt shall include the following information:

- (1) A complete, itemized schedule of fees and charges, and
- (2) Each individual Citation paid, listed by Citation number, the amount paid on that Citation, and the remaining balance due for any outstanding Citation.

6.7 Disclosure of Deficiency Claim at time of Vehicle Abandonment

Contractor shall provide written notice to Customers who abandon their vehicles at Contractor's facilities, informing the Customer that abandoning a vehicle is not sufficient to avoid towing and storage costs in excess of the vehicle's sale price, and that such towing and storage charges in excess of the vehicle's sale price may be subject to collection. Such notice shall include (1) the charges to date, and (2) the Lien Category expressed as the range of potential value of the vehicle. The form of such notice shall be subject to prior approval of the City. Contractor shall not provide such notice to owners of Lien 1 Vehicles, because Lien 1 Vehicles are not subject to deficiency claims.

6.8 Invalid Payments

Any costs incurred by or loss to Contractor resulting from the use of checks, counterfeit cash, credit cards, debit cards, or ATM cards, whether received in person, through the IVR or internet, shall be considered a business expense of Contractor and is not billable to the City under any circumstances.

7. Procedures for Lien Sales

7.1 Notification to Registered Owner

Contractor shall make a diligent effort to locate and contact the owner and any lienholder(s) for each impounded vehicle, in accordance with all state and local laws and regulations. Contractor shall request vehicle ownership information from the DMV for all vehicles stored at least seventy-two (72) hours, and shall, whenever ownership information is available, send lien notices to registered owners, lien holders and legal owners identified by the DMV between three (3) and seven (7) days from the date that the vehicle was towed. If Contractor is able to ascertain the identity of the owner of the vehicle and fails to send notice under this section within seven (7) days of the date that the vehicle was towed, Contractor shall waive storage fees for the vehicle for the eighth (8th) day of storage through the lien start date. Contractor shall use an electronic means of communicating its requests for vehicle license and ownership information to, and of receiving responsive

information from the DMV. Exceptions to the deadlines for providing prompt notice of storage to vehicle owners pursuant to this Section 7.1 include:

- (1) Vehicles subject to Investigative Police Holds. Ownership information for these vehicles should be requested and notices sent within seven (7) days after the hold is released by the SFPD, and storage charges shall begin to accrue as of the date of the release of the hold.
- (2) Vehicles identified as NO ID vehicles pursuant to Section 4.5 of this Appendix A which require SFPD inspection. Ownership information for these vehicles should be requested and notices sent within seven (7) days after vehicle identification has been provided by SFPD.
- (3) Vehicles with out-of-state license plates for which the DMV does not have ownership information. Lien notices for these vehicles must be sent to DMV in accordance with the requirements of the Vehicle Code.
- (4) Other cases of extenuating circumstances as approved by the City.

The form of notice sent to registered owners and lien holders shall be subject to Vehicle Code, Civil Code, and DMV requirements and prior approval by City. Lien sale notices shall include a statement that failure to claim a vehicle is not sufficient to avoid towing and storage costs in excess of the vehicle's sale price, and that, with the exception of Lien 1 Vehicles, such towing and storage charges in excess of the vehicle's sale price may be subject to collection. Lien sale notices shall also be accompanied by the following notice:

If your vehicle is sold at auction, you may be entitled to any proceeds that exceed the amount of the AutoReturn lien, any applicable City fees, and the amount of any unpaid parking citations on the vehicle. AutoReturn is required to send the excess proceeds to the California Department of Motor Vehicles to be deposited in the Motor Vehicle Account in the State Transportation Fund. If you do not claim the excess proceeds within three years of the date that the money is deposited into the Motor Vehicle Account, you may forfeit your right to the money under California Civil Code Section 3073. In order to find out if your car sold for more than our lien amount, you may contact the Lien Sale Unit at the Department of Motor Vehicles at the following address: California Department of Motor Vehicles, Lien Sale Unit, P.O. Box 932317, Sacramento, CA 94232-3170. Or, you may call the Lien Sale Unit at the following telephone number: 916-657-7976.

7.2 Lien Sales

(a) Lien Sale Procedures

Contractor shall comply with all state and local laws and regulations applicable to notice and conduct of lien sales of vehicles, including, but not limited to California Civil Code §§ 3068-3074 and Division 11, Chapter 10, Article 2 of the Vehicle Code (§§ 22650 *et seq.*), and any successor statutes. Contractor shall track lien-related dates, and process the official lien notification paperwork as required for lien sales in the state of California. Contractor's lien sale notice for Lien 2 and Lien 3 vehicles shall include the specific date that the vehicle is scheduled to be sold at auction, and Contractor may use this date to calculate the amount of storage charges due. Contractor shall not process the Bill of Sale and the Certificate of Lien Sale until the auction sale date and shall include on these forms the name of the buyer and the purchase price of the vehicle.

Unless City has given prior written approval to suspend a scheduled lien sale ("auction"), Contractor shall conduct lien sales at least weekly for vehicles that have been cleared for sale after the lien process is complete. The day of the week for these auctions is subject to approval by City. City hereby approves Wednesday as a regularly scheduled auction day. Contractor shall submit a Public Auction Plan for the City's review and approval upon the Effective Date of the Agreement. The Public Auction Plan shall be approved and adopted as part of the Operations Plan as provided in Section 14 of this Appendix A.

(b) Pre-Registration

Contractor shall require all persons who desire to purchase vehicles at auction to pre-register. Contractor shall require all registrants to provide photo identification with current address. Contractor shall maintain records of each purchaser's name and address. Contractor may develop lists of auction participants who demonstrate a tendency to purchase and subsequently abandon vehicles purchased at auction, and may prohibit such persons and any other persons who are known to have engaged in illegal conduct or conduct prohibited by this Agreement before, during or after an auction from submitting bids.

7.3 Post-Lien Sale Procedures

(a) Driver's License and Vehicle Registration

Contractor shall require the purchaser of a vehicle or the purchaser's agent to show a valid driver's license before driving the vehicle off of any Designated Facility and onto the public streets. Contractor shall comply with applicable Vehicle Code requirements for the transfer of title, including but not limited to the requirement of filing a notice of transfer of title pursuant to Vehicle Code § 5900. Contractor shall cooperate with City in implementing any program to provide on-site vehicle registration during vehicle auctions.

(b) Deficiency Claims

Following sale of any vehicle for which Contractor wishes to maintain a Deficiency Claim under California Civil Code § 3068.1, after the sale and before attempting collection of the Deficiency Claim Contractor shall send a notice to the registered owner of the amount of the Deficiency Claim, the basis of charges, including the dates and amounts of towing and storage fees, the make, model and license number of the vehicle that is the basis for the claim, and the amount of the debt, including the amount that is offset by money recovered from the sale or salvage of the vehicle. Documentation of any amounts received by Contractor for the sale or salvage of the vehicle shall be included with such notice. In the event that Contractor utilizes a third-party vendor to provide collection services for these Deficiency Claims, Contractor shall contractually require the third-party vendor to meet the notice requirements of this Section 7.3(b). The form of such notice, whether sent by Contractor or by a third-party vendor, shall be subject to prior approval by City. Contractor waives the right to collect any Deficiency Claim for which it has not complied with the notice requirement of this Section; however Contractor may pursue a Deficiency Claim in spite of a failure to provide adequate notice as required herein if it cures such failure by sending a notice that meets the requirements of this Section within ten (10) days of receiving a written request for documentation of the debt from the vehicle's legal owner, registered owner or the City. Contractor shall not maintain a Deficiency Claim for any Lien 1 Vehicle.

7.4 Disposal of Unsold Vehicles

Contractor shall ensure that Lien 1 and other vehicles not sold at auction are removed from the Secondary Storage Facility at least once per week.

8. Staffing Requirements

8.1 Adequate Staffing

Contractor shall employ adequate numbers of qualified personnel to perform the required services in accordance with the standards specified in this Agreement. The City may require Contractor to hire additional Employees if it reasonably determines that additional Employees are necessary to perform the services required under this Agreement to the specified standards. Criteria to be used in making this determination include but are not limited to any combination of the following:

- (1) Tow response times, or
- (2) Wait times for walk-in Customers at Customer Service Center, or
- (3) Vehicle delivery times to Customers, or
- (4) Live operator wait-times for phone Customers, or
- (5) Ten percent (10%) change in the number of Complaints, based on monthly Complaint report data, and at least ten (10) more Complaints than the prior month.

APPENDIX A: SCOPE OF WORK n:\ptc1\as2010\0900371\00630742.doc

Page 16 of 45

8.2 Subcontractors

(a) Subcontractor Designation

Contractor may subcontract with one or more towing service companies for the provision of towing services to Contractor in accordance with the standards of this Agreement. A list of the Contractor's pre-selected and prequalified subcontractors shall be submitted upon the Effective Date of this Agreement for City's approval. Consistent with the provisions of Sections 27 and 31 of the Agreement, Contractor shall notify the City in writing prior to deleting or adding any towing subcontractors. Contractor shall provide the City with copies of all subcontractor agreements within five (5) days of the Effective Date of this Agreement and required insurance certificates that identify the Contractor and City as co-insured parties for those subcontractors (not including Worker's Compensation Insurance). Contractor shall also provide copies of subsequent subcontract amendments within five (5) days of any such amendments to its subcontractor agreements.

(b) Subcontractor Compensation

Contractor shall not pay compensation to towing subcontractors based solely on a flat fee per tow. City may require Contractor to change any element of its compensation structure if it results in an undue number of subcontractor complaints or Customer service problems. Upon the Effective Date of this Agreement, Contractor shall submit a Subcontractor Compensation Plan to City for approval as part of the Operations Plan in accordance with the procedures set forth in Appendix A, Section 14.

(c) Subcontractor Performance Standards

Contractor shall define Subcontractor Performance Standards for all Tow Car operators. At a minimum, Performance Standards shall require compliance with all applicable Vehicle Code and San Francisco Police Code requirements for Tow Car operators and Tow Car firms. Subcontractor Performance Standards shall be provided to the City upon the Effective Date of this Agreement. The Subcontractor Performance Standards shall be approved and adopted as part of the Operations Plan as provided in Appendix A, Section 14. Contractor shall audit compliance with these standards quarterly, beginning on a date no later than one hundred and eighty (180) days from the Effective Date of this Agreement, and shall provide written audit results to the City. Contractor's quarterly audit shall, at a minimum, audit subcontractors for compliance with the requirement to maintain valid licenses and permits. The City reserves the right to require the Contractor to perform more frequent audits. Contractor is responsible for any and all claims arising out of the Contractor's failure to maintain current permits and licenses.

(d) Subcontractor Compliance with Licenses and Permits

Contractor shall maintain current and valid City licenses, permits and proof of insurance necessary to perform the services required for this Agreement. Contractor shall require in its subcontracting agreements that all subcontractors maintain current and valid City licenses, permits, and proof of insurance, including but not limited to Articles 30 (Permits for Tow Car Drivers) and 30.1 (Permits for Tow Car Firms) of the San Francisco Police Code or successor Tow Car or Tow Car firm permitting ordinances, and shall require all subcontractors to demonstrate evidence of such licenses, permits and insurance at the time of executing the subcontracts.

(e) Subcontractor Identifying Equipment and Uniforms

All subcontractor personnel who have regular, continuous contact with members of the public shall be neat in appearance and courteous to the public. Contractor shall ensure that all Tow Cars used by its subcontractors for the provision of towing services shall bear a permanently attached sign stating Contractor's and subcontractor's trade name and telephone number in characters at least two inches high and an identifying number in characters at least three inches high on both sides of the vehicle. Detachable magnetic signs may not be used. Contractor shall include in subcontracts the following uniform requirements for subcontractors:

- (1) All Tow Car operators must display photo ID badges at all times when on duty. The form and design of the photo ID badges must be approved by the City.
- (2) All Tow Car operators must wear at all times a standard colored shirt with the name of the employee affixed to right or left side of the shirt. The Contractor or

subcontractor's company name shall be displayed on the opposite side of the shirt and/or jacket.

8.3 Training

Contractor shall be responsible for all training costs for its Employees. City shall approve all training programs proposed by Contractor in advance of implementation. Contractor shall review the Employee Training Plan with the City annually. Contractor shall submit within ninety (90) days of the Effective Date of the Agreement an Employee Training Plan to the City for review and approval. The Employee Training Plan shall be approved and adopted as part of the Operations Plan as provided in Section 14 of this Appendix A. In approving the Employee Training Plan, City shall give Contractor credit for training provided to Employees during the term of the Emergency Interim Agreement to the extent that it meets the requirements of this Section.

Within thirty (30) days after the Effective Date of the Agreement, Contractor shall have every Employee attend twenty-four (24) hours of professional job training, including customer service training. In addition, Contractor shall conduct a minimum of eight (8) hours of professional customer service training annually for all Employees who perform duties involving significant Customer contact. Contractor shall require certificates of completion for each Employee and shall file these in Employee personnel files with copies sent to the City within five (5) days of completion of training. Contractor shall require in its subcontractor agreements with Tow Car operators that all Tow Car operators performing services for Contractor shall participate in one four (4) hour customer service training session annually. Contractor must provide one such four (4) hour training session annually for towing subcontractors at its own expense; however, when additional training sessions for new subcontractor employees are required Contractor may pass the cost of such trainings on to the subcontractors.

8.4 Policy and Procedure Manual

Contractor shall submit within ninety (90) days of the Effective Date of the Agreement a Policy and Procedure Manual to the City for review and approval. The Policy and Procedures Manual shall be approved and adopted as part of the Operations Plan as provided in Section 14 of this Appendix A. Contractor shall require in its subcontracting agreements that all subcontractors shall adhere to Contractor's Policy and Procedures Manual. The Policy and Procedures Manual shall differentiate between the responsibilities of each subcontractor where necessary.

8.5 Staffing Plan

Within forty-five (45) days of the Effective Date of the Agreement, Contractor shall provide a Staffing Plan to meet all service and performance requirements of the Agreement. Contractor's Staffing Plan shall indicate the number of people and positions it will provide to perform the services required in this Agreement. The Plan shall provide the name, title and time allocation (percentage of dedication to the proposed Agreement) for each staff person. The Staffing Plan shall be approved and adopted as part of the Operations Plan as provided in Section 14 of this Appendix A.

8.6 Management Changes

Contractor shall provide written notice for City approval prior to any permanent changes or substitutions of executive or management Employees, for any substitutions longer than six (6) weeks. SFMTA may request a change in contract personnel or reject any substitution. The Contractor shall notify the City within five (5) days of occurrence any terminations or resignations of Contractor's executive or management Employees.

8.7 On-Call Manager

Contractor shall have a manager with the authority to make decisions regarding Agreement-related issues who shall be available or on-call at all times through the use of mobile phones, pagers, or two-way radios for all operational functions. Contractor's system for providing the City with access to the manager at all times must be approved, in writing, by the City.

8.8 **Contract Monitor**

(a) Contract Monitor Payment Fund

APPENDIX A: SCOPE OF WORK n:\ptc1\as2010\0900371\00630742.doc

Page 18 of 45

Within 30 days of the Effective Date of this Amended and Restated Agreement, and within 30 days of each anniversary of that Effective Date during the Term of the Agreement (the "Anniversary Date"), Contractor shall make a payment to the City ("Contractor Monitor Payment") to be used by City for the sole purpose of employing a Contract Monitor who shall monitor Contractor's compliance with the Agreement, its Appendices, the Operations Plans and relevant California law as directed by City. The initial Contract Monitor Payment shall be \$140,000, which shall be adjusted annually-in direct proportion to the percentage increase in the current Consumer Price Index for Urban Wage Earners for the San Francisco Bay Area ("CPI") for the month immediately preceding the applicable Anniversary Date ("Current Index") over the CPI for the month of June 2010 ("Base Index"). In no case shall the Contract Monitor Payment, as adjusted, be less than the Contract Monitor Payment made on the previous Anniversary Date. If the Current Index has increased over the Base Index, the adjusted Contract Monitor Payment shall be determined by multiplying the initial Contract Monitor Payment by a fraction, the numerator of which is the Current Index and the denominator of which is the Base Index, as follows:

Current Index

Base Index x initial Contract Monitor Payment = Adjusted Contract Monitor Payment

(b) **Duties of Contract Monitor**

The Contract Monitor shall have broad audit and investigation authority. The Contract Monitor shall review the Contractor's classification of vehicles as Lien 1, Lien 2, or Lien 3 and has the authority to require Contractor to change the classification to protect vehicle owners prior to the Contractor sending the lien notice to the DMV. The Contract Monitor's decision to change the classification of a vehicle must be in accordance with Vehicle Valuation Standards Plan, which shall be approved and adopted as part of the Operations Plan as provided in Section 14 of this Appendix A.

The Contract Monitor's duties shall include such duties as City, in its sole discretion, shall specify. In addition, the Contract Monitor is hereby authorized to act as SFMTA's representative for the purpose of conducting any inspection authorized by Section 11.6 of Appendix A of this Agreement.

Contractor shall notify Contract Monitor of the day of regularly scheduled weekly auctions. Contractor also shall notify the Contract Monitor a minimum of three (3) days before any additional auctions will take place. Contractor shall provide the Contract Monitor a report of all vehicles being auctioned, which shall include at a minimum the vehicle makes, models and the minimum asking price as determined by Contractor. The Contract Monitor has the authority to require Contractor to reduce the minimum bid amounts, if any, prior to the auction.

8.9 Vehicle Auctioneer

(a) Auctioneer Staffing Plan

Contractor shall contract for the services of an independent auctioneer at Contractor's expense. Contractor shall replace a full-time auctioneer at least once per year except with the City's prior written approval. Upon Contractor's request, and with prior written approval by City, an individual who has served as a full-time auctioneer in the past, but not within the preceding year, may again act as a full-time auctioneer. A full-time auctioneer is defined as one who has performed more than 50% of the auctions within a given contract year. Within forty-five (45) days of the Effective Date of the Agreement, Contractor shall provide to City an Auctioneer Staffing Plan, which shall include the method for selection and screening of candidates for the position of auctioneer, the duties of the auctioneer, and the measures that Contractor shall take to ensure that no sales are made at auction in violation of Section 8.9(c) of this Agreement. The Auctioneer Staffing Plan shall be approved and adopted as part of the Operations Plan as provided in Section 14 of this Appendix A.

(b) Information Provided to Auctioneer

Contractor shall notify the auctioneer of the day of regularly scheduled weekly auctions. Contractor also shall notify the auctioneer a minimum of three (3) days before any additional auctions will take place. Contractor shall provide the auctioneer a report of all vehicles being auctioned, which shall include at a minimum the vehicle makes, models, and the minimum asking price as determined by Contractor ("Auctioneer Report").

(c) Vehicle Sales to Certain Individuals Prohibited

Contractor shall not knowingly sell vehicles at auction to any individuals meeting the following criteria (collectively, "Restricted Auction Participants"):

- (1) The Contract Monitor and any Close Family Member of the Contract Monitor;
- (2) Contractor's employees and any Close Family Member of Contractor's employees;
- (3) Any individual designated to provide auditing services under the Agreement as described in Section 12.6; and
- (4) Any person who is acting or has acted within the previous five years as Contractor's vehicle auctioneer, and any Close Family Member of a person who is acting or has acted within the past five years as Contractor's vehicle auctioneer.

Any auctioneer who knowingly has accepted bids on lien sale vehicles in violation of this Agreement, or who conducts an auction that in any way benefits the auctioneer's own financial interests or the financial interests of any Close Family Member, shall be immediately disqualified from conducting any future auctions. The Contractor must act immediately to dismiss an auctioneer if it has actual knowledge of any conduct on the part of the auctioneer that is prohibited under this Section, and must immediately notify the City if it has actual knowledge or suspicion of any conduct on the part of the auctioneer that is prohibited under this Section.

Contractor shall not be responsible for screening auction participants for improper relationships, except that Contractor must obtain and verify a list of the Restricted Auction Participants at the time of any changes to the current employee list, the Contract Monitor, the Auctioneer and/or the individuals designated to provide auditing services under the Agreement. On at least a quarterly basis, Contractor must utilize the most current list of Restricted Auction participants and a random sample set of vehicles sold through the auction during the given evaluation period to determine if any vehicle sales within the sample are prohibited under this Section. The sample should contain no less than 1% of the vehicles sold for the evaluation period.

Contractor must establish a clear employment policy that prohibits employees and their Close Family Members from making purchases of vehicles at Contractor's auctions. The policy must be included in Contractor's employee policies manual that all new employees are required to review and acknowledge acceptance of at the time of employment. The Contractor must act immediately initiate termination proceedings for any employee found to have violated this policy and must immediately notify the City if it has actual knowledge or suspicion of any employee violating this policy.

For the purpose of this Section, the term "Close Family Member" shall mean an individual's spouse, child or parent.

9. Equipment and Information Services Requirements

9.1 Tow Cars

Contractor shall provide regular and heavy duty Tow Cars staffed with trained operators twenty-four (24) hours a day, 365 days per year, within response time requirements, to perform any type of vehicle removal that is the subject of a Tow Request from any public street or highway or private property within the City in accordance with the requirements of the Vehicle Code and the Traffic Code, including, but not limited to, the following types of tows:

- (1) Towing of large and oversize vehicles
- (2) Towing from off-road areas
- (3) Towing from low-clearance areas and underground garages
- (4) Towing of evidentiary vehicles
- (5) Towing vehicles involved in collisions
- (6) Towing of vehicles with anti-theft locking devices.

All Tow Cars used in the performance of tow services under this Agreement shall be in good mechanical and operating condition and clean on the interior and exterior. Contractor's agreements with subcontractors for towing services must require subcontractors to comply with maintenance and cleanliness standards for Tow Cars and auxiliary equipment set forth on the California Highway Patrol form number 234 "Annual Tow Car Inspection Report" or successor California Highway Patrol form. Contractor shall conduct random inspections of all Tow Cars and Tow Equipment provided by subcontractors at least quarterly, on an inspection schedule that ensures that the entire fleet and all operators are inspected over a twelve (12) month period. Contractor shall submit all inspection results to the City.

9.2 GPS Tracking Systems for Tow Cars

(a) GPS Equipment

Subcontracts for towing services shall require that all Tow Cars or Tow Car operators used to provide services under this Agreement be equipped with a global positioning tracking system ("GPS") within ninety (90) days of the Effective Date of this Agreement. The GPS device selected for use in the Tow Cars must allow City to accurately track the location of the Tow Car operator in accordance with the specifications set forth in Section 15.5(5) of this Appendix A while it is used to provide services under this Agreement. If the selected GPS device does not meet the City's reasonable expectations based on the GPS standards set forth in the RFP, the City reserves the right to require an alternate device be used, including one that is affixed to the Tow Car.

(b) GPS Software

APPENDIX A: SCOPE OF WORK n:\ptc1\as2010\0900371\00630742.doc

Page 21 of 45
Contractor shall acquire and maintain all necessary software licenses for this GPS tracking system. Contractor shall provide to City, at Contractor's sole cost, access to the GPS tracking system using an internet based interface or client-server application that can operate on the City's WAN. Contractor shall install and configure GPS software, provide training for the use of the software, and support all activities related to the City's use of the GPS tracking system.

9.3 Two-Way Radio Communications Equipment

On the Effective Date of this Agreement, Contractor shall issue a two-way radio to each Tow Car operator that shall be used when direct voice communication is required or as a backup communication method, and to enable digital, hands-free communication between Central Dispatch and the Tow Car. All radio communications between Central Dispatch and a Tow Car shall be monitored by supervisory personnel and recorded (as described in Appendix A, Section 1.3). Contractor shall also issue a minimum of two (2) radios to SFMTA to be used as an alternative communications method between City dispatchers and Contractor dispatchers. Contractor shall be responsible for all costs associated with two-way radios that are issued to the City; however, Contractor may recover costs for radios distributed to subcontractors from those subcontractors.

9.4 Tow Car Data Terminal

Within 180 days of the Effective Date of this Agreement, Contractor shall send the initial basic information collected on every vehicle towed (field officer requested tows only) by each Tow Car in real-time to the TVMS using a data terminal or other device available in each Tow Car. Contractor shall train Employees and subcontractors on the manual procedure for inputting the initial basic information on towed vehicles if the Tow Car data terminals are not working. When Tow Car data terminals are not working, the information shall be communicated from the Tow Car operators in the field to Central Dispatch via two-way radio communications. This initial communication will create a new record indexed to the unique, system-generated field in the TVMS for every tow provided by Contractor, herein referred to as the "Tow Request ID" for the purpose of this Agreement only; Contractor may use any clearly defined field name in the TVMS. The basic information reported on every towed vehicle must at a minimum, include at least three (3) of any of the data elements from the following list; however City may, by written notice to Contractor, require that the vehicle license plate number be a mandatory data element for the purposes of this Section:

- (1) Vehicle license plate number
- (2) Vehicle make
- (3) Vehicle model
- (4) Vehicle color
- (5) Location of the tow (street and cross street)
- (6) Tow date and time.

9.5 Vehicle Tow Records and CMS

Within ninety (90) days of agreement by the parties on an interface design and joint project plan, Contractor shall create an interface between its TVMS and the City's CMS so that records on each towed vehicle are created in the City's CMS in real-time as the tows occur (field officer requested tows only). In addition, this interface must ensure that agreed-upon updates made by Contractor to a towed vehicle record in the TVMS are updated in the City's CMS in real-time for any data element that is shared by the CMS and the TVMS (both dispatch and field officer requested tows). Contractor must provide City with information to update the CMS with all intake information within one (1) hour of the intake of the vehicle. In any instance in which the CMS and TVMS have matching information for the vehicle. If the vehicle's identifying information cannot be matched between the CMS and the TVMS, Contractor shall report the discrepancy to SFMTA within two (2) business days, excluding weekends and holidays. Contractor shall update the City's CMS via a "simple screen swipe" method, for the purposes of inserting new tow information (field officer requested tows only) through

the interface; however, if Contractor identifies a more efficient, cost effective method which the City approves in writing, Contractor may implement such alternative method at its own cost.

At any time during the Term of this Agreement, with the City's prior, written approval, and subject to all conditions contained in such approval, the parties may agree to eliminate the use of the City's CMS for entering and tracking towed and impounded vehicles. Should the parties agree to eliminate the data entry of new tow information into the CMS, the interface requirements of this Section 9.5 would not longer apply. Likewise, Contractor would no longer be required to update CMS when a vehicle is released or sold (removed from the impound facilities). Instead, the sole record of towed and impounded vehicles would be the Contractor's TVMS system, and the City would have the right to audit this data at any time using reasonable auditing methods. If a record of towed and impounded vehicles is no longer maintained in CMS, the Contractor would be required, on a daily basis, to provide the City with a list, in an electronic format that is approved by the City, of all vehicles that are currently impounded.

9.6 Electronic Tow Inventory Slips

Within sixty (60) days of the Effective Date of this Agreement, Contractor shall provide electronic storage of all tow slip information in the TVMS system. The tow slips shall be processed and entered into the TVMS system as soon as the vehicle is delivered to one of the Designated Facilities, and never more than eight (8) hours after a vehicle is towed. Contractor shall ensure that City has remote electronic access to the tow inventory slip information at all times. Contractor shall also scan all manually written tow slips and store the tow slip as an electronically scanned image, cross-referenced to the tow record in the TVMS. Contractor shall provide a hotline for technical assistance related to electronic tow inventory slips Monday through Friday from 8:00 a.m. to 6:00 p.m., with pager or mobile phone access during all other hours.

10. Towed Vehicle Management System

Contractor shall maintain detailed electronic records of each tow in its TVMS. The TVMS shall assign a unique, system-generated Tow Request ID to each Tow Request made during the term of this Agreement. All information related to the towing, impoundment, and disposition of any vehicle currently impounded or previously released, sold or disposed of, including references to manually written (paper) tow slips, shall be searchable in the TVMS using the Tow Request ID. This system must be capable of providing to SFMTA a daily record containing information including the date of the Tow Request, the make, model, license plate number, and VIN of all towed vehicles and the current status of all towed vehicles. Towed vehicle information must be entered into the TVMS within three (3) days of the date of the tow.

For Dispatch Tows, information about vehicles that are dropped or not towed for some other reason shall be recorded in the Contractor's TVMS but Contractor shall not generate any record for the purpose of assessing charges for tows of such vehicles.

10.1 Computer Hardware and Software

Contractor shall use Microsoft Windows 2000 Server, or later version, VMware ESX Server 3.5, or later version, or CentOS 5.x, or later version, as the operating system of its servers. Contractor shall use Windows XP Professional, or later version, as the operating system for its desktop computer workstations. The location of the Contractor's data center in which information for the TVMS is hosted must be approved by the City. As of the Effective Date of the Agreement, City hereby approves the location of Contractor's data centers, at the following locations:

- (1) TVMS: SoftLayer Technologies, Inc., Seattle, Washington;
- (2) ADS: Evocative, Inc., Emeryville, California; and
- (3) Other Applications: the Evocative data center and the data center located at Contractor's Customer Service Center.

Contractor shall not under any circumstances maintain the TVMS using proprietary software that prevents data extraction and analyses into any general industry-wide database system protocol, such as Oracle or Microsoft SQL.

Contractor shall provide the following specified hardware for the City's use to connect the City network to the Contractor's network. All the equipment that is installed in the City's data center and required to terminate the secure, point-to-point connection between the City and Contractor networks shall become the property of the City at the expiration or termination of this Agreement. The requirement that Contractor provide this equipment is subject to the provisions of Section 10.2 If the City, in writing, removes the requirement for the T1 data line, the associated equipment will no longer be required.

(a) Firewall Configuration

Cisco (PIX-506E-BUN-K9)

Product	Description	Quantity
PIX-506E-BUN-K9	PIX 506E 3DES/AES Bundle (Chassis, SW, 2 FE Ports, 3DES/AES)	1
CAB-AC	Power Cord, 110V	1
SF-PIX-506-6.3	PIX v6.3 Software for the PIX 506E Chassis	1
PIX-506-SW-3DES	PIX 506E3DES/AES VPN/SSH/SSL encryption license	1
PIX-VPN-CLNT- K9	Cisco VPN Client Software (Windows, Linux, Solaris)	1
CON-OSP- PIX506BN	ONSITE 24x7x4 PIX 506E 3DES/AES Bundle (Chassis, SW, 2)	1

(b) Router Configuration

(CISCO 1760)

Product	Description	Quantity
CISCO1760	10/100 Modular Router w/ 4 slots, 19-in. Chassis, 32F/64D	1
S17B-12215T	Cisco 1700 IOS IP/IPX	1
WIC-1DSU-T1	1-Port T1/Fractional T1 DSU/CSU WAN Interface Card	1
CAB-AC	Power Cord, 110V	1
	,	
ROUTER-SDM	Device manager for routers	1
CON-SOP-1760	24x7x4 Onsite Svc, 10/100 Modular Router w/ 2WIC/VIC, 2VICs	1

See Figure 1 for further configuration details.

Page 25 of 45

Figure 1: Overview of Connectivity Between City and Contractor Networks



10.2 Network Connections

Within thirty (30) days of the Effective Date of the Agreement, or other date approved in writing by City, the Contractor shall provide data lines to connect the following locations to the Contractor's TVMS:

Contractor Division	Location
Customer Service Center	450 7th Street
Primary Storage Facility	450 7th Street
Secondary Storage Facility	Pier 70

Note: Alternate sites may be approved by the City.

Contractor shall provide T1 Internet data lines that connect the Customer Service Center and Primary Storage Facility to the Contractor's TVMS. Contractor shall provide data connections to the Primary and Secondary Storage Facilities using a T1 data line. Contractor shall install a firewall at all locations where there is a direct connect to the Internet to ensure the security of the data.

Within thirty (30) days of the Effective Date of this Agreement, or other date approved in writing by City, Contractor shall provide a T1 data line that connects the Customer Service Center with the SFMTA server headquarters at 1 South Van Ness Avenue. This connection will be used to provide a connection to the Contractor's TVMS for City staff and create a pathway for the Contractor's staff and systems to access the City's CMS. At any time during the Term of this Agreement, with the City's prior, written approval, and subject to all the conditions contained in such approval, the parties may agree to eliminate the requirement for the T1 data line that connects the networks of the City and Contractor. Because both the CMS and TVMS systems can now be accessed directly via the Internet, the parties may agree to eliminate the T1 data line to simplify the systems administration for the City and Contractor, and to eliminate the monthly costs of both the data connection and the associated networking equipment (see Section 10.1).

Contractor shall provide the following City locations with real-time access to the Contractor's TVMS within thirty (30) days of the Effective Date of this Agreement:

City Division	Location
SFMTA Enforcement	505 7th Street
SFMTA Citation Division	11 South Van Ness Avenue
SFMTA Hearing Division	11 South Van Ness Avenue
SFMTA Administration	1 South Van Ness Avenue, 8 th Floor
SFPD NOID Office	Pier 70
SFPD STOP Program	850 Bryant Street

For users located at the City locations listed above who cannot connect to the Contractor's system using either a direct connection to the Internet or the data line connecting the Customer Service Center to 1 South Van Ness Avenue, Contractor shall configure a single Virtual Private Network (VPN) utilizing 3DES encryption per location. Remote clients shall be able to connect to Contractor's network through remote VPN client software and DSL Internet connections. Contractor shall bear only the cost of the DSL Internet services and the corresponding DSL modems and/or routers. City shall identify and provide a computer (Windows XP, or later version) in each location on which the VPN client software will be installed, for which the DSL Internet service will be established, and for which the access to the Contractor's TVMS will be provided. The City shall also be responsible for any telecommunications cabling that is required for the DSL connections to be established in each location.

10.3 City Access to TVMS

Contractor shall provide the City with direct, real-time access to the TVMS. Access to the TVMS shall be controlled using sufficient security protocols and procedures to protect the security of the City's network.

Contractor shall issue user logins and passwords to authorized City staff members as needed and requested by the City. All costs associated with providing the City with access to the TVMS, including any required hardware and data lines (as described in Sections 10.1 and 10.2), shall be the sole responsibility of the Contractor (except as specified in Sections 10.1 and 10.2).

Contractor shall provide all associated hardware, software, data lines, and maintenance to ensure on-going City access to the TVMS at least ninety-nine percent (99%) of the time during a monthly reporting cycle and shall provide a hotline for technical assistance Monday through Friday from 8:00 a.m. to 6:00 p.m., with pager or mobile phone access during all other hours.

10.4 TVMS Functionality

Contractor shall provide the City with a proposed data model of the TVMS that shows all tables and fields for the City's review and approval within thirty (30) days of the Effective Date of this Agreement. Contractor's proposed data model for the TVMS shall include, at a minimum, the data fields listed in the table below.

Data Element	Required Field
Vehicle Information	Make
· · · · · · · · · · · · · · · · · · ·	Model
	Year
	Color
	Body style
	License number
	License state
	VIN
	Comments added by Customer service representatives from conversations with vehicle owner, lien holder, insurance agent, or any other applicable party
	Date of tow
	Time of tow
1	Location of tow - street
	Location of tow - cross street
	Tow Equipment type
	Reason for tow
	Time of arrival at Primary Storage Facility
· · · · · · · · · · · · · · · · · · ·	Date relocated to Secondary Storage Facility
	Time relocated to Secondary Storage Facility
	Storage location
	Dispatcher ID
	Tow subcontractor ID
	Disposition type (released or sold)

Data Element	Required Field
	Date of disposition
· · · ·	List of vehicle contents at time of tow
Photograph of Vehicle Condition	Digital photographs of vehicle condition at time of initial arrival at Designated Facility
Vehicle condition	Description of vehicle condition on arrival at Primary Storage Facility
Image of Vehicle Condition	Documentation of damage visible on vehicle at time of tow
Registered Owner Information	Name
	Address
Revenue	Sale revenue
	Salvage revenue
Fees Received	Tow fees
	SFMTA Administrative Fee
	Storage fees
	Transfer fees
· · · · · · · · · · · · · · · · · · ·	Lien fee
Lien Information	Date information sent to DMV
	Date of release (release date or sale date)
	Lien Category
· · · ·	Sale price
· · ·	Purchaser name
· ·	Purchaser address
	ID type provided
	ID information
City notification	Time of Tow Request
·	Date of Tow Request
	Agency that initiated Tow Request
·····	Officer badge number, if applicable
Contractor notification	Time that Tow Request was received
	Time that Tow Car arrived on site
Hold information	Field indicating whether vehicle on Hold
	Name and/or badge number of officer who authorized the release for holds requiring written release authorization

Data Element	Required Field
	Date of release from hold for holds requiring written release authorization
	Time of release from hold for holds requiring written release authorization

Once City has reviewed the initial data model, it shall request Contractor in writing to make any modifications it considers vital to the system. Contractor shall incorporate the City's comments and resubmit the data model for the City's approval.

10.5 Handheld Devices

Within 180 days of the Effective Date of this Agreement, or other date approved in writing by City, Contractor shall provide handheld devices to be used by its Employees at its Primary and Secondary Storage Facilities. These devices shall provide real-time access to a limited number of specified fields in the TVMS in order to allow Contractor's Employees to real-time query data on impounded vehicles in accordance with the standards set forth in Section 15.5(11) of this Appendix A. The fields available for such queries shall include but are not limited to vehicle license number, VIN, date of tow, and the Tow Request ID. Contractor's handheld devices shall also be able to scan and read ID tags (bar-code, RFID, etc.) so that the devices can be used for inventory management of towed vehicles stored at the Secondary Storage Facility. Contractor shall propose its recommended handheld device to the City for approval prior to purchase.

Contractor shall, at its sole expense, provide the City with up to three (3) handheld devices (as requested by the City) that are identical to the devices to be used by Contractor's Employees. Contractor shall be responsible for any ongoing license fees, airtime charges, or other related costs to use these devices.

11. Facilities Requirements

11.1 Designated Facilities

Contractor shall use the Designated Facilities for all service requirements of this Agreement, except as otherwise approved in advance in writing by City. The location or relocation of any Designated Facility shall be subject to prior written approval by City. City hereby approves the current location of all Designated Facilities. Contractor may use Designated Facilities for towing and storing vehicles that are not towed or stored at the request of the City with the prior written approval of City and subject to any conditions imposed in such approval. Contractor may allow Employee and vendor parking at Designated Facilities, subject to any limitations set forth in Appendix D, so long as it does not interfere with Contractor's performance of towing and impound services to all standards and requirements of this Agreement.

(a) Customer Service Center

The Customer Service Center shall provide a location for Customers recovering vehicles in person to pay for towing and storage charges, Citation fees and penalties, and other applicable fees, and/or to process any documentation required for vehicle release. The Customer Service Center shall be open to the public twenty-four (24) hours per day, 365 days per year. If the Customer Service Center is relocated outside of the Hall of Justice at 850 Bryant Street, Contractor shall provide a security guard in any area open to the public at all times that such facility is open to the public at its sole expense. The Customer Service Center must be located at or near the Primary Storage Facility.

(b) Primary Storage Facility

(i) Authorized Facility

City hereby approves the lot at 450 7th Street as a Primary Storage Facility.

(ii) Vehicle Storage and Retrieval

Contractor must provide a covered area at the Primary Storage Facility where Customers can wait while their vehicle is being retrieved. The Primary Storage Facility shall be open twenty-four (24) hours per day, 365 days per year.

(iii) Tows to Primary Storage

Contractor shall take all towed vehicles to the Primary Storage Facility for short-term storage if they are not subject to a Police Hold or are not taken directly to the Secondary Storage Facility. Contractor shall store all towed vehicles at the Primary Storage Facility for twenty-four (24) hours after being towed, unless vehicles are required by this Agreement to be directly towed to the Secondary Storage Facility. Vehicles stored at the Primary Storage Facility may be moved to the Secondary Storage Facility after the first twenty-four (24) hours. Contractor shall ensure that it transfers vehicles from Primary to Secondary Storage frequently enough to ensure that there is, at all times, sufficient space for newly-towed vehicles to be received at the Primary Storage Facility. Contractor shall not conduct any vehicle maintenance or vehicle parts sales at the Primary Storage Facility, except for maintenance of forklifts or other lot operations equipment.

(c) Secondary Storage Facility

(i) Authorized Facility

City hereby grants Contractor a license to occupy and use the Property at Pier 70 as a Secondary Storage Facility as of the Effective Date of the Agreement during the Term of and so long as the Contractor complies with the terms and conditions of Appendices C and D.

(ii) Vehicle Storage and Retrieval

Contractor shall use the Secondary Storage Facility to store vehicles, including vehicles towed directly to the Secondary Storage Facility and most vehicles which are not picked up by the public within twenty-four (24) hours. Contractor shall conduct vehicle lien sales at this location. This facility shall be open to the public from 8:00 a.m. to 6:00 p.m. Monday through Friday. Outside of operating hours, Contractor shall secure the Secondary Storage Facility using security personnel. All vehicles must be available for retrieval from the Secondary Storage Facility by Contractor's staff twenty-four (24) hours per day, 365 days per year, with the exception of the following categories of vehicles: (1) type II vehicles (more than 8,500 lbs. but less than 26,500 lbs., trucks, buses, and unattached trailers); (2) type V vehicles (more than 26,500 lbs., trucks, buses, and unattached trailers); (5) accident vehicles (vehicles involved in accidents and towed by the SFMTA or the SFPD); and (6) abandoned/missing parts vehicles (vehicles abandoned by owner/driver and missing parts).

(iii) Tows to Secondary Storage

Contractor shall tow all Scofflaw, arrest, accident, Abandoned, recovered stolen vehicles, oversized vehicles, SFPD STOP Administrative Police Hold vehicles, disabled vehicles, dilapidated vehicles and other vehicles as directed by the City directly to the Secondary Storage Facility. The specific policies for whether vehicles are towed to the Primary Storage Facility or directly to the Secondary Storage Facility may be changed at any time subject to City's prior written agreement.

(iv) Facility Management

Contractor agrees to assume all responsibilities for use of storage facilities at Pier 70 in accordance with the License Agreement attached hereto and incorporated by reference as Appendix D, and to be bound by all covenants, terms and conditions of the Port MOU, with the exception of SFMTA's obligations under Section 2 of the Port MOU, during the term of Appendix D.

Contractor shall manage the Secondary Storage Facilities to meet the following guidelines:

- (1) Contractor shall remove vehicles that have been legally cleared for disposal on a weekly basis.
- (2) Contractor shall, at a minimum, hold a vehicle lien sale auction once a week.

- (3) Vehicles shall be placed in such a way that no more than four (4) vehicles shall need to be moved to clear a passage for any vehicle.
- (4) Two (2) feet of clearance space shall be maintained between the sides of all vehicles.
- (5) The Secondary Storage Facility personnel shall comply with all municipal, state, and federal codes and safety regulations at all times.
- (6) The Secondary Storage Facility shall be clean and maintained at all times.
- (7) Facilities shall be screened from public view except for necessary gates, and except for, during the Term of Appendix D, the fence along the waterfront of Parcel B.
- (8) Gates shall be at least eight (8) feet high and maintained in good condition.
- (9) The parking and storage surface shall be maintained in accordance with all requirements of Appendices B, C and D.
- (10) Security systems, including ample lighting and a surveillance system, shall be in place and operational at all times for the entire occupied area.
- (11) Contractor shall not permit the public to walk through the lot unescorted by an employee of Contractor.

(d) Central Dispatch

Contractor's Central Dispatch facility shall operate twenty-four (24) hours per day, 365 days per year. City hereby approves any location of Central Dispatch at the Primary Storage Facility or at Contractor's Headquarters Office, so long as Central Dispatch is located within the City and County of San Francisco.

(e) Changes in Facilities

(i) Approval

City may approve relocation of Designated Facilities, including shifting Contractor's operations between existing Designated Facilities, terminating the use of one or more Designated Facilities, or adding new Designated Facilities. Any such relocation or change to Designated Facilities shall require prior written approval of City.

(ii) Service Standards

In the event that City approves the relocation of any Designated Facility, the parties acknowledge that certain response times and maximum charges contemplated by this Agreement may require modification to take into account the changed geographic circumstances of Contractor's operations. Any written approval of a change to the Designated Facilities listed in this Agreement shall include a revised schedule of fees and/or response times to which the parties have agreed as part of the relocation, if necessary.

(iii) Consolidation

Contractor has relocated the Customer Service Center and Primary Storage Facility to a single location at the 450 7th Street site. Should City cease to use Contractor for towing services prior to February 28, 2015, the terms for City's continued use of the 450 7th Street property are set forth in Appendix G.

11.2 Property Maintenance Requirements

All costs associated with maintenance of Designated Facilities shall be the sole responsibility of the Contractor.

All open areas within the Designated Facilities used for vehicle storage shall be maintained in a clean, secure, neat, and visually presentable manner. Contractor shall not dismantle or crush vehicles or remove vehicle fluids on the any of the facilities to be used in the performance of this Agreement except in compliance with environmental regulations and the applicable requirements of Appendices B, C, and D. Any removal of fluids from vehicles shall be conducted in a manner that complies with all requirements of this Agreement, and may

only be performed by a licensed contractor, and into portable containers that are immediately removed from the facility.

11.3 Facility Security

Contractor shall store vehicles in such a manner as to prevent damage to vehicles and to vehicle contents. Contractor shall provide adequate security at the Designated Facilities to ensure that vehicles and vehicle contents are protected at all times. Upon the Effective Date of the Agreement, Contractor shall maintain a camera-based security system for Designated Facilities at its sole expense which, within ninety (90) days of the Effective Date, shall be viewable by management at SFMTA. Contractor shall provide to the City a proposed plan for security systems at all Designated Facilities within ninety (90) days of the Effective Date of this Agreement ("Security Plan"). The Security Plan shall be approved and adopted as part of the Operations Plan as provided in Section 14 of this Appendix A. All costs associated with security at Designated Facilities shall be the sole responsibility of Contractor.

11.4 Protection of Vehicle and Contents

Contractor shall provide secure storage for any personal property removed from a vehicle in its possession. Within ninety (90) days of the Effective Date of this Agreement, Contractor shall submit a plan to securely store all personal property in a vehicle, to inventory and secure personal property that is stored outside of the vehicle if it cannot be securely stored inside the vehicle, and to dispose of unclaimed personal property ("Personal Property Plan"). The Personal Property Plan shall be approved as part of the Operations Plan as provided in Section 14 of this Appendix A. Personal property not claimed by the time the associated vehicle is lien sold shall be properly disposed of by any legally authorized disposal method approved by the City. Contractor shall not be responsible for retaining personal property (for the purposes of personal property release as defined in Section 4.3) after the DMV-issued lien sale authorization date (lien sale "clear date").

11.5 Lien Vehicle Storage

All vehicles upon which Contractor issues lien holds shall be stored primarily at the Secondary Storage Facility. With the exception of Lien 3 Vehicles, which must be held for ten days after the actual date of sale pursuant to Civil Code Section 3071(k), all vehicles that are auctioned or sold for dismantling shall be removed from the Designated Facilities within one (1) week after the date of sale. Lien 3 Vehicles shall be removed from the Designated Facilities within 14 days of the date of sale.

11.6 Right to Inspect

Any authorized representative of the City has the right to inspect the Designated Facilities at all times for the purpose of evaluating Contractor's performance pursuant to this Agreement. City officials and inspectors shall have the right to conduct periodic site visits, during reasonable business hours, to inspect for permit compliance or to respond to citizens' complaints. City officials and representatives shall have unrestricted access to all of the Designated Facilities subject to permits or licenses to make whatever announced or unannounced visits they deem appropriate.

12. Fees, Payments and Credits

12.1 Payments Due to City

(a) Referral Fee

Contractor shall submit to the City a Referral Fee of \$20 per tow, excluding dropped tows as described in Appendix A, Section 4.2. The Referral Fee shall be the same for every type of vehicle, and shall increase each twelve (12) month period on each July 1st by the Consumer Price Index for the San Francisco Region as published by the United States Department of Labor, Bureau of Labor Statistics on January 1. Adjustments will be rounded to the nearest twenty-five cents (\$0.25). No Referral Fee shall be paid for:

(1) Vehicles owned by the City under the jurisdiction of the SFMTA or the SFPD, or any other Courtesy Tow performed pursuant to Section 2.4 of this Appendix A, and

(2) Vehicles for which a waiver of towing, storage, transfer and/or lien fees is issued by SFMTA, DPH or SFPD.

(b) Percentage Fee

Contractor shall submit to the City a percentage fee of one percent (1%) on annual Gross Revenues from all money collected during the term of this Agreement. This fee shall be initially paid in the fifteenth (15^{th}) month after the Agreement is signed, in the thirteenth (13^{th}) month thereafter for the previous twelve (12) month term, and yearly thereafter.

(c) SFMTA Administrative Fees

Prior to releasing the vehicle, Contractor shall collect a pass-through SFMTA Administrative Towing Fee for all vehicles recovered by the vehicle owner. The amount of the SFMTA Administrative Towing Fee is subject to change in accordance with the provisions of San Francisco Transportation Code § 305. In addition, prior to releasing the vehicle, Contractor shall collect a pass-through daily SFMTA Administrative Storage Fee for stored vehicles based upon the number of days the vehicle has been stored prior to recovery by the vehicle owner. No Administrative Fee shall be collected for:

- (1) Any vehicle owned by the City under the jurisdiction of the SFMTA or the SFPD, or any other Courtesy Tow performed pursuant to Section 2.4 of this Appendix A.
- (2) Any vehicle for which the Customer produces a written waiver of the SFMTA Administrative Fee issued by SFMTA, DPH or the SFPD.

(d) SFPD Traffic Offender Fee

If applicable, when a vehicle is sold at a lien sale and there are funds to satisfy all other fees as defined in Section 12.3(c) of this Appendix A, then Contractor shall pay to the SFPD or into an account designated by the SFPD an SFPD Traffic Offender Fee in an amount set by the San Francisco Police Commission. City and Contractor may agree in writing to utilize a mechanism other than the process described in Section 12.1(g) for collection of the SFPD Traffic Offender Fee.

(e) Citation Fees

Contractor shall collect payments of Citation fees from Customers with towed vehicles and from members of the public whose vehicles have not been towed, in accordance with all requirements set forth in this Agreement.

(f) Liquidated Damages and Fines

Contractor shall pay to City the amounts of any liquidated damages or fines assessed pursuant to this Appendix A, Section 15 and Appendix D, Section 6.7.

(g) Deposit of Fees Due to City

Except as otherwise specified herein, Contractor shall deposit all funds collected under this Section 12 within twenty-four (24) hours of receipt into an account specified by the City, Monday through Friday, not including weekends and holidays. Any funds with a deadline for deposit which falls on a weekend or a holiday shall be deposited no later than the next business day. All funds due to City under this Section shall be paid by Contractor without prior demand by the City and without any deduction, setoff, or counterclaim whatsoever, except as expressly provided herein. The parties may agree upon alternative procedures for Contractor's payment to City, but any such change must be approved in advance, by City, in writing.

(h) Payment Shortages

If Contractor fails to collect all amounts due from a Customer, Contractor shall be responsible to reimburse the City for any amounts not collected as required herein, unless the failure is caused solely by the negligence of City or a failure of the CMS. Contractor shall follow any procedures required by the City to report overages or shortages.

12.2 Credits Due to Contractor

(a) City Waivers

In the event that (i) SFMTA, DPH or the SFPD determines pursuant to a post-storage hearing as required by Vehicle Code § 22852 that the towing, storage, transfer, lien and/or other fees shall be waived for a vehicle, or (ii) SFMTA, DPH or the SFPD waives the fees for the towing, storage, transfer and/or lien of a vehicle, or (iii) SFMTA or the SFPD waives the fees for the towing and storage of a vehicle for one of the reasons enumerated in §§ 10C.1, 10C.8 or 10C.8-1 of the San Francisco Administrative Code, then no such fees shall be charged by Contractor to the owner or operator of such vehicle. In the event that the owner or operator of a vehicle has paid Contractor for towing, storage, transfer, lien and/or other fees and the City subsequently waives the tow, storage, transfer, lien and/or other fees for that owner or operator, then Contractor shall directly reimburse the owner or operator in full the amounts previously paid to Contractor for such vehicle. When the City waives towing, storage, transfer and/or lien fees as provided for in this section, the City shall pay Contractor only the towing, storage, transfer and/or lien fees that would have been owed by the vehicle owner or operator, and such fees shall not include any SFMTA Administrative Fees as Contractor might otherwise charge. When the City waives storage fees as provided in this Section, the City shall pay Contractor storage fees for each such vehicle as set forth above for the first three (3) days of storage. There shall be no storage charge for the fourth (4th) through the ninth (9th) days of storage. For the tenth (10th) storage day and all days thereafter, City shall pay Contractor ten dollars (\$10.00) per day for the storage of such vehicles. Adjustments and credits and payments due to Contractor as a result of City waivers shall be calculated and submitted to the City through the City's claims process and are to be paid within one (1) month from the date of submission by the Contractor. There shall be no late payment charges or interest assessed against City for late payment.

When SFMTA or the SFPD orders Contractor to release a vehicle pursuant to Vehicle Code § 22654(e) relating to authorization for moving a vehicle otherwise lawfully parked, City shall pay the cost of the tow and storage charges for a period not to exceed seventy-two (72) hours.

(b) Police Investigative Hold Storage Fee

Contractor shall not charge or receive any fee or other reimbursement or credit from the City for the towing, storage, transfer or lien of any Police Investigative Hold vehicle, except when the number of Police Investigative Hold vehicles in storage on any day exceeds limit of 350 vehicles. SFPD shall pay Contractor a storage fee of ten dollars (\$10.00) per day per vehicle in excess of 350 Police Investigative Hold vehicles in storage at Contractor's facilities at any one time. The SFPD will be responsible for these fees and the Contractor shall not deduct any Police Hold storage fees due from any money owed to SFMTA under this contract. For the purposes of this Section 12.2(b), "Police Investigative Hold" vehicles shall not include NO ID vehicles described in this Appendix A, Section 4.5.

Notwithstanding the foregoing, in the event a vehicle is towed without a Police Investigative Hold and a Police Investigative Hold is subsequently placed on the vehicle, the SFPD shall immediately notify Contractor in writing of the Police Investigative Hold status change. The SFPD shall pay or require the owner or operator of the vehicle to pay the Contractor only the tow, transfer, lien and/or other fees accumulated from the date of tow to the date the Contractor is notified of the Police Investigative Hold by the SFPD and that would otherwise be owed by the vehicle owner or operator, and such fees shall not include any SFPD or SFMTA Administrative Fees. In addition, the SFPD shall pay or require the owner or operator to pay storage fees for any days from the tow date to the date Contractor is notified of the Police Investigative Hold by the SFPD. For storage amounts paid by the SFPD, storage charges shall be calculated for each such vehicle as set forth above for the first three (3) days of storage. There shall be no storage charge for the fourth (4th) through the ninth (9th) days of storage. For the tenth (10th) storage day and all days thereafter, the SFPD shall pay Contractor ten dollars (\$10.00) per day for the storage of such vehicles.

(c) Non-Towed Vehicle Citation Collection Fees

Contractor shall be entitled to a credit for any collection fees due pursuant to Section 6.2(c) of this Appendix A.

(d) Other Offset Allowances

At any time during the Term of this Agreement, City may elect to fund certain property maintenance, construction, improvements, systems development or staffing related to City towing and impoundment operations that are not the responsibility of Contractor under this Agreement ("Projects"). City may require Contractor to implement any such Project and offset actual costs of the Project against funds owed to the City.

No Project may be implemented without the prior written approval of the City, and all Project implementation must be in accordance with specifications, maximum costs and all other requirements provided in writing by City. Contractor shall comply with City's direction as to which category of funds collected pursuant to this Section 12 may be used to offset Project costs.

12.3 Charges to Customers

(a) Maximum Towing and Storage Charges

No lien for storage of a towed vehicle may exceed authorized charges for the maximum period of storage allowed by applicable laws, including but not limited to Vehicle Code §§ 22851.6, 22851.3, California Civil Code §§ 3067-3074 and any other applicable statutes enacted during the term of this Agreement. Contractor shall not charge any Customer amounts in excess of the amounts set forth in the rate schedule adopted pursuant to this Agreement and attached hereto as Appendix F without SFMTA's prior written approval of any change to existing rates and fees or the imposition of new fees. The rates and charges established by this Agreement shall apply to all vehicles handled by Contractor within the City, whether the vehicle is towed at the request of SFMTA, SFPD or DPH or is stored on any Designated Facility.

Towing and storage charges are subject to adjustment annually, on July 1, in direct proportion to the increase or decrease in the "Consumer Price Index for Urban Wage Earners and Clerical Workers", unadjusted data for all items for the twelve (12) months ending April of the current year, as published by the United States Department of Labor, Bureau of Labor Statistics. Adjusted charges will be rounded to the nearest twenty-five cents (\$0.25).

In the event storage charges accrue on a vehicle because of Contractor's failure to provide SFMTA or the SFPD, as applicable, with either an "Unreleased Vehicle Report" (as required in Appendix A, Section 13.2(a)(ii)), a vehicle description or a verification of the VIN, such charges shall be null and void and neither the City nor the owner of said vehicle shall be responsible for those charges.

(b) Vehicle Transfer Fee

If vehicle is not retrieved within twenty-four (24) hours of the tow, Contractor may charge a one-time vehicle transfer fee as set forth in Appendix F to move the vehicle between Designated Facilities. Contractor shall not assess a transfer fee for any vehicle that is towed directly to the Secondary Storage Facility.

(c) Application of Funds Collected

When vehicle title is transferred to the Contractor and the vehicle is subsequently sold, any funds collected from the sale of a vehicle shall be considered to have gone through the lien sale process using the average days from the tow date to lien sale date for the relevant vehicle classification (Lien 2 or 3) for the previous calendar year. The sale proceeds shall be applied first to paying storage, towing and legally authorized lien processing costs, and all remaining funds shall be applied toward payment of 1) Delinquent Citations, 2) the SFMTA Administrative Fee, and 3) the SFPD Traffic Offender Fee, in that order.

12.4 Additional Fees Proposed by Contractor

Except for periodic adjustments provided for in Section 12.3(a) above, Contractor must notify the City in advance of any proposed fee that relates to any services performed under this Agreement or services performed for owners of vehicles towed or buyers of vehicles sold pursuant to this Agreement. All Contractor fees must be approved by the City.

12.5 Account Reconciliation

Contractor shall monitor monthly any monies due to City and any monies due to Contractor and report such amounts on the Monthly Finance Report. At the end of each contract year, City and Contractor shall review all Monthly Finance Reports for the preceding year and conduct a formal reconciliation of payments, reimbursements and credits due under this Agreement. Upon verification by SFMTA in the course of the formal reconciliation, the City shall reduce any payments owed by Contractor to City by the amount of any credits due to Contractor from City. If the payments due to the City are greater than any credit due to the Contractor, SFMTA shall bill the Contractor for the remaining payment balance owing. Contractor shall remit payment for the balance due within ten (10) days of the billing date. If the payments due to the City are less than the total credits due to the Contractor, City shall either, in City's sole discretion, pay the Contractor for the amount owed or hold over a credit to be applied against the next payment due to the Contractor. Any challenges to the amount owed by City must be memorialized in writing within 60 days of the resolution of the formal reconciliation process. Nothing in this Section 12 waives any rights of City under Section 7 of the Master Agreement.

12.6 Contractor Compliance Audit Fund

(a) Contractor Contribution

Within 30 days of the effective date of this Second Amendment, and within 30 days of each anniversary of that effective date during the Term of the Agreement (the "Anniversary Date"), the Contractor shall deposit money (the "Audit Payment") into a Contractor Compliance Audit Fund to be used by the SFMTA for the purposes of auditing the Contractor in areas to be determined by the SFMTA at the time of each audit. The initial Audit Payment shall be \$40,000.

(b) Contribution Adjustment

The Audit Payment shall be adjusted in direct proportion to the percentage increase in the current Consumer Price Index for Urban Wage Earners for the San Francisco Bay Area ("CPI") for the month immediately preceding the applicable Anniversary Date ("Current Index") over the CPI for the month of June 2010 ("Base Index"). In no case shall the Audit Payment, as adjusted, be less than the Audit Payment for the previous year. If the Current Index has increased over the Base Index, the amount of the Audit Payment shall be determined by multiplying the amount of the initial Audit Payment by a fraction, the numerator of which is the Current Index and the denominator of which is the Base Index, as follows:

Current Index

Base Index x initial Audit Payment = Adjusted Audit Payment

Any monies in the Contractor Compliance Audit Fund that are unused during the term of the Agreement shall be returned to the Contractor within 60 days of termination of this Agreement.

13. Reporting and Records Requirements

13.1 TVMS Records

(a) Records of Transaction

Contractor shall maintain consecutively numbered electronic records of each transaction involving the removal, impoundment, and disposition of all vehicles towed pursuant to this Agreement. Each electronic record shall contain the following information:

- (1) Date and time of Tow Request;
- (2) Date, time and location of tow and identity of Tow Car operator;
- (3) Make, model, year and VIN of vehicle towed;
- (4) Name and address of individual to whom vehicle is released or sold;
- (5) Inclusive dates of and charges for impoundment; and
- (6) Date and manner of vehicle disposition and income received.

(b) Weekly Management Report

Contractor shall provide a weekly Management Report to include the number of Tow Requests, number of Claims filed, number of vehicles sold at lien sales and the number of vehicles returned to the owner during the previous seven (7) day period.

(c) Monthly Management Reports

APPENDIX A: SCOPE OF WORK n:\ptc1\as2010\0900371\00630742.doc

Page 37 of 45

Contractor shall also provide a monthly Management Report in a form approved by the City. The approved format for each reporting item may include (but is not limited to), standardized reports (in paper or electronic format), utilities to query and download data on a variable periodic basis, or direct access the TVMS querying and reporting capabilities for ad hoc use by the City. Except as otherwise authorized, in writing, by City, Contractor shall provide the monthly Management Reports by the 10th of each month, or on the next business day if the 10th is a weekend or a holiday recognized by the City. The Reports shall include, but shall not be limited to, items such as:

- (1) Tow and service response times (Monthly Response Time Report)
- (2) Number and type of tows (Monthly Tow Summary Report)
- (3) Information on vehicles retrieved by date and time (Monthly Released Vehicles Report)
- (4) Information on vehicles awaiting lien clearance (Monthly Lien Status Report)
- (5) Information on vehicles lien sold by number of days in storage and Lien Category (Monthly Lien Sold Summary Report)
- (6) Information on vehicles sold to dismantlers (Monthly Lien Sold Buyer Report)
- (7) Information on vehicles purchased by the public (Monthly Lien Sold Buyer Info Report)
- (8) Number of vehicles in storage by reason category for tow, by department requesting tow, or by date towed (Monthly Storage Summary Report)
- (9) Monthly listing of all details in the TVMS system for every tow performed that month (Monthly Tow Detail Report)
- (10) Report of transfers of vehicles between Designated Facilities (Monthly Vehicle Transfer Report)
 - (d) Auction Report

Contractor shall provide a weekly Auction Report to the City that includes detailed information for all lien sold vehicles. For each vehicle, the Report must include the following information:

- (1) Vehicle Identification Number (VIN)
- (2) License plate number
- (3) Year
- (4) Model
- (5) Lien Category
- (6) Actual sale amount
- (7) Purchaser name and address
- (8) A detailed description of the distribution of proceeds from vehicle sale
- (9) Identify vehicles not sold that are held for future lien sale or for disposal

13.2 Minimum Required Reporting

- (a) Daily Reports
- (1) Police Hold Report of Police Hold vehicles in excess of 325 vehicles (§ 4.6(d))
- (2) Unreleased Vehicle Report

- (b) Weekly Reports
- (1) Auctioneer Report (§ 8.9(b))
- (2) Weekly Management Report (§ 13.1(b))
- $(3) \qquad \text{Auction Report (§ 13.1(d))}$
- (4) Police Hold Report (§ 4.6(d))
- (5) Lien 1/Abandoned Vehicle Report
 - (c) Monthly Reports
- (1) Customer Service Report ($\S 5.1(c)(v)$)
- (2) Claim/Complaint Status Report (§5.2(c))
- (3) Management Contact List (§ 8.7)
- (4) Monthly Management Report (§ 13.1(c))
 - (d) Quarterly Reports
- (1) Subcontractor Performance Audit Report (§ 8.2(c))
 - (e) Additional Reports

City may provide Contractor with a list of any additional required reports. Once the City submits its list of required reports to Contractor, Contractor shall have thirty (30) days to provide the requested reports to the City unless otherwise specified. The City reserves the right to request up to twenty (20) new reports or modifications to existing reports during the term of this Agreement.

13.3 Records Maintenance

This Section 13.3 shall survive termination of this Agreement. Contractor shall maintain digital photos that are not associated with any Claim for a period of two (2) years, except as otherwise agreed by City in writing. Digital photos related to a Claim shall be retained with the Claim file for a period of five (5) years. The retention of audio tapes is governed by Section 1.3(b) of this Appendix A. Contractor shall maintain all other Records generated pursuant to this Agreement for a period of five (5) years following expiration of this Agreement.

Contractor shall respond to requests from City for information regarding services provided under this Agreement within forty-eight (48) hours. If the Records requested are not capable of being produced within that time, the forty-eight (48) hour response shall indicate where the Records are located and when they can be made available for City's review, which shall in no event be longer than fourteen (14) days unless otherwise agreed. Contractor shall respond to requests for Records from City by providing Records in any format in which they are maintained, including but not limited to paper, audio and electronic formats.

14. **Operations Plan**

14.1 General Provisions

Contractor shall submit the elements of an Operations Plan as listed in Appendix B in accordance with the requirements of this Agreement. The final approved version of any Operations Plan element and any subsequent modifications approved in accordance with this Section shall define service standards for the performance of this Agreement, and are hereby incorporated into this Agreement as Appendix B as though fully set forth herein.

14.2 Approval Process

All elements of the Operations Plan shall be subject to City review and approval. All Operations Plan elements must be initially submitted no later than the deadlines set forth in Appendix B for each Operations Plan element. The deadline for any Operations Plan element described in the Agreement may be extended by written approval

of SFMTA upon the request of and a showing of good cause by Contractor; however, the extension of any deadline for the Operation Plan elements designated in Appendix B as Pier 70 Operation Plan Elements also requires written approval of the Port during the term of Appendix D. City shall have sixty (60) days to review each element submitted, and either approve it as submitted or request revisions. Contractor shall respond to a request for revisions within twenty (20) days. City will have fifteen (15) days to either approve the revised Plan element or request further revisions. Contractor and City shall from this point on have five (5) days to either approve the revised Plan element as submitted, submit further requests for revisions or to respond to requests for revisions. Each revision must reflect tracking of document versions, including date and source of revisions, and each exchange of versions between the parties shall be accompanied by an executed document substantially in the form of Appendix B.

14.3 Line Item Approval

Pending the completion of the approval process of an entire Operations Plan element, Contractor may request line-item approval of certain portions of that Operations Plan element. If City does not respond to such request for line-item approval by Contractor within fourteen (14) days, the request for line-item approval shall be deemed denied.

14.4 Final Operations Plan

When an element of the Operations Plan is accepted by City, the final version of that element must be submitted to City in PDF format. Following City acceptance of Plan elements, the final Operations Plan and any subsequent modifications shall be distributed to all subcontractors.

14.5 Subsequent Modifications to Operations Plan

Contractor shall review the Operations Plan every six (6) months, and shall propose modifications as necessary to any element of the Operations Plan needed to improve service delivery. Modifications to the Operations Plan shall be approved through the process described in this Section 14. Each finally approved Operations Plan modification must identify the document version and date. Any subsequent modification of the Operations Plan shall supersede the prior version and be incorporated into this Agreement by reference when approved in accordance with this Section.

15. Liquidated Damages

15.1 Assessment of Liquidated Damages

Liquidated damages as described in this Section may be imposed by City for violations of the provisions of this Agreement. Failure by City to impose liquidated damages for specified violations shall not be a waiver of the right to enforce this Section, nor shall it constitute a waiver of any other right of City under this Agreement. No single act or omission by Contractor which incurs fines under Section 6.7 of Appendix D may be used as the basis for assessing any liquidated damages under this Section 15. The total amount of liquidated damages that City may collect under this Appendix A, Section 15 shall be limited to three hundred sixty thousand dollars (\$360,000) per year. For the purposes of this Section 15, written notice by City of a violation shall constitute enforcement even though the City may not assess liquidated damages at the time of such initial written notice of violation.

15.2 Damages Calculation

All contract violations listed in this Section 15 are subject to the \$360,000 per year limit set forth above. In addition, each type of violation which is subject to liquidated damages under this Section 15 is followed by the designation of one of the following categories: [A], [B], [C] or [D], and depending on the category shall be subject to the following definitions and limitations:

- [A]: The measure of liquidated damages in this category [A] shall be subject to no limitation other than the \$360,000 per year limitation stated above.
- **[B]:** Liquidated damages in this category **[B]** may only be enforced within forty-five (45) days of the act or omission which gave rise to the City's right to collect liquidated damages.

- [C]: Liquidated damages in this category [C] may be assessed for a period of no more than forty-five (45) days for continuing violations.
- **[D]**: Liquidated damages in this category **[D]** may only be assessed for the immediately preceding audit period and City must provide Contractor with notice of any violation within sixty (60) days of City's completion of the audit report pursuant to Appendix A, Section 12.6.

15.3 Staffing

If Contractor fails to comply with the following staffing requirements set out in the Agreement, excluding requirements related to subcontractors, City may collect damages of \$250 per occurrence, not to exceed \$1,000 per day for each day that the required staff is not on duty, and \$250 for each eight (8) hour training session per individual that is not provided as required by the Agreement. Requirements that are subject to this subsection include:

- (1) Failure to provide required staffing at Central Dispatch (§ 1.2) [B]
- (2) Failure to adhere to the window staffing requirements, except wait time requirement (§ 5.1(c)(ii)) [B]
- (3) Failure to have a Customer Service Representative on duty during designated hours (§ 5.1(c)(iii)) [B]
- (4) Failure to comply with the minimum training standards (§ 8.3) [D]
- (5) Failure of a manager to be available (\S 8.7) [B]

15.4 Subcontracting

If Contractor fails to comply with the following requirements for the use of subcontractors as set out in the Agreement, not including equipment and communication requirements related to subcontractors, City may collect damages in the amounts specified below:

- (1) Failure to include the requirement that subcontractors hold current Tow Car and Tow Firm permits in subcontracts and to check compliance at the time of executing the subcontract. (§ 8.2(d)): \$500 per occurrence **[D]**
- (2) Failure to perform quarterly audits of permits and license status of Tow Car operators (§ 8.2(c)): \$500 per occurrence [D]
- (3) Violation of the uniform requirements (§ 8. 2(e)): \$100 per occurrence [B]
- (4) Failure to replace the auctioneer annually without the City's prior, written approval (§8.9(a)): \$1000 per occurrence [**B**]

15.5 Equipment

If Contractor fails to comply with the following requirements for Tow Equipment set out in the Agreement, City may collect damages in the amounts specified below:

- Failure to provide two (2) dedicated telephone lines for more than an hour in a twenty-four (24) hour period within thirty (30) days of Effective Date of Agreement and every day thereafter (§ 1.3(a)): \$500 per day [B]
- (2) Failure to provide functional recording system and to store recordings for 120 days or longer time as required by City (§ 1.3(b)): \$500 per day [C]
- (3) Failure to provide Tow Cars at site of tow with appropriate equipment (§9.1): \$250, per tow [B]
- (4) Failure to have GPS tracking system in place in accordance with specified time limits and with all required software licenses in place for required GPS functionality (§ 9.2):
 \$500 per day of delay [B]

- (5) Failure of the GPS to operate ninety-seven percent (97%) of the time during a monthly reporting period (§ 9.2): \$500 per occurrence **[D]**
- (6) Failure to provide radios (§ 9.3): \$150 per reported occurrence [B]
- (7) Failure to provide the required hardware, software, and data lines to create and store electronic tow inventory slips within ninety (90) days of the Effective Date (§ 9.6):
 \$500 per day for each day of delay [B]
- (8) Failure to maintain adequate security in accordance with the Security Plan (§ 11.3):
 \$250 per occurrence, as defined in the Security Plan [D]
- (9) Failure to maintain standards and connections required for computer hardware and software system within specified time limits (§ 10): \$500 per day delayed [B]
- (10) Failure to have handheld device solution in place as required (§ 10.5): \$500 per day of delay [B]
- (11) Failure of the handheld device solution to operate ninety-seven percent (97%) of the time during a monthly reporting period (§ 10.5): \$500 per occurrence [D]
- (12) Failure to have IVR Telephone system in place within sixty (60) days of Effective Date (§ 6.3): \$500 per day delay [B]

15.6 Response Times

If Contractor fails to comply with the following response times or deadlines set forth in the Agreement City may collect damages in the amounts specified below:

- (1) Failure to answer call within 30 seconds or less with no busy signal for 95% of the calls, and within 90 seconds for the remaining 5% of the calls, during a given month (§ 1.2): \$250 for each month in which Contractor fails to answer 95% of the calls within 30 seconds, and an additional \$100 for each percentage point by which contractor fails to meet the 95% requirement in a given month; \$250 for each month in which Contractor fails to answer the remaining 5% of the calls within 90 seconds [B]
- (2) Failure to enter data into tow database within specified time limits ninety percent (90%) of the time (§ 3.1): \$500 per audit period in which the ninety percent (90%) goal is not met, plus \$250 for each additional percentage point by which Contractor fails to meet the ninety (90%) percent goal [D]
- (3) Failure during any given month to respond to at least 94% of all Dispatch Tows within the designated time requirements (§ 2.1(a)), including any approved extension of time: \$500 for each month in which Contractor fails to timely respond 94% of the time, and an additional \$250 for each percentage point by which Contractor fails to meet the 94% requirement in a given month [D]
- (4) Failure to respond to an Expedited Tow request within the designated time requirements, including any City-approved time extension (§ 2.2): \$500 per calendar day. [B]
- (5) Failure to respond to a Scheduled Tow Request (§2.1) or a request for a Regional Sweep (§ 2.3) at the agreed upon start time and location: \$100 credit for each twenty (20) minute period or portion thereof, not to exceed \$500 per calendar day. **[B]**

15.7 Record Keeping and Reporting Requirements

If Contractor fails to meet reporting and record keeping requirements listed below, City may collect damages in the following amounts:

- (1) Failure to submit any report required by Section 13 or maintain any record required by this Agreement: \$50 per day for each day that the record is not provided or maintained or the required report is overdue, not to exceed \$250 per month per report. City agrees to notify Contractor if City becomes aware of any report required by Section 13 of this Appendix A that is overdue. [B]
- (2) Failure to provide audio records within twenty-four (24) hours of City's request (§ 1.3(b)): \$50 per day for each day that the record is not provided [C]
- (3) Failure to furnish audit or waiver (authorizing DMV to release audits of Contractor) to the City within specified time limits (§ 12.6): \$100 per day [C]
- (4) Failure to submit copies of subcontractor agreements with required proof of insurance documents within five (5) days of the Effective Date of Agreement, or failure to submit subcontract amendments within five (5) days of any such amendments (§ 8.2(a)): \$100 per day per subcontract, up to \$3,000 per month for all subcontracts [D]
- (5) Failure to notify the City prior to additions and deletions of towing subcontractor (§ 8.2(a)): \$500 per occurrence [D]
- (6) Failure to comply with notice requirements for any personnel changes within specified time limits (§ 8.6): \$100 per occurrence [D]
- (7) Failure to notify the Contract Monitor of an auction at least three (3) days prior to any auction, as specified, or to provide said Contract Monitor with the report information requested (§ 8.8(b)) \$100 per occurrence [**B**]
- (8) Failure to enter a towed vehicle record within three (3) days of the date of the tow (§ 10): \$100 per towed vehicle [A]
- (9) Failure to maintain the functionality of the TVMS ninety-nine percent (99%) of the time during a monthly reporting period as specified (§ 10.3): \$500 per monthly reporting period in which the ninety-nine percent (99%) goal is not met, plus \$250 for each additional percentage point by which Contractor fails to meet the ninety-nine (99%) percent goal [B]
- (10) Failure to print and distribute receipts and notices as required by the City or any local, state or federal laws (§§ 6.6, 7.3): \$250 per occurrence **[D]**

15.8 Plan Submittals

If Contractor fails to submit any element of the Operations Plan in accordance with the requirements of this Agreement and the deadlines for initial document submittals and revisions set forth in Section 14, Appendix A of this Agreement, City may collect damages of \$250 per day for each day that the Operations Plan element is overdue. **[B]**

15.9 Customer Service Standards

If Contractor fails to meet the following Customer service standards as defined in the Agreement City may collect damages in the amounts specified below:

- (1) Failure to have functional internet site elements implemented within specified time limits (§ 5.1(d), 6.4): \$500 per day delay **[B]**
- (2) Failure to keep internet site functional for public and City use at least ninety-seven percent (97%) of the time during a monthly reporting period between the hours of 6:00 a.m. and 12:00 a.m. (§§ 5.1(d), 6.4): \$250 per month in which the ninety-seven percent (97%) goal is not met, plus \$250 for each additional percentage point by which Contractor fails to meet the ninety-seven percent (97%) goal **[D]**
- (3) Failure to accept the specified forms of payment (§ 6.1): \$500 per occurrence [D]

- (4) Failure to accept a payment of Citation fees for vehicles that have not been towed (§ 6.2(c)): \$100 per transaction [D]
- (5) Failure to have information available to the public as required, or as required by any local, state or federal laws (§ 5.3): \$250 per posting requirement per day **[B]**
- (6) Failure to meet the standards for telephone operator response time for customer service calls(§ 5.1(b)): \$100 for each percentage point by which Contractor fails to meet any of the percentage level requirements in a given month **[B]**
- (7) Failure to meet the standards for "in person" customer service wait times (§5.1(c)(ii):
 \$100 for each percentage point by which Contractor fails to meet any of the percentage level requirements in a given month [B]
- (8) Failure to respond to a Customer Complaint within seven (7) days, or a Customer Claim within fourteen (14) days (§ 5.2): \$100 per day for each day delayed [B]
- (9) Failure to release a vehicle to Customer's possession within one (1) hour of a Customer's compliance with all requirements for vehicle release (§ 5.1(c)(i), 4.4):
 \$100 per occurrence, \$50 credit to Customer per hour/fraction of hour of delay, not to exceed \$45,000 per audit cycle [D]
- (10) Failure to provide free shuttle service, one-way taxi fare or vehicle retrieval service (§ 4.4): \$250 per occurrence, not to exceed \$45,000 per audit cycle [D]
- (11) Failure to provide hotline service for technical assistance to City. (§§ 9.6, 10.3): \$200 per day [B]
- (12) Failure to have remote electronic access tow inventory slip information available to City for more than one (1) day (§ 9.6): \$200 per day [B]
- (13) At any time following the implementation of the IVR system, failure of the IVR system to function three percent (3%) of the time from 6:00 a.m. to 12:00 a.m. during a monthly reporting period (§ 6.3): \$250 per occurrence \$250 per month in which the ninety-seven percent (97%) goal is not met, plus \$250 for each additional percentage point by which Contractor fails to meet the ninety-seven percent (97%) goal [D]

15.10 Vehicle Handling Requirements

If Contractor fails to meet the following vehicle intake or handling requirements set forth in the Agreement City may collect damages in the amounts specified below:

- (1) Contractor's failure to drop a vehicle when instructed to do so by the City officer present at the scene of the tow (§ 4.2): \$500 per occurrence **[B]**
- (2) Failure to provide Courtesy, Tow or roadside assistance services to SFMTA or SFPD vehicles (§ 2.4): \$100 per occurrence plus reimbursement of any expense associated with City's procurement of towing or roadside assistance services for vehicles subject to Courtesy Tows **[B]**
- (3) Failure to hold a vehicle with Delinquent Citations that have been assessed penalties (§ 6.2(b)): an amount equal to the total amount of Delinquent Citations owed but not collected, plus \$100 [D]
- (4) Failure to visually inspect impounded vehicles and collect or confirm VIN within specified deadlines (§ 3.3(a)): \$500 per occurrence, not to exceed \$45,000 per audit cycle [D]
- (5) Failure to notify the SFPD within specified time limits of any impounded vehicles in its possession where the license plate and the VIN do not match (§ 3.3(a)): \$500 per occurrence **[D]**

- (6) Failure to hold weekly lien sale auctions (§ 7.2): \$1,000 per occurrence [B]
- (7) If Contractor releases, sells or disposes of any vehicle in violation of the requirements of the Vehicle Code, or otherwise loses possession of or is unable to locate within its possession a vehicle that it has towed under this Agreement and if City directs Contractor to resolve a Claim after sixty (60) days by paying blue book value of the vehicle to the owner (§ 4.1): \$1,000 per occurrence [A]
- (8) Failure to provide City with information to update CMS with information on intake or release of vehicles within specified time limits (§9.5): \$150 per day that release information is not provided, and \$20 for each one (1) hour period that intake information is not provided, not to exceed \$100 per towed vehicle, not to exceed \$45,000 per year [B]
- (9) Failure to maintain accurate personal property inventory in accordance with Personal Property Plan (§ 11.4): \$250 per occurrence as defined in the Personal Property Plan, not to exceed \$45,000 per audit cycle [D]
- (10) Failure to provide adequate security for personal property removed from towed vehicle (§ 11.4): \$500 per occurrence as defined in the Personal Property Plan [D]
- (11) Failure to provide secured Police Hold storage facility as specified (§ 4.6(a)): \$1000 per day [A]
- (12) Failure to remove vehicle from Secondary Storage Facility within one (1) week of sale (§ 11.5): current daily storage fee per vehicle per day [B]
- (13) Contractor's sale of a vehicle at an auction conducted under the Agreement to a Restricted Auction Participant in conjunction with Contractor's violation of the provisions of Section 8.9(c): \$1,000 per prescribed sale or the blue book value of the vehicle sold, whichever is greater, plus all outstanding administrative fees and citations, if any, on the vehicle. [A]

15.11 Financial Obligations

If Contractor fails to meet the following financial obligations set forth in the Agreement City may collect damages in the amounts specified below:

- (1) Failure to reimburse the City within five (5) days of the due date for the cost of the Contract Monitor (§ 8.8): \$50 per day [B]
- (2) Failure to reimburse the City within five (5) days of the due date for the cost of a City-appointed vehicle auctioneer (§ 8.9): \$50 per day [B]
- (3) Failure to deposit Referral Fee within specified time limits (§ 12.1(a), § 12.1(g)):
 \$100 per day of delay in depositing fee [B]
- (4) Failure to deposit percentage fee within specified time limits (§ 12.1(b)): \$100 per day of delay in depositing fee, beginning on the fifteenth (15th) month after the Effective Date of the Agreement **[B]**
- (5) Failure to deposit SFMTA Administrative Fees within specified time limits (§ 12.1(c), § 12.1(g): \$100 per day of delay in depositing fee [**B**]
- (6) Failure to pay balance due within ten (10) days of receiving bill from City (§ 12.5):
 \$100 for each day of delay in paying billed amount [B]
- (7) Failure to collect any Mandatory Fee due to internet site or IVR system errors (§§ 6.3, 6.4): amount of uncollected Mandatory Fees due to the City which have not been assessed as liquidated damages pursuant to another subsection of this Section 15 [D]

- (8) Failure to deposit all or any part of collected funds not identified in any other paragraph of this Section 15.11 within specified time limits (§12.1(g)): \$500 per day delayed [B]
- (9) Failure to maintain minimum balance in Claims Fund (Master Agreement § 12.4):
 \$500 per day that balance is below minimum requirement [D]
- (10) Failure to maintain minimum balance in Maintenance Deposit (Master Agreement § 12.3): ten percent (10%) APR or the maximum allowed by California law, whichever is greater, per day that balance is below minimum requirement, paid on the deficiency [D]
- (11) Failure to timely make the annual payment to reimburse the City for the cost of the Contract Monitor (§ 8.9(a)): \$100 per day for each day that payment is delayed [B]
- (12) Failure to make timely annual payment into the Contractor Compliance Audit Fund (§12.6): \$100 per day for each day that payment is delayed [B]

APPENDIX H TO THE TOWING AGREEMENT BY AND BETWEEN THE CITY AND COUNTY OF SAN FRANCISCO AND

TEGSCO, LLC, d.b.a. SAN FRANCISCO AUTORETURN Granting a Revocable License for the Use of Certain Property

TABLE OF CONTENTS

1. BASIC LICENSE INFORMATION 2 2. PREMISES 3 2.1 LICENSE PREMISES 3 2.2 SUBORDINATE TO LEASE 3 2.3 PERFORMANCE OF LEASE OBLIGATIONS. 3 2.4 AMENDMENTS TO LEASE 3 2.5 CONTACTS WITH PROLOGIS. 3 2.6 PROLOGIS DUTIES. 3 2.7 TERMINATION OF LEASE 3 3. INSPECTION OF PROPERTY; AS IS CONDITION 4 3.1 INSPECTION OF PROPERTY. 4 3.2 As IS CONDITION. 4 3.1 INSPECTION OF PROPERTY. 5 3.3 ACCESS INSPECTION DISCLOSURE 5 3.4 LICENSE TERM 5 4 LICENSE TERM 5 5.1 BASE FEE 5 5.1 BASE FEE 5 5.2 ADJUSTMENTS IN BASE FEE 5 5.4 DEPAULT INTEREST 6 5.5 DEDUCTION FROM AMOUNTS DUE 6 5.6 ABATEMENT OF BASE FEE 6 6.7 USE OF PREMISES 7 6.1 DEPAULT INTEREST 6 6.2 USE OF PREMISES 7 6.1 USE NEES 7 6.2 USE OF PREMISES 7 6.3	RECITALS			
2.1 LICENSE PREMISES. 3 2.2 SUBORDINATE TO LEASE. 3 2.3 PERFORMANCE OF LEASE OBLIGATIONS. 3 2.4 AMERDMENTS TO LEASE. 3 2.5 CONTACTS WITH PROLOGIS. 3 2.6 PROLOGIS DUTIES. 3 2.7 TERMINATION OF FRASE 3 3. INSPECTION OF PROPERTY; AS IS CONDITION 4 3.1 INSPECTION OF PROPERTY 4 3.2 As IS CONDITION. 5 3.3 ACCESS INSPECTION DISCLOSURE. 5 4 LICENSE TERM 5 4.1 INSPECTION OF PROPERTY 4 4.2 EARLY TERMINATION 5 5 FEE 5 5.1 BASE FEE 5 5.2 ADJUSTMENTS IN BASE FEE 5 5.3 LATE CHARGES 6 5.4 DEFAULT INTEREST 6 5.5 DEDUCTION FROM AMOUNTS DUE 6 5.6 ABATEMENT OF BASE FEE 6 6.6 USE OF FREMISES 7 6.1 PERMITTE	1. BA	SIC LICENSE INFORMATION2		
2.2 SUBORDINATE TO LEASE. 3 2.3 PERFORMANCE OF LEASE OBLIGATIONS. 3 2.4 AMENDMENTS TO LEASE. 3 2.5 CONTACTS WITH PROLOGIS. 3 2.6 PROLOGIS DUTIES. 3 2.7 TERMINATION OF LEASE. 3 3. INSPECTION OF PROPERTY; AS IS CONDITION 4 3.1 INSPECTION OF PROPERTY 4 3.2 As IS CONDITION. 5 3.3 ACCESS INSPECTION DISCLOSURE. 5 4 LICENSE TERM 5 4.1 INSPECTION OF PROPERTY 4 4.2 EARLY TERMINATION. 5 5 FEE 5 5.1 BASE FEE 5 5.2 ADJUSTMENTS IN BASE FEE 5 5.4 DEFAULT INTEREST 6 5.5 DEDUCTION FOR MACOUNTS DUE 6 5.6 ABATEMENT OF BASE FEE 6 6.6 USE OF FREMISES 7 6.1 PERMITTED USE 7 6.2 LIDENSES, NUISANCES OR WASTE 8 6.3	2. PR			
3.1 INSPECTION OF PROPERTY 4 3.2 AS IS CONDITION. 5 3.3 ACCESS INSPECTION DISCLOSURE. 5 4 LICENSE TERM. 5 4.1 INSPECTION OF PROPERTY. 4 4.2 EARLY TERMINATION. 5 5 FEE 5 5.1 BASE FEE 5 5.2 ADJUSTMENTS IN BASE FEE 6 5.3 LATE CHARGES. 6 5.4 DEFAULT INTEREST 6 5.5 DEDUCTION FROM AMOUNTS DUE 6 5.6 ABATEMENT OF BASE FEE 6 6 USE OF PREMISES 7 6.1 PERMITTED USE. 7 6.1 PERMITTED USE. 7 6.2 Use of EQUIPMENT AND MACHINERY. 8 6.3 LIMITATION TO DESCRIBED PURPOSE. 8 6.4 NO UNLAWFUL USES, NUISANCES OR WASTE 8 6.5 SECURITY. 8 6.6 FINES 8 7 ALTERATIONS 9 7.1 LICENSEE'S ALTERATIONS 9	2.2 2.3 2.4 2.5 2.6 2.7	SUBORDINATE TO LEASE. 3 PERFORMANCE OF LEASE OBLIGATIONS. 3 AMENDMENTS TO LEASE. 3 CONTACTS WITH PROLOGIS. 3 PROLOGIS DUTIES. 3 TERMINATION OF LEASE. 3		
3.2 AS IS CONDITION 5 3.3 ACCESS INSPECTION DISCLOSURE 5 4 LICENSE TERM 5 4.1 INSPECTION OF PROPERTY 4 4.2 EARLY TERMINATION 5 5 FEE 5 5.1 BASE FEE 5 5.2 ADJUSTMENTS IN BASE FEE 6 5.3 LATE CHARGES 6 5.4 DEFAULT INTEREST 66 5.5 DEDUCTION FROM AMOUNTS DUE 6 5.6 ABATEMENT OF BASE FEE 6 6 USE OF PREMISES 7 6.1 PERMITTED USE 7 6.1 PERMITTED USE 7 6.2 Use of EQUIPMENT AND MACHINERY 8 6.3 LIMITATION TO DESCRIBED PURPOSE 8 6.4 NO UNLAWFUL USES, NUISANCES OR WASTE 8 6.5 SECURITY. 8 6.6 FINES 9 7.1 LICENSEE'S ALTERATIONS 9 7.2 TITLE TO IMPROVEMENTS. 9 7.3 LICENSEE'S PERSONAL PROPERTY.		·		
4.1 INSPECTION OF PROPERTY 4 4.2 EARLY TERMINATION. 5 5 FEE 5 5.1 BASE FEE 5 5.2 ADJUSTMENTS IN BASE FEE 6 5.3 LATE CHARGES. 6 5.4 DEFAULT INTEREST 6 5.5 DEDUCTION FROM AMOUNTS DUE 6 5.5 DEDUCTION FROM AMOUNTS DUE 6 5.6 ABATEMENT OF BASE FEE. 6 6 USE OF PREMISES 7 6.1 PERMITTED USE. 7 6.1 PERMITTED USE. 7 6.2 USE OF EQUIPMENT AND MACHINERY. 8 6.3 LIMITATION TO DESCRIBED PURPOSE. 8 6.4 NO UNLAWFUL USES, NUISANCES OR WASTE 8 6.5 SECURITY. 8 6.6 FINES 8 7 ALTERATIONS 9 7.1 LICENSEE'S ALTERATIONS 9 7.3 LICENSEE'S PERSONAL PROPERTY. 9 7.4 CITY'S ALTERATIONS OF THE BUILDING AND BUILDING SYSTEMS. 10 7.5 <td>3.2</td> <td>As Is Condition</td> <td></td>	3.2	As Is Condition		
4.2 EARLY TERMINATION. 5 5 FEE. 5 5.1 BASE FEE. 5 5.2 ADJUSTMENTS IN BASE FEE 6 5.3 LATE CHARGES. 6 5.4 DEFAULT INTEREST 6 5.5 DEDUCTION FROM AMOUNTS DUE 6 5.6 ABATEMENT OF BASE FEE. 6 6 USE OF PREMISES 7 6.1 PERMITTED USE. 7 6.1 PERMITTED USE. 7 6.2 USE OF EQUIPMENT AND MACHINERY. 8 6.3 LIMITATION TO DESCRIBED PURPOSE. 8 6.4 NO UNLAWFUL USES, NUISANCES OR WASTE 8 6.5 SECURITY. 8 6.6 FINES 8 7 ALTERATIONS 9 7.1 LICENSEE'S ALTERATIONS 9 7.2 TITLE TO IMPROVEMENTS. 9 7.3 LICENSEE'S PERSONAL PROPERTY. 9 7.4 CITY'S ALTERATIONS OF THE BUILDING AND BUILDING SYSTEMS. 10 7.5 REMOVAL OF ALTERATIONS. 10	4 LIC	CENSE TERM		
5.1 BASE FEE	4.2	EARLY TERMINATION		
5.2 ADJUSTMENTS IN BASE FEE 6 5.3 LATE CHARGES. 6 5.4 DEFAULT INTEREST 6 5.5 DEDUCTION FROM AMOUNTS DUE 6 5.6 ABATEMENT OF BASE FEE. 6 6 USE OF PREMISES 7 6.1 PERMITTED USE. 7 6.2 USE OF EQUIPMENT AND MACHINERY. 8 6.3 LIMITATION TO DESCRIBED PURPOSE. 8 6.4 NO UNLAWFUL USES, NUISANCES OR WASTE 8 6.5 SECURITY. 8 6.6 FINES 8 7 ALTERATIONS 9 7.1 LICENSEE'S ALTERATIONS 9 7.1 LICENSEE'S PERSONAL PROPERTY. 9 7.3 LICENSEE'S PERSONAL PROPERTY. 9 7.4 CITY'S ALTERATIONS of THE BUILDING AND BUILDING SYSTEMS. 10 7.5 REMOVAL OF ALTERATIONS. 10	5 FEI			
6.1 PERMITTED USE	5.2 5.3 5.4 5.5	ADJUSTMENTS IN BASE FEE		
6.1 PERMITTED USE	6 USI			
7.1LICENSEE'S ALTERATIONS	6.2 6.3 6.4 6.5 6.6	PERMITTED USE		
7.2 TITLE TO IMPROVEMENTS	7 AL			
	7.2 7.3 7.4 7.5	TITLE TO IMPROVEMENTS))))	

APPENDIX H: 2650 Bayshore Boulevard License Page i

 7.5.2 Removal of Non-Permitted Improvements. 7.5.3 Alterations Not Subject to Removal. 	
8 REPAIRS AND MAINTENANCE	
 8.1 LICENSEE'S REPAIRS. 8.1.1 Removal of Refuse. 8.1.2 Storm Water Pollution Prevention. 8.1.3 Repair of Any Damage. 8.2 APPROVED OPERATIONS PLAN. 	
9 LIENS AND ENCUMBRANCES	
9.1 LIENS 9.2 ENCUMBRANCES	
10 UTILITIES AND SERVICES	
10.1 Utilities and Services	
11 COMPLIANCE WITH LAWS AND RISK MANAGEMENT REQUIREMEN	NTS 12
11.1 COMPLIANCE WITH LAWS. 11.2 REGULATORY APPROVALS. 11.2.1 Responsible Party. 11.2.2 City Acting in Proprietary Capacity. 11.3 COMPLIANCE WITH CITY'S RISK MANAGEMENT REQUIREMENTS	
12 SUBORDINATION	
13 INABILITY TO PERFORM	
14 DAMAGE AND DESTRUCTION	14
 14.1 DAMAGE AND DESTRUCTION 14.2 CITY REPAIRS 14.3 TERMINATION BY CITY 14.4 LICENSEE WAIVER 	
15 EMINENT DOMAIN	
 15.1 DEFINITIONS	
16 ASSIGNMENT AND SUBLETTING	
17 DEFAULT; REMEDIES 17.1 Events of Default	
17.1 EVENTS OF DEFAULT. 17.2 CITY RIGHTS UPON DEFAULT	
18 WAIVER OF CLAIMS; INDEMNIFICATION	
18.1 LIMITATION ON CITY'S LIABILITY: WAIVER OF CLAIMS 18.2 LICENSEE'S INDEMNITY	

APPENDIX H: 2650 Bayshore Boulevard License Page ii

2

1.07 $\{[1, l_{ij}] \in \{j_i\} \}$ 1.5.1.2.

19	INSURANCE	.19
19.	LICENSEE'S INSURANCE	19
19.		.19
19.3		
19.4		
20	ACCESS BY CITY	.19
21	LICENSEE'S CERTIFICATES	.20
22	REPAIR STANDARDS AND MAINTENANCE REQUIREMENTS	
22.		
22.	2 PLAN AND REPORTING	.21
23	SURRENDER OF PREMISES	.21
24	HAZARDOUS MATERIALS	22
24.		
24.		
24.		
24.		. 22
24.		
24.		
24.		
24.		
24.		
24.		
24.		
24.		
24.		
24.		25
24.		
24.		. 26
24.		
24.		. 27
	10 Environmental Security Deposit	
	11 Environmental Oversight Deposit	
24.	12 HAZARDOUS SUBSTANCE DISCLOSURE	28
25	GENERAL PROVISIONS	
25.	1 NOTICES.	28
25.	2 No Implied Waiver	29
25.	3 <u>Amendments</u>	29
25.	4 AUTHORITY.	29
25.	5 PARTIES AND THEIR AGENTS; APPROVALS	29
25.	6 INTERPRETATION OF LICENSE.	29
25.		
25.		
25.		
25.		
	11 ATTORNEYS' FEES	
25.	12 HOLDING OVER ERROR! BOOKMARK NOT DEFIN	ED.
25.	13 TIME OF ESSENCE.	31
25.	14 CUMULATIVE REMEDIES.	31

APPENDIX H: 2650 Bayshore Boulevard License Page iii

25.15	PROVISIONS OF LICENSE SURVIVING TERMINATION.	31
	SIGNS	
25.17	RELATIONSHIP OF THE PARTIES	31
25.18	LIGHT AND AIR	32
25.19	No Recording	32
25.21	PUBLIC TRANSIT INFORMATION	32
25.22	TAXES, ASSESSMENTS, LICENSES, PERMIT FEES AND LIENS.	32
25.23	WAGES AND WORKING CONDITIONS,	32
25.24	NON-DISCRIMINATION IN CITY CONTRACTS AND BENEFITS ORDINANCE	32
25.24.	1Covenant Not to Discriminate	32
25.24.2	2Subcontracts	33
25.24.3	3Non-Discrimination in Benefits	<u>33</u>
25.24.4	4HRC Form	33
25.24.	5Incorporation of Administrative Code Provisions by Reference.	33
25.25	NON-LIABILITY OF CITY OFFICIALS, EMPLOYEES AND AGENTS.	33
25.26	NO RELOCATION ASSISTANCE: WAIVER OF CLAIMS.	33
	MACBRIDE PRINCIPLES - NORTHERN IRELAND	
25.28	TROPICAL HARDWOOD AND VIRGIN REDWOOD BAN	34
25.29	PESTICIDE PROHIBITION.	34
25.30	FIRST SOURCE HIRING ORDINANCE.	34
25.31		
25.32	CONFLICTS OF INTEREST	34
25.33	CHARTER PROVISIONS	
25.34	PROHIBITION OF TOBACCO ADVERTISING	35
25.35	REQUIRING HEALTH BENEFITS FOR COVERED EMPLOYEES	35
25,36	NOTIFICATION OF LIMITATIONS ON CONTRIBUTIONS	36
25.37	PRESERVATIVE-TREATED WOOD CONTAINING ARSENIC	
25.38	References.	37
25.39	COUNTERPARTS,	37

ATTACHMENTS:

Attachment 1 - Lease Attachment 2 - Depiction of Premises

APPENDIX H

TO THE TOWING AGREEMENT BY AND BETWEEN THE CITY AND COUNTY OF SAN FRANCISCO AND TEGSCO, LLC, d.b.a. SAN FRANCISCO AUTORETURN

AMENDED AND RESTATED REVOCABLE LICENSE TO ENTER AND USE PROPERTY

THIS AMENDED AND RESTATED REVOCABLE LICENSE TO ENTER AND USE PROPERTY ("License"), between the CITY AND COUNTY OF SAN FRANCISCO, a municipal corporation ("City"), acting by and through its Municipal Transportation Agency ("SFMTA"), and TEGSCO, LLC, a California limited liability company, d.b.a. San Francisco AutoReturn ("Licensee"), and dated as of May , 2014, is Appendix H to the Amended and Restated Agreement and Property Use License for Towing, Storage and Disposal of Abandoned and Illegally Parked Vehicles between City and Licensee, dated for convenience as July 31, 2010 (the "Original Towing Agreement"), as amended by the First Amendment to Amended and Restated Agreement and Property Use License dated as of May 1, 2014 (the "Towing Amendment"), which are incorporated herein by reference as if fully set forth herein.

RECITALS

This License is made with reference to the following facts:

A. City and Licensee are parties to the Original Towing Agreement and the Towing Amendment (together, the "Towing Agreement") for the towing and storage of abandoned and illegally parked vehicles.

B. As part of the Towing Agreement, City granted Licensee a license to use certain City property pursuant to a Revocable License for the Use of Certain Property on Pier 70, San Francisco, California, between Licensee and City, with an effective date of July 31, 2010 (as amended, the "Original License Agreement"), for the premises described in the Original License Amendment (the "Pier 70 License Area"), subject to the terms of a Memorandum of Understanding between the San Francisco Port Commission and SFMTA with an effective date of July 31, 2005, as amended.

C. City leases that certain real property commonly known as 2650 Bayshore Boulevard, Daly City, California (the "Property") from Prologis, L.P., a Delaware limited partnership ("Prologis") pursuant to an Industrial Lease between City and Prologis dated as of October 29, 2012, a copy of which is attached to this License as <u>Attachment 1</u> (the "Lease").

D. Pursuant to Section 2.2.1 and Section 4.1 of the Original License Agreement, City relocated License from the Pier 70 License Area to the portion of the Property shown on <u>Attachment 2</u> (the "Premises") and agrees to reduce the Base Fee (as defined in the Original License Agreement) through a rent credit to reflect Licensee's relocation costs from the Pier 70 License Area to the Premises and Licensee's increased operating cost at the Premises, and City and Licensee wish to amend and restate the Original License Agreement in its entirety in connection with such relocation and rent credit.

Therefore, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

APPENDIX H: 2650 Bayshore License Agreement

Page 1 of 39

AGREEMENT

1. BASIC LICENSE INFORMATION

The following is a summary of Basic License information (the "Basic License Information"). Each item below shall be deemed to incorporate all of the terms in this License pertaining to such item. In the event of any conflict between the information in this Section and any more specific provision of this License, the more specific provision shall control.

Licensor:

Licensee:

Property (Section 2.1):

Building (Section 2.1):

Premises (Section 2.1):

Term (Section 4):

Base Fee (Section 5):

Permitted Use (Section 6.1):

Utilities and Services (Section 10);

Security (Section 6.5):

Security Deposit (Section 24.10):

Notices to the Parties: (Section 25.1)

CITY AND COUNTY OF SAN FRANCISCO, A MUNICIPAL CORPORATION

TEGSCO, LLC, A CALIFORNIA LIMITED LIABILITY COMPANY D.B.A. SAN FRANCISCO AUTORETURN

That certain real property commonly known as 2650 Bayshore Boulevard, Daly City, California, as fully described in Exhibit A to the Lease (the "Property").

The building located on the Property.

The portions of the Property and Building depicted on <u>Attachment 2</u> and comprised of approximately 330,771 square feet

Commencement Date: The date this License is fully executed.

Expiration Date: July 31, 2015, subject to any earlier termination of this License pursuant to the terms hereof.

Initial Monthly Fee: \$150,378.75.

Parking space for the storage and transfer of vehicles, public lien sale auctions and office space for the administration of Licensee's operations under the Towing Agreement.

Obtained and paid by Licensee at its sole cost, provided that Licensee shall reimburse City pursuant to <u>Section 10</u> for the gas and electricity utilities provided by City to the Premises.

Licensee shall be solely responsible for the security of the Premises.

Provided under Section 12 [Financial Assurances] of the Towing Agreement.

Any notice, demand, consent or approval required under this License must be sent by first class certified U.S. mail with return receipt requested, or by overnight courier,

APPENDIX H: 2650 Bayshore License Agreement

Page 2 of 39

San Francisco AutoReturn Attention: John Wicker 945 Bryant Street San Francisco, CA 94103 telephone : (415) 575-2355 facsimile: (415) 575-2341 Email: jwicker@autoreturn.com

City and County of San Francisco Attention: Lorraine Fuqua 1 South Van Ness Avenue, Suite 800 San Francisco, CA 94102 telephone: 415-701-4678 facsimile: 415-701-4736 Email: lorraine.fuqua@sfinta.com

2. PREMISES

2.1 License Premises.

2.1.1 City confers to Licensee a revocable, personal, non-exclusive and non-possessory privilege to enter upon and use those certain premises identified in the Basic License Information and shown on <u>Attachment 2</u>, attached hereto and incorporated by reference as though fully set forth herein (collectively, the "Premises"), for the limited purpose and subject to the terms, conditions and restrictions set forth below. This License gives Licensee a license only, revocable at any time at the will of City, and notwithstanding anything to the contrary herein, this License does not constitute a grant by City of any ownership, leasehold, easement or other property interest or estate whatsoever in the Premises, or any portion thereof. The privilege given to Licensee under this License is effective only insofar as the rights of City in the Premises are concerned, and Licensee shall obtain any further permission necessary because of any other existing rights affecting the Premises. The area of the Premises, land upon which the Premises is located and all other improvements on and appurtenances to such land are referred to collectively as the "Premises."

2.1.2 City may, at City's sole and absolute discretion, relocate Licensee from any portion or all of the Premises to another location on the Property or other City property that City in its sole and absolute discretion deems suitable for the uses permitted hereunder; provided that such relocation shall not materially interfere with Licensee's ability to meet its obligations under the Towing Agreement. In the event of any such relocation, the new location shall become part or all of the Premises hereunder.

2.1.3 City may, at City's sole and absolute discretion, modify the original configuration of the Premises; provided that such modification shall not materially interfere with Licensee's ability to meet its obligations under the Towing Agreement, unless such modification is required under the Lease.

2.1.4 Licensee acknowledges that the interest of City in the Premises is limited to those rights conveyed to City pursuant to the Lease. Licensee hereby agrees to assume all

APPENDIX H: 2650 Bayshore License Agreement

Page 3 of 39

To Licensee:

To City:

responsibility for and be bound by all covenants, terms and conditions made by or applicable to City under the Lease, and shall not take any actions that would cause the City to be in default under the Lease. In the event there are any inconsistencies between the provisions of this License and the Lease, the provisions of the Lease shall govern Licensee's use of the Premises hereunder.

2.2 Subordinate to Lease. This License is expressly made subject and subordinate to all the terms, covenants and conditions of the Lease, which are incorporated herein by reference (collectively, the "Lease Terms"). Licensee agrees to use the Premises in accordance with the Lease Terms and not take or fail to take any act that City would be required to not take or take under the Lease to comply with the Lease Terms.

2.3 Performance of Lease Obligations. Licensee further agrees to assume the obligation for performance of all City's obligations under the Lease with respect to the Premises, except as may be specifically modified by this License and excluding City's obligation to pay rent to Prologis under the Lease.

2.4 Amendments to Lease. Licensee agrees that City shall have the right to enter into amendments or modifications to the Lease without Licensee's prior written consent; provided, however, that if such proposed amendment or modification would materially affect Licensee's rights under this License, Licensee shall not be subject to such amendment or modification unless it consents to be subject thereto in writing. In the event Licensee fails within fifteen (15) business days to respond in writing to City's written request for such consent, then Licensee shall be deemed without further notice to have consented to City's request for consent.

2.5 Contact with Prologis. Licensee has no authority to contact Prologis or make any agreement with Prologis concerning the Premises or the Lease without City's prior written consent, and Licensee shall make payments of the Base Fee and any other charges payable by Licensee under this License only to City.

2.6 Prologis Duties. The Lease describes Prologis' duties with respect to the Property. City is not obligated to perform such Prologis' duties. If Prologis fails to perform its duties, Licensee shall provide notice to City. In no event shall City incur any liability, or otherwise be responsible, nor shall there be any set-off, deduction or abatement of any Base Fee or other amounts owed by Licensee to City pursuant to this License arising from Prologis' failure to comply with its duties. Notwithstanding anything to the contrary in the foregoing, if Prologis fails to perform its duties under the Lease and such failure materially interferes with Licensee's use of the Premises, Licensee shall be entitled to a proportionate abatement of the Base Fee to the extent City receives an abatement of rent for such failure under the Lease.

2.7 Termination of Lease. If the Lease is terminated for any reason during the Term, this License shall automatically terminate as of such Lease termination date.

3. INSPECTION OF PROPERTY; AS IS CONDITION

3.1 Inspection of Premises. Licensee represents and warrants that Licensee has conducted a thorough and diligent inspection and investigation of the Property and the suitability of the Premises for Licensee's intended use, either independently or through its officers, directors, employees, agents, affiliates, subsidiaries licensees and contractors, and their respective heirs, legal representatives, successors and assigns, and each of them. Licensee is fully aware of the needs of its operations and has determined, based solely on its own investigation, that the Premises is suitable for its operations and intended uses. Licensee further represents and warrants that Licensee has occupied the Premises since May 1, 2013.

Page 4 of 39

3.2 As Is Condition. WITHOUT WAIVING ANY OF LICENSEE'S RIGHTS ESTABLISHED IN <u>SECTIONS 24.3</u> AND <u>24.8</u> BELOW, LICENSEE ACKNOWLEDGES AND AGREES THAT THE PREMISES ARE BEING LICENSED AND HAVE BEEN ACCEPTED IN THEIR <u>"AS IS"</u> AND "WITH ALL FAULTS" CONDITION, WITHOUT REPRESENTATION OR WARRANTY OF ANY KIND, AND SUBJECT TO ALL APPLICABLE LAWS, RULES AND ORDINANCES GOVERNING THEIR USE, OCCUPANCY AND POSSESSION. LICENSEE ACKNOWLEDGES AND AGREES THAT NEITHER CITY NOR ANY OF ITS AGENTS HAVE MADE, AND CITY HEREBY DISCLAIMS, ANY REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED, CONCERNING THE RENTABLE AREA OF THE PREMISES, THE PHYSICAL, SEISMOLOGICAL OR ENVIRONMENTAL CONDITION OF THE PREMISES OR THE PROPERTY, THE PRESENT OR FUTURE SUITABILITY OF THE PREMISES FOR LICENSEE'S BUSINESS, OR ANY OTHER MATTER WHATSOEVER RELATING TO THE PREMISES, INCLUDING, WITHOUT LIMITATION, AND IMPLIED WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE.

3.3 Accessibility Inspection Disclosure. Licensee is hereby advised that the Premises have not been inspected by a Certified Access Specialist under California Civil Code Section 1938.

4. LICENSE TERM

4.1 Original Term. The privilege given to Licensee pursuant to this License is temporary only and shall commence upon the Commencement Date (as described in the Basic License Information) and shall terminate on July 31, 2015, or the date of earlier termination of this License pursuant to the terms of this License or the Towing Agreement (the "Expiration Date"). Without limiting any of its rights hereunder, City may at its sole option freely revoke this License at any time, without cause and without any obligation to pay any consideration to Licensee. Licensee acknowledges its receipt and acceptance of the Premises on May 1, 2013.

4.2 Early Termination. Without limiting any of its rights hereunder, City may at its sole option freely terminate this License as to all or a portion of the Premises without cause and without any obligation to pay any consideration to Licensee. In the event that City terminates this License as to a portion of the Premises under this Paragraph, the Base Fee will be reduced in proportion to the amount of square footage removed from the Premises and Licensee shall be solely responsible for all costs associated with such modifications or reconfiguration that City in its sole discretion deems necessary, including all costs incurred to relocate the operations, Premises, fences, gates, lights, driveways, and other improvements; provided, however, that City shall make good faith efforts to reach an agreement with Licensee regarding the nature and extent of such necessary modifications or reconfiguration.

5. FEE

5.1 Base Fee; Rent Credit. Throughout the Term commencing on the first day of the Term, Licensee shall pay to City the monthly Base Fee specified in the Basic License Information (the "Base Fee"), subject to the adjustments made pursuant to <u>Section 5.2</u>. Licensee shall pay the Base Fee to City monthly, in advance, on or before the first of the month for which the fee is due, without prior demand and without any deduction, setoff or counterclaim whatsoever, except as such deduction or setoff is specifically provided for in <u>Section 5.6</u> [Abatement of Base Fee and Base Fee Credits]. The Base Fee shall be paid by wire transfer to the account designated by City in writing. If the first day of the Term occurs on a day other than the first day of a calendar month, then the Base Fee for such fractional month shall be prorated based on a thirty (30) day month.

Page 5 of 39

City and Licensee acknowledge that, pursuant to Section 2.1.2 of the Original License Agreement, Licensee relocated from the Pier 70 License Area to the Premises on May 1, 2013, and continued to pay the full monthly license fee required under the Original License Agreement from May 1, 2013, through the first day of the month in which the Commencement Date occurs. To reimburse Licensee for its costs in relocating from the Pier 70 License Area to the Premises, and Licensee's increased monthly operating expenses at the Premises, on the Commencement Date, Licensee shall be entitled to a one-time rent credit equal to (i) \$6,226.47, multiplied by (ii) the number of months between the Commencement Date and May 1, 2013. Such one-time rent credit shall be offset against the first monthly Base Fee payment made by Licensee pursuant to this License. City and Licensee further acknowledge that to reimburse Licensee for such relocation costs and increased monthly operating expenses, Licensee shall have the right to apply a rent credit in the amount of \$6,226.47 per month against the monthly payments of the Base Fee for the period commencing on the first month immediately following the Commencement Date and terminating on the Expiration Date.

5.2 Adjustments in Base Fee. Commencing on August 1, 2014, the monthly Base Fee shall be adjusted to \$154,212.23.

5.3 Late Charges. If Licensee fails to pay all or any portion of any payment to be made by Licensee to City pursuant to this License within five (5) days following the due date for such payment, such unpaid amount shall be subject to a late payment charge equal to six percent (6%) of the unpaid amount in each instance. The late payment charge has been agreed upon by City and Licensee, after negotiation, as a reasonable estimate of the additional administrative costs and detriment that City will incur as a result of any such failure by Licensee, the actual costs thereof being extremely difficult if not impossible to determine. The late payment charge constitutes liquidated damages to compensate City for its damages resulting from such failure to pay and shall be paid to City together with such unpaid amount.

5.4 Default Interest. Any payment to be made by Licensee to City pursuant to this License, if not paid within five (5) days following the due date, shall bear interest from the due date until paid at the rate of ten percent (10%) per year or, if a higher rate is legally permissible, at the highest rate an individual is permitted to charge under law. However, interest shall not be payable on late charges incurred by Licensee nor on any amounts on which late charges are paid by Licensee to the extent this interest would cause the total interest to be in excess of that which an individual is lawfully permitted to charge. Payment of interest shall not excuse or cure any default by Licensee.

5.5 Deduction from Amounts Due. In the event Licensee fails to pay any payment due hereunder for more than ten (10) days following the due date, City may deduct and withhold the amount of such payment, together with the amount of applicable late charges and default interest as provided herein, from any monies in City's possession due Licensee pursuant to the Towing Agreement.

5.6 Abatement of Base Fee and Base Fee Credits. During the Term, and subject to City's prior written approval (which shall not be unreasonably withheld), City shall grant abatement of the Base Fee or allow Licensee a credit that may offset from the Base Fee under the circumstances listed in this Section 5.6. All Base Fee abatements and credits available to Licensee shall be applied against the Base Fee payment obligation during the Term at a rate not greater than one half ($\frac{1}{2}$) of the applicable month Base Rent payment and shall be applied if and only if Licensee is in good standing and is not in default of any of the terms of this License. In the event that the total of Base Fee abatements or credits available to Licensee pursuant to this Section 5.6 exceeds an amount equal to one half ($\frac{1}{2}$) of the Base Fee payment for any one calendar month, the remaining available Base Fee credit shall be carried forward to successive calendar months at a rate not to exceed one half ($\frac{1}{2}$) of the applicable Base Fee payment, until all

APPENDIX H: 2650 Bayshore License Agreement

Page 6 of 39

available Base Fee credits have been fully applied. In no event, however, shall Licensee be entitled to the application of any Base Fee credits or the value thereof beyond the expiration or earlier termination of this License.

5.6.1 If the Premises ceases to be used for towing operations at any time due to damage sustained during the Term by fire, earthquake, or other casualty rendering the Premises unsuitable for occupancy as determined by the City of Daly City pursuant to the Daly City Building Code, or are otherwise deemed legally not useable for any reason, Licensee shall be entitled to an abatement in Base Fee to the same extent that City receives a Rent abatement pursuant to Section 12 of the Lease.

5.6.2 At any time during the Term of this License, City may elect to fund certain property maintenance, construction, improvements, systems development or staffing related to City towing and impoundment operations that are not the responsibility of Licensee under the Towing Agreement ("Projects"). City may require Licensee, upon ninety (90) day's written notice, to implement any such Project. In consideration for implementation of the Project by Licensee, City shall issue an appropriate Base Fee credit in the amount of the reasonable actual costs of the Project. No Project may be implemented without the prior written approval of the City, and all Project implementation must be in accordance with the scope, specifications, maximum costs and all other requirements provided in writing by City. Licensee shall be responsible for obtaining all required permits, governmental approvals and insurance for any Project, the reasonable actual costs of which shall be included in the Base Fee credit. After the completion of the Project, Licensee must deliver to City an itemized statement of the actual costs of the Project, accompanied by documentation substantiating all said expenditures. Such documentation of expenditures shall include: (i) copies of executed contracts; (ii) copies of invoices for labor, services and/or materials, copies of bills of lading, and/or copies of other bills or receipts for goods, materials and/or services; (iii) copies of canceled checks, and (iv) such other proofs of expenditure as may by reasonably requested by City. Such appropriate proofs of expenditure may include copies of canceled checks, copies of contracts or invoices for labor, services and/or materials marked "Paid", or otherwise evidenced as having been paid bills of lading marked "Paid"; other bills, contracts, receipts for goods materials and/or services marked "Paid" and such other proofs of expenditure as may be reasonably approved by City. All such proofs of expenditure must be directly attributable to the approved Project and may include materials purchased by City for installation by City but not the City's cost if it undertakes such installation.

6. USE OF PREMISES

6.1 Permitted Use. Licensee shall use and continuously occupy the Premises during the Term solely for temporary storage and transfer of vehicles, public lien sale auctions and related office use as necessary to meet its obligations under the Towing Agreement and for such other uses, if any, as may be specified in the Basic License Information, all to the extent permitted under the Lease Terms. Licensee shall not use the Premises for any other purpose without the prior written approval of City, including, without limitation, the following: (a) crushing or dismantling; (b) maintenance or fueling of vehicles, except as otherwise may be permitted under the Approved Operations Plan (as defined in Section 8.2); (c) selling vehicle parts from the Premises, (d) parking or storage of vehicles not covered under the Towing Agreement, or (e) parking for Licensee's employees, without the prior written approval of City and subject to availability of space necessary to fully perform Licensee's obligations under the Towing Agreement. All available space for vehicle parking shall be used for the purposes set forth in the Towing Agreement, except as otherwise expressly approved by City pursuant to this License. The washing of vehicles shall be with cleansing agents that are bio-degradable and non-toxic, and shall be in compliance with the Approved Operations Plan. No advertising or signage may be placed in or about the Premises without the prior written permission of City.

APPENDIX H: 2650 Bayshore License Agreement

Page 7 of 39
6.2 Use of Equipment and Machinery. Licensee shall have the right to place on the Premises all necessary equipment and machinery in connection with the permitted use of the Premises. It is understood and agreed that City is not responsible for loss of or damage to any Licensee-owned equipment herein involved, unless caused by the sole negligence of City's officers, agents, and employees.

6.3 Limitation to Described Purpose. Licensee may occupy and use the Premises solely for the purpose of fulfilling its obligations under the Towing Agreement to store and auction any vehicles towed pursuant to the terms of the Towing Agreement, and for incidental purposes related thereto. Adequate drop-off space must be provided so that tow and transport trucks can load and unload on the Premises. No loading, unloading, queuing, parking or storage of vehicles will be permitted on any remaining portion of the Property or any public streets or rights-of-way adjacent to the Property. All storage activities authorized by this License shall be restricted to the designated enclosed and visually screened area. Any use of the Premises by Licensee shall be subject to the requirements of the Approved Operations Plan.

6.4 No Unlawful Uses, Nuisances or Waste. Without limiting the foregoing, Licensee shall not use, occupy or permit the use or occupancy of any of the Premises in any unlawful manner or for any illegal purpose, or permit any offensive, noisy or hazardous use or any waste on or about the Premises. Licensee shall take all precautions to eliminate any nuisances or hazards relating to its activities on or about the Premises. Licensee shall not conduct any business, place any sales display, or advertise in any manner in areas outside the Premises or on or about the Property without the prior written permission of City.

6.5 Security. Licensee shall at all times provide security at a level acceptable to the City to protect the Premises and all vehicles stored therein, and the persons and property of owners of towed vehicles, against damage, injury, theft or other loss.

6.6 Fines. Without limiting City's other rights and remedies set forth in this License, if Licensee violates any of the following provisions governing its use of the Premises contained in this License or the Towing Agreement, City may impose a fine of \$300 per day during which Licensee is in violation of any of the specified provisions: <u>Sections 2</u>, 6, 7, 8, 11, 19, 22, and 24 of this License; and Towing Agreement section 12.2. City may also impose this fine for Licensee's failure to submit any documents, reports or other items as and when required by any provision of this License.

The additional charges described in this <u>Section 6.6</u> shall run from the date of City's notice to Licensee of the violation and shall continue until the violation is cured. All such accrued amounts under this Section shall be payable to City monthly in arrears at the same time, place and manner as the Base Fee is payable unless otherwise specified herein. City shall have the same remedies for a default in the payment of any such amounts as for a default in the payment of Base Fee. The parties agree that the charges described above represent a fair and reasonable estimate of the administrative cost and expense which City will incur due to such violations by reason of its inspections, issuance of charges and other costs.

If City initiates notice of a charge under this Section, Licensee may appeal such charge to SFMTA's Director of Transportation within fourteen (14) days of the notice with evidence supporting Licensee's claim for relief from the charge imposed. SFMTA's Director of Transportation will respond within fourteen (14) days. Any failure of SFMTA's Director of Transportation to respond within the fourteen (14) day period shall be deemed a rejection of Licensee's claim for relief from the charges imposed. The provisions of Section 46 of the Towing Agreement shall not apply to charges imposed under this Section.

APPENDIX H: 2650 Bayshore License Agreement

Page 8 of 39

City's right to impose the foregoing charges shall be in addition to and not in lieu of any and all other rights under this License or at law or in equity; provided, however, that City agrees that once it has declared an "Event of Default" pursuant to Section 17 of this License, it will no longer impose any new charges with respect to such default. City shall have no obligation to Licensee to impose charges on or otherwise take action against any other person.

7. ALTERATIONS

7.1 Licensee's Alterations. Licensee shall not make, nor cause or suffer to be made, any alterations, installations, improvements, or additions to any improvements or to the Premises. (including demolition or removal), installations, additions or improvements to the Premises, including but not limited to the installation of any appurtenances or trade fixtures affixed to the Premises, constructed by or on behalf of Licensee pursuant to the Towing Agreement, or any trailers; signs, roads, trails, driveways, parking areas, curbs, walks fences walls, stairs, poles, plantings or landscaping, (collectively, "Improvements" or Alterations," which words are interchangeable as used in this License) without first obtaining City's written approval and any required approvals of regulatory agencies having jurisdiction over the Premises. All Alterations shall be done at Licensee's expense in accordance with plans and specifications approved by City, only by duly licensed and bonded contractors or mechanics approved by City, and subject to any conditions that City may reasonably impose. City may require Licensee, at Licensee's expense, to obtain the prior written approval of City's Arts Commission with respect to any Alterations, to the extent the Arts Commission has jurisdiction over the design of such proposed alterations under City's Charter Section 5.103. All Alterations shall be subject to the following conditions:

7.1.1 All Alterations shall be constructed in a good and workerlike manner and in compliance with all applicable building, zoning and other laws, and in compliance with the terms of and the conditions imposed in any regulatory approval;

7.1.2 All Alterations shall be performed with reasonable dispatch, delays beyond the reasonable control of Licensee excepted; and

7.1.3 At the completion of the construction of the Alterations, Licensee shall furnish one (1) set of "as-built" drawings of the same made on or to the Premises. Unless otherwise stated as a condition of the regulatory approval, this requirement may be fulfilled by the submittal after completion of the Alterations of a hand-corrected copy of the approved permit drawing(s).

7.2 Title to Improvements. Except for Licensee's Personal Property (as defined in <u>Section 7.3</u>), or as may be specifically provided to the contrary in approved plans, all Alterations, equipment, or other property attached or affixed to or installed in the Premises at the Commencement Date or, by Licensee with the advance approval of City during the Term, shall, at City's sole discretion, remain City's property without compensation to Licensee or be removed at the termination of this License. Licensee may not remove any such Alterations at any time during or after the Term unless City so requests pursuant to <u>Section 23</u> [Surrender of Premises], below.

7.3 Licensee's Personal Property. All furniture, trade fixtures, office equipment and articles of movable personal property installed in the Premises by or for the account of Licensee, without expense to City, and that can be removed without structural or other damage to the Premises (collectively, "Licensee's Personal Property") shall be and remain Licensee's property. Licensee may remove Licensee's Personal Property at any time during the Term, subject to the provisions of Section 23 [Surrender of Premises], below. Licensee shall pay any taxes or other

APPENDIX H: 2650 Bayshore License Agreement

Page 9 of 39

impositions levied or assessed upon Licensee's Personal Property, at least ten (10) days prior to delinquency, and shall deliver satisfactory evidence of such payment to City upon request.

7.4 City's Alterations of the Building and Building Systems. City reserves the right at any time to make alterations, additions, repairs, deletions or improvements to the common areas of the Premises or any other part of the Building or the heating, ventilating, air conditioning, plumbing, electrical, fire protection, life safety, security and other mechanical, electrical and communications systems located at the Premises ("Building Systems"), provided that any such alterations or additions shall not materially adversely affect the functional utilization of the Premises for the Permitted Use set forth in Section 6.1 [Permitted Use].

7.5 **Removal of Alterations.** At City's election made in accordance with this <u>Section</u> 7.5, Licensee shall be obligated at its own expense to remove and relocate or demolish and remove (as Licensee may choose) any or all Alterations which Licensee has made to the Premises, including without limitation all telephone wiring and equipment installed by Licensee. Licensee shall repair, at its own expense, in good workerlike fashion any damage occasioned thereby.

7.5.1 Notice of Removal. Prior to the termination of this License, City shall give written notice to Licensee specifying the Alterations or portions thereof that Licensee shall be required to remove and relocate or demolish and remove from the Premises, in accordance with this <u>Section 7.5</u> (herein "Notice of Removal"). If termination is the result of loss or destruction of the Premises or any improvements thereon, City shall deliver said Notice of Removal to Licensee within a reasonable time after the loss or destruction. If Licensee fails to complete such demolition or removal on or before the termination of this License, City may perform such removal or demolition at Licensee's expense, and Licensee shall reimburse City upon demand therefor.

7.5.2 Removal of Non-Permitted Improvements. If Licensee constructs any Alterations to the Premises without City's prior written consent or without complying with this <u>Section 7</u>, then, in addition to any other remedy available to City, City may require Licensee to remove, at Licensee's expense, any or all such Alterations and to repair, at Licensee's expense and in good workerlike fashion, any damage occasioned thereby. Licensee shall pay any special inspection fees required by the City of Daly City for inspecting any Alterations performed by or for Licensee without required permits.

7.5.3 Alterations Not Subject to Removal. In conjunction with a request to make an Alteration under <u>Section 7.1</u> above, Licensee may submit a request for a City determination of whether a proposed Alteration would or would not be required to be removed upon expiration or termination of this License. Licensee acknowledges that such a determination will be based, in part, on whether Prologis would require the removal of the proposed Alteration upon expiration or termination of the Lease. This <u>Section 7.5.3</u> shall not apply to Alterations that are required by any regulatory authority to conform the Premises or any building thereon to a requirement of statute, ordinance or regulation.

8. REPAIRS AND MAINTENANCE

8.1 Licensee's Repairs. Licensee shall maintain, at its sole expense, the Premises in good repair and working order and in a clean, secure, safe and sanitary condition. Licensee shall promptly make all repairs and replacements: (a) at its sole expense, (b) by licensed contractors or qualified mechanics approved by City, (c) so that the same shall be at least equal in quality, value and utility to the original work or installation, (d) in a manner and using equipment and materials that will not interfere with or impair the use or occupation of the Premises, and (e) in accordance with the Lease and all applicable laws, rules and regulations (collectively,

APPENDIX H: 2650 Bayshore License Agreement

Page 10 of 39

"Applicable Law"). Licensee hereby waives all rights to make repairs at City's expense under Sections 1941 and 1942 of the California Civil Code or under any similar law, statute or ordinance now or hereafter in effect.

8.1.1 Removal of Refuse. All refuse, including tires, non-salvageable vehicle parts and litter, shall be removed from the Premises on a regular basis by an authorized refuse collection company. All trash areas shall be effectively screened from view and maintained in orderly manner. All trash and refuse containers shall be maintained in approved enclosures.

8.1.2 Storm Water Pollution Prevention. Licensee agrees to effect mechanisms to control stormwater pollution at the Premises to the reasonable satisfaction of City, which mechanism may include (by way of example and not limitation) good housekeeping and materials management practices, preventing run-on and run-off from materials storage areas, maintenance areas, or areas where contaminants may be present, installation and maintenance of catchments or absorbent pads in stormwater drains located at or servicing the Premises, or other pollution prevention practices appropriate to the facility and operations. Documentation of Licensee's pollution prevention practices shall be provided as part of the Approved Operations Plan. Licensee shall comply with all stormwater pollution control regulations applicable to the Property, and shall prepare and submit all stormwater permit applications and stormwater pollution control plans required for the Premises under any Applicable Law on or before the forty-fifth (45th) day immediately following the Commencement Date.

8.1.3 Repair of Any Damage. In the event that damage to any of the improvements to the Premises which are Licensee's obligation to maintain by reason of ordinary wear and tear or deterioration results in such improvements not meeting the standard of maintenance required by City for such uses as Licensee is making of the Premises, then Licensee shall have the independent responsibility for, and shall promptly undertake such maintenance or repair and complete the same with due diligence. If Licensee fails to do so after reasonable notice in writing from City, then in addition to any other remedy available to City, City may make such maintenance or repairs and Licensee shall reimburse City therefor. Work performed by the City pursuant to this <u>Section 8.1.3</u> shall not be subject to any abatement of the Base Fee. The City, in its sole discretion, may obtain reimbursement for damages from Licensee's Letter of Credit required by Section 12.2 of the Towing Agreement ("Security Deposit"). Should the City obtain reimbursement for damages from the Security Deposit to its original amount.

8.2 Approved Operations Plan. The 2005 Operations Plan approved by City pursuant to the Towing Agreement shall be revised for the Premises as follows. Within sixty (60) days of the Commencement Date, City shall review the proposed Operations Plan for 2650 Bayshore previously submitted by Licensee to City, and either approve it as submitted or request revisions. Licensee shall respond to a request for revisions within twenty (20) days. City shall have fifteen (15) days to either approve the revised Operations Plan as submitted, submit further requests for revisions, or respond to requests for revisions. Each revision must reflect tracking of document versions, including the date and source of revisions, and each exchange of versions between the parties shall be accompanied by an executed document substantially in the form of Appendix B to the Towing Agreement. The Operations Plan approved by City for the Premises shall be the "Approved Operations Plan".

9. LIENS AND ENCUMBRANCES

9.1 Liens. Licensee shall keep the Premises and the rest of the Property free from any liens arising out of any work performed, material furnished or obligations incurred by or for Licensee. In the event Licensee does not, within twenty (20) days following the imposition of any such lien, cause the lien to be released of record by payment or posting of a proper bond, City shall

APPENDIX H: 2650 Bayshore License Agreement

Page 11 of 39

have, in addition to all other remedies, the right, but not the obligation, to cause the lien to be released by such means as it shall deem proper, including, but not limited to, payment of the claim giving rise to such lien. All such sums paid by City and all expenses reasonably incurred by it in connection therewith (including, without limitation, reasonable attorneys' fees) shall be payable to City by Licensee within thirty (30) days of demand by City. City shall have the right to post on the Premises any notices that City may deem proper for the protection of City, the Premises from mechanics' and materialmen's liens. Licensee shall give to City at least fifteen (15) days' prior written notice of commencement of any repair or construction on the Premises.

양가 가슴을 가슴을 가슴을 가슴을 가슴을 다.

9.2 Encumbrances. Licensee shall not create, permit or suffer any liens or encumbrances affecting any portion of the Premises, the Property or City's interest therein or under this License.

10. UTILITIES AND SERVICES

an an an an an Arrange Arrange an Ar

10.1 Utilities and Services. Sewer, water, janitor service, telecommunications services and any other utilities (other than gas and electricity) or services shall be acquired and paid by Licensee, including the initial hook up to said utilities and services. City shall provide electricity and gas to the Premises, and on a quarterly basis, Licensee shall reimburse City in an amount equal to the lesser of (i) \$3,000 per month, and (ii) fifty-nine and one-half percent (59.5%) of the gas and electricity charges incurred by City for the Property during such month. Licensee shall deliver such quarterly utility reimbursements to City within thirty (30) days of receiving City's invoice therefor, which shall include reasonable evidence of the gas and electricity charges incurred by City for the Property. If the Commencement Date occurs on a day other than the first day of a quarterly period, then the utility reimbursement to City for such fractional quarterly period shall be prorated based on a ninety (90) day quarter.

10.2 Utility Maintenance. Licensee shall be obligated, at its sole cost and expense, to repair and maintain in good operating condition all utilities located within the Premises and all utilities installed by Licensee (whether within or outside the Premises). If Licensee requests City to perform such maintenance or repair, whether emergency or routine, City may charge Licensee for the cost of the work performed at the then prevailing standard rates, and Licensee agrees to pay said charges to City promptly upon billing. Licensee shall pay for repair of utilities located outside the Premises (regardless of who installed the same) which are damaged by or adversely affected by Licensee's use of such utility and shall be responsible for all damages, liabilities and claims arising therefrom. The parties agree that any and all utility improvements shall become part of the realty and are not trade fixtures.

11. COMPLIANCE WITH LAWS AND RISK MANAGEMENT REQUIREMENTS

11.1 Compliance with Laws.

11.1.1 Licensee shall promptly comply, at its sole expense, with all present or future laws, judicial decisions, orders, regulations and requirements of all governmental authorities relating to the Premises or the use or occupancy thereof, whether in effect at the time of the execution of this License or adopted at any time thereafter and whether or not within the present contemplation of the parties.

11.1.2 Licensee further understands and agrees that it is Licensee's obligation, at its sole cost, to cause the Premises and Licensee's uses thereof to be conducted in compliance with the Americans with Disabilities Act, 42 U.S.C.A.§§ 12101 et seq. and any other disability access laws, rules, and regulations. Licensee shall not be required to make any structural alterations in order to comply with such laws unless such alterations shall be occasioned, in whole or in part, directly or indirectly, by Licensee's Alterations, Licensee's manner of using the Premises, or any act or omission of Licensee, its Agents or Invitees. Any Alteration made by or on behalf of

APPENDIX H: 2650 Bayshore License Agreement

Page 12 of 39

Licensee pursuant to the provisions of this Section shall comply with the provisions of <u>Section</u> <u>8.1</u> [Licensee's Repairs], above.

11.1.3 Licensee shall comply with all Fire Code requirements in its use and occupancy of the Premises.

11.1.4 The parties acknowledge and agree that Licensee's obligation to comply with all laws as provided herein is a material part of the bargained for consideration under this License. Licensee's obligation under this Section shall include, without limitation, the responsibility of Licensee to comply with Applicable Law by making substantial or structural repairs and modifications to the Premises (including any of Licensee's Alterations), regardless of, among other factors, the relationship of the cost of curative action to the fee under this License, the relative benefit of the repairs to Licensee or City and the degree to which the curative action may interfere with Licensee's use or enjoyment of the Premises. This section shall not apply to any non-compliance with laws relating to changes in use or configuration of the Premises requested by City.

11.2 Regulatory Approvals.

11.2.1 Responsible Party. Licensee understands and agrees that Licensee's use of the Premises and construction of any Alterations permitted hereunder may require authorizations, approvals or permits from governmental regulatory agencies with jurisdiction over the Premises. Licensee shall be solely responsible for obtaining any and all such regulatory approvals. Licensee shall not seek any regulatory approval without first obtaining the written consent of City hereunder. Licensee shall bear all costs associated with applying for and obtaining any necessary or appropriate regulatory approval and shall be solely responsible for satisfying any and all conditions imposed by regulatory agencies as part of a regulatory approval, other than any conditions that may arise out of Hazardous Materials in, on, or under any part of the Building or other portion of the Premises that were present immediately prior to May 1, 2013, to the extent that such regulatory conditions relate to property conditions existing at such time, and except to the extent that the regulatory conditions relate to Licensee's exacerbation of any preexisting condition; provided, however, that City shall not be required to engage in any work or incur any costs necessary to secure any regulatory approval or satisfy any condition imposed by a regulatory agency. Any fines or penalties levied as a result of Licensee's failure to comply with the terms and conditions of any regulatory approval shall be immediately paid and discharged by Licensee, and City shall have no liability, monetary or otherwise, for any such fines or penalties. As defined in Section 18.2 herein, Licensee shall Indemnify City and the other Indemnified Parties hereunder against all Losses arising in connection with Licensee's failure to obtain or comply with the terms and conditions of any regulatory approval.

11.2.2 City Acting as Leasehold Owner of Real Property. Licensee further understands and agrees that City is entering into this License in its capacity as a leasehold owner with a proprietary interest in the Premises and not as a regulatory agency with police powers. Nothing in this License shall limit in any way Licensee's obligation to obtain any required approvals from City departments, boards, agencies, or commissions having jurisdiction over the Premises or Licensee's activities at the Premises. By entering into this License, City is in no way modifying or limiting Licensee's obligation to cause the Premises to be used and occupied in accordance with all Applicable Law, as provided further above.

11.3 Compliance with City's Risk Management Requirements. Licensee shall not do anything, or permit anything to be done, in or about the Premises which would be prohibited by or increase the rates under a standard form fire insurance policy or subject City to potential premises liability. Licensee shall faithfully observe, at its expense, any and all requirements of

APPENDIX H: 2650 Bayshore License Agreement

Page 13 of 39

No 815 Bén crea

12. SUBORDINATION

This License is and shall be subordinate to the Lease (including Prologis' rights and City's obligations thereunder) and any reciprocal easement agreement, ground lease, facilities lease or other underlying leases or licenses and the lien of any mortgage or deed of trust, that may now exist or hereafter be executed affecting the Property, or any part thereof, or City's interest therein. Notwithstanding the foregoing, City or the holder shall have the right to subordinate any such interests to this License. Licensee agrees, however, to execute and deliver, upon demand by City and in the form requested by City, any additional documents evidencing the priority or subordination of this License.

13. INABILITY TO PERFORM

If City is unable to perform or is delayed in performing any of City's obligations under this License, by reason of acts of God, accidents, breakage, repairs, strikes, lockouts, other labor disputes, protests, riots, demonstrations, inability to obtain utilities or materials or by any other reason beyond City's reasonable control, no such inability or delay shall constitute an actual or constructive eviction, in whole or in part, or entitle Licensee to any abatement or diminution of fee or relieve Licensee from any of its obligations under this License, or impose any liability upon City or its Agents by reason of inconvenience, annoyance, interruption, injury or loss to or interference with Licensee's business or use and occupancy or quiet enjoyment of the Premises or any loss or damage occasioned thereby. Licensee hereby waives and releases any right to terminate this License under Section 1932, subdivision 1 of the California Civil Code or any similar law, statute or ordinance now or hereafter in effect.

14. DAMAGE AND DESTRUCTION

14.1 Damage and Destruction. If all or any portion of the Premises is damaged by fire or other casualty, City shall have no obligation to repair the Premises. City shall provide Licensee with a copy of the notice City receives from Prologis of Prologis' estimated time to restore such damage (the "Prologis Repair Notice") within ten (10) days of City's receipt of the Prologis Repair Notice. If the damage is not repaired within thirty (30) days, or the damage reasonably prevents Licensee from using a portion of the Premises in excess of 250 square feet, Licensee shall be entitled to a proportionate reduction of the Base Fee until the damage is repaired.

If the restoration time set forth in the Prologis Repair Notice (the "Repair Period") is estimated to exceed two hundred ten (210) days, City shall have the right to terminate this License by delivering written notice of such termination to Licensee within thirty (30) days of City's delivery of the Prologis Repair Notice to Licensee, in which event this License shall terminate as of the date specified in such termination notice.

If at any time during the last twelve (12) months of the Term of this License all or any portion of the Premises is damaged or destroyed, then Licensee may terminate this License by giving written notice to City of its election to do so within thirty (30) days after the date of the occurrence of such damage; provided, however, Licensee may terminate only if such damage or destruction substantially impairs its use or occupancy of the Premises for general office purposes. The effective date of termination shall be specified in the notice of termination, which date shall not be more than thirty (30) days from the date of the notice.

APPENDIX H: 2650 Bayshore License Agreement Page 14 of 39

14.2 City Repairs. Notwithstanding anything to the contrary in this License, City shall have no obligation to repair the Premises in the event the damage or destruction, and in no event shall City be required to repair any damage to Licensee's Personal Property or any paneling, decorations, railings, floor coverings, or any Alterations installed or made on the Premises by or at the expense of Licensee.

14.3 Termination by City. In the event the Premises are substantially damaged or destroyed, Prologis intends to restore the Premises pursuant to the Lease, and City intends to use the restored Premises for public purposes inconsistent with this License, City may terminate this License upon written notice to Licensee.

14.4 Licensee Waiver. City and Licensee intend that the provisions of this Section govern fully in the event of any damage or destruction and accordingly, City and Licensee each hereby waives the provisions of Section 1932, subdivision 2, Section 1933, subdivision 4, Section 1941, and Section 1942 of the Civil Code of California or under any similar law, statute or ordinance now or hereafter in effect, to the extent such provisions apply.

15. EMINENT DOMAIN

15.1 Definitions.

15.1.1 "Taking" means a taking or damaging, including severance damage, by eminent domain, inverse condemnation or for any public or quasi-public use under law. A Taking may occur pursuant to the recording of a final order of condemnation, or by voluntary sale or conveyance in lieu of condemnation or in settlement of a condemnation action.

15.1.2 "Date of Taking" means the earlier of (i) the date upon which title to the portion of the Premises taken passes to and vests in the condemner or (ii) the date on which Licensee is dispossessed.

15.1.3 "Award" means all compensation, sums or anything or value paid, awarded or received for a Taking, whether pursuant to judgment, agreement, settlement or otherwise.

15.2 General. If during the Term there is any Taking of all or any part of the Premises or any interest in this License, the rights and obligations of Licensee shall be determined pursuant to this Section. City and Licensee intend that the provisions hereof govern fully Licensee's rights in the event of a Taking and accordingly, Licensee hereby waives any right to terminate this License in whole or in part under Sections 1265.110, 1265.120, 1265.130 and 1265.140 of the California Code of Civil Procedure or under any similar law now or hereafter in effect.

15.3 Total Taking; Automatic Termination. If there is a total Taking of the Premises, then this License shall terminate as of the Date of Taking.

15.4 Partial Taking; Election to Terminate.

15.4.1 If there is a Taking of any portion (but less than all) of the Premises, then this License shall terminate in its entirety under either of the following circumstances: (i) if all of the following exist: (A) the partial Taking renders the remaining portion of the Premises unsuitable for continued use by Licensee for the permitted uses described in <u>Section 6.1</u>, (B) the condition rendering the Premises unsuitable either is not curable or is curable but neither City nor Prologis is willing or able to cure such condition, and (C) Licensee elects to terminate; or (ii) if City elects to terminate.

APPENDIX H: 2650 Bayshore License Agreement

Page 15 of 39

15.4.2 If Licensee elects to terminate under the provisions of this <u>Section 15</u>, Licensee shall do so by giving written notice to the City before or within thirty (30) days after the Date of Taking, and thereafter this License shall terminate upon the later of the thirtieth (30th) day after such written notice is given or the Date of Taking.

22월21 - 18월 22일 - 18월 22일 - 18월 22일 - 18일 - 1 18일 - 18g - 18g - 18g - 18g 18일 - 18g - 18g

15.5 License Fee: Award. Upon termination of this License pursuant to an election under <u>Section 15.4</u> above, then: (i) Licensee's obligation to pay the Base Fee shall continue up until the date of termination, and thereafter shall cease, except that fee shall be reduced as provided in <u>Section 15.6</u> below for any period during which this License continues in effect after the Date of Taking, and (ii) City shall be entitled to the entire Award in connection therewith (including, but not limited to, any portion of the Award made for the value of Licensee's interest under this License, provided that Licensee may make a separate claim for compensation, and Licensee shall receive any Award made specifically to Licensee, for Licensee's relocation expenses or the interruption of or damage to Licensee's business or damage to Licensee's Personal Property.

15.6 Partial Taking: Continuation of License. If there is a partial Taking of the Premises under circumstances where this License is not terminated in its entirety under <u>Section</u> <u>15.4</u> above, then this License shall terminate as to the portion of the Premises so taken, but shall remain in full force and effect as to the portion not taken, and the rights and obligations of the parties shall be as follows: (a) the Base Fee shall be reduced by an amount that is in the same ratio to the Base Fee as the area of the Premises taken bears to the area of the Premises prior to the Date of Taking; and (b) City shall be entitled to the entire Award in connection therewith (including, but not limited to, any portion of the Award made for the value of the License interest created by this License), and Licensee shall have no claim against City for the value of any unexpired term of this License, provided that Licensee may make a separate claim for compensation, and Licensee shall receive any Award made specifically to Licensee, for Licensee's Personal Property.

15.7 Temporary Takings. Notwithstanding anything to contrary in this Section, if a Taking occurs with respect to all or any part of the Premises for a limited period of time not in excess of one hundred eighty (180) consecutive days, this License shall remain unaffected thereby, and Licensee shall continue to pay all fees and to perform all of the terms, conditions and covenants of this License. In the event of such temporary Taking, Licensee shall be entitled to receive that portion of any Award representing compensation for the use or occupancy of the Premises during the Term up to the total fee owing by Licensee for the period of the Taking, and City shall be entitled to receive the balance of any Award.

16. ASSIGNMENT AND SUBLETTING

Licensee shall not directly or indirectly (including, without limitation, by merger, acquisition or other transfer of any controlling interest in Licensee), voluntarily or by operation of law, sell, assign, encumber, pledge or otherwise transfer (collectively, "Assignment") any part of its interest in or rights with respect to the Premises, or permit any portion of the Premises to be occupied by anyone other than itself, or sublet or license any portion of the Premises (collectively, "Subletting"), without City's prior written consent in each instance, as provided herein.

17. DEFAULT; REMEDIES

17.1 Events of Default. Any of the following shall constitute an event of default by Licensee hereunder:

APPENDIX H: 2650 Bayshore License Agreement

Page 16 of 39

17.1.1 A failure to pay the Base Fee or any other amount payable under this License when due, and such failure continues for three (3) days after the date of written notice by City. However, City shall not be required to provide such notice with respect to more than two delinquencies and any such failure by Licensee after Licensee has received two (2) such notices shall constitute a default by Licensee hereunder without any further action by City or opportunity of Licensee to cure except as may be required by Section 1161 of the California Code of Civil Procedure.

14

17.1.2 A failure to comply with any other covenant, condition or representation under this License and such failure continues for fifteen (15) days after the date of written notice by City, provided that if such default is not capable of cure within such fifteen (15) day period, Licensee shall have a reasonable period to complete such cure if Licensee promptly undertakes action to cure such default within such 15-day period and thereafter diligently prosecutes the same to completion within sixty (60) days after the receipt of notice of default from City. City shall not be required to provide such notice with respect to more than two defaults and after the second notice any subsequent failure by Licensee shall constitute an event of default hereunder;

17.1.3 A vacation or abandonment of the Premises for a continuous period in excess of five (5) business days;

17.1.4 An uncured event of default under the Towing Agreement; or

17.1.5 An appointment of a receiver to take possession of all or substantially all of the assets of Licensee, or an assignment by Licensee for the benefit of creditors, or any action taken or suffered by Licensee under any insolvency, bankruptcy, reorganization, moratorium or other debtor relief act or statute, whether now existing or hereafter amended or enacted, if any such receiver, assignment or action is not released, discharged, dismissed or vacated within sixty (60) days.

17.2 City Rights Upon Default. Upon the occurrence of an event of default by Licensee, City shall have the right to terminate the License in addition to the following rights and all other rights and remedies available to City at law or in equity:

17.2.1 The rights and remedies provided by California Civil Code Section 1951.2 (damages on termination for breach), including, but not limited to, the right to terminate Licensee's right to possession of the Premises and to recover the worth at the time of award of the amount by which the unpaid Base Fee for the balance of the Term after the time of award exceeds the amount of rental loss for the same period that Licensee proves could be reasonably avoided, as computed pursuant to subsection (b) of such Section 1951.2. City's efforts to mitigate the damages caused by Licensee's breach of this License shall not waive City's rights to recover damages upon termination.

17.2.2 The rights and remedies provided by California Civil Code Section 1951.4 (continuation of lease after breach and abandonment), allowing City to continue this License in effect and to enforce all its rights and remedies under this License, including the right to recover the Base Fee as it becomes due, for so long as City does not terminate Licensee's right to possession, if Licensee has the right to sublet or assign, subject only to reasonable limitations. For purposes hereof, none of the following shall constitute a termination of Licensee's right of possession: acts of maintenance or preservation; efforts to relet the Premises or the appointment of a receiver upon City's initiative to protect its interest under this License; withholding consent to an Assignment or Sublicense, or terminating an Assignment or Sublicense, if the withholding or termination does not violate the rights of Licensee specified in subdivision (b) of California Civil Code Section 1951.4. If City exercises its remedy under California Civil Code Section 1951.4. City may from time to time sublet or license the Premises or any part thereof for such

APPENDIX H: 2650 Bayshore License Agreement

Page 17 of 39

방 옷 옷 사람을 물 것을 물 것이 가지 않는 것이 나라.

term or terms (which may extend beyond the Term) and at such rent or fee and upon such other terms as City may in its sole discretion deem advisable, with the right to make alterations and repairs to the Premises. Upon each such subletting or sublicensing, Licensee shall be liable for Base Fee and any other amounts due hereunder, as well as the cost of such subletting or sublicensing and such alterations and repairs incurred by City and the amount, if any, by which fee owing hereunder for the period of such subletting or sublicensing (to the extent such period does not exceed the Term) exceeds the amount to be paid as rent or fee for the Premises for such period pursuant to such subletting or sublicensing. No action taken by City pursuant to this subsection shall be deemed a waiver of any default by Licensee, or to limit City's right to terminate this License at any time.

17.2.3 The right to have a receiver appointed for Licensee upon application by City to take possession of the Premises and to apply any fees or rental collected from the Premises and to exercise all other rights and remedies granted to City pursuant to this License.

17.3 City's Right to Cure Licensee's Defaults. If Licensee defaults in the performance of any of its obligations under this License, then City may, at its sole option, remedy such default for Licensee's account and at Licensee's expense by providing Licensee with three (3) days' prior written or oral notice of City's intention to cure such default (except that no such prior notice shall be required in the event of an emergency as determined by City). Such action by City shall not be construed as a waiver of such default or any rights or remedies of City, and nothing herein shall imply any duty of City to do any act that Licensee is obligated to perform. Licensee shall pay to City upon demand, all reasonable costs, damages, expenses or liabilities incurred by City, including, without limitation, reasonable attorneys' fees, in remedying or attempting to remedy such default. Licensee's obligations under this Section shall survive the termination of this License.

18. WAIVER OF CLAIMS; INDEMNIFICATION

18.1 Limitation on City's Liability: Waiver of Claims. Except as provided for in <u>Sections 24.3</u> and <u>24.8</u>, below, City shall not be responsible for or liable to Licensee, and Licensee hereby assumes the risk of, and waives and releases City and its Agents from all Claims (as defined below) for, any injury, loss or damage to any person or property in or about the Premises by or from any cause whatsoever including, without limitation, (i) any act or omission of persons occupying adjoining premises or any part of the Building adjacent to or connected with the Premises which are not occupied by City, (ii) theft, (iii) explosion, fire, steam, oil, electricity, water, gas or rain, pollution or contamination, (iv) stopped, leaking or defective Building Systems, (v) Building defects, and (vi) any other acts, omissions or causes. Nothing herein shall relieve City from liability caused solely and directly by the gross negligence or willful misconduct of City or its Agents, but City shall not be liable under any circumstances for any consequential, incidental or punitive damages.

18.2 Licensee's Indemnity. Except as provided for in <u>Sections 24.3</u> and <u>24.8</u>, below, Licensee, on behalf of itself and its successors and assigns, shall indemnify, defend and hold harmless ("Indemnify") the City and County of San Francisco, including, but not limited to, all of its boards; commissions, departments, agencies and other subdivisions, and Prologis, and all of their Agents, and their respective heirs, legal representatives, successors and assigns (individually and collectively, the "Indemnified Parties"), and each of them, from and against any and all liabilities, losses, costs, claims, judgments, settlements, damages, liens, fines, penalties and expenses, including, without limitation, direct and vicarious liability of every kind (collectively, "Claims"), incurred in connection with or arising in whole or in part from: (a) any accident, injury to or death of a person, including, without limitation, employees of Licensee, or loss of or damage to property, howsoever or by whomsoever caused, occurring in or about the Property; (b) any default by Licensee in the observation or performance of any of

APPENDIX H: 2650 Bayshore License Agreement

Page 18 of 39

the terms, covenants or conditions of this License to be observed or performed on Licensee's part; (c) the use or occupancy or manner of use or occupancy of the Premises by Licensee, its Agents or Invitees or any person or entity claiming through or under any of them; (d) the condition of the Premises; (e) any construction or other work undertaken by Licensee on the Premises whether before or during the Term of this License; or (f) any acts, omissions or negligence of Licensee, its Agents or Invitees, in, on or about the Premises or the Property, all regardless of the active or passive negligence of, and regardless of whether liability without fault is imposed or sought to be imposed on, the Indemnified Parties, except to the extent that such Indemnity is void or otherwise unenforceable under any Applicable Law in effect on or validly retroactive to the date of this License and further except only such Claims as are caused exclusively by the willful misconduct or gross negligence of the Indemnified Parties. The foregoing Indemnity shall include, without limitation, reasonable fees of attorneys, consultants and experts and related costs and the Indemnified Party's costs of investigating any Claim. Licensee specifically acknowledges and agrees that it has an immediate and independent obligation to defend the Indemnified Parties from any claim which actually or potentially falls within this indemnity provision even if such allegation is or may be groundless, fraudulent or false, which obligation arises at the time such claim is tendered to Licensee by an Indemnified Party and continues at all times thereafter. Licensee's obligations under this Section shall survive the termination of the License.

19. INSURANCE

19.1 Licensee's Insurance. Licensee, at its sole cost, shall procure and keep in effect at all times during the Term insurance for the Premises in the form and amounts and under the terms and conditions specified in the Section 11 of the Towing Agreement [Required Insurance].

19.2 Licensee's Personal Property. Licensee shall be responsible, at its expense, for separately insuring Licensee's Personal Property.

19.3 City's Self Insurance. Licensee acknowledges that City self-insures against casualty, property damage and public liability risks and agrees that City may at its sole election, but shall not be required to, carry any third party insurance with respect to the Premises or otherwise.

19.4 Waiver of Subrogation. Notwithstanding anything to the contrary contained herein, to the extent permitted by Licensee's policies of insurance and any third party insurance that City elects to carry with respect to the Premises, City and Licensee each hereby waive any right of recovery against the other party and against any other party maintaining a policy of insurance covering the Premises or the contents, or any portion thereof, for any loss or damage maintained by such other party with respect to the Premises or any portion thereof or the contents of the same or any operation therein, whether or not such loss is caused by the fault or negligence of such other party. If any policy of insurance relating to the Premises carried by Licensee does not permit the foregoing waiver or if the coverage under any such policy would be invalidated due to such waiver, Licensee shall obtain, if possible, from the insurer under such policy a waiver of all rights of subrogation the insurer might have against City or any other party maintaining a policy of insurance covering the same loss, in connection with any claim, loss or damage covered by such policy.

20. ACCESS BY CITY

City reserves for itself and any of its designated Agents, the right to enter the Premises at all reasonable times, with or without advance notice, including, without limitation, in order to (i) oversee or inspect Licensee's operations or conduct any business with Licensee; (ii) show the

APPENDIX H: 2650 Bayshore License Agreement

Page 19 of 39

Premises to prospective Licensees or other interested parties, to post notices of nonresponsibility, to conduct any environmental audit of Licensee's use of the Premises, to repair, alter or improve any part of the Building, Building Systems or the Premises, and for any other lawful purpose; or (iii) whenever City believes that emergency access is required. City shall have the right to use any means that it deems proper to open doors in an emergency in order to obtain access to any part of the Premises, and any such entry shall not be construed or deemed to be a forcible or unlawful entry into or a detainer of, the Premises, or an eviction, actual or constructive, of Licensee from the Premises or any portion thereof. Licensee shall not alter any lock or install any new or additional locking devices without the prior written consent of City. City shall at all times have a key with which to unlock all locks installed in the Premises (excluding Licensee's vaults, safes or special security areas, if any, designated by Licensee in writing to City).

21. LICENSEE'S CERTIFICATES

Licensee, at any time, and from time to time upon not less than ten (10) days' prior notice from City, shall execute and deliver to City or to any party designated by City a certificate stating: (a) that Licensee has accepted the Premises, (b) the Commencement Date and Expiration Date of this License, (c) that this License is unmodified and in full force and effect (or, if there have been modifications, that the License is in full force and effect as modified and stating the modifications), (d) whether or not there are then existing any defenses against the enforcement of any of Licensee's obligations hereunder (and if so, specifying the same), (e) whether or not there are any defaults then existing under this License (and if so specifying the same), (f) the dates, if any, to which the Base Fee and any other amounts owing under this License have been paid, and (g) any other information that may be required.

22. PREMISES MAINTENANCE REQUIREMENTS

Until the Approved Operations Plan is finalized pursuant to <u>Section 8.2</u>, the "Maintenance Plan" shall mean the Maintenance Plan attached as an exhibit to the 2005 Operations Plan approved by City pursuant to the Towing Agreement, modified as reasonably necessary (and with City's prior written consent to apply to the Premises instead of the Pier 70 License Area, provided Licensee obtains City's prior written consent to such modification. Once the Approved Operations Plan is finalized pursuant to <u>Section 8.2</u>, the "Maintenance Plan" shall mean the Maintenance Plan that is attached as an exhibit to the Approved Operations Plan. Licensee shall faithfully comply with the Maintenance Plan, and any violation of the Maintenance Plan shall be a violation of this <u>Section 22</u>.

22.1 Maintenance of Pavement. Licensee shall maintain the pavement in the Premises in good condition, including the vehicle and parts storage area, in order to prevent Releases of Hazardous Materials (as those terms are defined below) into or onto the Premises, the remainder of the Property, or the environment. Licensee shall inspect the pavement at least quarterly and shall record in written form the dates and times of such inspections, the name or names of the persons conducting the inspections, and any damage discovered to the pavement and its location. Licensee shall promptly repair any cracked or broken pavement and shall report such damage and repair to City. City shall have the right to enter and inspect the Premises from time to time to ensure Licensee's compliance with the terms of this License, including, without limitation, this <u>Section 22.1</u> and <u>Section 22.2</u> below.

(a) Licensee must furnish at its own cost sealed concrete pads and hazardous waste containment systems for removing and storing residual fluids and batteries from vehicles;

APPENDIX H: 2650 Bayshore License Agreement

Page 20 of 39

- (b) Licensee shall clean up and remove all leaked or spilled fluids immediately upon discovery or upon notice by City in accordance with the Maintenance Plan.
- (c) Licensee shall only store vehicles and parts in areas with pavement in good condition. Draining must take place on a sealed concrete pad with a containment system to collect residual fluids.
- (d) Licensee must ensure that paving, including maintenance and repair, shall protect existing or future groundwater monitoring wells on the Premises.

22.2 Plan and Reporting. In addition to the requirements in <u>Section 22.1</u> above, the Maintenance Plan shall provide for ongoing inspection, spill and drip response procedures, a maintenance schedule for pavement maintenance and repair of cracks and other identified deficiencies, staff training protocols, and supervised video or photo documentation of initial surface conditions and exit surface conditions. In addition to pavement maintenance, the Maintenance Plan shall include other property management protocols, including but not limited to, maintenance of fencing, lighting, signage and permanent or temporary buildings. The Maintenance Plan shall also include a reporting schedule, with submittal of reports at least quarterly, documenting maintenance performed. Such reports shall include, the following information:

(a) An initial survey of pavement condition as of May 1, 2013;

- (b) Surface type and surface conditions at time of repair, including photographs of pre- and post-repair conditions;
- (c) Repair procedure performed;
- (d) Cost of repairs performed; and
- (e) A final survey of pavement condition at the time of termination of Term.

23. SURRENDER OF PREMISES

Upon the Expiration Date or other termination of the Term of this License, Licensee shall immediately peaceably quit and surrender to City the Premises together with all Alterations approved by City in good order and condition, free of debris and any Hazardous Materials deposited on the Premises on or after May 1, 2013, except for normal wear and tear after Licensee's having made the last necessary repair required on its part under this License, and further except for any portion of the Premises condemned and any damage and destruction for which Licensee is not responsible hereunder. The Premises shall be surrendered free and clear of all liens and encumbrances other than liens and encumbrances existing as of the May 1, 2013, and any other encumbrances created by City.

Immediately before the Expiration Date or other termination of this License, Licensee shall remove all of Licensee's Personal Property as provided in this License, and repair any damage resulting from the removal. Notwithstanding anything to the contrary in this License, City can elect at any time prior to the Expiration Date or within thirty (30) days after termination of this License, to require Licensee to remove, at Licensee's sole expense, all or part of Licensee's Alterations or other improvements or equipment constructed or installed by or at the expense of Licensee, or any vehicles that City may in its sole and absolute discretion authorize to be stored

APPENDIX H: 2650 Bayshore License Agreement

Page 21 of 39

on the Premises after termination of this License, together with any Hazardous Materials contained within such vehicles.

Licensee shall promptly remove such items and shall repair, at its expense, any damage to the Premises resulting from the performance of its removal obligations pursuant to this Section. Licensee's obligations under this Section shall survive the Expiration Date or other termination of this License. Any items of Licensee's Personal Property remaining in the Premises after the Expiration Date or sooner termination of this License may, at City's option, be deemed abandoned and disposed of in accordance with Section 1980 et seq. of the California Civil Code or in any other manner allowed by law. Any expenses or costs incurred by City to discharge liens, remove Licensee's Personal Property or Alterations, or repair any damage for which Licensee is responsible shall be charged against Licensee's Security Deposit.

Concurrently with the surrender of the Premises, Licensee shall, if requested by City, execute, acknowledge and deliver to any instrument reasonably requested by City to evidence or otherwise effect the termination of Licensee's interest to the Premises and to effect such transfer or vesting of title to the Alterations or equipment which remain part of the Premises.

24. HAZARDOUS MATERIALS

24.1 Definitions. As used herein, the following terms shall have the meanings set forth below:

24.1.1 "Environmental Laws" shall mean any present or future federal, state, local or administrative law, rule, regulation, order or requirement relating to Hazardous Material (including, without limitation, their generation, use, handling, transportation, production, disposal, discharge or storage), or to health and safety, industrial hygiene or the environment, including, without limitation, soil, air and groundwater conditions, including without limitation Article 21 of the San Francisco Health Code.

24.1.2 "Hazardous Material" shall mean any material that, because of its quantity, concentration or physical or chemical characteristics, is at any time now or hereafter deemed by any federal, state or local governmental authority to pose a present or potential hazard to human health, welfare or safety or to the environment. Hazardous Material includes, without limitation, any material or substance listed or defined as a "hazardous substance", or "pollutant" or "contaminant" pursuant to the Comprehensive Environmental Response, Compensation and Liability Act of 1980 ("CERCLA", also commonly known as the "Superfund" law), as amended, (42 U.S.C. Sections 9601 et seq.) or pursuant to the Carpenter-Presley-Tanner Hazardous Substance Account Act, as amended, (Cal. Health & Safety Code Sections 25300 et seq.) or pursuant to the Hazardous Waste Control Law, as amended, (Cal. Health & Safety Code Sections 25100 et seq.) or pursuant to the Porter-Cologne Water Quality Control Act, as amended, (Cal. Water Code Sections 13000 et seq.) or pursuant to Section 25501(o) of the California Health and Safety Code; and petroleum, including crude oil or any fraction thereof, natural gas or natural gas liquids.

24.1.3 "Indemnify" shall mean, whenever any provision of this <u>Section 24</u> requires a person or entity (the "Indemnitor") to Indemnify any other entity or person (the "Indemnitee"), the Indemnitor shall be obligated to defend, indemnify, hold harmless and protect the Indemnitee, its officers, employees, agents, stockholders, constituent partners, and members of its boards and commissions harmless from and against any and all Losses arising directly or indirectly, in whole or in part, out of the act, omission, event, occurrence or condition with respect to which the Indemnitor is required to Indemnify such Indemnitee, whether such act, omission, event, occurrence or condition is caused by the Indemnitor or its agents, employees or contractors, or by any third party or any natural cause, foreseen or

APPENDIX H: 2650 Bayshore License Agreement

Page 22 of 39

unforeseen; provided that no Indemnitor shall be obligated to Indemnify any Indemnitee against any Loss from the negligence or intentional wrongful acts or omissions of such Indemnitee, or such Indemnitee's agents, employees or contractors. If a Loss is attributable partially to the negligent or intentionally wrongful acts or omissions of the Indemnitee (or its agents, employees or contractors), such Indemnitee shall be entitled to Indemnification for that part of the Loss not attributable to such Indemnitee's (or its agents, employees or contractors) negligent or intentionally wrongful acts or omissions.

24.1.4 "Investigate and Remediate" (also "Investigation" and "Remediation") shall mean the undertaking of any activities to determine the nature and extent of Hazardous Material that may be located in, on, under or about the Premises or surrounding property or that has been, is being or threaten to be Released into the environment, and to clean up, remove, contain, treat, stabilize, monitor or otherwise control such Hazardous Material.

24.1.5 "Losses" shall mean any and all claims, demands, losses, damages, liens, liabilities, injuries, deaths, penalties, fines, lawsuits and other proceedings, judgments and awards rendered therein, and costs and expenses, including, but not limited to, reasonable attorneys' fees.

24.1.6 "Release" when used with respect to Hazardous Material shall include any actual, threatened or imminent spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, dumping, or disposing into or inside the Premises, or in, on, under or about any other part of the Premises or into the environment.

24.2 No Hazardous Materials. Licensee covenants and agrees that neither Licensee nor any of its Agents, Employees or Invitees shall cause or permit any Hazardous Material to be brought upon, kept, used, stored, generated, processed, produced, packaged, treated, emitted, discharged or disposed of in, on or about the Premises, or transported to or from the Premises without the prior written consent of City, which consent shall not be unreasonably withheld so long as Licensee demonstrates to City's reasonable satisfaction that such Hazardous Material is necessary to Licensee's business, will be handled in a manner which strictly complies with all Environmental Laws and will not materially increase the risk of fire or other casualty to the Premises, and without the prior written consent of Prologis. City and Licensee understand that the vehicles transported to and stored at the Property will contain and may partially consist of Hazardous Materials. Licensee shall immediately notify City if and when Licensee learns or has reason to believe a Release of Hazardous Material on or about any part of the Premises or the remainder of the Property has occurred that may require any Investigation or Remediation. Licensee shall not be responsible for the safe handling of Hazardous Materials introduced on the remainder of the Property during the Term of the Towing Agreement by City or its Agents.

Without limiting any other obligation of Licensee, if acts or omissions of Licensee results in any Hazardous Materials Release or contamination of the Premises, Licensee shall, at its sole expense, promptly take all action that is necessary to return the Premises to the condition existing prior to the introduction of such Hazardous Material in, on, under or about the Premises; provided that City approval of such actions shall first be obtained, which approval shall not be unreasonably withheld so long as such actions could not potentially have any material adverse effect upon the Premises.

24.3 Licensee's Environmental Indemnity. If Licensee breaches any of its obligations contained in this <u>Section 24</u>, or, if any act, omission or negligence of Licensee, its Agents, Employees or Invitees, results in any Release of Hazardous Material in, on or under any part of the Premises, or the remainder of the Property or the Building, then, Licensee shall, on behalf of itself and its successors and assigns, Indemnify the City from and against all Claims (including, without limitation, claims for injury to or death of a person, damages,

APPENDIX H: 2650 Bayshore License Agreement

Page 23 of 39

liabilities, losses, judgments, penalties, fines, regulatory or administrative actions, damages for decrease in value of the Premises or the remainder of the Building or the Property, the loss or restriction of the use of rentable or usable space or of any amenity of the Premises, or the remainder of the Building or the Property, damages arising from any adverse impact on marketing of any such space, restoration work required by any federal, state or local governmental agency or political subdivision because of Hazardous Material present in the soil or groundwater in, on or under the Premises, the remainder of the Property, or in any Improvements, and sums paid in settlement of claims, attorneys' fees, consultants' fees and experts' fees and costs) arising during or after the Term of the Towing Agreement and relating to such breach or Release. The foregoing Indemnity includes, without limitation, costs incurred in connection with activities undertaken to Investigate and Remediate any Release of Hazardous Material, and to restore the Premises or the remainder of the Building or Property to their prior condition. This indemnification of City by Licensee includes, but is not limited to, costs incurred in connection with any investigation of site conditions or any clean-up, remediation, removal or restoration work requested by Prologis or required by any federal, state or local governmental agency or political subdivision because of Hazardous Material present in the soil or groundwater in, on or under the Premises, the remainder of the Property or in any Alterations. Licensee's obligations hereunder shall survive the termination of this License. Licensee's obligations under this Section do not include an indemnity for Claims arising as a result of Hazardous Materials (or other conditions alleged to be in violation of any Environmental Law) in, on, or under any part of the Premises or the remainder of the Building or Property that were present prior to May 1, 2013, or to the extent that such Claims relate to conditions existing prior to May 1, 2013, except to the extent that Licensee exacerbates any pre-existing condition or introduces such Hazardous Materials or conditions. In the event any action or proceeding is brought against City by reason of a claim arising out of any Loss, Claim, injury or damage suffered on or about the Premises or the remainder of the Building or the Property for which Licensee has Indemnified the City and upon written notice from the City, Licensee shall at its sole expense answer and otherwise defend such action or proceeding using counsel approved in writing by the City. City shall have the right, exercised in its sole discretion, but without being required to do so, to defend, adjust, settle or compromise any claim, obligation, debt, demand, suit or judgment against the City in connection with this License or the Premises or the remainder of the Building or Property. The provisions of this paragraph shall survive the termination of this License with respect to any Loss occurring prior to or upon termination. Licensee and City shall afford each other a full opportunity to participate in any discussions with governmental regulatory agencies regarding any settlement agreement, cleanup or abatement agreement, consent decree, or other compromise or proceeding involving Hazardous Material.

24.4 Compliance with Environmental Laws. In addition to its obligations under <u>Section 11</u> above, and without limiting any such obligations or the foregoing, Licensee shall comply with the following requirements or more stringent requirements in any Environmental Laws:

24.4.1 Any Hazardous Materials found and identified as such in the towed vehicles which are not typically part of a towed vehicle, will be removed from the vehicle to an appropriate storage location within 72 hours.

24.4.2 No Hazardous Materials shall be voluntarily or involuntarily disposed of onto or into the ground or into the sewer system.

24.4.3 In no event shall Hazardous Waste (as defined by Title 22 of the California Code of Regulations, as amended) accumulate on the Premises for longer than 90 days. Drums used to store Hazardous Materials shall not be stacked more than two drums

APPENDIX H: 2650 Bayshore License Agreement

Page 24 of 39

high. City will not consider fluids that are normally contained within a vehicle to be Hazardous Wastes so long as they remain contained within the vehicle.

24.4.4 Licensee shall store all Hazardous Materials above ground, not in underground storage tanks.

24.4.5 An emergency response plan, emergency response employee training plan and an inventory of Hazardous Materials stored at the Premises by or for Licensee shall be provided to City.

24.5 Information Requests. City may from time to time request, and Licensee shall be obligated to provide, information reasonably adequate for City to determine that any and all Hazardous Materials are being handled in a manner which complies with all Environmental Laws.

24.6 Damaged Vehicles. Licensee shall inspect all vehicles before storing them on the Premises to make a good faith effort to determine that vehicles are not leaking fluids, including but not limited to gasoline, battery acid, oil, transmission and transfer case fluids, brake and clutch fluids, and coolant. Licensee shall secure vehicles that have been severely damaged due to collision or vandalism so that parts do not fall off and fluids do not leak. Leaking vehicles shall be drained of the leaking fluid on a sealed cement pad, which Licensee shall maintain free of build-up of Hazardous Materials. Licensee shall immediately clean-up and remove all leaked or spilled fluids, whether within or outside of sealed or contained areas. Licensee shall treat all such fluids and used cleaning materials as hazardous waste and shall dispose of them in accordance with Environmental Laws and Section 24.3 above. Parts that have fallen off a vehicle shall be placed inside the vehicle in a manner that minimizes damage to the vehicle. Licensee shall not be deemed an owner or operator of any damaged vehicle, but shall be deemed the owner of any fluids that leak from such damaged vehicles on or about the Premises.

24.7 Fire Prevention Measures. Licensee shall comply with the following fire prevention measures.

24.7.1 Welding and torch cutting shall be in conformance with the Daly City Fire Code.

24.7.2 No smoking will be allowed in the Premises except in designated areas consistent with Applicable Law.

24.7.3 No crushing, burning of wrecked or discarded motor vehicles or waste materials shall be allowed.

24.7.4 Motor vehicles, parts of motor vehicles, junk, waste, or other materials shall not be stored, displayed, or kept in a manner that could hinder or endanger firefighting efforts and operations.

24.7.5 One or more aisles, at least 30 inches wide (or any greater width required under Applicable Law), must be maintained in the area where vehicles are stored, to permit access by the Daly City Fire Department to all parts of the vehicle storage area. Entrances and exits to the area shall be at least 15 feet in width (or any greater width required under Applicable Law).

24.8 Requirement to Remove. Prior to termination of this License or during the Term if required by a governmental agency, Licensee, at its sole cost and expense, shall remove any

APPENDIX H: 2650 Bayshore License Agreement

Page 25 of 39

and all Hazardous Materials introduced in, on, under or about the Premises by Licensee, its Agents or Invitees during the Term or during any prior time in which Licensee occupied the Premises. Licensee shall not be obligated to remove any Hazardous Material introduced onto the Premises before, during, or after the Term of the Towing Agreement by (1) City or its officers, directors, employees, or Agents or (2) any prior occupants, tenants, property owners, individuals, corporations or entities. If Licensee demonstrates its compliance with the property maintenance requirements of this License, the Maintenance Plan described in <u>Section 22</u> above, there shall be a rebuttable presumption that any Hazardous Materials in, on, under or about the Premises were not introduced by Licensee, its Agents or Invitees. However, if Licensee does not demonstrate its compliance with the property maintenance requirements of this License or of the Maintenance Plan, then there shall be a rebuttable presumption that such Hazardous Materials are Licensee's responsibility to the extent that the presence of such Hazardous Materials bear a reasonable causal relationship to Licensee's non-compliance in their composition and location.

Prior to the termination of this License, at Licensee's expense, City and Licensee shall conduct a joint inspection of the Premises for the purpose of identifying Hazardous Materials on the Premises which can be determined to have been introduced by the Licensee and which Licensee is therefore required to remove. City's failure to conduct an inspection or to detect conditions if an inspection is conducted shall not be deemed to be a release of any liability for environmental conditions subsequently determined to be Licensee's responsibility.

24.9 Licensee's Environmental Condition Notification Requirements.

24.9.1 Notification of Any Release or Discharge. Licensee shall notify City in writing as required in the Maintenance Plan if Licensee learns or has reason to believe that a Release of any Hazardous Materials on or about any part of the Premises has occurred, whether or not the Release is in quantities that under any law would require the reporting of such Release to a governmental or regulatory agency.

24.9.2 Notification of Any Notice, Investigation, or Claim. Licensee shall also immediately notify City in writing of, and shall contemporaneously provide City with a copy of:

- (a) Any written notice of Release of Hazardous Materials on the Premises that is provided by Licensee or any subtenant or other occupant of the Premises to a governmental or regulatory agency;
- (b) Any notice of a violation, or a potential or alleged violation, of any Environmental Law that is received by Licensee or any subtenant or other occupant of the Premises from any governmental or regulatory agency;
- (c) Any inquiry, investigation, enforcement, cleanup, removal or other action that is instituted or threatened by a governmental or regulatory agency against Licensee or subtenant or other occupant of the Premises and that relates to the Release of Hazardous Materials on or from the Premises;
- (d) Any claim that is instituted or threatened by any third party against Licensee or any subtenant or other occupant of the Premises and that relates to any Release of Hazardous Materials on or from the Premises; and

APPENDIX H; 2650 Bayshore License Agreement

Page 26 of 39

(e) Any notice of the loss of any environmental operating permit by Licensee or any subtenant or other occupant of the Premises.

24.9.3 Notification of Regulatory Actions.

- (a) Licensee shall immediately notify City in writing of any inspection by any governmental or regulatory agency with jurisdiction over Hazardous Materials and shall provide City with a copy of any inspection record, correspondence, reports and related materials from or to the agency.
- (b) Licensee must notify City of any meeting, whether conducted face-toface or telephonically, between Licensee and any regulatory agency regarding an environmental regulatory action. City will be entitled to participate in any such meetings at its sole election.
- (c) Licensee must notify City of any environmental regulatory agency's issuance of an environmental regulatory approval. Licensee's notice to City must state the issuing entity, the environmental regulatory approval identification number, and the date of issuance and expiration of the environmental regulatory approval. In addition, Licensee must provide City with a list of any environmental regulatory approval, plan or procedure required to be prepared and/or filed with any regulatory agency for operations on the Property, including a "Spill Pollution Control and Countermeasure Plan." Licensee must provide City with copies of any of the documents within the scope of this Section upon City's request.
- (d) Licensee must provide City with copies of all communications with regulatory agencies and all non-privileged communications with other persons regarding potential or actual Hazardous Materials Claims arising from Licensee's or its Agents' or Invitees' operations at the Property. Upon City's request, Licensee must provide City with a log of all communications withheld under a claim of privilege that specifies the parties to and subject of each withheld communication.
- (e) City may from time to time request, and Licensee will be obligated to provide, information reasonably adequate for City to determine that any and all Hazardous Materials are being handled in a manner that complies with all Environmental Laws.

24.10 Environmental Security Deposit. Upon the Commencement Date, Licensee shall provide to the City, and shall maintain throughout the Term of this License and for a period of at least ninety (90) days after expiration of this License, a Security Deposit in the form of a Letter of Credit as described in the Towing Agreement Section 12.2. If Licensee receives a notice of violation or other regulatory order from a governmental or regulatory agency with jurisdiction over the Premises and or its operations and Licensee does not achieve compliance with the notice of violation or order to the satisfaction of the issuing agency within the time specified by the agency or by the City if the agency does not specify a timeframe, the City may draw upon the Letter of Credit for purposes of ensuring regulatory compliance. In addition, the City may draw upon the Letter of City related to any notice of violation or other regulatory order issued to Licensee. The City may also draw upon the Letter of Credit in order

APPENDIX H: 2650 Bayshore License Agreement

Page 27 of 39

to reimburse the City for costs associated with City's environmental assessments or corrective action, which may be performed at the City's sole discretion.

24.11Environmental Oversight Deposit. Upon execution of the Towing Agreement, Licensee shall provide to the City, and shall maintain and replenish throughout the Term of this License and for a period of at least ninety (90) days after expiration of this License, an "Environmental Oversight Deposit" in the amount of \$10,000, which shall be deposited in an account specified by City. If Licensee receives a notice of violation or other regulatory order from a governmental or regulatory agency with jurisdiction over the site and or its operations and such notice is not cured within fourteen (14) days, the City may draw from this deposit to reimburse the City for staff costs incurred by the City while inspecting site conditions and enforcing and administering the Hazardous Materials provisions of the License. If Licensee receives a notice of violation or other regulatory order from a governmental or regulatory agency with jurisdiction over the site and or its operations, and such notice is cured within fourteen (14) days, the City may draw from this deposit in an amount not to exceed \$500 to reimburse the City for staff costs incurred by the City. City will submit an invoice to Licensee for any such costs, and Licensee will pay such invoiced amounts within thirty (30) days to replenish the Environmental Oversight Deposit. Licensee's failure to pay such costs within thirty (30) days, or to replenish the Environmental Oversight Deposit if drawn upon, will constitute an Event of Default.

24.12Hazardous Substance Disclosure. California law requires landlords to disclose to tenants the presence or potential presence of certain hazardous materials and hazardous substances prior to lease. Accordingly, Licensee is hereby advised that occupation of the Premises may lead to exposure to Hazardous Materials such as, but not limited to, gasoline, diesel and other vehicle fluids, vehicle exhaust, office maintenance fluids, asbestos, PCBs, tobacco smoke, methane and building materials containing chemicals, such as formaldehyde. Further, there were and may be Hazardous Materials located on the Premises, which are described in that certain Phase I Environmental Site Assessment prepared by URS Corporation Americas, dated July 29, 2011, and the memorandum from Robert Begley, of City's Department of Public Works, to Kerstin Magary, of SFMTA, dated May 11, 2012, (collectively, the "Environmental Reports"). Licensee acknowledges that copies of the Environmental Reports have been provided to it.

By execution of this License, Licensee agrees that the Environmental Reports, notices and warnings set forth in this <u>Section 24</u> have been provided pursuant to California Health and Safety Code Sections 25359.7 and related statutes. City agrees to provide additional information that comes into its possession regarding hazardous substances on the Premises upon request of Licensee.

25. GENERAL PROVISIONS

25.1 Notices. Any notice, demand, consent or approval required under this License shall be effective only if in writing and given by delivering the notice in person or by sending it first-class certified mail with a return receipt requested or by overnight courier, return receipt requested, with postage prepaid, to: (a) Licensee (i) at the Premises, or (ii) at any place where Licensee or any Agent of Licensee may be found if sent subsequent to Licensee's vacating, abandoning or surrendering the Premises; or (b) City at City's address set forth in the Basic License Information; or (c) to such other address as either City or Licensee may designate as its new address for such purpose by notice given to the other in accordance with the provisions of this Section at least ten (10) days prior to the effective date of such change. Any notice hereunder shall be deemed to have been given two (2) days after the date when it is mailed if sent by first class or certified mail, one day after the date it is made if sent by overnight courier, or upon the date personal delivery is made. For convenience of the parties, copies of such

APPENDIX H: 2650 Bayshore License Agreement

Page 28 of 39

notices, demands, consents or approvals may also be given by telefacsimile to the telephone number set forth in the Basic License Information or such other number as may be provided from time to time; however, neither party may give official or binding notice by facsimile. All other written communications may be by first class U.S. mail, postage pre-paid, by email or by facsimile addressed with the contact information set forth in the Basic License Information.

25.2 No Implied Waiver. No failure by City to insist upon the strict performance of any obligation of Licensee under this License or to exercise any right, power or remedy arising out of a breach thereof, irrespective of the length of time for which such failure continues, no acceptance of full or partial Base Fee or any other amounts owing under this License during the continuance of any such breach, and no acceptance of the keys to or possession of the Premises prior to the expiration of the Term by any Agent of City, shall constitute a waiver of such breach or of City's right to demand strict compliance with such term, covenant or condition or operate as a surrender of this License. No express written waiver of any default or the performance of any provision hereof shall affect any other default or performance, or cover any other period of time, other than the default, performance or period of time specified in such express waiver. One or more written waivers of a default or the performance. Any consent by City hereunder shall not relieve Licensee of any obligation to secure the consent of City in any other or future instance under the terms of this License.

25.3 Amendments. Neither this License nor any term or provisions hereof may be changed, waived, discharged or terminated, except by a written instrument signed by both parties hereto.

25.4 Authority. Each of the persons executing this License on behalf of Licensee does hereby covenant and warrant that Licensee is a duly authorized and existing entity, that Licensee has and is qualified to do business in California, that Licensee has full right and authority to enter into this License, and that each and all of the persons signing on behalf of Licensee are authorized to do so. Upon City's request, Licensee shall provide City with evidence reasonably satisfactory to City confirming the foregoing representations and warranties.

25.5 Parties and Their Agents; Approvals. The words "City" and "Licensee" as used herein shall include the plural as well as the singular. If there is more than one Licensee, the obligations and liabilities under this License imposed on Licensee shall be joint and several. As used herein, the term "Agents" when used with respect to either party shall include the agents, employees, officers, contractors and representatives of such party, and the term "Invitees" when used with respect to Licensee shall include the clients, customers, invitees, guests, licensees, assignees or sublicensees of Licensee. All approvals, consents or other determinations permitted or required by City hereunder shall be made by or through SFMTA unless otherwise provided in this License, subject to Applicable Law.

25.6 Interpretation of License.

25.6.1 The captions preceding the articles and sections of this License and in the table of contents have been inserted for convenience of reference only and such captions shall in no way define or limit the scope or intent of any provision of this License.

25.6.2 This License has been negotiated at arm's length and between persons sophisticated and knowledgeable in the matters dealt with herein and shall be interpreted to achieve the intents and purposes of the parties, without any presumption against the party responsible for drafting any part of this License.

APPENDIX H: 2650 Bayshore License Agreement

Page 29 of 39

25.6.3 Provisions in this License relating to number of days shall be calendar days, unless otherwise specified, provided that if the last day of any period to give notice, reply to a notice or to undertake any other action occurs on a Saturday, Sunday or a bank or City holiday, then the last day for undertaking the action or giving or replying to the notice shall be the next succeeding business day.

25.6.4 Use of the word "including" or similar words shall not be construed to limit any general term, statement or other matter in this License, whether or not language of non-limitation, such as "without limitation" or similar words, are used.

25.6.5 Any capitalized term used herein shall be interpreted in accordance with the definition set forth in this License. If the capitalized term is not defined in this License Agreement, it shall be interpreted in accordance with the definition set forth in the Towing Agreement or the Lease.

25.6.6 Any inconsistency between this License, the Towing Agreement, and the Lease with respect to Licensee's performance of its obligations under the Towing Agreement shall be resolved by giving precedence in the following order: (a) the Towing Agreement; (b) the Lease; (c) this License.

25.7 Successors and Assigns. Subject to the provisions of this License relating to Assignment and subletting, the terms, covenants and conditions contained in this License shall bind and inure to the benefit of City and Licensee and, except as otherwise provided herein, their personal representatives and successors and assigns; provided, however, that upon any sale, assignment or transfer by City named herein (or by any subsequent Licensor) of its interest in the Premises as owner or lessee, including any transfer by operation of law, City (or any subsequent Licensor) shall be relieved from all subsequent obligations and liabilities arising under this License subsequent to such sale, assignment or transfer.

25.8 Severability. If any provision of this License or the application thereof to any person, entity or circumstance shall, to any extent, be invalid or unenforceable, the remainder of this License, or the application of such provision to persons, entities or circumstances other than those as to which it is invalid or unenforceable, shall not be affected thereby, and each other provision of this License shall be valid and be enforceable to the fullest extent permitted by law.

25.9 Governing Law. This License shall be construed and enforced in accordance with the laws of the State of California.

25.10Entire Agreement. This License, together with all exhibits hereto, which are made a part of this License, and the Towing Agreement, constitute the entire agreement between City and Licensee about the subject matters hereof and may not be modified except by an instrument in writing signed by the party to be charged. In the event of any conflict between the terms of the Towing Agreement and the terms of this License with respect to Licensee's activities and obligations at the Premises, the terms of the License shall control. All prior written or oral negotiations, understandings and agreements are merged herein. The parties further intend that this License shall constitute the complete and exclusive statement of its terms and that no extrinsic evidence whatsoever (including prior drafts hereof and changes therefrom) may be introduced in any judicial, administrative or other legal proceeding involving this License. Licensee hereby acknowledges that neither City nor City's Agents have made any representations or warranties with respect to the Premises or this License except as expressly set forth herein, and no rights, easements or licenses are or shall be acquired by Licensee by implication or otherwise unless expressly set forth herein.

APPENDIX H: 2650 Bayshore License Agreement Page 30 of 39

25.11Attorneys' Fees. In the event that either City or Licensee fails to perform any of its obligations under this License or in the event a dispute arises concerning the meaning or interpretation of any provision of this License, the defaulting party or the party not prevailing in such dispute, as the case may be, shall pay any and all costs and expenses incurred by the other party in enforcing or establishing its rights hereunder (whether or not such action is prosecuted to judgment), including, without limitation, court costs and reasonable attorneys' fees. For purposes of this License, reasonable fees of attorneys of City's Office of the City Attorney shall be based on the fees regularly charged by private attorneys with the equivalent number of years of experience in the subject matter area of the law for which the City Attorney's services were rendered who practice in the City of San Francisco in law firms with approximately the same number of attorneys as employed by the Office of the City Attorney.

25.12Holding Over. Should Licensee hold over without City's consent, such holding over shall not be deemed to extend the Term or renew this License, but such term thereafter shall continue as a month-to-month occupancy. Such occupancy shall be on all the terms and conditions set forth in this License and the Base Fee payable by Licensee during the period of such holding over shall be one hundred fifty percent (150%) of the monthly Base Fee in effect during the last month of the Term of this License.

All other payments shall continue under the terms of this License. In addition, Licensee shall be liable for all damages incurred by City as a result of such holding over. No holding over by Licensee, whether with or without consent of City, shall operate to extend this License except as otherwise expressly provided, and this Section shall not be construed as consent for Licensee to retain possession of the Premises beyond such month to month holdover occupancy. For purposes of this Section, "possession of the Premises" shall continue until, among other things, Licensee has delivered all keys to the Premises to City, has fully vacated the Premises, and completely fulfilled all obligations required of it upon termination of the License as set forth in this License, including, without limitation, those concerning the condition and repair of the Premises.

25.13Time of Essence. Time is of the essence with respect to all provisions of this License in which a definite time for performance is specified.

25.14Cumulative Remedies. All rights and remedies of either party hereto set forth in this License shall be cumulative, except as may otherwise be provided herein.

25.15Provisions of License Surviving Termination. Termination of this License shall not affect the right of either party to enforce any and all indemnities and representations and warranties given or made to the other party under this License. This Section and the following Sections of this License shall survive termination or expiration of this License: Sections 2.1.4, 5, 6.6, 7, 12, 14, 17.2, 17.3, 18, 23, 24.1, 24.2, 24.3, 24.5, 24.8, 24.9, 24.10, 24.11, 25.6, 25.7, 25.8, 25.9, 25.10, 25.11, 25.12, 25.14, 25.22, 25.26, 25.31.

25.16Signs. Licensee agrees that it will not erect or maintain, or permit to be erected or maintained, any signs, notices or graphics upon or about the Premises which are visible in or from public corridors or other portions of any common areas of the Building or from the exterior of the Premises without City's prior written consent, which City may withhold or grant in its sole discretion.

25.17Relationship of the Parties. City is not, and none of the provisions in this License shall be deemed to render City, a partner in Licensee's business, or joint venturer or member in any joint enterprise with Licensee. Neither party shall act as the agent of the other party in any respect hereunder. This License is not intended nor shall it be construed to create any third party beneficiary rights in any third party, unless otherwise expressly provided.

APPENDIX H: 2650 Bayshore License Agreement

Page 31 of 39

25.18Light and Air. Licensee covenants and agrees that no diminution of light, air or view by any structure that may hereafter be erected (whether or not by City) shall entitle Licensee to any reduction of the Base Fee under this License, result in any liability of City to Licensee, or in any other way affect this License or Licensee's obligations hereunder.

25.19No Recording. Licensee shall not record this License or any memorandum hereof in the public records.

25.20Drug-Free Workplace. Licensee acknowledges that pursuant to the Federal Drug-Free Workplace Act of 1989, the unlawful manufacture, distribution, possession or use of a controlled substance is prohibited on City premises. Licensee agrees that any violation of this prohibition by Licensee, its Agents or assigns shall be deemed a material breach of this License.

25.21Public Transit Information. Licensee shall establish and carry on during the Term a program to encourage maximum use of public transportation by personnel of Licensee employed on the Premises, including, without limitation, the distribution to such employees of written materials explaining the convenience and availability of public transportation facilities adjacent or proximate to the Premises and encouraging use of such facilities, all at Licensee's sole expense.

25.22Taxes, Assessments, Licenses, Permit Fee's and Liens. (a) Licensee recognizes and understands that this License may create a possessory interest subject to property taxation and that Licensee may be subject to the payment of property taxes levied on such interest. (b) Licensee agrees to pay taxes of any kind, including possessory interest taxes, that may be lawfully assessed on the License interest hereby created and to pay all other taxes, excises, licenses, permit charges and assessments based on Licensee's usage of the Premises that may be imposed upon Licensee by law, all of which shall be paid when the same become due and payable and before delinquency. (c) Licensee agrees not to allow or suffer a lien for any such taxes to be imposed upon the Premises or upon any equipment or property located thereon without promptly discharging the same, provided that Licensee, if so desiring, may have reasonable opportunity to contest the validity of the same.

25.23Wages and Working Conditions. Licensee agrees that any person performing labor in the construction of any Alterations to the Premises, which Licensee provides under this License, shall be paid not less than the highest prevailing rate of wages as required by Section 6.22(E) of the San Francisco Administrative Code, shall be subject to the same hours and working conditions, and shall receive the same benefits as in each case are provided for similar work performed in San Francisco County. Licensee shall include in any contract for construction of such Alterations a requirement that all persons performing labor under such contract shall be paid not less than the highest prevailing rate of wages for the labor so performed. Licensee shall require any contractor to provide, and shall deliver to City upon request, certified payroll reports with respect to all persons performing labor in the construction of such any Alterations to the Premises.

25.24Non-Discrimination in City Contracts and Benefits Ordinance.

25.24.1 Covenant Not to Discriminate. In the performance of this License, Licensee covenants and agrees not to discriminate on the basis of the fact or perception of a person's race, color, creed, religion, national origin, ancestry, age, sex, sexual orientation, gender identity, domestic partner status, marital status, disability or Acquired Immune Deficiency Syndrome, HIV status (AIDS/HIV status), weight, height, association with members of classes protected under this chapter or in retaliation for opposition to any practices forbidden under this chapter against any employee of, any City employee working with, or applicant for employment with

APPENDIX H: 2650 Bayshore License Agreement

Page 32 of 39

such contractor and shall require such contractor to include a similar provision in all subcontracts executed or amended thereunder.

25.24.2 Subcontracts. Licensee shall include in all assignment, subleases or other subcontracts relating to the Premises a non-discrimination clause applicable to such assignee, sublicensee or other subcontractor in substantially the form of subsection (a) above. In addition, Licensee shall incorporate by reference in all assignments, subleases, and other subcontracts the provisions of Sections 12B.2(a), 12B.2(c)-(k), and 12C.3 of the San Francisco Administrative Code and shall require all assignees, sublicensees and other subcontractors to comply with such provisions. Licensee's failure to comply with the obligations in this subsection shall constitute a material breach of this License.

25.24.3Non-Discrimination in Benefits. Licensee does not as of the date of this License and will not during the Term, in any of its operations in San Francisco, where the work is being performed for the City, or elsewhere within the United States, discriminate in the provision of bereavement leave, family medical leave, health benefits, membership or membership discounts, moving expenses, pension and retirement benefits or travel benefits, as well as any benefits other than the benefits specified above, between employees with domestic partners and employees with spouses, and/or between the domestic partners and spouses of such employees, where the domestic partnership has been registered with a governmental entity pursuant to state or local law authorizing such registration, subject to the conditions set forth in Section 12B.2(b) of the San Francisco Administrative Code.

25.24.4HRC Form. As a condition to this License, Licensee shall execute the "Chapter 12B Declaration: Nondiscrimination in Contracts and Benefits" form (Form HRC-12B-101) with supporting documentation and secure the approval of the form by the San Francisco Human Rights Commission.

25.24.5Incorporation of Administrative Code Provisions by Reference. The provisions of Chapters 12B and 12C of the San Francisco Administrative Code relating to nondiscrimination by parties contracting for the License of City property are incorporated in this <u>Section 25.24</u> by reference and made a part of this License as though fully set forth herein. Licensee shall comply fully with and be bound by all of the provisions that apply to this License under such Chapters of the Administrative Code, including but not limited to the remedies provided in such Chapters. Without limiting the foregoing, Licensee understands that pursuant to Section 12B.2(h) of the San Francisco Administrative Code, a penalty of \$50 for each person for each calendar day during which such person was discriminated against in violation of the provisions of this License may be assessed against Licensee and/or deducted from any payments due Licensee.

25.25Non-Liability of City Officials, Employees and Agents. No elective or appointive board, commission, member, officer, employee or other Agent of City shall be, personally liable to Licensee, its successors and assigns, in the event of any default or breach by City or for any amount which may become due to Licensee, its successors and assigns, or for any obligation of City under this License.

25.26No Relocation Assistance: Waiver of Claims. Licensee acknowledges that it will not be a displaced person at the time this License is terminated or expires by its own terms, and Licensee fully RELEASES, WAIVES AND DISCHARGES forever any and all Claims against, and covenants not to sue, City, its departments, commissions, officers, directors and employees, and all persons acting by, through or under each of them, under any laws, including, without limitation, any and all claims for relocation benefits or assistance from City under federal and state relocation assistance laws (including, but not limited to, California Government Code

APPENDIX H: 2650 Bayshore License Agreement

Page 33 of 39

Section 7260 et seq.), except as otherwise specifically provided in this License with respect to a Taking.

25.27MacBride Principles - Northern Ireland. The City and County of San Francisco urges companies doing business in Northern Ireland to move toward resolving employment inequities and encourages then to abide by the MacBride Principles as expressed in San Francisco Administrative Code Section 12F.1 et seq. The City and County of San Francisco also urges San Francisco companies to do business with corporations that abide by the MacBride Principles. Licensee acknowledges that it has read and understands the above statement of the City and County of San Francisco concerning doing business in Northern Ireland.

25.28Tropical Hardwood and Virgin Redwood Ban. The City and County of San Francisco urges companies not to import, purchase, obtain or use for any purpose, any tropical hardwood, tropical hardwood wood product, virgin redwood or virgin redwood wood product. Except as expressly permitted by the application of Sections 802(b) and 803(b) of the San Francisco Environment Code, Licensee shall not provide any items to the construction of Alterations, or otherwise in the performance of this License which are tropical hardwoods, tropical hardwood wood products, virgin redwood, or virgin redwood wood products. In the event Licensee fails to comply in good faith with any of the provisions of Chapter 8 of the San Francisco Environment Code, Licensee shall be liable for liquidated damages for each violation in any amount equal to Licensee's net profit on the contract, or five percent (5%) of the total amount of the contract dollars, whichever is greater.

25.29Pesticide Prohibition. Licensee shall comply with the provisions of Section 308 of Chapter 3 of the San Francisco Environment Code (the "Pesticide Ordinance") which (i) prohibit the use of certain pesticides on City property, (ii) require the posting of certain notices and the maintenance of certain records regarding pesticide usage and (iii) require Licensee to submit an integrated pest management ("IPM") plan to SFMTA that (a) lists, to the extent reasonably possible, the types and estimated quantities of pesticides that Licensee may need to apply to the Premises during the terms of this License, (b) describes the steps Licensee will take to meet the City's IPM Policy described in Section 300 of the Pesticide Ordinance and (c) identifies, by name, title, address and telephone number, an individual to act as the Licensee's primary IPM contact person with the City. In addition, Licensee shall comply with the requirements of Sections 303(a) and 303(b) of the Pesticide Ordinance.

25.30First Source Hiring Ordinance. The City has adopted a First Source Hiring Ordinance, San Francisco Administrative Code, Chapter 83, which establishes specific requirements, procedures and monitoring for first source hiring of qualified economically disadvantaged individuals for entry level positions. Upon request when applicable, Licensee shall enter into a First Source Hiring Agreement that meets the requirements of Section 83.9 of the First Source Hiring Ordinance.

25.31Sunshine Ordinance. In accordance with Section 67.24(e) of the San Francisco Administrative Code, contracts, Licensees' bids, leases, agreements, responses to Requests for Proposals, and all other records of communications between City and persons or firms seeking contracts will be open to inspection immediately after a contract has been awarded. Nothing in this provision requires the disclosure of a private person's or organization's net worth or other proprietary financial data submitted for qualification for a contract, lease, agreement or other benefit until and unless that person or organization is awarded the contract, lease, agreement or benefit. Information provided which is covered by this Section will be made available to the public upon request.

25.32Conflicts of Interest. Through its execution of this License, Licensee acknowledges that it is familiar with the provisions of Section 15.103 of the San Francisco Charter, Article III,

APPENDIX H: 2650 Bayshore License Agreement

Page 34 of 39

Chapter 2 of City's Campaign and Governmental Conduct Code, and Section 87100 et seq. and Section 1090 et seq. of the Government Code of the State of California, and certifies that it does not know of any facts which would constitute a violation of said provision, and agrees that if Licensee becomes aware of any such fact during the Term, Licensee shall immediately notify the City.

25.33Charter Provisions. This License is governed by and subject to the provisions of the Charter of the City and County of San Francisco.

25.34Prohibition of Cigarette or Tobacco Advertising. Licensee acknowledges and agrees that no advertising of cigarettes or tobacco products is allowed on the Premises. This advertising prohibition includes the placement of the name of a company producing, selling or distributing cigarettes or tobacco products or the name of any cigarette or tobacco product in any promotion of any event or product. This advertising prohibition does not apply to any advertisement sponsored by a state, local, nonprofit or other entity designed to (i) communicate the health hazards of cigarettes and tobacco products, or (ii) encourage people not to smoke or to stop smoking.

25.35Prohibition of Alcoholic Beverage Advertising. Licensee acknowledges and agrees that no advertising of alcoholic beverages is allowed on the Premises. For purposes of this section, "alcoholic beverage" shall be defined as set forth in California Business and Professions Code Section 23004, and shall not include cleaning solutions, medical supplies and other products and substances not intended for drinking. This advertising prohibition includes the placement of the name of a company producing, selling or distributing alcoholic beverages or the name of any alcoholic beverage in any promotion of any event or product. This advertising prohibition does not apply to any advertisement sponsored by a state, local, nonprofit or other entity designed to (i) communicate the health hazards of alcoholic beverages, (ii) encourage people not to drink alcohol or to stop drinking alcohol, or (iii) provide or publicize drug or alcohol treatment or rehabilitation services

25.36Requiring Health Benefits for Covered Employees. Unless exempt, Licensee agrees to comply fully with and be bound by all of the provisions of the Health Care Accountability Ordinance (HCAO), as set forth in San Francisco Administrative Code Chapter 12Q, including the remedies provided, and implementing regulations, as the same may be amended from time to time. The provisions of Chapter 12Q are incorporated herein by reference and made a part of this License as though fully set forth. The text of the HCAO is available on the web at <u>http://www.sfgov.org/olse/hcao</u>. Capitalized terms used in this Section and not defined in this License shall have the meanings assigned to such terms in Chapter 12Q.

25.36.1For each Covered Employee, Licensee shall provide the appropriate health benefit set forth in Section 12Q.3 of the HCAO. If Licensee chooses to offer the health plan option, such health plan shall meet the minimum standards set forth by the San Francisco Health Commission.

25.36.2Notwithstanding the above, if the Licensee is a small business as defined in Section 12Q.3(d) of the HCAO, it shall have no obligation to comply with <u>Subsection 25.36.1</u> above.

25.36.3Licensee's failure to comply with the HCAO shall constitute a material breach of this License. City shall notify Licensee if such a breach has occurred. If, within thirty (30) days after receiving City's written notice of a breach of this License for violating the HCAO, Licensee fails to cure such breach or, if such breach cannot reasonably be cured within such period of thirty (30) days, Licensee fails to commence efforts to cure within such period, or thereafter fails diligently to pursue such cure to completion, City shall have the right to pursue

APPENDIX H: 2650 Bayshore License Agreement

Page 35 of 39

the remedies set forth in Section 12Q.5(f)(1-5). Each of these remedies shall be exercisable individually or in combination with any other rights or remedies available to City.

经财产的现在分词 机可能的 网络拉拉

25.36.4Any Subcontract entered into by Licensee shall require the Subcontractor to comply with the requirements of the HCAO and shall contain contractual obligations substantially the same as those set forth in this Section. Licensee shall notify City when it enters into such a Subcontract and shall certify to the City that it has notified the Subcontractor of the obligations under the HCAO and has imposed the requirements of the HCAO on Subcontractor through the Subcontract. Each Licensee shall be responsible for its Subcontractors' compliance with this Chapter. If a Subcontractor fails to comply, the City may pursue the remedies set forth in this Section against Licensee based on the Subcontractor's failure to comply, provided that City has first provided Licensee with notice and an opportunity to obtain a cure of the violation.

25.36.5Licensee shall not discharge, reduce in compensation, or otherwise discriminate against any employee for notifying City with regard to Licensee's compliance or anticipated compliance with the requirements of the HCAO, for opposing any practice proscribed by the HCAO, for participating in proceedings related to the HCAO, or for seeking to assert or enforce any rights under the HCAO by any lawful means.

25.36.6Licensee represents and warrants that it is not an entity that was set up, or is being used, for the purpose of evading the intent of the HCAO.

25.36.7Licensee shall keep itself informed of the current requirements of the HCAO.

25.36.8Licensee shall provide reports to the City in accordance with any reporting standards promulgated by the City under the HCAO, including reports on Subcontractors and Sublicensees, as applicable.

25.36.9Licensee shall provide City with access to records pertaining to compliance with HCAO after receiving a written request from City to do so and being provided at least five (5) business days to respond.

25.36.10 City may conduct random audits of Licensee to ascertain its compliance with HCAO. Licensee agrees to cooperate with City when it conducts such audits.

25.36.11 If Licensee is exempt from the HCAO when this License is executed because its amount is less than Twenty-Five Thousand Dollars (\$25,000), but Licensee later enters into an agreement or agreements that cause Licensee's aggregate amount of all agreements with City to reach Seventy-Five Thousand Dollars (\$75,000), all the agreements shall be thereafter subject to the HCAO. This obligation arises on the effective date of the agreement that causes the cumulative amount of agreements between Licensee and City to be equal to or greater than Seventy-Five Thousand Dollars (\$75,000) in the fiscal year.

25.37Notification of Limitations on Contributions. Through its execution of this License, Licensee acknowledges that it is familiar with Section 1.126 of the San Francisco Campaign and Governmental Conduct Code, which prohibits any person who contracts with the City for the selling or leasing of any land or building to or from the City, whenever such transaction would require approval by a City elective officer, the board on which that City elective officer serves, or a board on which an appointee of that individual serves, from making any campaign contribution to (a) the City elective officer, (b) a candidate for the office held by such individual, or (c) a committee controlled by such individual or candidate, at any time from the commencement of negotiations for the contract until the later of either the termination of negotiations for such contract or six (6) months after the date the contract is approved. Licensee acknowledges that the foregoing restriction applies only if the contract or a combination or series

APPENDIX H: 2650 Bayshore License Agreement

Page 36 of 39

of contracts approved by the same individual or board in a fiscal year have a total anticipated or actual value of \$50,000 or more. Licensee further acknowledges that the prohibition on contributions applies to each Licensee; each member of Licensee's board of directors, and Licensee's chairperson, chief executive officer, chief financial officer and chief operating officer; any person with an ownership interest of more than twenty percent (20%) in Licensee; any subcontractor listed in the contract; and any committee that is sponsored or controlled by Licensee. Additionally, Licensee acknowledges that Licensee must inform each of the persons described in the preceding sentence of the prohibitions contained in Section 1.126. Licensee further agrees to provide to City the name of each person, entity or committee described above.

25.38Preservative-Treated Wood Containing Arsenic. Licensee may not purchase preservative-treated wood products containing arsenic in the performance of this License unless an exemption from the requirements of Environment Code Chapter 13 is obtained from the Department of Environment under Section 1304 of the Environment Code. The term "preservative-treated wood containing arsenic" shall mean wood treated with a preservative that contains arsenic, elemental arsenic, or an arsenic copper combination, including, but not limited to, chromated copper arsenate preservative. Licensee may purchase preservative-treated wood products on the list of environmentally preferable alternatives prepared and adopted by the Department of Environment. This provision does not preclude Licensee from purchasing preservative-treated wood containing arsenic for saltwater immersion. The term "saltwater immersion" shall mean a pressure-treated wood that is used for construction purposes or facilities that are partially or totally immersed in saltwater.

25.39Resource Efficient City Buildings and Pilot Projects. Licensee acknowledges that the City and County of San Francisco has enacted San Francisco Environment Code Sections 700 to 713 relating to the environmental design and construction of City buildings. Licensee hereby agrees that it shall comply with all applicable provisions of such code sections.

25.40Food Service Waste Reduction. Licensee agrees to comply fully with and be bound by all of the provisions of the Food Service Waste Reduction Ordinance, as set forth in the San Francisco Environment Code, Chapter 16, including the remedies provided therein, and implementing guidelines and rules. The provisions of Chapter 16 are incorporated herein by reference and made a part of this License as though fully set forth herein. This provision is a material term of this License. By entering into this License, Licensee agrees that if it breaches this provision, City will suffer actual damages that will be impractical or extremely difficult to determine. Without limiting City's other rights and remedies, Licensee agrees that the sum of One Hundred Dollars (\$100.00) liquidated damages for the first breach, Two Hundred Dollars (\$200.00) liquidated damages for the second breach in the same year, and Five Hundred Dollars (\$500.00) liquidated damages for subsequent breaches in the same year is a reasonable estimate of the damage that City will incur based on the violation, established in light of the circumstances existing at the time of the Commencement Date. Such amounts shall not be considered a penalty, but rather agreed monetary damages sustained by City because of Licensee's failure to comply with this provision.

25.41References. No reference to this License Agreement is necessary in any instrument or document at any time referring to the Agreement. Any future reference to the Agreement shall be deemed a reference to such document as amended hereby.

25.42Counterparts. This License may be executed in two or more counterparts, each of which shall be deemed an original, but all of which taken together shall constitute one and the same instrument.

25.43Effective Date. This License shall be effective, and the Original License Agreement shall be superseded and replaced in its entirety by this License, as of the Commencement Date.

25.44Cooperative Drafting. This License has been drafted through a cooperative effort of both parties, and both parties have had an opportunity to have the License reviewed and revised by legal counsel. No party shall be considered the drafter of this License, and no presumption or rule that an ambiguity shall be construed against the party drafting the clause shall apply to the interpretation or enforcement of this License.

NOTWITHSTANDING ANYTHING TO THE CONTRARY CONTAINED IN THIS LICENSE, LICENSEE ACKNOWLEDGES AND AGREES THAT NO OFFICER OR EMPLOYEE OF CITY HAS AUTHORITY TO COMMIT CITY TO THIS LICENSE UNLESS AND UNTIL THE BOARD OF DIRECTORS OF CITY'S MUNICIPAL TRANSPORTATION AGENCY SHALL HAVE DULY ADOPTED A RESOLUTION APPROVING THE TOWING AMENDMENT AND AUTHORIZING THE TRANSACTIONS CONTEMPLATED IN THIS LICENSE. THEREFORE, ANY OBLIGATIONS OR LIABILITIES OF CITY HEREUNDER ARE CONTINGENT UPON ADOPTION OF SUCH A RESOLUTION, AND THIS LICENSE SHALL BE NULL AND VOID IF THE BOARD OF DIRECTORS OF CITY'S MUNICIPAL TRANSPORTATION AGENCY DO NOT APPROVE THE TOWING AMENDMENT AND THIS LICENSE BY ITS RESPECTIVE SOLE DISCRETION. APPROVAL OF THIS LICENSE BY ANY DEPARTMENT, COMMISSION OR AGENCY OF CITY SHALL NOT BE DEEMED TO IMPLY THAT SUCH RESOLUTION WILL BE ENACTED, NOR WILL ANY SUCH APPROVAL CREATE ANY BINDING OBLIGATIONS ON CITY.

[REMAINDER OF PAGE INTENTIONALLY BLANK]

IN WITNESS WHEREOF, the parties hereto have executed this License on the day first mentioned above.

CITY:

CITY AND COUNTY OF SAN FRANCISCO, a municipal corporation, acting by and through its Municipal Transportation Agency

Bv Edward D. Reiskin

Director of Transportation

San Francisco Municipal Transportation Agency Board of Directors Resolution No: 14-050 Adopted: 412014 Attest: 2014 Secretary, SFMTA Board of Directors

Approved as to Form:

Dennis J. Herrera, City Attorney

By: Carol Wong, Deputy City Attorney

LICENSEE:

TEGSCO, LLC, a California limited liability company d.b.a. San Francisco AutoReturn

By

John Wicker, President and CEO

APPENDIX H: 2650 Bayshore Property License

FIRST AMENDMENT TO MOU

Port Reference MOU M-13828

This First Amendment to Memorandum of Understanding Port Reference M-13828 ("First Amendment") is entered into by and between the Municipal Transportation Agency, an agency of the City and County of San Francisco ("MTA") and the San Francisco Port Commission, an agency of the City and County of San Francisco ("Port").

RECITALS

A. Effective on July 30, 2005, Port and MTA entered into a lease of certain real property located at Pier 70 in the City and County of San Francisco, State of California as more particularly described in the "Original MOU". With Port's consent, as required by the Original MOU, MTA simultaneously entered into an agreement with Tegsco, LLC, dba San Francisco AutoReturn ("AutoReturn") to conduct automobile towing and storage operations for the City, which included a license to use the Port property for storage and other required services. MTA's agreement with AutoReturn expires on July 30, 2010. The Original MOU had a five year term expiring on July 30, 2010 and an extension option through March 1, 2012. Through correspondence (including a letter dated February 17, 2009 from Port to MTA), Port and MTA agreed to extend the Original MOU until March 1, 2012.

B. The Original MOU contemplated that it would be concurrent with the initial agreement with AutoReturn. MTA has determined that it will renew or extend its contract with AutoReturn ("Licensee") until July 31, 2015 without a competitive bid, subject to approval by the Board of Supervisors. Accordingly, the parties have agreed to extend the Original MOU to cover such term and possibly additional term with AutoReturn or another MTA contractor, subject to Port's consent to any license or agreement to allow the use of the premises at Pier 70.

C. On May 11, 2010, the Port Commission adopted Resolution 10-27 by which it (1) endorsed the vision, goals, objectives, and design criteria for the Preferred Master Plan for Pier 70; and (2) authorized Port staff to prepare and issue a competitive solicitation for a private development partner for the waterfront site ("Waterfront Site") described in the accompanying Port Commission staff report which encompasses the premises contemplated by this First Amendment. Port staff expects the following timeline for the competitive solicitation authorized by the Port Commission and for subsequent development activities: issue Waterfront Site competitive solicitation in June 2010; Port Commission selection of Waterfront Site developer in late 2010; initiate environmental review in 2011; commence infrastructure/remediation work requiring site access in late 2012; commence construction on Waterfront Site in 2013 or later; and occupy Waterfront Site in 2015 or later.

D. Port and MTA are negotiating the First Amendment for their mutual benefit and to benefit the citizens of San Francisco and the State of California, and this First Amendment speaks to this goal while assisting each agency with their respective needs, including a steady income stream at fair market value rent, space that is convenient and appropriate in a San Francisco location for tow customers and contractors and continuity of operations.

E. Port and MTA now desire to amend the Original MOU to: (i) permit the Port to reconfigure the Premises with one hundred-eighty (180) days notice from Port to MTA, with MTA's approval, and partial rent credits for some relocation costs; (ii) add an additional five (5) year term from the original expiration date and to allow a year to year holdover tenancy at an increased monthly base rate of 110% in the first year, and increased by 110% in the eighteenth month, with an annual CPI increase in each year of the term and annually during any holdover period of longer than 12 months; (iii) confirm that all of the requirements of the Original MOU apply to the renewed or extended agreement and license with AutoReturn and any other MTA contractor providing the services described; (iv) provide for partial or complete termination by either party with twelve (12) months notice; (v) allow Port to access up to 15% of the Premises

Port/MTA MOU M-13828 1st Amendment 5/17/11

N:\PORT\A\$2010\9690151\00637004.DOC

•

to facilitate development of Pier 70, with ninety (90) days notice from Port to MTA, with rent credits or third party reimbursement for specified relocation costs; (vi) require MTA to conduct a relocation study; (vii) add a requirement that MTA comply with Port's Southern Waterfront Beautification Policy, which was adopted subsequent to the effectiveness of the Original MOU, through specified actions and to authorize rent credits for a portion of same; and (viii) make other changes consistent with the above.

F. The Original MOU and this First Amendment shall collectively be referred to as the "MOU." All capitalized terms used herein but not otherwise defined shall have the meaning given to them in the Original MOU.

NOW THEREFORE, in consideration of the mutual covenants herein contained and for other good and valuable consideration, the Port and MTA hereby amend the Original MOU as follows:

AGREÈMENT

1. Recitals. The foregoing recitals are true and correct and are incorporated herein by this reference as if fully set forth herein.

2. Pursuant to Section 16 of the Original MOU, the parties agree that MTA is a successor of DPT and that MTA will fulfill all of the obligations and responsibilities and have all of the rights of DPT as set forth in the Original MOU. All references in the MOU to "DPT" shall now be to "MTA".

3. The parties agree that all of the provisions of the Original MOU remain in full force and effect with respect to the new, renewed or extended MTA license with AutoReturn and/or other future licenses or similar agreements to use the Premises or portions thereof with other future providers of tow services under contract with MTA. This includes without limitation, securing and maintaining a current BCDC permit (Sections 3(b) and 14(i)); compliance with the permitted uses and property use conditions, including Port approval of an Operations Plan (Sections 7 and 14); compliance with the terms of the MOU (Section 12); and Port's consent to any MTA license or similar agreement that allows use of the Premises (Section 12).

4. On the Effective Date of this First Amendment, Paragraph 3 of the Original MOU shall be amended to add a new subparagraph (f) as follows:

3. Premises/Condition.

<u>"f.</u> Reconfiguration of Premises. Upon one hundred eighty (180) days notice from Port to MTA, and subject to MTA approval, Port may reconfigure the Premises by altering the boundaries of Parcel A and/or Parcel B, such that the new Parcel A and/or Parcel B contain the same approximate square footage shown and outlined on Exhibit A of this MOU. MTA shall be solely responsible for relocating vehicles and its other operations to accommodate such a reconfiguration. MTA shall be entitled to rent credits for half of the costs associated with relocating the fences, gates, lights, driveways, and other improvements pursuant to Section 5(e). Port may not require MTA to incur relocation costs that are eligible for rent credits under this section if One Hundred Fifty percent (150%) of the total value of all rent credits claimed under this MOU exceeds the total rent due for the Term. Port shall not be liable in any manner, and MTA and its licensee hereby waive any claims, for any inconvenience, disturbance, loss of business, nuisance or other damage arising out of Port's or its designees' entry onto the Premises under this Paragraph, except damage resulting directly and exclusively from the gross negligence or willful misconduct of Port or its designees and not contributed to by the acts, omissions or negligence of MTA or its licensees, Contactors or Invitees.

In order for rent credits to be authorized by Port for relocation costs under this section, MTA or its licensee must first obtain written approval from Port that the proposed costs are reasonable and MTA or its licensee must obtain all required governmental approvals including, but not limited to Port building permits for the work. After the completion of the work, as

Port/MTA MOU M-13828

N:\PORT\AS2010\9690151\00637004.DOC

evidenced by a certificate of completion or its equivalent by the Port's Chief Harbor Engineer, MTA must deliver to Port an itemized statement of the actual costs expended, accompanied by documentation substantiating all said expenditures. Such documentation of expenditures shall include: (i) copies of executed contracts; (ii) copies of invoices for labor, services and/or materials, copies of bills of lading, and/or copies of other bills or receipts for goods, materials and/or services; (iii) copies of canceled checks, and (iv) such other proofs of expenditure as may be reasonably requested by Port. Such appropriate proofs of expenditure may include copies of canceled checks; copies of contracts or invoices for labor, services and/or materials marked "Paid", or otherwise evidenced as having been paid; bills of lading marked "Paid"; other bills, contracts, receipts for goods materials and/or services marked "Paid"; and such other proofs of expenditure as may be reasonably approved by Port. All such proofs of expenditure must be directly attributable to the approved project."

5. On the Effective Date of this First Amendment, Paragraph 4 of the Original MOU shall be deleted and replaced with the following:

"4. <u>Term, Early Termination; Holdover</u>.

Date").

(a) <u>Term</u>. The Term of this MOU shall expire on July 31, 2015 ("Expiration

(b) <u>Holdover</u>. Any holding over after the Expiration Date ("Holdover Period") shall not constitute a renewal of this MOU, but be deemed a holdover tenancy upon the terms, conditions, and covenants of this MOU, except as provided in Section 5(c). Either party may cancel the holdover tenancy upon twelve (12) months written notice to the other party. "Term" shall refer to the total time period during which this MOU is effective, including any holdover period.

(c) <u>Early Termination</u>. Either the Port Executive Director or the MTA Executive Director or their designees shall have the right to terminate the MOU as to all or a portion of the Premises at any time for any cause or without cause during the Term (including any holdover period) upon providing twelve (12) months written notice to the other party specifying the portion(s) of the Premises affected. Concurrent with the effective date either party's early termination of all or a portion of the Premises, rent will be reduced in proportion to the amount of square footage removed from the Premises and MTA shall amend or terminate the license with AutoReturn or any current licensee accordingly. In the event that Port seeks a partial termination under this paragraph, MTA shall be solely responsible for all costs associated with such modifications or reconfiguration that MTA in its sole discretion deems necessary, including all costs incurred by MTA or its licensee to relocate the operations, Premises, fences, gates, lights, driveways, and other improvements. "

6. On the Effective Date of this First Amendment, Paragraph 5(b) of the Original MOU shall be deleted and replaced with the following:

"b. <u>Rent Adjustment</u>. Commencing on the Effective Date of this MOU and on each anniversary date (the "Anniversary Date") thereafter, including any Anniversary Date during any holdover period, the Rent shall be adjusted on the first day of the month that immediately follows the Effective Date and on that same date in each succeeding year in direct proportion to the percentage increase in the SF Bay Area CPI for the month immediately preceding the applicable Anniversary Date ("Current Index") over the SF Bay Area CPI index for the month of June 2004 ("Base Index"). In no case shall the Rent, as adjusted, be less than the Rent in effect immediately prior to the Anniversary Date. If the Current Index has increased over the Base Index, the Adjusted Rent shall be determined by multiplying the Rent by a fraction, the

Port/MTA MOU M-13828 1st Amendment 5/17/11 N:\PORT\AS2010\9690151\00637004.DOC

numerator of which is the Current Index and the denominator of which is the Base Index, as follows:

Rent

Current Index

Base Index

Adjusted Rent"

7. On the Effective Date of this First Amendment, Paragraph 5(c) of the Original MOU shall be deleted and replaced with the following:

"c. Holdover Period Rent. If neither the Port nor MTA provides notice of Early Termination by July 31, 2014 and MTA holds over, monthly Rent shall increase as provided in Section 5(b) on the 5th Anniversary Date, with the Adjusted Rent so derived multiplied by one hundred ten percent (110%) to determine the monthly Rent for the first twelve (12) months of the Holdover Period. Rent will be adjusted as provided in Section 5(b) effective on the thirteenth (13th) month of the Holdover Period. If MTA holds over more than eighteen (18) months, monthly Rent shall increased to one hundred ten percent (110%) of the monthly rent in the seventeenth (17th) month of the Holdover Period and shall be subject to further increases every 12 months as provided in Section 5(b) on each Anniversary Date."

8. On the Effective Date of this First Amendment, Paragraph 5(d)(1) of the Original MOU shall be deleted and replaced with the following:

"1. <u>Rent Abatement and Credits</u>. If the Premises cease to be used for towing operations at any time due to damage sustained during the Term by fire, earthquake, or other casualty rendering the Premises unsuitable for occupancy, as determined by the Port's Chief Harbor Engineer pursuant to the Port Building Code, or are otherwise deemed legally not useable in either case for reasons not attributable to MTA's or its licensee's acts or omissions, Rent hereunder shall be abated and MTA shall have the option to terminate the MOU and shall be entitled to a prorated refund of any Rent or deposits paid. In the event the Premises cease to be used for more than two (2) consecutive months for towing operations, Port, at its option, may terminate this MOU."

9. On the Effective Date of this First Amendment, Paragraph 5(d)(2) of the Original MOU shall be deleted and replaced with the following:

"2. MTA shall be entitled to a proportional abatement in the Rent if the exercise of Port's rights under section 13(b) of this MOU results in the loss of use of the Premises or any portion thereof for a period in excess of thirty (30) days or in an area in excess of 250 square feet, or if MTA surrenders the possession of Parcel B to the Port for the sole reason that it is unable to obtain a BCDC permit for use of Parcel B consistent with this MOU. The opening of 22nd Street by the Port for non-exclusive, general circulation through the Pier 70 area may occur at the Port's sole option without Rent abatement. The planned alignment for the 22nd Street Extension is as shown on Department of Parking and Traffic, Division of Traffic Engineering drawing on file with the Port's Chief Harbor Engineer, entitled "New Pier 70 Roadway – Preliminary Striping Plan", dated June 28, 2001, File Name swl349sv newroad final.dwg. The parties recognize that the final alignment may deviate as necessary to effect the right of way improvement.

10. On the Effective Date of this First Amendment, Paragraph 5(d)(3) shall be deleted and replaced with the following:

"3. The parties agree that costs incurred to modify the original configuration of the Premises due to a surrender of possession of Parcel B to the Port due to MTA's inability to obtain a BCDC permit for use of Parcel B consistent with this MOU may not be offset from the Rent."

Port/MTA MOU M-13828 1st Amendment 5/17/11 N:\PORT\AS2010\9690151\00637004.DOC
11. On the Effective Date of this First Amendment, Paragraph 5(d)(5) shall be added to the MOU to read as follows:

115. Upon ninety (90) days prior written notice to MTA, Port may access up to fifteen percent (15%) of the Premises for purposes related to the development of Pier 70. MTA will cooperate to ensure that Port or its licensees, Contactors or Invitees have adequate access to the designated area(s) and shall be solely responsible for costs incurred by MTA or its licensee to relocate vehicles or its other operations to accommodate Port's access. MTA shall be entitled to rent credits pursuant to Section 5(e) or third-party reimbursement arranged by Port for all costs incurred by MTA or its licensee to relocate fences, gates, lights, driveways, and other improvements. Port may not require MTA to incur relocation costs that are eligible for rent credits under this section if One Hundred Fifty percent (150%) of the total value of all rent credits claimed under this MOU exceeds the total rent due for the Term. Notwithstanding the time and square footage limitations of Paragraph 5(d)(2), if the rights exercised by Port hereunder result in the loss of use of the designated area(s) of the Premises, MTA shall be entitled to a proportional abatement in Rent. Port shall not be liable in any manner, and MTA and its licensee hereby waive any claims, for any inconvenience, disturbance, loss of business, nuisance or other damage arising out of Port's or its designees' entry onto the Premises under this Paragraph, except damage resulting directly and exclusively from the gross negligence or willful misconduct of Port or its designees and not contributed to by the acts, omissions or negligence of MTA or its licensees, Contactors or Invitees.

In order for rent credits or third party reimbursement to be authorized by Port for relocation costs under this section, MTA or its licensee must first obtain written approval from Port that the proposed costs are reasonable and MTA or its licensee must obtain all required governmental approvals including, but not limited to Port building permits for the work. After the completion of the work, as evidenced by a certificate of completion or its equivalent by the Port's Chief Harbor Engineer, MTA must deliver to Port an itemized statement of the actual costs expended, accompanied by documentation substantiating all said expenditures. Such documentation of expenditures shall include: (i) copies of executed contracts; (ii) copies of invoices for labor, services and/or materials, copies of bills of lading, and/or copies of other bills or receipts for goods, materials and/or services; (iii) copies of canceled checks, and (iv) such other proofs of expenditure as may be reasonably requested by Port. Such appropriate proofs of expenditure may include copies of canceled checks; copies of contracts or invoices for labor, services and/or materials marked "Paid" or otherwise evidenced as having been paid; bills of lading marked "Paid"; other bills, contracts, receipts for goods materials and/or services marked "Paid"; and such other proofs of expenditure as may be reasonably approved by Port. All such proofs of expenditure must be directly attributable to the approved project."

12. On the Effective Date of this First Amendment, Paragraph 5(e) of the Original MOU shall be deleted and replaced with the following:

"e. Formula for Abatement/Credit. All rent credits available to MTA permitted by Section 3(f), Section 5(d)(4) and Section 5(d)(5) shall be applied against Rent payment obligation during the Term at a rate not greater than one half ($\frac{1}{2}$) of the applicable month Rent payment and shall be applied if and only if MTA is in good standing and is not in default of any of the terms of this MOU. In the event that the total of rent credits available to MTA pursuant to Section 3(f), Section 5(d)(4) and Section 5(d)(5) of this MOU exceeds an amount equal to one half ($\frac{1}{2}$) of the Rent payment for any one calendar month, the remaining available Rent Credit shall be carried forward to successive calendar months at a rate not to exceed one half ($\frac{1}{2}$) of the applicable Rent payment, until all available rent credits have been fully applied. In no event, however, shall MTA be entitled to the application of any rent credits or the value thereof, beyond the expiration or earlier termination of this MOU."

Port/MTA MOU M-13828 1st Amendment 5/17/11

13. Section 15, Notices, shall be revised by replacing MTA's address with "And if to MTA to:

San Francisco Municipal Transportation Agency Attention: Steve Lee One South Van Ness Avenue, 7th Floor San Francisco, California 94103"

14. <u>Presence of Hazardous Materials</u>. California Law requires landlords to disclose to tenants the presence or potential presence of certain Hazardous Materials. Accordingly, MTA is hereby advised that the reports listed in *Schedule 1*, copies of which have been made available to MTA describe known or suspected Hazardous Materials (as defined in the License) on or near the Premises. MTA acknowledges that the notice set forth in this section satisfies the requirements of California Health and Safety Code Section 25359.7 and related Laws. MTA must disclose the information contained in this Section to any subtenant, licensee, transferee, or assignee of MTA's interest in the Premises. MTA also acknowledges its own obligations pursuant to California Health and Safety Code Section 25359.7 as well as the penalties that apply for failure to meet such obligations.

15. <u>Southern Waterfront Benefits.</u> The Port's "Policy for Southern Waterfront Community Benefits and Beautification" identifies beautification and related projects in the Southern Waterfront (from Mariposa Street in the north to India Basin) that require funding. Under this policy, MTA shall provide the following community benefits and beautification measures in consideration for the use of the Premises. All improvements must be performed in accordance with Paragraph 14(f) of this MOU.

(a) Not sooner than Eighteen (18) months after the Effective Date of this First Amendment and not later than Twenty Four (24) months after the Effective Date of this First Amendment unless otherwise approved by Port, MTA or its licensee shall expend not less than Fifty Thousand Dollars (\$50,000) to design, produce and install signs and other interpretive devices describing the historic significance of the Port's Pier 70 site. The form, content and placement of such signs and other devices are subject to Port's consent in its sole discretion.

(b) MTA shall require its licensee to perform repairs as specified by Port to Building 12 Complex with costs not to exceed Four Hundred Fifty Thousand dollars (\$450,000,00). Port shall not require repairs to be conducted at the same time that it implements an extension of 22^{nd} Street under Paragraph 5(d)(4). In consideration for completion of the repairs and expenditures related thereto, and upon approval by Port of Construction Costs, Port shall issue an appropriate rent credit (herein "Building 12 Complex Repair Rent Credit") to MTA in accordance with the terms of this Paragraph. "Construction Costs" are the actual costs incurred for labor, materials, contractor fees, and reasonable architecture and engineering fees in connection with the project pursuant to a bid obtained under a guaranteed not-to-exceed construction bid that is approved by the Port prior to issuance. The Building 12 Complex Repair Rent Credit shall be a sum equal to the Construction Costs or a sum not to exceed Four Hundred Fifty Thousand dollars (\$450,000.00), whichever is the lesser amount. The timing, scope and specification of the repairs shall be in the Port's sole discretion except that Port may not require Building 12 Complex repairs if the remaining rent owed by MTA for the Term minus rent credits owed to MTA pursuant to Section 3(f) and Section 5(d)(5) is less than One Hundred Fifty percent (150%) of the guaranteed not-to-exceed construction bid.

All rent credits available to MTA under this Paragraph shall be applied against Rent during the Term at a rate of 100% of the applicable month Rent payment and shall be applied if and only if MTA is in good standing and is not in default of any of the terms of this MOU. In the event that the total of rent credits available to MTA pursuant to this Paragraph exceeds an

Port/MTA MOU M-13828 1st Amendment 5/17/11

amount equal to 100% of the Rent payment for any one calendar month, the remaining available Rent Credit shall be carried forward to successive calendar months at a rate not to exceed 100% of the applicable Rent payment, until all available rent credits have been fully applied. In no event, however, shall MTA be entitled to the application of any rent credits or the value thereof, beyond the expiration or earlier termination of this MOU.

In order for construction of the project to be authorized by Port: MTA must first obtain. prior to commencing the project, written approval from Port that the Construction Costs of the proposed project are reasonable; and MTA, or its licensee, must obtain all required governmental approvals, including, but not limited to building permits from the Port. After the completion of the approved project as evidenced by a certificate of completion or its equivalent by the Port's Chief Harbor Engineer, MTA must deliver to Port an itemized statement of the actual Construction Costs expended on the approved project, accompanied by documentation substantiating all said expenditures. Such documentation of expenditures shall include: (i) copies of executed contracts; (ii) copies of invoices for labor, services and/or materials, copies of bills of lading, and/or copies of other bills or receipts for goods, materials and/or services; (iii) copies of canceled checks, and (iv) such other proofs of expenditure as may be reasonably requested by Port. Such appropriate proofs of expenditure may include copies of canceled checks; copies of contracts or invoices for labor, services and/or materials marked "Paid", or otherwise evidenced as having been paid; bills of lading marked "Paid"; other bills, contracts, receipts for goods materials and/or services marked "Paid"; and such other proofs of expenditure as may be reasonably approved by Port. All such proofs of expenditure must be directly attributable to the approved project."

16. <u>Relocation Study</u>. Within six (6) months of a written request by the Port, MTA will examine options to relocate AutoReturn or its current licensee and deliver a relocation study report to the Port in a form and manner jointly agreed to by Port and MTA.

17. <u>Contract Monitor</u>. The Contract Monitor, as defined in Section 8.8 of Appendix A, Scope of Work, of the Towing Agreement shall cooperate with and provide to Port such information in a form and frequency as Port reasonably requests to aid Port's determination of compliance with this MOU.

18. <u>Operations Plan</u>. Within One Hundred Twenty Days after the Effective Date, MTA shall require AutoReturn to update its Operations Plan and submit it to Port for approval by Port's Executive Director or her designee. The current Port-approved Operations Plan will continue to apply until the revised Operations Plan is approved by Port.

19. <u>Entire Agreement.</u> This First Amendment contains all of the representations and the entire agreement between the parties with respect to the subject matter of this agreement.

20. <u>Miscellaneous</u>. This First Amendment shall bind, and shall inure to the benefit of, the successors and assigns of the parties hereto. This First Amendment is made for the purpose of setting forth certain rights and obligations of MTA and the Port, and no other person shall have any rights hereunder or by reason hereof as a third party beneficiary or otherwise. As amended hereby, the MOU is hereby ratified and confirmed in all respects and shall remain in full force and effect. In the event of any inconsistencies between the terms of this First Amendment and the MOU, the terms of this Amendment shall prevail. Neither this First Amendment nor any of the terms hereof may be amended or modified except by a written instrument signed by all the parties hereto.

21. Effective Date. The Effective Date of this First Amendment is July 31, 2010.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

Port/MTA MOU M-13828 1st Amendment 5/17/11

IN WITNESS WHEREOF, the parties have caused this First Amendment to be executed as of the date first written above.

CITY AND COUNTY OF SAN FRANCISCO, a municipal corporation operating by and through THE SAN FRANCISCO PORT COMMISSION CITY AND COUNTY OF SAN FRANCISCO, a municipal corporation operating by and through its SAN FRANCISCO MUNICIPAL TRANSPORTATION AGENCY

By: MONIQUE MOYER

Executive Director Port of San Francisco

By: NATHANIEL FORD

Executive Director San Francisco Municipal Transportation Agency

N:\PORT\AS2010\9690151\00637004:DOC

Dated: 05 - 31

Dated: 6 10 201

Port Commission Reso. 10-41

MTA Board Reso. 10-08-

Adopted: 6 15 2010 Attest:

Secretary, MTA Board

REVIEWED: DENNIS J. HERRERA, City Attorney

By:

Deputy City Attorney

Amendment Prepared By: Brad Benson

(initial)

Port/MTA MOU M-13828 1st Amendment 5/17/11

SCHEDULE 1

ENVIRONMENTAL REPORTS AND DOCUMENTS REGARDING HAZARDOUS MATERIALS

SF MTA / AUTORETURN

MAY 11, 2010

<u>PIER 70</u>

1999 Annual Ground Water Monitoring Report For Piers 70 And 94, Tetra Tech, Inc., February 2000.

1999 Annual Ground Water Monitoring Report Piers 70 Tetra Tech Inc 02-2000, Tetra Tech, Inc., February 2000.

Activity Summary Report, Pier 70, Sca Environmental, Inc., 8/5/02.

Amended Report Of Waste Discharge Pier 70 Solid Waste Disposal Site, The Mark Group 8-20-90, Mark Group, August 20, 1990.

Appendices F And G - Chemical And Physical Data Tables, Volume 2 Of 3, Subchapter 15 Compliance, Pier 70, Solid Waste Disposal Site, Geo/Resource Consultants, Inc., December 1989.

Appendix H - Laboratory Data Reports, Volume 3 Of 3, Subchapter 15 Compliance, Pier 70, Solid Waste Disposal Site, Geo/Resource Consultants, Inc., December 1989.

Article 22a Compliance; City Tow, Pier 70, San Francisco; Pr0249798, Iris Environmental, January 9, 2004.

Article 22a Compliance; Former Car Crusher Building

Former City Tow, Pier 70; Ehs-Hwu Case Number 69, San Francisco Department Of Public Health, September 1, 2005.

Asbestos Abatement Report, Pick Your Part, Inc. Pier 70, Iris Environmental, May 21, 2001, Iris Environmental, May 21, 2001.

Assessment Of Suspect Asbestos-Containing Materials At The Pier 70 Office Building, 20th And Illinois Streets, Clayton Environmental Consultants, April 9, 1996.

Certification Report; Pick Your Part, Inc.; Pier 70, San Francisco, Ca, Iris Environmental, December 19, 2003.

Data Compilation Report & Appendices A Through E Data Logs, Vol. 1 Of 3, Subchapter 15 Compliance, Pier 70 Solid Waste Disposal Site, Geo/Resource Consultants, Inc., December 1989.

Final Summary Report, Soil Boring Sampling And Well Installation For Pier 70 Building Pcb Remedial Investigation, Pier 70 Mixed Use Opportunity Area

Engineering/Remediation Resources Group, Inc. Nov. 2005, Engineering/Remediation Resources Group, Inc., November 2005.

Port/MTA MOU M-13828 1st Amendment 5/17/11

Georesource Data Compilation & Appendices A Through E Data Logs Report Vol. 1 Of 3 Subchapter 15 Pier 70 Solid Waste Disposal Site, December 1989, Geo/Resource Consultants, Inc., December 1989.

Groundwater Data Summary And Request To Rescind Waste Discharge Requirements; Pier 70 (Order 87-060) And Pier 94 (Order 87-061 Disposal Sites / Port Of San Francisco, Port Of San Francisco, August 17, 1999.

Hazardous Materials Investigation Of The Mariposa Facilities Project Area, July 1990, Erm-West, Inc., July, 1990.

Health And Safety Plan & Dust Control Mitigation And Monitoring Plan For Utility Installation And Soil Load Out Activities, Building 116 At Pier 70, Sca Environmental, Inc., 10/13/2009.

Order No. 00-030 / Rescission Of Waste Discharge Requirements In Order No. 87-060; City And County Of San Francisco / Pier 70 Class Iii Landfill, California Regional Water Quality Control Board, April 20, 2000.

Order No. 87-060 Updated Requirements; City And County Of San Francisco / Pier 70 Class Iii Landfill, California Regional Water Quality Control Board; 1987.

Pcb Removal, Corrective Action Plan, Pier 70, Building 50, Engineering-Remediation Resources Group, Inc. August 2003, Engineering/Remediation Resources Group, Inc., August 2003.

Phase 1 Environmental Site Assessment For Pier 70, Mixed Use Opportunity Area, Corner Of Illinois Street And 20th Street, Vol. I Of Ii, Environmental Site Assessment, Tetra Tech, Inc., August 1998.

Phase 1 Environmental Site Assessment For Pier 70, Mixed Use Opportunity Area, Corner Of Illinois Street And 20th Street, Vol. Ii Of Ii, Supplemental Documents, Tetra Tech, Inc., August 1998.

Phase I Brownfields, Environmental Site Assessment Report Pier 70, Ecology And Environment, Inc., March 2000.

Phase I Environmental Site Assessment For Pier 70 Maritime Use Area. Ecology And Environmental, Inc. March 2001, Ecology And Environment, Inc., March 2001.

Phase I Environmental Site Assessment For Pier 70, Mixed Use Opportunity Area, Corner Of Illinois Street And 20th Street. Tetra Tech, Inc. August 1998 (Vol. I), Tetra Tech, Inc., August 1998.

Phase I Environmental Site Assessment For Pier 70, Mixed Use Opportunity Area, Corner Of Illinois Street And 20th Street. Tetra Tech, Inc. August 1998 (Vol. Ii), Tetra Tech, Inc., August 1998.

Phase I Environmental Site Assessment, One-Acre Parcel On Swl 349, Pier 70. Iris Environmental, October 23, 2003, Iris Environmental, October 23, 2003.

Phase Ii Corrective Action Plan, Pcb Removal Action, Port Of San Francisco, Pier 70, Building 50. Engineering-Remediation Resources Group, Inc. June 2006, Engineering/Remediation Resources Group, Inc., June 2006.

Port/MTA MOU M-13828 1st Amendment 5/17/11

Pier 70 Mixed Use Opportunity Area, Pcb Assessment, Targeted Brownfields Assessment Report, Ecology And Environment, Inc., 1/9/03.

Pier 70 Mixed Use Opportunity Area, Phase Ii, Targeted Brownfields Assessment Report. Ecology And Environment, Inc. November 2000, Ecology And Environment, Inc., November 2000.

Pier 70 Mixed Used Opportunity Area Pcb Assessment, Targeted Brownfields Assessment Report, Ecology And Environmental, Inc. January 2003, Ecology And Environment, Inc., January 2003.

Pier 70 Mixed Used Opportunity Area, Expanded Phase Ii Target Brownfields Assessment Report. Ecology And Environment, Inc. January 2004, Ecology And Environment, Inc., January 2004.

Pier 70, Building 101 Cleanup, Acc Environmental Consultants, February 22, 1996.

Pier 70, Mixed Use Opportunity Area, Phase Ii Brownfields Targeted Site Assessment, Draft Sampling And Analysis Plan, Ecology And Environment, Inc., August 2002.

Pier 70, Mixed Use Opportunity Area, Phase Ii Brownfields, Targeted Site Assessment Report, Ecology And Environment, Inc., November 2000.

Pier 70, Mixed Use Opportunity Area, Phase Ii Brownfields, Targeted Site Assessment, Sampling And Analysis Plan, Ecology And Environment, Inc., June 2000.

Removal Of Underground Tanks, Pier 70, Harding Lawson Associates, March 31, 1988.

Sample Analysis Plan, Pier 70 Building 50 Pcb Remedial Investigation, Pier 70 Mixed Use Opportunity Area, Engineering Remediation Resources Group Inc, September 2004, Engineering/Remediation Resources Group, Inc., September 2004.

Sampling And Analytical Report, Pier 70, Building 6, Curtis & Tompkins, Ltd, November 1983.

Site History Report And Work Plan City Tow, Pier 70, Iris Environmental, Inc. February 7, 2002, Iris Environmental, February 7, 2002.

Site History Report And Work Plan, City Tow, Pier 70, Iris Environmental, February 7, 2002.

Site Investigation Report, Former Crusher Building, Pick Your Part, Inc. Pier 70, Iris Environmental, July 13, 2004, Iris Environmental, July 13, 2004.

Soil Remediation Report, Pick Your Part, Inc. Pier 70, Iris Environmental, December 12, 2001, Iris Environmental, December 12, 2001.

Soil Remediation Report, Pick Your Part, Inc., Pier 70, Iris Environmental, December 12, 2000.

Soils Analysis Report And Mitigation Plan For The City Tow Facility; Pier 70, San Francisco, Ca, San Francisco Department Of Public Health, July 29, 2002.

Soils Analysis Report And Mitigation Plan, City Tow, Pier 70, Iris Environmental, 7/10/02.

Port/MTA MOU M-13828 1st Amendment 5/17/11

Soils Analysis Report And Mitigation Plan; City Tow; Pier 70, San Francisco Pr0249798, San Francisco Department Of Public Health, October 10, 2002.

Soils Analysis Report And Site Mitigation Plan, Pick Your Part, Inc. [City Tow], Pier 70, Iris Environmental, July 10, 2002, Iris Environmental, July 10, 2002.

Submittals For: Pier 70, Corner Of 20th And Illinois, Bluewater Environmental Services, 1992-1996.

Subsurface Investigation At Illinois And 20th Streets (Site), Tetra Tech, Inc., 4/13/99.

Subsurface Investigation For Port Of San Francisco, Pier 70. Tetra Tech, Inc., December 1997, Tetra Tech, Inc., December 1997.

Subsurface Investigation, Pier 70, Tetra Tech, Inc., December 1997.

Summary Report: Asbestos And Lead-Based Paint Survey, Pier 70, Buildings 36, 40, 101, 102, 104 And 109, 20th And Illinois Streets, Sca Environmental, Inc., May 1998.

Summary Report: Bulk Asbestos And Exterior Paint Survey, Pier 70, Building 104, Sca Environmental, Inc., October 1999.

Summary Report: Bulk Asbestos And Lead-Based Paint Survey, Building 111, Pier 70, Sca Environmental, Inc., March 1995.

Summary Report: Bulk Asbestos And Lead-Based Paint Survey, Pier 70 Crane #30, Sca Environmental, Inc., February 1996.

Summary Report: Hazardous Materials Assessment, Port of SF Pier 70 Building 21, Sca Environmental, Inc., 4/2008.

Supplement Investigation Report Former Car Crusher Building Former City Tow-Pier 70, Iris Environmental, March 3, 2005.

Survey For Asbestos And Lead-Based Paint; Noonan Building (Building #11); Pier 70; Sca Project No: B-7265, Sca Environmental, Inc., 9/2/2005.

Pier 70 Site Investigation and Risk Assessment Report, Treadwell and Rollo, June 2010.

Seawall Lot 349

Analysis Of Bay Discharge, North Embarcadero Roadway Project, Camp Dresser & Mckee, 9/27/94.

Bunker Oil Characterization, North Embarcadero Roadway And F-Line Extension, Camp Dresser & Mckee, 12/29/95.

Embarcadero Roadway Urban Design Project, Hazardous Waste Investigation, Site History Report, Bechtel Environmental, Inc., August 11, 1989.

Embarcadero Roadway Urban Design Project, Rincon Point Block, Sampling And Testing Plan, Geo/Resource Consultants, Inc., October 16, 1989.

Geotechnical Investigation, Mariposa Storage/Transport Facilities, Ags, Inc., June 1989.

Port/MTA MOU M-13828 1st Amendment 5/17/11

10

Groundwater Data Summary And Request To Rescind Waste Discharge Requirements: Pier 70 (Order 87-060) And Pier 94 (Order 87-061 Disposal Sites / Port Of San Francisco, Port Of San Francisco, August 17, 1999.

Hazardous Materials Monitoring And Management Plans, Mariposa Facilities Project, Erm-West, Inc., 1/31/92.

Railway Track Ballast Survey, Sampling Results And Summary, North Embarcadero Roadway Project, Cky Environmental Service, 3/1/93.

Site History Report, Embarcadero Roadway Urban Design Project, Hazardous Waste Investigation, Appendix Iv, Agency Site Files, Volume 1 Of 4, Bechtel Environmental, Inc., August 1989.

Site History Report, Embarcadero Roadway Urban Design Project, Hazardous Waste Investigation, Appendix Iv, Agency Site Files, Volume 3 Of 4, Bechtel Environmental, Inc., August 1989.

Site History Report, Embarcadero Roadway Urban Design Project, Hazardous Waste Investigation, Appendix IV, Agency Site Files, Volume 4 Of 4, Bechtel Environmental, Inc., August 1989.

The Industrial History Of The Proposed Mariposa Facilities Project Area, Erm-West, Inc., August 1989.

Waste Classification Application, Embarcadero Roadway Project, Baseline Environmental Consulting, April 1993.

Waste Classification, Embarcadero Roadway Project (Waste Classification Unit File Number F134), Baseline Environmental Consulting, August 9, 1993.

Port/MTA MOU M-13828 1st Amendment 5/17/11

11

• • •

CITY AND COUNTY OF SAN FRANCISCO

GAVIN NEWSOM, MAYOR

MEMORANDUM OF UNDERSTANDING PORT REF. NO. MOU M-13828

BY AND BETWEEN

THE SAN FRANCISCO PORT COMMISSION

AND

THE SAN FRANCISCO DEPARTMENT OF PARKING AND TRAFFIC, A DIVISION OF THE MUNICIPAL TRANSPORTATION AGENCY

PORT OF SAN FRANCISCO

SAN FRANCISCO MUNICIPAL TRANSPORTATION AGENCY

Michael T. Burns Director of Transportation

BOARD OF DIRECTORS

Cleopatra Vaughns, Chairman Michael Kasolas, Vice-Chairman Shirley Breyer Black Wil Din Rev. Dr. James McCray, Jr. Peter Mezey

Monique Moyer Executive Director

PORT COMMISSION

Wilfred W. Hsu, President Michael Hardeman, Vice President Susan J. Bierman Kimberly K. Brandon Ann Lazarus

MEMORANDUM OF UNDERSTANDING PORT REF. NO. MOU M-13828

THIS MEMORANDUM OF UNDERSTANDING ("MOU") is entered into by and between the Department of Parking and Traffic ("DPT"), a division of the Municipal Transportation Agency, an agency of the City and County of San Francisco ("MTA") and the San Francisco Port Commission, an agency of the City and County of San Francisco ("Port").

RECITALS

A. Pursuant to an MOU dated June 1, 1994, as amended March 7, 1996 ("1994 MOU"), Port leased to DPT approximately 13.66 acres of property for the purpose of storing and handling abandoned automobiles, and related uses. The 1994 MOU replaced an earlier 1987 license between the Port and the San Francisco Police Commission for a similar use at the same location.

B. The initial term of the 1994 MOU expired on May 31, 1999, after which date the 1994 MOU has continued on a month-to-month basis in accordance with its terms. The 1994 MOU premises are presently being used by TEGSCO, LLC, a California limited liability company, dba San Francisco AutoReturn ("AutoReturn") for its automobile towing, storage and lien sale operations pursuant to an Emergency Interim Service Agreement and Property Use License for Towing, Storage and Disposal of Abandoned and Illegally Parked Vehicles between DPT and AutoReturn ("Emergency Interim Agreement"), approved by the Board of Supervisors by Resolution No. 423-04 on July 13, 2004, File No. 040638.

C. DPT is currently negotiating a long term towing services agreement ("Towing Agreement") pursuant to which DPT anticipates licensing the Port property shown in Exhibit A to this MOU ("License") to the long term towing services provider ("Licensee") and the Licensee is expected to continue use of the real property described herein for the purpose of automobile towing and storage and to conduct weekly public lien sales or auctions, subject to all of the applicable terms and conditions of this MOU, the Towing Agreement and License.

D. Under the Charter of San Francisco, the administration and control of real property transferred to the City of San Francisco by the State of California pursuant to the legislative trust grant, commonly referred to as the Burton Act (Chapter 1333 of Statutes of 1968), as amended including the area encompassing the real property which is the subject of this MOU, is vested in the Port.

NOW, THEREFORE, IT IS AGREED AS FOLLOWS:

AGREEMENT

1. <u>Recitals.</u> The foregoing recitals are true and correct and are incorporated herein by this reference.

2. <u>Termination of 1994 MOU/Effective Date</u>. The Port and DPT hereby mutually agree that the 1994 MOU will be terminated on the Effective Date of the Towing Agreement as defined therein, which date shall be the effective date ("Effective Date") of this MOU. The 1994 MOU contained the following acknowledgment: "Because the rental fees due hereunder are not intended to cover risks associated with DPT's use of the Premises and because of certain funding restrictions imposed on Port funds due to public trust restraints, it is the understanding of the parties that Port shall not expend any funds due to or in connection with DPT's, or its contractor's, use of the Premises." Accordingly, DPT agreed to hold the Port harmless from claims, damages, liabilities or losses as specified in Section 12 of the 1994 MOU, and Section 12 further provided that: "The foregoing obligation of DPT shall survive termination of this agreement." Therefore, the parties hereto recognize that DPT's obligations under Section 12 of the 1994 MOU survive the termination of the 1994 MOU.

3. <u>Premises/Condition</u>.

a. <u>Parcel A</u>. For the Rent and subject to the terms and conditions of this MOU, Port hereby authorizes DPT to use the area located at Pier 70 and SWL 349 in the City and County of San Francisco, California, shown outlined and marked Parcel A on Exhibit A, attached hereto and made a part hereof, including approximately 406,810 square feet of paved land and 112,518 square feet of shed space contained in 5 buildings that are as shown on Exhibit A, and 3,967.8 square feet of Unusable Area as shown on Exhibit A and reserving a non-exclusive right of way in the general area shown on Exhibit A as Reserved 22nd Street Extension, an area extending from the east end of 22nd Street through Parcel A and connecting with the east end of 20th Street, together with any improvements located thereon (the "Parcel A Premises").

b. <u>Parcel B</u>. For the Rent and subject to the terms and conditions of this MOU, including but not limited to the obligation to obtain a permit from the San Francisco Bay Conservation and Development Commission ("BCDC") for the uses contemplated under this MOU, Port hereby authorizes DPT to use the area located at Pier 70 and SWL 349 in the City and County of San Francisco, California, shown outlined and marked Parcel B on Exhibit A attached hereto, including approximately 17,241 square feet of paved land (the "Parcel B Premises").

c. <u>Definition of Premises</u>. The Parcel A Premises and the Parcel B Premises are hereinafter collectively referred to as the "Premises."

d. <u>Condition</u>. DPT hereby acknowledges that hazardous substances may exist on the Premises, and that it will be responsible for notifying its Licensee as required by Health and Safety Code Section 25359.7. DPT further acknowledges that DPT has permitted, and will permit its Licensees to use the Premises for the purposes authorized herein. DPT agrees that the Port shall have no responsibility for: (i) terminating Licensee's use of the Premises, (ii) removing Licensee's possessions, (iii) removing Licensee's vehicles and (iv) removing other materials (including Hazardous Materials if any) put on the Premises by Licensee(s).

e. <u>Requirement to Fence</u>. DPT shall have the affirmative obligation to install and maintain Port-approved fencing along the perimeter of the Premises. Port agrees that the existing perimeter fencing meets its requirements; provided, however, that if DPT fails to satisfy the condition of obtaining BCDC approval for use of Parcel B, and surrenders possession of Parcel B to the Port, DPT shall install and maintain Port approved fencing along the adjusted perimeter of the Premises. The installation cost of the fencing needed as a result of the Port extending 22nd Street through the Premises shall be the responsibility of the Port.

Term. DPT will use the Premises from the Effective Date through a date five (5) years 4. from the Effective Date ("Expiration Date"), but not later than September 1, 2010. DPT shall have the right to terminate the MOU at any time during the Term of the MOU provided that DPT also terminates the License, upon providing to the Port one hundred eighty (180) days prior written notice. On a date no sooner than three (3) years and no later than two (2) years prior to the Expiration Date DPT shall send written notice to the Port Executive Director stating whether DPT intends to extend the MOU beyond the Expiration Date and requesting a meeting between DPT and the Port to discuss the parties' plans for the Premises. In the event that DPT wishes to extend this MOU beyond the Expiration Date, it shall so notify Port in writing at least one hundred eighty (180) days prior to the Expiration Date, and thereupon DPT's use of the Premises shall continue on all of the terms and conditions stated herein, except for Rent which shall be governed by Section 5(c), unless the Port has notified DPT one hundred eighty (180) days prior to the Expiration Date in writing that the MOU may not be extended. If the MOU is extended as provided above beyond the Expiration Date, either party hereto may terminate the MOU upon providing to the other party one hundred eighty (180) days prior written notice. In no event shall any hold over period extend beyond March 1, 2012. The definition "Term" shall refer to the total time period during which this MOU exists as a legally binding agreement between the parties, including all month-to-month extensions.

5. <u>Rental Payments</u>.

a. <u>Rent</u>. DPT will cause to be paid a monthly rental fee of \$118,830.00 (one hundred eighteen thousand eight hundred thirty dollars) for Parcel A and \$2,850 (two thousand eight hundred fifty dollars) for Parcel B ("Rent") to Port in lawful money of the United States of America at Port's address for notices, as set forth herein, on or before the 1st of the month for which the Rent is due, without prior demand and without any deduction, setoff or counterclaim whatsoever, except as such deduction or setoff is specifically provided for in Section 5(d) [Rent Abatement and Credits]. Use of the Premises for any partial month will be prorated at the rate of one-thirtieth (1/30) of the monthly charge.

Appendix C – MOU between DPT and Port Page 2 of 18 If DPT fails to pay Rent or any portion of Rent within ten (10) days following the due date, such unpaid amount shall be subject to a late payment charge (the "Late Charge") equal to one and one-half percent $(1\frac{1}{2}\%)$ of all undisputed amounts which remain unpaid, and/or one and one-half percent $(1\frac{1}{2}\%)$ of all undisputed amounts which are more than five (5) business days late. Such Late Charge may be assessed without notice and cure periods. The Late Charge shall start accruing as of the original due date of the owed amount notwithstanding the five (5) calendar day grace period provided herein.

b. <u>Rent Adjustment</u>. Commencing on the Effective Date of this MOU and on each anniversary date (the "Anniversary Date") thereafter, the Rent shall be adjusted on the first day of the month that immediately follows the Effective Date and on that same date in each succeeding year in direct proportion to the percentage increase in the Current CPI Index for the month immediately preceding the applicable Anniversary Date ("Current Index") over the CPI index for the month of June, 2004 ("Base Index"). In no case shall the Rent, as adjusted, be less than the Rent in effect immediately prior to the Anniversary Date. If the Current Index has increased over the Base Index, the adjusted Base Rent shall be determined by multiplying the Base Rent by a fraction, the numerator of which is the Current Index and the denominator of which is the Base Index, as follows:

 $\frac{\text{Current Index}}{\text{Base Index}} X \qquad \text{Base Rent} = Adjusted Base Rent}$

c. <u>Extension Period Rent</u>. Should DPT hold over beyond the Expiration Date specified in Section 4 of this MOU, Rent shall be increased to an amount equal to one hundred twenty percent (120%) of the Rent payable by DPT on the last calendar month of the initial five-year term on a month-to-month basis until such time as the parties renegotiate the Rent payments for any additional term, or the MOU terminates. The Extension Period Rent shall be subject to further Rent Adjustment per Section 5(b) of this MOU.

d. <u>Rent Abatement and Credits</u>.

1. If the Premises cease to be used for towing operations at any time due to damage sustained during the Term by fire, earthquake, or other casualty rendering the Premises unsuitable for occupancy, as determined by the Director of Building Inspection pursuant to the San Francisco Building Code, or are otherwise deemed legally not useable for any reason, Rent hereunder shall be abated and DPT shall have the option to terminate the MOU and shall be entitled to a prorated refund of any Rent or deposits paid. In the event the Premises ceases to be used for more than two (2) consecutive months for towing operations, Port, at its option, may terminate this MOU.

2. DPT shall be entitled to a proportional abatement in the Rent if the exercise of any rights reserved to the Port in this MOU, results in the loss of use of the Premises or any portion thereof for a period in excess of thirty (30) days or in an area in excess of 250 square feet, or if DPT surrenders the possession of Parcel B to the Port for the sole reason that it is unable to obtain a BCDC permit for use of Parcel B consistent with this MOU. The opening of 22nd Street by the Port for Appendix C – MOU between DPT and Port

Page 3 of 18

non-exclusive, general circulation through the Pier 70 area may occur at the Port's sole option without Rent abatement. The planned alignment for the 22^{nd} Street Extension is as shown on Department of Parking and Traffic, Division of Traffic Engineering drawing on file with the Port's Chief Harbor Engineer, entitled "New Pier 70 Roadway – Preliminary Striping Plan", dated June 28, 2001, File Name swl349sv newroad final.dwg. The parties recognize that the final alignment may deviate as necessary to effect the right of way improvement.

3. In the event Port requires the modification of the original configuration of the Premises, any reasonable and actual costs incurred by Licensee to relocate the Premises, fences, gates, lights, driveways and other improvements in order to comply with such requirement may be offset from the Rent, except for such costs incurred with respect to the opening of 22^{nd} Street by the Port for non-exclusive, general circulation through the Pier 70 area, which may occur at the Port's sole option without Rent abatement; however in such event Port shall be responsible for the cost of any fencing that Port requires along the extended 22^{nd} Street corridor. The parties agree that any such costs incurred to modify the original configuration of the Premises due to a surrender of possession of Parcel B to the Port due to DPT's inability to obtain a BCDC permit for use of Parcel B consistent with this MOU may not be offset from the Rent.

4. At any time during the Term of the MOU, the Port may elect to implement the extension of 22nd Street in the approximate location shown on Exhibit A. In order to fully implement the extension of 22nd Street, a fence will have to be installed around certain areas demising the Premises from the new roadway to provide for adequate security for the Premises. The responsibility to construct the fence is at the sole cost and expense of the Port. Both Port and DPT herein agree that, at the Port's option, Port Executive Director may require DPT's Licensee, upon ninety (90) days written notice, to construct the fence at a cost not to exceed the Construction Costs defined below. The scope and specification of the fence shall be in the Port's sole discretion. In consideration for installation of the fence by DPT's Licensee, and expenditures related thereto, and upon approval by Port of Construction Costs, Port shall issue an appropriate rent credit (herein "Rent Credit") to DPT. The Rent Credit shall be a sum equal to the Construction Costs, or a sum not to exceed one hundred fifty thousand dollars (\$150,000.00), whichever is the lesser amount. In the event that Licensee determines that the fence specified by the Port does not meet its security or related needs, it may elect to apply the Rent Credit for estimated Construction Costs of the fence specified by the Port to the costs of an alternate fence, provided that: (i) it obtains a building permit from the Port for construction of the alternate fence, (ii) the alternate fence is more expensive than the fence initially specified by the Port, and (iii) the Port Executive Director approves the plan for the alternate fence. "Construction Costs" are defined to be the lesser of the estimated or actual costs incurred for labor, materials, contractor fees, and reasonable architecture and engineering fees in connection with the project pursuant to a bid obtained under a guaranteed not-to-exceed construction bid for the fence project that is approved by the Port Executive Director. In order for construction of the project to be authorized by Port: DPT must first obtain, prior to commencing the project, written approval from Port Executive Director that the Construction Costs of the proposed project are reasonable; and DPT, or its Licensee, must obtain all required governmental approvals, including, but not limited to building permits. After the completion of the Appendix C – MOU between DPT and Port

Page 4 of 18

approved project, DPT must deliver to Port an itemized statement of the actual Construction Costs expended on the approved project, accompanied by documentation substantiating all said expenditures. Such documentation of expenditures shall include: (i) copies of executed contracts; (ii) copies of invoices for labor, services and/or materials, copies of bills of lading, and/or copies of other bills or receipts for goods, materials and/or services; (iii) copies of canceled checks, and (iv) such other proofs of expenditure as may by reasonably requested by Port. Such appropriate proofs of expenditure may include copies of canceled checks; copies of contracts or invoices for labor, services and/or materials marked "Paid", or otherwise evidenced as having been paid bills of lading marked "Paid"; other bills, contracts, receipts for goods materials and/or services marked "Paid" and such other proofs of expenditure as may be reasonably approved by Port. All such proofs of expenditure must be directly attributable to the approved project and may include materials purchased by DPT for installation by Port but not the Port's cost if it undertakes such installation.

e. Formula for Abatement/Credit. All rent credits available to DPT permitted by Section 5(d) shall be applied against Base Rent payment obligation during the Term at a rate not greater than one half ($\frac{1}{2}$) of the applicable month Base Rent payment and shall be applied if and only if DPT is in good standing and is not in default of any of the terms of this MOU. In the event that the total of rent credits available to DPT pursuant to Section 5(d) and any other sections of this MOU exceeds an amount equal to one half ($\frac{1}{2}$) of the Base Rent payment for any one calendar month, the remaining available Rent Credit shall be carried forward to successive calendar months at a rate not to exceed one half ($\frac{1}{2}$) of the applicable Base Rent payment, until all available rent credits have been fully applied. In no event, however, shall DPT be entitled to the application of any rent credits or the value thereof, beyond the expiration or earlier termination of this MOU.

6. <u>Financial Assurances and Fines for Violations.</u>

a. DPT agrees that it shall require Licensee to provide DPT and the Port with a security deposit of at least \$1,000,000.00 to secure Licensee's obligations under this MOU, the Towing Agreement and the License ("Security Deposit").

b. The parties agree that at least \$250,000.00 of the Security Deposit shall be reserved as a security for rental payments ("Rental Payment Security Deposit") due to Port hereunder, and DPT shall require Licensee to provide the Rental Payment Security Deposit in the form of a financial guarantee that is directly accessible to the Port at all times during the term of the License. The amount of the Rental Payment Security Deposit shall be increased in accordance with increases in Rent so as to equal at least two months' Rent at all times.

c. The parties agree that at least \$400,000.00 of Licensee's Security Deposit shall be reserved for the purposes of (i) ensuring regulatory compliance and to pay Port and DPT costs in the event that Licensee receives a notice of violation or other regulatory order from a governmental or regulatory agency with jurisdiction over the Premises and or Licensee's operations (other than from the Port) and Licensee does not achieve compliance with the notice of violation or order to the satisfaction of the issuing agency within the time specified by the agency or by the City if the agency does not Appendix C - MOU between DPT and Port

Page 5 of 18

specify a timeframe; or (ii) reimbursement of any fine or other charge assessed against the City related to any notice of violation or other regulatory order issued to Licensee; or (iii) reimbursing the City for costs associated with City's environmental assessments or corrective action, which may be performed at the City's sole discretion; or (iv) for property damage to the Premises ("Environmental and Property Security Deposit"). DPT shall require Licensee to provide the Environmental and Property Security Deposit in the form of a financial guarantee that is directly accessible to the Port at all times during the term of the License.

d. DPT agrees that it shall require Licensee to provide, maintain and replenish throughout the Term of the License and for a period of at least ninety (90) days after expiration of the License, an Environmental Oversight Deposit in the amount of \$10,000, which shall be deposited in an account to be specified by and accessible to the Port. If Licensee receives a notice of violation or other regulatory order from a governmental or regulatory agency with jurisdiction over the site and or its operations (other than from the Port), and such notice is not cured within fourteen (14) days, the City may draw from this deposit to reimburse the City for staff costs incurred by the City while inspecting site conditions and enforcing and administering the Hazardous Materials provisions of the License. If Licensee receives a notice of violation or other regulatory order from a governmental or the site and or its operations (other than from the Port), and such notice is not cured within fourteen (14) days, the City may draw from this deposit in an amount not to exceed \$500 to reimburse the City for staff costs incurred by the City, and such notice is cured within fourteen (14) days, the City may draw from this deposit in an amount not to exceed \$500 to reimburse the City for staff costs incurred by the City. The City will submit an invoice to Licensee for any such costs, and Licensee will pay such invoiced amounts within thirty (30) days to replenish the Environmental Oversight Deposit.

e. Port and DPT agree to cooperate in enforcement of the License requirements, including the imposition of any fines allowed under Section 6.7 of the License as appropriate. DPT shall have primary responsibility for enforcement of fines and providing notification of violations to Licensee. If Port identifies violations of the License requirements, it may notify DPT and DPT shall notify Licensee of the violation and imposition of fines within three (3) business days. In the event that Port notifies DPT of a violation of the License which results in the imposition of fines pursuant to Section 6.7, the fines for such violations shall be paid to Port.

7. <u>Permitted Uses.</u> The Premises shall be used for temporary storage and transfer of motor vehicles towed pursuant to the Towing Agreement between DPT and Licensee, for related office use as necessary to meet its obligations under the Towing Agreement and to conduct weekly public auction "lien sales" required by state laws and related uses, and for no other purpose. The buildings identified as Buildings 12, 15, 16, 31, and 32 on Exhibit A shall only be used to store vehicles and shall not be used as office space. Occupancy of the loft area of Building 12 will not be permitted unless all State and City Code requirements are satisfied, as determined by the Port. DPT and Licensee shall not use the Premises for any other purpose, including, without limitation, vehicle crushing and dismantling. Licensee may seek authorization from the Port Executive Director and DPT to maintain, fuel or wash vehicles and/or to sell vehicle parts from the Premises for the purpose of making auctioned or lien sold vehicles operable. These activities may only be conducted on the Premises after both of the following

Appendix C – MOU between DPT and Port Page 6 of 18 occur: (i) the Port Executive Director and DPT authorize the specific activity in writing, and (ii) the specific activity is incorporated as authorized into the appropriate section of the Operations Plan as described in Section 14(k) herein, including a description of the manner in which the activity will be conducted on the Premises, and such addition to the Operations Plan is approved as required by Section 14(k) herein, the Towing Agreement, Appendix A Section 14 and Appendix B. Once approved as provided herein, such approval of Operations Plan elements listed in Appendix B as "Pier 70 Operation Plan Elements" may not be revoked except as part of a regular amendment of the Operations Plan under the Towing Agreement. DPT, SFPD, its Licensee and their respective agents or employees shall use 20th and/or 22nd Street for access and egress to and from the Premises. No loading, unloading, queuing, parking or storage of vehicles will be permitted on any adjacent Port property, public streets or rightsof-way in the vicinity unless provided for in a separate agreement executed by both parties. DPT shall not authorize the placement of advertising or signage in areas on or about the Premises without the prior written permission of the Port, subject to the Port sign guidelines. DPT agrees to require and enforce an Operations Plan through the Towing Agreement, which will incorporate the provisions required in Section 14(k) of this MOU.

8. <u>BCDC Permit.</u> Port shall cooperate with DPT and its Licensee(s) in obtaining any necessary BCDC permits and approvals, including, if necessary, joining with DPT or Licensee as a co-applicant, provided that DPT or Licensee pay all fees and satisfy all conditions incurred in the BCDC process.

9. <u>Maintenance, Improvements, Utilities, Surrender of Premises.</u> DPT will be strictly responsible for the security of the Premises, and the maintenance of the Premises and all utilities thereon in good order and repair. DPT agrees not to make or authorize any improvements or alterations to the Premises without the prior written consent of Port, which consent shall not be unreasonably withheld. DPT will pay, or cause to be paid for, all utility services provided to the Premises, including but not limited to electricity, water, sewer, gas and telephone, and will provide any scavenger service necessitated by its use of the Premises. At the expiration or sooner termination of this MOU, DPT shall surrender the Premises clean and free of any vehicles, auto parts, Hazardous Materials or other materials introduced and stored on the Premises by DPT or its Licensee(s) and shall repair any damage to the Premises occasioned by DPT's use, excluding ordinary wear and tear. DPT shall require the Licensee to have a letter of credit or other guarantee securing Licensee's obligation to remove all vehicles, auto parts, Hazardous Materials introduced and stored on the Premises, improvements, alterations or other materials introduced and stored on the guarantee securing Licensee's obligation to remove all vehicles, auto parts, Hazardous Materials, improvements, alterations or other materials introduced and stored on the remises by Licensee, in a form and amount acceptable to the Port, in its reasonable discretion.

10. <u>Insurance</u>. DPT shall require, pursuant to the Towing Agreement and/or any other written instrument acceptable to the Port, that Licensee and any other subtenant or any agent, contractor or subcontractor (collectively referred to as "Licensee" for the purpose of this section) it hires in connection with its use of the Premises shall secure such insurance as is recommended by the City Risk Manager and approved by Port, which approval shall not be unreasonably withheld. Such required insurance shall include, at a minimum, Commercial General Liability coverage with policy limits of five million dollar (\$5,000,000) per occurrence. All liability insurance policies shall be endorsed to

Appendix C - MOU between DPT and Port

Page 7 of 18

name the City and County of San Francisco, the Port Commission and their officers, directors and employees as additional insureds. DPT shall require the Licensee to provide to the Port certificates of insurance and policy endorsements upon execution of the Towing Agreement, and as the insurance and endorsements are renewed during the Term. For Commercial General Liability insurance required of the Licensee, DPT's Towing Agreement shall require the Licensee to provide an endorsement in the form of ISO Form 2026 or its equivalent, without modification.

11. Damages and Indemnity. DPT, directly and through its Licensee, agrees to be responsible for all costs associated with all claims, judgments, damages, penalties, fines, costs, liabilities or losses, including costs of defense and attorneys fees, which arise as the result of presence of Hazardous Materials (as defined in the License) on the Premises or out of any injury or death of any person or damage of any property occurring in, on or about the Premises, as the result of any other act or omission of DPT or the Licensee or the officers, directors, employees, agents, or invitees of DPT or the Licensee. The foregoing obligations of DPT shall survive the termination of this agreement. Furthermore, this MOU imposes no responsibility on DPT or Licensee for any claims, judgments, including costs of defense and attorneys' fees, damages, penalties, fines, costs, liabilities or losses which arise from existence of Hazardous Materials introduced onto the Premises by (1) the Port or its officers, directors, employees, agents, or invitees, corporations or entities, except as otherwise expressly provided in Section 2 of this MOU.

Compliance of Licensee(s) with Terms; Port Consent; Assignment. During the Term 12. of this MOU, DPT may issue a Request for Proposals ("RFP") or other solicitation to award a Licensee the right to enter into a Towing Agreement with DPT for the purpose of towing, storing and disposing of abandoned and illegally parked vehicles. DPT hereby agrees to include this MOU in any such RFP or solicitation. DPT further agrees that the Towing Agreement and any other agreement entered into by and between DPT and another party granting such party use of the Premises shall incorporate all of the terms and conditions of this MOU, and require Licensee's or such other party's agreement to comply with all of terms and conditions set forth hereunder. Prior to submitting Towing Agreement for approval, DPT shall provide Port with a copy of such Towing Agreement and the License to use the MOU Premises, and Port shall review the Towing Agreement and License for consistency with this MOU. Any License or other agreement granting a party use of the Premises shall be subject to the prior written consent of the Port Director, in the Port Director's reasonable discretion. Consent to the License shall not be construed as a waiver or release of any DPT obligation under this MOU, nor shall it create a privity of contract between Port and Licensee. DPT shall not consent to the assignment or assign the Towing Agreement or enter into any new agreement during the Term without the prior written consent of Port pursuant to this section. Prior to entering into a new agreement or assigning the Towing Agreement or License to a new Licensee, DPT will disclose to the Port any outstanding obligations of the previous Licensee. and present a plan describing how such obligations will be met by DPT and/or the Licensee or assigned to the new Licensee.

13. <u>Entry</u>.

a. Port may enter upon the Premises at any reasonable time with or without notice to DPT or the present occupant of the Premises when deemed reasonably necessary to oversee or inspect Licensee's operations or conduct any business with Licensee, to show the Premises to prospective tenants or other interested parties, to post notices of non-responsibility, to conduct any environmental audit of Licensee's use of the Premises, to repair, alter or improve any part of the Buildings, Building Systems or the Premises, whenever City believes that emergency access is required, for the protection of the Port's interests and for any other lawful purpose. In the event Port chooses to install or replace metal plates and bull rails around the trenches located on the Premises, DPT and any present occupant of the Premises shall permit such entry and shall cooperate with and not interfere with such work.

b. The Port reserves for itself and its designees, the right to enter the Premises and any portion thereof at all reasonable times upon reasonable advance oral or written notice to Licensee (except in the event of an emergency) for the following purposes: (a) to conduct inspections or inventories; (b) to install, inspect, sample, monitor, close and abandon permanent or temporary groundwater wells; (c) to obtain environmental samples from all media, from both on and offshore areas; (d) to conduct utility clearances; and (e) to conduct cleanup activities. Port shall use its reasonable good faith efforts to conduct any activities on the Premises allowed under this Section in a manner that, to the extent practicable, will minimize any disruption to Licensee's use hereunder. Licensee will cooperate with Port by complying with reasonable requests to temporarily relocate vehicles and other operations to accommodate the Port, at no expense to the Port. Except as otherwise specified herein, Port shall not be liable in any manner, and Licensee hereby waives any claims, for any inconvenience, disturbance, loss of business, nuisance or other damage arising out of Port's or its designees' entry onto the Premises, except damage resulting directly and exclusively from the gross negligence or willful misconduct of Port or its designees and not contributed to by the acts, omissions or negligence of Licensee or Contactor's Invitees.

14. <u>Property Use Conditions</u>. DPT shall include the following conditions in any property lease or license with any Licensee:

a. <u>Requirement that Premises be Used</u>. Licensee shall continuously use the Premises for the uses specified in this MOU.

b. <u>Compliance with Laws and Regulations</u>. Licensee, at Licensee's cost and expense, shall comply with all laws, ordinances, judicial decisions, orders and regulations of federal, state, county and municipal governments and the departments, courts, commissions, boards and officers thereof pertaining to Licensee's use and occupation of the Premises in effect either at the time of execution of the Towing Agreement or at any time during the Term. Licensee further understands and agrees to be responsible and comply within its leased Premises with 42 USCS 12101, et seq., commonly known as the Americans with Disabilities Act.

c.

Mineral Rights. The State of California, pursuant to Section 2 of Chapter 1333 of Appendix C – MOU between DPT and Port Page 9 of 18 the Statutes of 1968, as amended, has reserved all subsurface mineral deposits, including oil and gas deposits, on or underlying the Premises. In accordance with the provisions of said statute, Port and Licensee shall and hereby do grant to the State of California the right to explore, drill for and extract said subsurface minerals, including oil and gas deposits, from an area located by the California Grid System, Zone 3, beginning at a point where X equals 1,456,113 and Y equals 463,597, extending 1,000 feet south, thence 1,000 feet east, thence 1,000 feet north, and thence 1,000 feet west, ending at said point of beginning.

d. <u>Possessory Interest Tax</u>. Licensee acknowledges and understands that a possessory interest subject to property taxation may be created by the Towing Agreement and that Licensee may be subject to the payment of property taxes levied on such possessory interest. Licensee further acknowledges that Licensee is familiar with San Francisco Administrative Code Sections 23.38-23.39, which require that Port submit a report, which includes specified information relating to the creation, renewal, sublease, or assignment of any such possessory interest, to the County Assessor within 60 days after any such transaction. Licensee agrees to provide to Port the information required by Sections 23.38-23.39 within 30 days of a request in writing by Port to do so.

e. <u>Hazardous Materials</u>

1. Each party to the MOU will cooperate to affirmatively enforce the License to the fullest extent possible and in a manner that protects the Premises and the City from potential liability related to Hazardous Materials and Environmental Laws.

2. The License shall require the Licensee, prior to termination of the License or during the Term if required by a governmental agency, at its sole cost and expense, to remove any and all Hazardous Materials introduced in, on, under or about the Premises by Licensee, its agents or invitees during the term of the License. Prior to the termination of the License, Port, DPT and Licensee shall conduct a joint inspection of the Premises for the purpose of identifying Hazardous Materials on the Premises which Licensee is required to remove.

3. If Licensee fails to fulfill its removal obligations, DPT shall undertake such obligations, or make funds available to the Port to fulfill such obligations.

f. Improvements and Alterations.

1. <u>Definitions</u>. "Alterations" means any alterations, including, but not limited to demolition, removal, installations or additions to or improvements to the Property, including the installation of any appurtenances or trade fixtures affixed to the Property, including those constructed by or on behalf of Licensee to the Improvements or to the Premises. "Improvements" means any and all buildings, structures, fixtures or other improvements constructed or installed on the Premises, including those constructed by or on behalf of Licensee pursuant to the Towing Agreement (including, without limitation, any trailers, signs, roads, trails, driveways, parking areas, curbs, walks, fences, walls, stairs, poles, plantings and landscaping).

Appendix C – MOU between DPT and Port

Page 10 of 18

2. Construction Requirements. All Alterations or Improvements to the Premises made by or on behalf of Licensee shall be subject to the following conditions, which Licensee covenants faithfully to perform. Licensee shall not make, nor cause or suffer to be made, any Alterations or Improvements to the Premises until Licensee shall have first obtained DPT and Port written approval, and if granted, then obtained a permit therefore from the Port Engineering Department, with respect to the Premises, and any other permits or approvals as the Port's Chief Harbor Engineer deems necessary, and any required approvals of regulatory agencies having jurisdiction over the Premises. All Alterations or Improvements shall be done at Licensee's expense in accordance with plans and specifications approved by, and subject to any conditions imposed by, DPT and Port, and only by duly licensed and bonded contractors or mechanics approved by DPT. Any proposed Alterations to Buildings 12,15,16, 31, and 32 must comply with the federal Secretary of the Interior Standards of Rehabilitation, as determined by the Port. DPT may require Licensee, at Licensee's expense to obtain the prior written approval of City's Arts Commission with respect to any Alterations or Improvements, to the extent that Arts Commission has jurisdiction over the design of such proposed alterations under City's Charter Section 5.103. Licensee shall pay to Port any applicable permit fees for such Alterations in accordance with standard permit fees generally charged to Port tenants, as adopted by the Port Commission.

(i) All Alterations and Improvements shall be constructed in a good and workmanlike manner and in compliance with all applicable building, zoning and other applicable Laws, and compliance with the terms of and the conditions imposed in any Regulatory Approval;

(ii) All Alterations and Improvements shall be performed with reasonable dispatch, delays beyond the reasonable control of Licensee excepted; and

(iii) At the completion of the construction of the Alterations or Improvements, Licensee shall furnish one (1) set of "as-built" drawings of the same made on or to the Premises. Unless otherwise stated as a condition of the Regulatory Approval, this requirement may be-fulfilled by the submittal after completion of the Alterations or Improvements of a hand-corrected copy of the approved permit drawings.

(iv) Licensee shall have procured and paid for all Regulatory Approvals (as defined in paragraph 14(i) below) required to be obtained for such Alterations or Improvements, including, but not limited to, any building or similar permits required by Port or its Chief Harbor Engineer in the exercise of its jurisdiction with respect to the Premises.

3. <u>Improvements Part of Realty</u>. All Alterations or Improvements to the Premises made by or on behalf of Licensee which may not be removed without substantial injury to the Premises shall become part of the realty, shall be owned by Port and shall, at the end of the Term hereof, remain on the Premises without compensation to Licensee, unless Port first waives its right to the Alterations or Improvements in writing.

4. <u>Removal of Improvements</u>. At Port's election made in accordance with Section 14(f)5 hereof, Licensee shall be obligated at its own expense to remove and relocate or demolish and remove (as Licensee may choose) any or all Alterations or Improvements which Licensee has made to the Premises, including without limitation all telephone wiring and equipment installed by Licensee. Licensee shall repair, at its own expense, in good workmanlike fashion any damage occasioned thereby, except as may be provided in Section 14(f)(7) below.

5. Notice of Removal. Prior to the termination of the Towing Agreement, Port shall give written notice to Licensee (herein "Notice of Removal") specifying the Alterations or Improvements or portions thereof which Licensee shall be required to remove and relocate or demolish and remove from the Premises, in accordance with Section 14(f)4. If termination is the result of loss or destruction of the Premises or any Improvements thereon, Port shall deliver said Notice of Removal to Licensee within a reasonable time after the loss or destruction. If Licensee fails to complete such demolition or removal on or before the termination of the Towing Agreement, Port may perform such removal or demolition at Licensee's expense, and Licensee shall reimburse Port upon demand therefor. The Notice of Removal shall not include any Alterations which were previously designated as not subject to removal in accordance with Section 14(f)(7) below.

6. <u>Removal of Non-Permitted Improvements</u>. If Licensee constructs any Alterations or Improvements to the Premises without Port's prior written consent or without complying with Section 14(f)2 hereof, then, in addition to any other remedy available to Port, Port may require Licensee to remove, at Licensee's expense, any or all such Alterations or Improvements and to repair, at Licensee's expense and in good workmanlike fashion, any damage occasioned thereby. Licensee shall pay to Port all special inspection fees as set forth in the San Francisco Building Code for inspection of work performed without required permits.

7. <u>Alterations Not Subject to Removal</u>. Licensee may submit a request for a City determination to Port and DPT prior to commencing work on an Alteration, for the purpose of requesting a determination of whether the Alteration would or would not be required to be removed upon expiration or termination of this License. Such request for determination shall be submitted to DPT and to the Executive Director of the Port. The Executive Director of the Port shall have sixty (60) days to review the request and to respond to Licensee with a determination of whether the proposed Alteration would or would not be required to be removed upon termination of this License. This Section shall not apply to Alterations that are required by any regulatory authority to conform the Premises or any building thereon to a requirement of statute, ordinance or regulation.

g. <u>Suitability: Acceptance.</u> Licensee acknowledges that Port has made no representations or warranties concerning the Premises and accepts the Premises in its "As Is" "With All Faults" condition.

h. <u>Liens</u>. Licensee shall keep the Premises free from any liens arising out of any work performed, materials furnished or obligations incurred by Licensee or its Agents. In the event that Appendix C – MOU between DPT and Port

Page 12 of 18

Licensee shall not, within twenty (20) days following the imposition of any such lien, cause the same to be released of record, DPT and Port shall have, in addition to all other remedies provided by the Towing Agreement or by Law, the right but not the obligation to cause the same to be released by such means as it shall deem proper, including without limitation, payment of the claim giving rise to such lien. All sums paid by DPT or Port for such purpose and all reasonable expenses incurred by Port in connection therewith shall be payable to Port by Licensee within thirty (30) days following written demand by DPT or Port.

i. Regulatory Approvals. Licensee understands that Licensee's operations on the Premises, changes in use, or Improvements or Alterations to the Premises may require an authorization, approval or a permit required by any governmental agency having jurisdiction over the Premises, including but not limited to the Bay Conservation and Development Commission ("BCDC") ("Regulatory Approval"). Licensee shall be solely responsible for obtaining any such Regulatory Approval, and Licensee shall not seek any Regulatory Approval without first obtaining the approval of DPT and the Port. All costs associated with applying for and obtaining any necessary Regulatory Approval shall be borne by Licensee. Licensee shall be solely responsible for complying with any and all conditions imposed by regulatory agencies as part of a Regulatory Approval. Any fines or penalties imposed as a result of the failure of Licensee to comply with the terms and conditions of any Regulatory Approval shall be paid and discharged by Licensee, and Port shall have no liability, monetary or otherwise, for said fines and penalties. To the fullest extent permitted by Law, Licensee agrees to indemnify and hold City, Port and their Agents harmless from and against any loss, expense, cost, damage, attorneys' fees, penalties, claims or liabilities which City or Port may incur as a result of Licensee's failure to obtain or comply with the terms and conditions of any Regulatory Approval.

j. <u>Utilities</u> Licensee shall make arrangements and shall pay all charges for all utilities to be furnished on, in or to the Premises or to be used by Licensee, including, without limitation, gas, electrical, water, sewer, scavenger service and telecommunications services. Licensee shall pay all charges for said utilities, including charges for the connection and installation of the utilities.

Licensee shall be obligated, at its sole cost and expense, to repair and maintain in good operating condition all utilities located within the Premises and all utilities installed by Licensee (whether within or outside the Premises). If Licensee requests Port or City to perform such maintenance or repair, whether emergency or routine, either Port or DPT may, in their sole discretion, elect to do so, and Port or City shall charge Licensee for the cost of the work performed at the then prevailing standard rates, and Licensee agrees to pay said charges to Port or City promptly upon billing. Licensee shall pay for repair of utilities located outside the Premises (regardless of who installed the same) which are damaged by or adversely affected by Licensee's use of such utility and shall be responsible for all damages, liabilities and claims arising therefrom. The parties agree that any and all utility improvements shall become part of the realty and are not trade fixtures.

k. <u>Operations Plan</u>. DPT shall cause the Licensee to submit to the Port the Operations Plan elements identified below, pursuant to Section 14 of Appendix A to the Towing Appendix C – MOU between DPT and Port Page 13 of 18 Agreement. The Licensee shall be required to submit those proposed Operations Plan elements to the Port and DPT at the same time, and the Port and DPT will confer in a timely manner so that DPT may satisfy the time limits established in Towing Agreement Appendix A, Section 14, and DPT will forward Port requested revisions to Licensee, and DPT shall not accept an Operations Plan element that is inconsistent with this MOU or the License Agreement for Pier 70.

1. <u>Pollution Prevention Plan</u>: Providing for: (i) the proper storage, handling and disposal of hazardous materials and hazardous waste, including volumes and locations where such materials will be handled and stored; (ii) spill prevention and response, including prevention measures and equipment and training to respond to spills; (iii) wastewater management practices; (iv) solid waste management plan, including disposal of regulated materials (e.g. tires) and recycling; (v) management practices for any authorized vehicle or equipment maintenance or other activities posing a potential for environmental impact; and (vi) related administrative controls. This Plan should include information related to best management practices employed by the Licensee related to any of the activities described above.

2. <u>Utilities Maintenance Plan</u>

3. <u>Pier 70 Premises Maintenance Plan</u>: Providing for the maintenance of the surface of the Premises, including the initial seal coating of paved areas, supervised video or photo documentation of initial surface conditions, ongoing inspection, spill and drip response procedures, maintenance of cracks and other identified deficiencies, re-application of seal coat, staff training protocols, and supervised video or photo documentation of exit surface conditions. This plan shall include other property management protocols, including but not limited to, maintenance of fencing, lighting, signage and buildings, permanent or temporary.

4. <u>Facility Plan</u>: Providing for the Contractor's conduct of a clean, wellmaintained and efficient business operations program to minimize or avoid external effects on surrounding streets and parking supply, property, businesses and residents in the neighborhood. This Plan should include information provided to employees, tow truck operators and auction attendees to educate them on these issues and recommended actions to appropriately address them.

5. <u>Public Auction Plan</u>: Describing auction procedures designed to minimize or avoid external effects on surrounding streets, property, businesses and neighbors, including a parking plan for auction participants designed to reduce parking impacts of auction events, and information to be provided to all auction attendees to inform them of prohibitions on vehicle abandonment. This Plan shall address community concerns and provide for communication with community representatives in designing signage and information provided to educate customers and other site visitors of local safety, parking and neighborhood concerns. Licensee shall commit to facilitate a positive relationship with the community by offering to meeting regularly or as needed with neighborhood representatives.

15. <u>Notices</u>. All notices, demand, consents or approvals which are or may be required to be given by either party to the other under this MOU shall be in writing and shall be deemed to have been fully given when delivered in person to such representatives of Port and DPT as shall from time to time

Appendix C – MOU between DPT and Port Page 14 of 18 be designated by the parties for the receipt of notice, or when deposited in the United States mail, postage prepaid, and addressed, if to Port to:

Director of Real Estate Port of San Francisco Pier 1 San Francisco, CA 94111 (415) 274-0578 (415) 274-0510

FAX No: Telephone No:

And if to DPT to:

Department of Parking and Traffic City and County of San Francisco Attention: Steve Bell 25 Van Ness Avenue, Suite 230 San Francisco, CA 94102 (415) 554-9825 (415) 252-3272

Telephone No: Fax:

16. <u>Successors and Assigns</u>. The covenants and conditions contained herein shall inure and bind the heirs, successors, executors and assigns of the Port, DPT and the Licensee.

17. <u>Severability</u>. The invalidity or unenforceability of a particular provision of this MOU shall not affect the other provisions hereof.

Appendix C – MOU between DPT and Port Page 15 of 18

IN WITNESS WHEREOF, the parties have caused this agreement to be executed as of the date first written above.

AGREED TO AS WRITTEN ABOVE: SAN FRANCISCO PORT COMMISSION

MONIQUE MOYER

Port of San Francisco

Executive Director

AGREED TO AS WRITTEN ABOVE: **DEPARTMENT OF PARKING & TRAFFIC**

Recommended By:

By:

BOND YEE Acting Director

Authorized for execution by Port Commission Resolution Number:

Date:

By:

Date:

Approved:

By:

MICHAEL T. BURNS Director of Transportation Municipal Transportation Agency

Date:_____

Municipal Transportation Agency Board of Directors Resolution No.

Adopted:

Attest: Secretary, MTA Board

Appendix C – MOU between DPT and Port Page 16 of 18

REVIEWED: **DENNIS J. HERRERA**, City Attorney

By:

Deputy City Attorney

AMENDED AND RESTATED SERVICE AGREEMENT AND PROPERTY USE LICENSE

FOR TOWING, STORAGE AND DISPOSAL

OF ABANDONED AND ILLEGALLY PARKED VEHICLES

by and between the

CITY AND COUNTY OF SAN FRANCISCO

AND

TEGSCO, LLC

d.b.a. SAN FRANCISCO AUTORETURN, a California Limited Liability Company

Dated: July 31, 2010

1.	DE	FINITIONS
1.	.1	ABANDONED VEHICLE
1.	2	ADMINISTRATIVE HOLD
1.	3	AGREEMENT
1.	4	AUTOMATED DISPATCH SYSTEM (ADS)
1.	.5	CENTRAL DISPATCH
1.	6	CITATION
i.	7	C TTY
1.	8	CLAIM
1.9	9	COMPLAINT
1,	10	CONTROLLER
1.	11	CITATION AND TOWING MANAGEMENT SYSTEM (CTMS)
1.	12	CUSTOMER
1.	13	CUSTOMER SERVICE CENTER
1.	14	COURTESY TOWS
1.	15	DAYS
1.	16	DEFICIENCY CLAIM
1.	17	DELINQUENT CITATION
1.	18	DESIGNATED FACILITIES
1.2	19	DIRECTOR OF TRANSPORTATION
1.2	20	DISPATCH TOWS
1.2	21	DMV
1.2	22	DPH
1.2	23	EFFECTIVE DATE
1,2	24	EMPLOYEES
1.2	25	Expedited Tow4
1.2	26	GROSS REVENUES
1.2	27	HEADQUARTERS OFFICE
1.2	28 ⁻	INVESTIGATIVE POLICE HOLD
1.2	29	LETTER OF CREDIT
1.3	30	LIEN 1 VEHICLE
1.3	31	LIEN 2 VEHICLE
1.3	32	LIEN 3 VEHICLE
1.3	33	LIEN CATEGORY
1.3	34	MANDATORY FEE

TABLE OF CONTENTS

1.35	MANDATORY FEE
1,36	OCCURRENCE
1.37	OPERATIONS PLAN
1.38	PEAK HOURS
1.39	PEAK HOURS
1.40	PEO CONTRACT
1,41	PEO CONTRACTOR
1.42	PEO CONTRACTOR
1.43	POLICE HOLD VEHICLE
1.44	Port
1.45	PORT MOU
1.46	PRIMARY STORAGE FACILITY
1.47	Ргојест 206
1.48	PROPERTY, PROPERTIES
1.49	Proposal
1,50	RECORDS
1.51	REFERRAL FEE
1.52	RFP
1.53	REGIONAL SWEEPS
1.54	ROUTINE TOWS
1.55	Scheduled Tows
1,56	Scofflaw
1.57	SECONDARY STORAGE FACILITY
1.58	MTA
1.59	SFPD
1.60	SFPD TRAFFIC OFFENDER FEE
1,61	SPECIAL EVENTS TOWS
1.62	Term
1.63	Tow Car
1.64	TOW EQUIPMENT
1.65	Tow Request
1.66	TOWED VEHICLE MANAGEMENT SYSTEM (TVMS)7
1.67	TRAFFIC CODE
1.68	TREASURER
1.69	UNAVOIDABLE DELAYS
1 70	VEHICLE CODE 8

• 1.71	VIN	8
1.72	WAN	
1.73	ZONE TOWS	8
	ERM OF THE AGREEMENT	8
3. EF	FECTIVE DATE OF AGREEMENT	8
4. SE	RVICES CONTRACTOR AGREES TO PERFORM	8
5. SU	JBMITTING FALSE CLAIMS; MONETARY PENALTIES	9
6. TA	XES AND ASSESSMENTS	9
7. PA	AYMENT DOES NOT IMPLY ACCEPTANCE OF WORK	
8. QT	UALIFIED PERSONNEL	
8.1	CONTRACTOR IS THE EMPLOYER OF THE EMPLOYEES	
8.2	PEO CONTRACT SHALL NOT CONTROL OR MODIFY THE AGREEMENT	
8.3	PEO CONTRACT IS A SUBCONTRACT TO THE AGREEMENT	
8.4	CONTRACTOR IS LIABLE FOR ALL ACTIONS OF PEO CONTRACTOR	11
8.5	CITY IS NOT A GENERAL OR SPECIAL EMPLOYER TO THE EMPLOYEES	
8.6	PREVAILING WAGES	•
9. RE	ESPONSIBILITY FOR EQUIPMENT	11
10.]	INDEPENDENT CONTRACTOR; PAYMENT OF TAXES AND OTHER EXPENSE	
10.1	INDEPENDENT CONTRACTOR	11
10.2	PAYMENT OF TAXES AND OTHER EXPENSES	
10.3	CONTRACTOR'S EMPLOYEES	
10.4	PAYROLL AND TAXATION	
	REQUIRED INSURANCE	
	FINANCIAL ASSURANCES	
12.1	REQUIREMENT TO PROVIDE FINANCIAL GUARANTEES	
12.2	LETTER OF CREDIT AND PERFORMANCE BOND	
12.3	MAINTENANCE DEPOSIT	
12.4	CLAIMS FUND	
12.5	ENVIRONMENTAL OVERSIGHT DEPOSIT	
	INSURANCE AND PERFORMANCE GUARANTEE REQUIREMENTS	
	CONTRACTOR'S REPRESENTATIONS AND WARRANTIES	
14.1		
14.2	FORMATION	
14.3	AUTHORITY	
14.4	CONFLICTS AND CONSENTS	
14.5	CONFLICT WITH ORDERS	19

AMENDED AND RESTATED AGREEMENT n:\ptc1\as2010\0900371\00637281.doc Page iii of v

14.	5 LITIGATION	19
15.	INDEMNIFICATION	19
16.	INCIDENTAL AND CONSEQUENTIAL DAMAGES	20
17.	LIABILITY OF CITY	
18.	DEFAULT; REMEDIES	20
19.	TERMINATION	21
19.1	1 TERMINATION EFFECTIVE UPON NOTICE	22
19.2	2 TERMINATION NOT EXCLUSIVE REMEDY	22
19.3	3 DUTIES UPON TERMINATION	22
20.	PROVISIONS OF AGREEMENT SURVIVING TERMINATION OR EXPIRATION	22
21.	CONFLICT OF INTEREST	22
22.	PROPRIETARY OR CONFIDENTIAL INFORMATION	22
23.	NOTICES TO THE PARTIES	23
24.	OWNERSHIP OF RESULTS	24
25.	WORKS FOR HIRE	24
26.	AUDIT AND INSPECTION OF RECORDS	24
26.	1 Records	24
26.2	2 CITY'S RIGHT TO INSPECT	24
26.3	3 AUDIT	24
26.4	4 RETENTION OF RECORDS	25
27.	SUBCONTRACTING	25
28.	ASSIGNMENT	25
29.	NON-WAIVER OF RIGHTS	25
30.	EARNED INCOME CREDIT (EIC) FORMS	25
31.	MINORITY/WOMEN/LOCAL BUSINESS UTILIZATION; LIQUIDATED DAMAGE	S.26
31.		
31.2	2 ENFORCEMENT	26
31.	·	26
31.4		
31.:	5 PAYMENT OF SUBCONTRACTORS	27
32.	NONDISCRIMINATION; PENALTIES	
32.	1 CONTRACTOR SHALL NOT DISCRIMINATE	27
32.2	2 NONDISCRIMINATION IN BENEFITS	27
32.	3 CONDITION TO CONTRACT	27
32.4	4 INCORPORATION OF ADMINISTRATIVE CODE PROVISIONS BY REFERENCE	27
33,	MACBRIDE PRINCIPLES-NORTHERN IRELAND	27

34.	TROPICAL HARDWOOD AND VIRGIN REDWOOD BAN2	8
35. ·	DRUG-FREE WORKPLACE POLICY	8
36.	RESOURCE CONSERVATION	8
37.	COMPLIANCE WITH AMERICANS WITH DISABILITIES ACT2	8
38.	SUNSHINE ORDINANCE	8
3 8.]	DOCUMENTS SUBJECT TO DISCLOSURE BY THE CITY	8
38.2		
38.3	3 PUBLIC RECORDS REQUESTS FROM PUBLIC	9
39.	NOTIFICATION OF LIMITATIONS ON CONTRIBUTIONS	9
40.	REQUIRING MINIMUM COMPENSATION FOR COVERED EMPLOYEES	9
41.	REQUIRING HEALTH BENEFITS FOR COVERED EMPLOYEES	1
42.	FIRST SOURCE HIRING PROGRAM	
43.	PROHIBITION ON POLITICAL ACTIVITY WITH CITY FUNDS	6
44.	PRESERVATIVE-TREATED WOOD CONTAINING ARSENIC	6
45.	MODIFICATION OF AGREEMENT	6
46.	ADMINISTRATIVE REMEDY FOR AGREEMENT INTERPRETATION	6
47.	AGREEMENT MADE IN CALIFORNIA; VENUE	7.
48.	CONSTRUCTION	
48. 48.1	CONSTRUCTION	7
	CONSTRUCTION	7 7 7
48.1	CONSTRUCTION	7 7 7
48.1 48.2	CONSTRUCTION	7 7 7 7
48.1 48.2 49.	CONSTRUCTION	7 7 7 7
48.1 48.2 49. 50.	CONSTRUCTION	7 7 7 7 8 8
48.1 48.2 49. 50. 51.	CONSTRUCTION	7 7 7 7 8 8
48.1 48.2 49. 50. 51. 52.	CONSTRUCTION	7 7 7 7 8 8 8
48.1 48.2 49. 50. 51. 52. 53.	CONSTRUCTION	7 7 7 7 8 8 8 8
48.1 48.2 49. 50. 51. 52. 53. 54.	CONSTRUCTION3'SYNTAX3'References3'ENTIRE AGREEMENT3'COMPLIANCE WITH LAWS3'SERVICES PROVIDED BY ATTORNEYS3'UNAVOIDABLE DELAYS3'SEVERABILITY3'TRANSITION PERIOD3'	7 7 7 7 8 8 8 8 8
48.1 48.2 49. 50. 51. 52. 53. 54. 55.	CONSTRUCTION3'SYNTAX3'References3'ENTIRE AGREEMENT3'COMPLIANCE WITH LAWS3'SERVICES PROVIDED BY ATTORNEYS3'SERVICES PROVIDED BY ATTORNEYS3'SEVERABILITY3'SEVERABILITY3'ADMINISTRATION OF CONTRACT3'	7 7 7 7 8 8 8 8 8 8 9
48.1 48.2 49. 50. 51. 52. 53. 54. 55. 56.	CONSTRUCTION3'SYNTAX3'References3'ENTIRE AGREEMENT3'COMPLIANCE WITH LAWS3'SERVICES PROVIDED BY ATTORNEYS3'SERVICES PROVIDED BY ATTORNEYS3'SEVERABILITY3'SEVERABILITY3'READINISTRATION OF CONTRACT3'REQUIRED ACTIONS3'	7 7 7 7 8 8 8 8 8 9 9
48.1 48.2 49. 50. 51. 52. 53. 54. 55. 56. 57.	CONSTRUCTION3'SYNTAX3'REFERENCES3'ENTIRE AGREEMENT3'COMPLIANCE WITH LAWS3'SERVICES PROVIDED BY ATTORNEYS3'SERVICES PROVIDED BY ATTORNEYS3'SEVERABILITY3'SEVERABILITY3'SEVERABILITY3'REQUIRED ACTIONS3'NON-IMPAIRMENT OF CITY-OWNED TOWING EQUIPMENT AND FACILITIES3'	7 7 7 7 8 8 8 8 8 9 9 9
48.1 48.2 49. 50. 51. 52. 53. 54. 55. 56. 57. 58.	CONSTRUCTION3'SYNTAX3'References3'ENTIRE AGREEMENT3'COMPLIANCE WITH LAWS3'SERVICES PROVIDED BY ATTORNEYS3'SERVICES PROVIDED BY ATTORNEYS3'SEVERABILITY3'SEVERABILITY3'REQUIRED ACTIONS3'NON-IMPAIRMENT OF CITY-OWNED TOWING EQUIPMENT AND FACILITIES3'DELETED BY AGREEMENT OF PARTIES3'	7 7 7 7 8 8 8 8 8 9 9 9 9
48.1 48.2 49. 50. 51. 52. 53. 54. 55. 56. 57. 58. 59.	CONSTRUCTION3'SYNTAX3'REFERENCES3'ENTIRE AGREEMENT3'COMPLIANCE WITH LAWS3'SERVICES PROVIDED BY ATTORNEYS3'SERVICES PROVIDED BY ATTORNEYS3'SEVERABILITY3'SEVERABILITY3'SEVERABILITY3'REQUIRED ACTIONS3'NON-IMPAIRMENT OF CITY-OWNED TOWING EQUIPMENT AND FACILITIES3'PROTECTION OF PRIVATE INFORMATION3'	7 7 7 7 8 8 8 8 8 8 9 9 9 9 9 9 9

AMENDED AND RESTATED AGREEMENT AND PROPERTY USE LICENSE FOR TOWING, STORAGE, AND DISPOSAL OF ABANDONED AND ILLEGALLY PARKED VEHICLES BY AND BETWEEN THE CITY AND COUNTY OF SAN FRANCISCO AND TEGSCO, LLC, D.B.A. SAN FRANCISCO AUTORETURN

This Amended and Restated Service Agreement And Property Use License for Towing, Storage, and Disposal of Abandoned and Illegally Parked Vehicles ("Amended and Restated Agreement"), dated for convenience as July 31, 2010, is entered into by and between the City and County of San Francisco, a municipal corporation, acting by and through its Municipal Transportation Agency, hereinafter referred to as "City", and TEGSCO, LLC, a California limited liability company doing business as San Francisco AutoReturn, hereinafter referred to as "Contractor", in the City and County of San Francisco, State of California, for the services and under the terms described herein.

Recitals

WHEREAS, The City issued a Request for Proposals for Towing, Storage and Disposal of Abandoned and Illegally Parked Vehicles on September 18, 2002 ("RFP"); and

WHEREAS, Contractor submitted a proposal dated March 10, 2003 ("Proposal") which was selected by City as the highest-ranked proposal among the proposals submitted in response to the RFP; and

WHEREAS, In the course of negotiating a Service Amended and Restated Agreement And Property Use License For Towing, Storage and Disposal of Abandoned and Illegally Parked Vehicles the City's existing towing services contractor, The City Tow, informed the City on January 23, 2004 that, it would cease to provide towing services as of March 21, 2004; and

WHEREAS, The cessation of towing services under the then-existing contract with The City Tow created emergency circumstances that required the City and Contractor, with the approval of the San Francisco Board of Supervisors, to enter into an Emergency Interim Service Amended and Restated Agreement and Property Use License for Towing, Storage and Disposal of Abandoned and Illegally Parked Vehicles ("Emergency Interim Agreement") pending negotiation of the long-term contract pursuant to the RFP, and Contractor began providing towing services to the City on March 22, 2004; and

WHEREAS, The City and Contractor entered into the Service Agreement and Property Use License for Towing, Storage and Disposal of Abandoned and Illegally Parked Vehicles (the "Original Agreement"), which had a five-year term ending on July 31, 2010 and an option to extend; and

WHEREAS, On June 30, 2007 the City and Contractor entered into an amendment to the Original Agreement (the "First Amendment"), which revised the rate schedule for towing, impounding, lien, and administrative towing fees; and

WHEREAS, the City and Contractor now wish to amend the Original Agreement as amended by the First Amendment to: (1) extend the Original Agreement for five years, until July 31, 2015, (2) revise the rates to be charged for towing and storage of vehicles under the Original Agreement, as amended by the First Amendment, (3) require Contractor to contribute to an audit fund and a fund to be used by City to hire an employee to monitor Contractor's performance, (4) update the insurance and performance guarantees required by the Original Agreement, (5) make changes to the procedures that

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 1 of 42 Contractor must follow under the Original Agreement; and 6) amend other provisions of the Original Agreement as amended, by the First Amendment; and

WHEREAS, For simplicity of reference and ease of use by the parties, the parties wish to amend and restate the Original Agreement as amended by the First Amendment by means of this Amended and Restated Service Agreement and Property Use License for Towing, Storage and Disposal of Abandoned and Illegally Parked Vehicles; and

WHEREAS, The parties acknowledge and agree that this Amended and Restated Agreement amends and restates the Original Agreement as amended by the First Amendment, in its entirety, contains the entire understanding of the parties, and thereby supersedes and replaces the Original Agreement as amended by the First Amendment;

NOW, THEREFORE, based on the foregoing, and in consideration of the promises, covenants, and undertakings contained herein, the City and Contractor hereby agree as follows:

AMENDED AND RESTATED AGREEMENT

1. **Definitions**

The following words and phrases shall have the meanings set forth below when used in this Amended and Restated Agreement:

1.1 <u>Abandoned Vehicle</u>: A vehicle that qualifies to be towed pursuant to San Francisco Traffic Code § 159, or successor section of the San Francisco Municipal Code, because it is parked or left standing on a public street or highway in the City for more than 72 consecutive hours, or is deemed to be abandoned pursuant to Vehicle Code § 22669(a), or is found to be a public nuisance on private property pursuant to Vehicle Code § 22660 and Traffic Code § 234, or successor statutes or local laws..

1.2 <u>Administrative Hold</u>: A hold placed on a vehicle impounded by SFMTA or SFPD pursuant to which the vehicle may not be released prior to the passage of a specified period of time, fulfillment of statutory requirements or upon written authorization by the impounding agency. The procedures and requirements for Administrative Holds are further described in Section 4.7 of Appendix A to this Amended and Restated Agreement.

1.3 <u>Agreement</u>: This Amended and Restated Agreement, its Appendices, Exhibits and Attachments to Appendices, the Operations Plan as adopted in accordance with Section 14 of Appendix A, the RFP, the Proposal, and other documents attached hereto or specifically incorporated herein by reference.

1.4 <u>Automated Dispatch System (ADS)</u>: Contractor's automated dispatch management system that is GPS-enabled and provides electronic dispatch communications.

1.5 <u>Central Dispatch</u>: Contractor's primary facility for taking Tow Requests and dispatching Tow Cars, as further defined in Section 11.1(d) of Appendix A.

1.6 Citation: A ticket for a parking violation processed by SFMTA.

1.7 <u>City</u>: The City and County of San Francisco, acting by and through its Municipal Transportation Agency.

1.8 <u>Claim</u>: Any claim brought against the City or Contractor or their respective agents, contractors or employees for theft of property or injury to any property or person arising out of Contractor's performance of services under this Agreement. For the purpose of Appendix A, § 5.2, "Claim" shall mean a request for compensation for personal injury, loss from or damage to towed

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 2 of 42
vehicle and/or personal property, costs of recovering a vehicle that was not dropped in violation of Appendix A, Section 4.2, or the value of a vehicle that was lost in violation of Section 4.1 of Appendix A.

1.9 <u>Complaint</u>: Any problem with service provided by Contractor within the scope of this Agreement that is communicated to Contractor or City by a Customer and which is not a Claim. This does not include issues reported to the Contractor by Customers that the Contractor is able to resolve to the satisfaction of the Customer through internal issue resolution procedures.

1.10 <u>Controller</u>: The Controller of the City and County of San Francisco or any duly authorized agent thereof.

1.11 <u>Citation Management System (CMS)</u>: SFMTA's Ticket Information Management System (TIMS) or successor system(s) used by SFMTA for tracking Tow Requests and Citations.

1.12 <u>Customer</u>: A member of the public who is associated with an impounded vehicle, which may include the registered owner, the driver of the vehicle at the time that it is stopped or towed, a person who appears on behalf of the owner or the driver of the vehicle, or a purchaser of a vehicle at lien sale auction.

1.13 <u>Customer Service Center</u> The location for Customers recovering vehicles in person to pay fees and process required forms for release of a vehicle, as further described in Appendix A Section 11.1(a) and located at 450 7th Street, San Francisco.

1.14 <u>Courtesy Tows</u>: Tows requested for disabled vehicles owned or used in an official capacity by either SFPD or SFMTA.

1.15 <u>Days</u>: Consecutive calendar days, including weekends and holidays, unless otherwise specified.

1.16 Deficiency Claim: A claim by Contractor against a registered vehicle owner equal to towing and storage charges, less any amount received from the sale of the vehicle, and which is subject to all rights and limitations set forth in California Civil Code § 3068.2 or any successor statute that creates, defines and limits Contractor's right to such claim.

1.17 <u>Delinquent Citation</u>: A Citation that was unpaid past the original due date for payment, upon which penalties for overdue payment have accrued, and which is not scheduled for administrative review or hearing by SFMTA.

1.18 <u>Designated Facilities</u>: Real property and any buildings and improvements thereon used by Contractor in the performance of this Agreement, as further described in Appendix A, Section 11.1.

1.19 <u>Director of Transportation</u>: The Chief Executive Officer of the Municipal Transportation Agency of the City and County of San Francisco.

1.20 <u>Dispatch Tows</u>: Individual Tow Requests that are communicated to Contractor from the Enforcement tow desk or other dispatch center operated by the City.

1.21 DMV: California Department of Motor Vehicles.

1.22 DPH: Department of Public Health of the City and County of San Francisco.

1.23 <u>Effective Date</u>: July 31, 2010, or the date upon which all required approvals are obtained and all signatures of the parties have been affixed hereto, whichever date is later.

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 3 of 42 **1.24** <u>Employees:</u> The persons directed and controlled by Contractor and/or its PEO Contractor who are providing work and services on behalf of Contractor in order to perform the work required under this Agreement, not including entities that are independent subcontractors. The classes and job descriptions of persons who are Employees of Contractor may include, but are not limited to:

- (a) Cashiers, bookkeepers, and accountants.
- (b) Persons who move vehicles at and between Designated Facilities (also known as "Storage Specialists").
- (c) Forklift and heavy equipment operators.
- (d) Clerks, secretaries, telephone operators and administrative and information technology/computer support personnel.
- (e) Tow Car operators directly employed by Contractor and/or its PEO Contractor.
- (f) Guards and security personnel.

1.25 Expedited Tow: A tow needed to ensure public or officer safety and/or to eliminate a hazard.

1.26 <u>Gross Revenues</u>: Contractor's gross receipts from all fees and proceeds under this Agreement, including all revenues derived from lien sales conducted pursuant to this Agreement. "Gross Revenues" do not include SFMTA or SFPD Traffic Offender Fees or fees payable by vehicle owners or operators collected by Contractor on behalf of the City pursuant to the terms of this Agreement.

1.27 <u>Headquarters Office</u>: Contractor's administrative office, currently located at 945 Bryant Street, Suite 350 San Francisco, California.

1.28 Investigative Police Hold: A Police Hold imposed on an evidentiary vehicle for the purpose of criminal investigation, as further described in Appendix A, Section 4.6.

1.29 Letter of Credit: The letter of credit which Contractor is required to maintain to guarantee the performance of Contractor's obligations under this Agreement, as further described in Section 12.1 and 12.2 and in Appendix C, Section 6 of this Agreement. The Letter of Credit and all replacement Letters of Credit provided by Contractor during the Term of this Agreement shall be attached hereto as Appendix E and is incorporated by reference as though fully set forth herein.

1.30 Lien 1 Vehicle: A low-value vehicle, including a vehicle valued at five hundred dollars (\$500) or less, in accordance with Vehicle Code § 22670 (requiring valuation of any vehicle towed by a public agency) and § 22851.2 (regarding vehicles valued at an amount not exceeding five hundred dollars (\$500) on less pursuant to towed for being abandoned) or a vehicle valued at five hundred dollars (\$500) or less pursuant to § 22851.3 (regarding vehicles towed for being abandoned) and §§ 22851.6 - 22851.10 (regarding disposal procedures for low-value vehicles). If the Vehicle Code is amended subsequent to the Effective Date to change the dollar amounts that trigger requirements for low-value vehicles, this Agreement shall incorporate such amendments by reference as though fully set forth herein for the purpose of defining dollar-value thresholds and legally required procedures for handling and disposal of low-value vehicles.

1.31 Lien 2 Vehicle: A medium-value vehicle valued at more than five hundred dollars (\$500) and up to and including four thousand dollars (\$4,000) in accordance with Vehicle Code § 22670 (requiring valuation of any vehicle towed by a public agency), or over five hundred dollars (\$500) and up to and including four thousand dollars (\$4,000) for the purpose of Vehicle Code (\$500) and up to and including four thousand dollars (\$4,000) for the purpose of Vehicle Code AMENDED AND RESTATED AGREEMENT

§ 22851.3 (regarding vehicles towed for being abandoned), and California Civil Code §§ 3067-3075 (setting forth legally required procedures for lien sales of towed vehicles). If the Vehicle Code or the California Civil Code are amended subsequent to the Effective Date to change the dollar amounts which trigger requirements for medium-value vehicles, this Agreement shall incorporate such amendments by reference as though fully set forth herein for the purpose of defining dollar-value thresholds and legally required procedures for handling and disposal of medium-value vehicles.

1.32 Lien 3 Vehicle: A high-value vehicle valued at more than four thousand dollars (\$4,000), in accordance with Vehicle Code § 22670, requiring valuation of any vehicle towed by a public agency, and California Civil Code Sections 3067-3075, setting forth required procedures for lien sales of vehicles. If the Vehicle Code or the California Civil Code are amended subsequent to the Effective Date to change the dollar amounts which trigger requirements for high-value vehicles, this Agreement shall incorporate such amendments by reference as though fully set forth herein for the purpose of defining dollar-value thresholds and legally required procedures for handling and disposal of high-value vehicles.

1.33 <u>Lien Category</u>: The classification of a vehicle as a Lien 1, Lien 2 or Lien 3 vehicle in accordance with its appraised value.

1.34 <u>Mandatory Fee</u>: All fees that must be paid by Customer before Contractor may release any vehicle to the owner or purchaser of a vehicle. Mandatory Fees include: boot fees, SFMTA Administrative Fee, SFPD Traffic Offender Fee, towing, storage, transfer and lien fees, returned check charges and Delinquent Citation fees, as applicable.

1.35 <u>Monthly Finance Report</u>: Monthly report that summarizes administrative and referral fees paid by the Contractor to the City and waiver activity.

1.36 <u>Occurrence</u>: Any accident or incident occurring in a single place at a single time from a single event that results in one or more claims for injury to persons or property. An accident or incident that results in multiple claims shall be considered a single Occurrence for purposes of applying any deductible provisions of Contractor's insurance coverages.

1.37 Operations Plan: The collected written procedures manuals, lists and schedules required to be submitted by Contractor and approved by City as set forth in Appendix A, Section 14, and which are listed in Appendix B to this Agreement. When approved by City, the Operations Plan shall define the service standards for the work to be performed under this Agreement, and is hereby incorporated into this Agreement as though fully set forth herein.

1.38 <u>Peak Service Hours</u>: Monday through Friday, 7:00 a.m. to 8:00 p.m., excluding City holidays.

1.39 <u>Peak Towing Hours:</u> Monday through Friday, 7:00 a.m. to 9:00 a.m. and 3:00 p.m. to 7:00 p.m., excluding City holidays.

1.40 <u>PEO Contract</u>: Any agreement with any professional employment organization (PEO) which provides payroll, workers' compensation, work place safety, staffing and other human resources services, for the purpose of performing the services required by this Agreement. "PEO Contract" does not include contracts with subcontractors for the provision of Tow Cars, Tow Equipment and related towing services.

1.41 <u>PEO Contractor</u>: An entity with whom Contractor has entered into a PEO Contract.</u>

1.42 <u>Performance Bond</u>: The Performance Bond which Contractor is required to maintain to guarantee the performance of Contractor's obligations under this Agreement, as further AMENDED AND RESTATED AGREEMENT described in Section 12.1 and 12.2 and in Appendix C, Section 6 of this Agreement. The Performance Bond and all replacement Performance Bonds provided by Contractor during the Term of this Agreement shall be attached hereto as Appendix F and incorporated by reference as though fully set forth herein.

1.43 <u>Police Hold</u>: A hold placed on a vehicle by the SFPD in writing which requires a vehicle to be processed in accordance with the Police Hold procedures specified in this Agreement, or a vehicle with no visible VIN which is held for inspection by the SFPD.

1.44 Port: The San Francisco Port Commission, an agency of the City and County of San Francisco.

1.45 <u>Port MOU</u>: That certain Memorandum of Understanding by and between SFMTA and the Port, a copy of which is attached hereto as Appendix C and incorporated by reference as though fully set forth herein.

1.46 <u>Primary Storage Facility</u>: The facilities primarily used for short term storage of impounded vehicles currently located at 450 7th Street, San Francisco, California, as further described in Section 11.1(b) of Appendix A.

1.47 <u>Project 20</u>: The SFMTA program which allows vehicle owners and operators to perform community service in lieu of paying Citation fees, in accordance with Vehicle Code § 40215(c)(7).

1.48 Property, Properties: Real property owned by the City and licensed to Contractor for the purpose of this Agreement, Pier 70 as of the Effective Date of this Agreement, and any other properties that may be licensed to Contractor by City for the purpose of this Agreement during its Term.

1.49 Proposal: The proposal submitted by Contractor in response to the RFP, dated March 10, 2003, incorporated by reference into this Agreement as though fully set forth herein.

1.50 <u>Records</u>: The documents Contractor is required to create and maintain under this Agreement, including but not limited to: (1) complete and accurate books, accounts and documentation of financial transactions relating to all items of income received and expenses incurred in the performance of this Agreement; (2) documentation of all vehicles towed; (3) documentation of all vehicles stored; (4) documentation of all Claims; (5) all Monthly Management Reports and other reports Contractor is required to submit to City; (6) charts and diagrams of any property licensed to Contractor by City to fulfill the obligations of this Agreement; (7) other documents or reports as City may require Contractor to produce in the course of performing work under the Agreement; and (8) the Records described in Section 13 of Appendix A to this Agreement.

1.51 <u>Referral Fee</u>: The fee established in Section 12.1(a) of Appendix A to this Agreement.

1.52 <u>**RFP**</u>: The City and County of San Francisco's Request for Proposals for Towing, Storage and Disposal of Abandoned and Illegally Parked Vehicles, dated September 18, 2002, and all addenda thereto, incorporated by reference into this Agreement as though fully set forth herein.

1.53 <u>Regional Sweeps</u>: Pre-arranged parking or traffic enforcement operations in designated areas. These tow requests are generally communicated directly by SFPD officers or SFMTA Parking Enforcement Officers to Tow Car operators who are deployed in the field and assigned to specific officer(s).

1.54 Routine Tows: Scheduled Tows and Dispatch Tows.

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 6 of 42 **1.55** <u>Scheduled Tows</u>: Pre-arranged tow events at times and in places designated by City, including but not limited to scheduled commuting hour ("towaway") tows, Zone Tows, Special Events Tows and Abandoned Vehicle tows. These tow requests are generally communicated directly by SFPD officers or SFMTA Parking Enforcement Officers to Tow Car operators who are deployed in the field and assigned to specific officer(s).

1.56 <u>Scofflaw Vehicle</u>: A vehicle towed for multiple Delinquent Citations pursuant to subsection (i) of Vehicle Code § 22651.

1.57 <u>Secondary Storage Facility</u>: The facility primarily used for long term storage of impounded vehicles, located at Pier 70, San Francisco, California as of the Effective Date of this Agreement, as further described in Appendix A, Section 11.1(c).

1.58 SFMTA: The San Francisco Municipal Transportation Agency an agency of the City and County of San Francisco

1.59 <u>SFPD</u>: The Police Department of the City and County of San Francisco or any duly authorized officer or member thereof.

1.60 SFPD Traffic Offender Fee: The fee described in Section 12.1(d) of Appendix A to this Agreement, as authorized by Traffic Code §§ 170.2-A and 170.2-B and Vehicle Code § 22850.5, or successor statutes or ordinances.

1.61 <u>Special Events Tows</u>: Tows of vehicles parked in violation of temporary parking restrictions, as authorized by Traffic Code §§ 33(c) and 130 or successor ordinances.

1.62 <u>Term</u>: The duration of this Agreement as established in Section 2 herein, and any additional period during which Contractor completes repair, remediation or other work required for the termination of the License agreement for Pier 70, as set forth in Appendices C and D of this Agreement, and the tasks listed in Section 54 of this Agreement for transition to a successor contractor or successor agreement.

1.63 <u>Tow Car</u>: A motor vehicle which has been altered or designed and equipped for and exclusively used in the business of towing vehicles by means of a crane, hoist, tow bar, tow line, or dolly or is otherwise exclusively used to render assistance to other vehicles, and as defined in Vehicle Code § 615.

1.64 <u>Tow Equipment</u>: Tow Cars and all appurtenant computer systems, communications devices, hand tools, electric tools and towing hardware, whether or not expressly listed in this Agreement, which are necessary to perform towing services to the standards of the towing industry and as set forth in this Agreement.

1.65 <u>Tow Request</u>: A request directed to Contractor from SFMTA, DPH or SFPD for service by a Tow Car or Tow Equipment for the removal or relocation of a vehicle.

1.66 <u>Towed Vehicle Management System (TVMS)</u>: The electronic database system to be used by Contractor to meet record keeping, reporting and vehicle handling requirements of this Agreement, as further described in Section 10 of Appendix A.

1.67 Traffic Code: The Traffic Code of the City and County of San Francisco.

1.68 Treasurer: The Treasurer of the City and County of San Francisco.

1.69 <u>Unavoidable Delays</u>: With respect to a delay in performance, "Unavoidable Delay" shall mean any delay that is attributable to any: (a) strike, lockout or other labor or industrial disturbance (whether or not on the part of the employees of either party hereto or their contractors),

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 7 of 42 civil disturbance, future order claiming jurisdiction, act of the public enemy, war, riot, sabotage, blockage, embargo, inability to secure customary materials, supplies or labor through ordinary sources by reason of regulation or order of any government or regulatory body; (b) changes in any applicable laws or the interpretation thereof; or (c) lightning, earthquake, fire, storm, hurricane, tornado, flood, washout, explosion; or (d) a governmental taking by eminent domain of all or a substantial portion of a Designated Facility; or (e) any interference with Contractor's right to access a substantial portion of a Designated Facility owned or provided for Contractor's use by City, or any other cause beyond the reasonable control of the party from whom performance is required, or any of its contractors or other representatives.

1.70 Vehicle Code: The Vehicle Code of the State of California.

1.71 <u>VIN</u>: The distinguishing number or other mark used for the purpose of uniquely identifying a vehicle or vehicle part, as further defined in Vehicle Code § 671.

1.72 WAN: Wide Area Network.

1.73 Zone Tows: Tows of vehicles parked in zones listed in Traffic Code § 71b, or successor ordinance.

GENERAL PROVISIONS

2. Term of the Agreement

The Term of this Agreement shall be five years, from the July 31, 2010 through July 31, 2015. The City agrees to submit a request for an increase in rates to the SFMTA Board to reflect any increased cost to Contractor for increased rental payments for the property at Pier 70 from August 1, 2010 through July 31, 2015 that is in excess of CPI adjustments to be applied during the terms of Appendices C and D.

3. Effective Date of Amended and Restated Agreement

This Amended and Restated Agreement shall commence upon the Effective Date.

4. Services Contractor Agrees to Perform

The Contractor shall tow and store any vehicles that SFMTA, DPH or the SFPD orders removed from any public street or highway or from private property within the City, in accordance with the requirements of the San Francisco Transportation Code and the Vehicle Code, and shall perform such other related services as are described in this Agreement, in accordance with Appendix A, "Scope of Work," and Appendix B "Operations Plan". Appendix A, Appendix B, and the Operations Plan to be adopted and amended as provided in Section 14 of Appendix A are all attached hereto and incorporated by reference as though fully set forth herein. Rates and charges to the public for services under this Agreement shall be as set forth in Appendix F, attached hereto and incorporated by reference as though fully set forth herein.

All vehicle handling and storage required by this Agreement shall be conducted at one of the Designated Facilities. As of the Effective Date of this Agreement, City hereby licenses to Contractor the use of one of the real properties on which the Designated Facilities are currently located -- Pier 70, San Francisco. The license agreement for this property is attached hereto as Appendix D and is incorporated by reference as though fully set forth herein. The parties acknowledge that Contractor's Customer Service Center and Primary Storage Facility operations are conducted at a property owned

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 8 of 42 by Caltrans and leased by Contractor -- 450 7th Street. Should City cease to use Contractor for towing services, the terms for the City's continued use of the 450 7th Street property are attached hereto as Appendix G and are incorporated by reference as though fully set forth herein.

City hereby consents to Contractor's towing and storage of vehicles that are not towed or stored pursuant to a Tow Request with the prior written approval of City and subject to any conditions imposed in such written approval; provided however, that (i) Contractor shall at all times conduct itself in accordance with the Customer service standards of this Agreement so as not create any negative effect on Contractor's public image and reputation as the City's towing contractor, and (ii) Contractor's operations with respect to such vehicles shall not create any adverse impact on its performance of all requirements of this Agreement. City may revoke its consent at any time without cause by written notice to Contractor. The following sections of this Agreement shall apply to any towing or storage of vehicles by Contractor within the City and County of San Francisco that are not towed or stored pursuant to a Tow Request: Service Agreement Sections 8, 9, 11, 12, 13, 14, 15, 16, 17, 20, 22, 26, 27, 28, 29, 31, 32, 35, 37, 38, 40, 41, 42, 50; Appendix A Sections 1, 3, 4, 5, 6, 7, 9, 11, 12.1(b), 12.3, 12.5, 12.6, 13, 14; Appendix B, Appendices C and D during their respective terms, and Appendices E and F.

5. Submitting False Claims; Monetary Penalties

Pursuant to San Francisco Administrative Code §21.35, any contractor, subcontractor or consultant who submits a false claim shall be liable to the City for the statutory penalties set forth in that section. The text of Section 21.35, along with the entire San Francisco Administrative Code is available on the web at http://www.municode.com/Library/clientCodePage.aspx?clientID=4201. A contractor, subcontractor or consultant will be deemed to have submitted a false claim to the City if the contractor, subcontractor or consultant: (a) knowingly presents or causes to be presented to an officer or employee of the City a false claim or request for payment or approval; (b) knowingly makes, uses, or causes to be made or used a false record or statement to get a false claim paid or approved by the City; (c) conspires to defraud the City by getting a false claim allowed or paid by the City; (d) knowingly makes, uses, or causes to be made or used a false record or statement to conceal, avoid, or decrease an obligation to pay or transmit money or property to the City; or (e) is a beneficiary of an inadvertent submission of a false claim to the City within a reasonable time after discovery of the false claim.

6. Taxes

Payment of any taxes, including possessory interest taxes and California sales and use taxes, levied upon or as a result of this Amended and Restated Agreement, or the services delivered pursuant hereto, shall be the obligation of Contractor. Contractor recognizes and understands that this Agreement may create a "possessory interest" for property tax purposes. Generally, such a possessory interest is not created unless the Agreement entitles the Contractor to possession, occupancy, or use of City property for private gain. If such a possessory interest is created, then the following shall apply:

(1) Contractor, on behalf of itself and any permitted successors and assigns, recognizes and understands that Contractor, and any permitted successors and assigns, may be subject to real property tax assessments on the possessory interest;

(2) Contractor, on behalf of itself and any permitted successors and assigns, recognizes and understands that the creation, extension, renewal, or assignment of this Amended and Restated Agreement may result in a "change in ownership" for purposes of real property taxes, and therefore may result in a revaluation of any possessory interest created by this Agreement. AMENDED AND RESTATED AGREEMENT Contractor accordingly agrees on behalf of itself and its permitted successors and assigns to report on behalf of the City to the County Assessor the information required by Revenue and Taxation Code section 480.5, as amended from time to time, and any successor provision.

(3) Contractor, on behalf of itself and any permitted successors and assigns, recognizes and understands that other events also may cause a change of ownership of the possessory interest and result in the revaluation of the possessory interest. (See, e.g., Rev. & Tax. Code section 64, as amended from time to time). Contractor accordingly agrees on behalf of itself and its permitted successors and assigns to report any change in ownership to the County Assessor, the State Board of Equalization or other public agency as required by law.

(4) Contractor further agrees to provide such other information as may be requested by the City to enable the City to comply with any reporting requirements for possessory interests that are imposed by applicable law.

7. Payment Does Not Imply Acceptance of Work

The acceptance of any payment by City and/or Contractor's deduction or offset of funds (that would otherwise be paid by City to Contractor as payment for services) from revenues generated under the Agreement shall in no way lessen the liability of Contractor to replace unsatisfactory work, equipment, or materials, although the unsatisfactory character of such work, equipment or materials may not have been apparent or detected at the time such payment was received. Work that does not conform to the requirements of this Agreement may be rejected by City and in such case Contractor must take immediate corrective action.

8. Qualified Personnel

Work under this Agreement shall be performed only by competent personnel under the supervision of and in the employment of Contractor, or through subcontracts entered into pursuant to Section 27. Contractor will comply with City's reasonable requests regarding assignment of personnel, including the removal of specified personnel upon request of City, but all personnel, including those assigned at City's request, must be supervised by Contractor. Contractor shall commit adequate resources to meet its obligations under this Amended and Restated Agreement.

8.1 Contractor Is the Employer of the Employees

Notwithstanding any language in any PEO Contract, and notwithstanding the intent of Contractor and/or any contractor for PEO Contract services in entering into a PEO Contract or as memorialized within a PEO Contract or as a PEO Contract may be construed by any person, Contractor affirms, agrees and warrants that for all purposes under the Agreement and under relevant law applicable to the Agreement, Contractor is the employer of the Employees. Contractor affirms, agrees and warrants that for all purposes of the Agreement, specifically including but not limited to Contractor's duties under Sections 4, 5, 8, 9, 10, 14, 15, 16, 18, 21, 22, 26, 30, 32, 35, 37, 40, 41, 42, 43, 49, 50 of this Agreement, and Section 8 of Appendix A to this Agreement, Contractor is the employer of the Employees.

8.2 PEO Contract Shall Not Control or Modify the Agreement

Notwithstanding any language in any PEO Contract, and notwithstanding the intent of Contractor and/or a PEO Contractor in entering into a PEO Contract or as memorialized within a PEO Contract or as a PEO Contract may be construed by any person, Contractor affirms, agrees and warrants that all provisions of this Agreement are in full force and effect, and that a PEO Contract does not modify Contractor's obligations and duties under this Agreement. Contractor further agrees, affirms and AMENDED AND RESTATED AGREEMENT

> n:\ptc1\as2010\0900371\00637281.doc Page 10 of 42

warrants that a PEO Contract shall have no effect upon the City's rights under this Agreement and a PEO Contract does not alter the City's right to enforce all provisions of this Agreement against Contractor directly.

8.3 PEO Contract is a Subcontract to the Agreement

Notwithstanding any language in any PEO Contract, and notwithstanding the intent of Contractor and/or a PEO Contractor in entering into a PEO Contract or as memorialized within a PEO Contract or as a PEO Contract may be construed by any person, Contractor affirms, agrees and warrants that for all purposes under this Agreement and under relevant law applicable to this Agreement, any PEO Contract is a subcontract for the purpose of providing payroll, workers' compensation, work place safety and other human resources services to Contractor.

8.4 Contractor is Liable for All Actions of PEO Contractor

Notwithstanding any language in any PEO Contract, and notwithstanding the intent of Contractor and/or a PEO Contractor in entering into a PEO Contract or as memorialized within a PEO Contract, or as any PEO Contract may be construed by any person, Contractor affirms, agrees and warrants that, Contractor shall be liable for all actions of its PEO Contractor insofar as said actions affect the City or Contractor's performance of its duties under the Agreement. For purposes of the Agreement, Contractor shall not assign and has not assigned any of its duties under the Agreement to its PEO Contractor.

8.5 City is not a General or Special Employer to the Employees

The parties agree no employment relationship exists between the City and the Employees. As to the Employees, the City is not a "General Employer" or "Special Employer", as those terms are defined under California law, for any purpose, including but not limited to application of California Labor Code §§ 6300 *et seq.* and California Insurance Code § 11663, or any other applicable statute. Should a court or administrative agency having jurisdiction over the issue determine that the City is a "General Employer" or "Special Employer" of the Employees, Contractor shall fully indemnify the City for all costs and liabilities arising from that finding, including but not limited to consequential and incidental damages, temporary disability indemnity, permanent disability indemnity, medical costs, penalties and fines, vocational rehabilitation costs, and attorneys fees. Contractor's duties and obligations under this section of the Agreement are material obligations guaranteed by the performance guarantee described in Section 12.

8.6 Prevailing Wages

Contractor hereby acknowledges that it has read completely and fully understands San Francisco Administrative Code Section 21.25-2 and agrees that this Agreement shall be subject to, and Contractor shall comply with, all applicable obligations and requirements imposed by that Section.

9. Responsibility for Equipment

City shall not be responsible for any damage to persons or property as a result of the use, misuse or failure of any equipment used by Contractor, or by any of its Employees, even though such equipment be furnished, rented or loaned to Contractor by City.

10. Independent Contractor; Payment of Taxes and Other Expenses

10.1 Independent Contractor. Contractor or any agent or Employee of Contractor shall be deemed at all times to be an independent contractor and is wholly responsible for the manner in which it performs the services and work requested by City under this Agreement. Contractor or any agent or Employee of Contractor shall not have employee status with City, nor be entitled to

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 11 of 42 participate in any plans, arrangements, or distributions by City pertaining to or in connection with any retirement, health or other benefits that City may offer its employees. Contractor or any agent or Employee of Contractor is liable for the acts and omissions of itself, its employees and its agents. Contractor shall be responsible for all obligations and payments, whether imposed by federal, state or local law, including, but not limited to, FICA, income tax withholdings, unemployment compensation, insurance, and other similar responsibilities related to Contractor's performing services and work, or any agent or Employee of Contractor providing same. Nothing in this Agreement shall be construed as creating an employment or agency relationship between City and Contractor or any agent or Employee of Contractor.

Any terms in this Agreement referring to direction from City shall be construed as providing for direction as to policy and the result of Contractor's work only, and not as to the means by which such a result is obtained. City does not retain the right to control the means or the method by which Contractor performs work under this Agreement.

10.2 Payment of Taxes and Other Expenses. Should City, in its discretion, or a relevant taxing authority such as the Internal Revenue Service or the State Employment Development Division, or both, determine that Contractor or any of its agents or Employees is an employee of the City for purposes of collection of any employment taxes, any amounts offset from revenues paid by Contractor to the City under this Agreement shall be reduced by amounts equal to both the employee and employer portions of the tax due (and offsetting any credits for amounts already paid by Contractor which can be applied against this liability). City shall then forward those amounts to the relevant taxing authority.

Should a relevant taxing authority determine a liability for past services performed by Contractor for City, upon notification of such fact by City, Contractor shall promptly remit such amount due to the City (again, offsetting any amounts already paid by Contractor that can be applied as a credit against such liability).

A determination of employment status pursuant to the preceding two paragraphs shall be solely for the purposes of the particular tax in question, and for all other purposes of this Agreement, Contractor shall not be considered an employee of City. Notwithstanding the foregoing, should any court, arbitrator, or administrative authority determine that Contractor is an employee for any other purpose, then Contractor agrees to a reduction in City's financial liability so that City's total expenses under this Agreement are not greater than they would have been had the court, arbitrator, or administrative authority determined that Contractor was not an employee.

10.3 Contractor's Employees. Contractor shall be solely responsible for all matters relating to payment of Contractor's Employees, including compliance with Social Security, withholding and payment of any and all federal, state and local personal income taxes, disability insurance, unemployment, and any other taxes for such persons, including any related assessments or contribution required by law and all other regulations governing City Health Service System, vacation, holiday, retirement or other programs, nor, in the event that City terminates this Agreement shall Contractor have recourse of rights of appeal under City's rules and regulations which are applicable to employees.

10.4 Payroll and Taxation. Contractor shall make or cause to be made all necessary payroll deductions for disability and unemployment insurance, social security, withholding taxes and other applicable taxes, and prepare, maintain and file or cause to be filed all necessary reports with respect to such taxes or deductions, and all other necessary statements and reports pertaining to labor employed in the performance of this Agreement.

11. Required Insurance

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 12 of 42 Subject to approval by the City's Risk Manager of the insurers and policy forms, Contractor shall place and maintain throughout the Term of this Agreement, and pay the cost thereof, the following insurance policies:

11.1 Comprehensive general liability insurance with limits not less than \$1,000,000 (such limit may be provided through a primary and excess policy) each Occurrence, combined single limit for bodily injury and property damage, or in such greater amount and limits as SFMTA may reasonably require from time to time, including coverage for contractual liability, personal injury, broadform property damage, products and completed operations, independent contractors (excluding towing and dismantling subcontractors), and mobile equipment.

11.2 Sudden and accidental pollution insurance with limits not less than \$1,000,000 for each occurrence.

11.3 Comprehensive business/commercial automobile liability insurance with limits not less than \$1,000,000 for each Occurrence combined single limit for bodily injury and property damage, including coverage for owned, non-owned and hired automobiles. If Contractor does not own or lease company vehicles that are subject to motor vehicle registration, then only non-owned and hired coverage is required.

11.4 Garage-keeper's legal liability insurance with limits not less than \$5,000,000 (such limit may be provided through a primary and excess policy) for each Occurrence combined single limit for loss and damage to vehicles in Contractor's care, custody or control caused by fire, explosion, theft, riot, civil commotion, malicious mischief, vandalism or collision, with any deductible not to exceed \$25,000 for each Occurrence. Contractor may insure or self-insure loss of non-automobile property in the care, custody, or control of the garage keeper with a limit of \$5000.

11.5 Workers' Compensation Insurance, including Employers' Liability, with limits not less than \$1,000,000 for each accident, covering all Employees employed by Contractor in the performance of this Agreement to provide statutory benefits as required by the laws of the State of California. Said policy shall be endorsed to provide that the insurer waives all rights of subrogation against the City.

11.6 Environmental impairment liability insurance with limits not less than \$1,000,000 each occurrence, covering the sudden and accidental release of hazardous materials and the resulting costs of clean up.

Except as set forth above, any deductibles in the policies listed above shall not exceed \$25,000 each occurrence. The insurance policies shall be endorsed to name as an additional insured the City and County of San Francisco and its respective departments, commissioners, officers, agents and employees.

The agreements between the Contractor and its towing and dismantling subcontractors (as applicable) shall require that reasonable insurance is maintained and that the City will be named as an additional insured on the general liability policy.

11.7 Commercial General Liability and Commercial Automobile Liability Insurance policies must be endorsed to provide:

(a) Name as Additional Insured the City and County of San Francisco, its Officers, Agents, and Employees.

(b) That such policies are primary insurance to any other insurance available to the Additional Insureds, with respect to any claims arising out of this Agreement; and that insurance

AMENDED AND RESTATED AGREEMENT

applies separately to each insured against whom claim is made or suit is brought, but the inclusion of more than on insured shall not operate to increase the insurer's limits of liability.

(c) Should any of the required insurance be provided under a form of coverage that includes a general annual aggregate limit or provides that claims investigation or legal defense costs be included in such general annual aggregate limit, such general annual aggregate limit shall be double the occurrence or claims limits specified above.

11.8 Regarding Workers' Compensation, Contractor hereby agrees to waive subrogation which any insurer of Contractor may acquire from Contractor by virtue of the payment of any loss. Contractor agrees to obtain any endorsement that may be necessary to effect this waiver of subrogation. The Workers' Compensation policy shall be endorsed with a waiver of subrogation in favor of the City for all work performed by the Contractor, its employees, agents and subcontractors.

11.9 Should any of the required insurance be provided under a claims-made form, Contractor shall maintain such coverage continuously throughout the term of this Agreement and, without lapse, for a period of three years beyond the expiration of this Agreement, to the effect that, should occurrences during the contract term give rise to claims made after expiration of the Agreement, such claims shall be covered by such claims-made policies.

11.10 Subject to the provisions of Section 18.1, should any required insurance lapse during the term of this Agreement, requests for payments originating after such lapse shall not be processed until the City receives satisfactory evidence of reinstated coverage as required by this Agreement, effective as of the lapse date. If insurance is not reinstated, the City may, at its sole option, terminate this Agreement effective on the date of such lapse of insurance.

11.11 Before commencing any operations under this Agreement, Contractor shall furnish to City certificates of insurance and additional insured policy endorsements with insurers with ratings comparable to A-, VIII or higher, that are authorized to do business in the State of California, and that are satisfactory to City, in form evidencing all coverages, including additional insured endorsements and the policy declaration page for any umbrella policies, set forth above. Failure to maintain insurance shall constitute a material breach of this Agreement.

11.12 Approval of the insurance by City shall not relieve or decrease the liability of Contractor hereunder.

12. Financial Assurances

12.1 Requirement to Provide Financial Guarantees

Upon the Effective Date of this Agreement, Contractor shall provide, and shall maintain for the time periods specified herein, financial instruments and funds described in this Section 12 as security to ensure Contractor's performance of all terms and conditions of this Agreement and to compensate for any damage to City property and/or other actual costs to City or for reimbursement to Customers for Contractor's violation of the terms of this Agreement, as further described below.

12.2 Letter of Credit and Performance Bond

12.2.1 Performance Guarantee. Upon the Effective Date of this Agreement, Contractor shall provide to the City, and shall maintain throughout the Term of this Agreement and for a period of at least ninety (90) days after expiration or termination of this Agreement, or until all of Contractor's obligations have been performed under this Agreement (including but not limited to investigation and remediation obligations under Appendices C and D), whichever date is later, a performance guarantee of two million dollars (\$2,000,000), which shall consist of a confirmed, irrevocable Letter of Credit in an amount between one million dollars (\$1,000,000) and two million AMENDED AND RESTATED AGREEMENT dollars (\$2,000,000) and, if Contractor chooses to satisfy the performance guarantee, in part, with a Performance Surety Bond, a Performance Surety Bond of up to one million dollars (\$1,000,000) in favor of the City and County of San Francisco, a municipal corporation, acting by and through its Municipal Transportation Agency (and during the Term of Appendix D, the Port of San Francisco), guarantying the faithful performance by Contractor of this Agreement and of the covenants, terms and conditions of this Agreement, including all monetary obligations set forth herein, and including liquidated damages and any dishonesty on the part of Contractor.

The City may draw upon such Performance Surety Bond and/or Letter of Credit in circumstances which include, but are not limited to:

(a) To ensure regulatory compliance in the event that Contractor receives a notice of violation or other regulatory order from a governmental or regulatory agency with jurisdiction over Contractor's operations or the Properties used by Contractor for the performance of this Agreement and Contractor does not achieve compliance with the notice of violation or order to the satisfaction of the issuing agency within the time specified by the agency or by the City if the agency does not specify a timeframe.

(b) To reimburse the City for any fine or other charge assessed against the City related to any notice of violation or other regulatory order issued to Contractor.

(c) To reimburse the City for costs associated with City's environmental assessments or corrective action related to Contractor's violation of any of the requirements of Appendix D, which may be performed at the City's sole discretion.

(d) To satisfy rental payment obligations for City-owned property licensed to Contractor in Appendix D.

(e) To satisfy fines assessed by City against Contractor pursuant to Appendix D of the Agreement.

(f) To compensate City for losses or damage to property caused by Contractor.

12.2.2 Performance Surety Bond/Letter of Credit Requirements. The Performance Surety Bond and/or Letter of Credit required by this Section 12 shall be issued on a form and issued by a financial institution acceptable to the City in its sole discretion, which financial institution shall (a) be a bank, insurance or trust company doing business and having an office in the State of California, (b) have a combined capital and surplus of at least \$25,000,000, and (c) be subject to supervision or examination by federal or state authority. If Contractor defaults with respect to any provision of this Agreement, City may, but shall not be required to, make its demand under said Performance Surety Bond or Letter of Credit for all or any portion thereof, to compensate City for any loss or damage which City may have incurred by reason of Contractor's default or dishonesty, including (but not limited to) any claim for fines or liquidated damages; provided, however, that City shall present its written demand to said bank, insurance or trust company for payment under said Performance Surety Bond or Letter of Credit only after City first shall have made its demand for payment directly to Contractor, and five (5) full days have elapsed without Contractor having made payment to City.

12.2.3 Expiration or Termination of Letter of Credit.

In the event the City receives notice from the issuer of the Letter of Credit that the Letter of Credit will be terminated, not renewed or will otherwise be allowed to expire for any reason during the period from the commencement of the Term of this Agreement to ninety (90) days after the expiration or termination of this Agreement or the conclusion of all of Contractor's obligations under

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 15 of 42 the Agreement, whichever occurs last, and Contractor fails to provide the City with a replacement Letter of Credit (in a form and issued by a financial institution acceptable to the City) within ten (10) days following the City's receipt of such notice, such occurrence shall be an Event of Default as defined in Section 18 of this Agreement, and, in addition to any other remedies the City may have due to the default, including the right to terminate this Agreement, the City shall be entitled to draw down the entire amount of the Letter of Credit (or any portion thereof) and hold such funds in an account with the Treasurer in the form of cash guarantying Contractor's obligations under this Agreement under the terms of this Section 12. In such event, the cash shall accrue interest to the Contractor at a rate equal to the average yield of Treasury Notes with one-year maturity, as determined by the Treasurer. In the event the Letter of Credit is converted into cash pursuant to this paragraph, upon termination of this Agreement Contractor shall be entitled to a full refund of the cash (less any demands made thereon by the City pursuant to Section 12.2.4,) within ninety (90) days of the termination date, including interest accrued through the termination date.

12.2.4 Expiration or Termination of Performance Surety Bond. The term of the Performance Surety Bond shall apply for individual one-year periods, and may be extended by the insurance, bank or trust company by Continuation Certificate. The insurance, bank or trust company herein may, if it so elects, terminate its obligation under this bond by serving at least forty five (45) days written notice of its intention to do so upon the SFMTA. In the event the City receives notice from the issuer of the Performance Surety Bond that the Performance Surety Bond will be terminated, not renewed or will otherwise be allowed to expire for any reason during the period from the commencement of the Term of this Agreement to ninety (90) days after the expiration or termination of this Agreement or the conclusion of all of Contractor's obligations under the Agreement, whichever occurs last, and Contractor fails to provide the City with a replacement Performance Surety Bond (in a form and issued by a financial institution acceptable to the City) within ten (10) days following the City's receipt of such notice, such occurrence shall be an Event of Default as defined in Section 18 of this Agreement. However, neither nonrenewal by the insurance, bank or trust company, nor the failure or inability of the Contractor to file a replacement bond in the event of nonrenewal, shall itself constitute a loss to the City recoverable under the Performance Surety Bond or any renewal or continuation thereof. Insurance, bank or trust company's liability under the Performance Surety Bond and all continuation certificates issued in connection therewith shall not be cumulative and shall in no event exceed the amount as set forth in the Performance Surety Bond or in any additions, riders, or endorsements properly issued by the insurance, bank or trust company as supplements thereto.

12.2.5 Demands Upon Performance Surety Bond or Letter of Credit. City may use all or any portion of the Performance Surety Bond or Letter of Credit to compensate City for any loss or damage that it may have incurred by reason of Contractor's negligence or breach. Such loss or damage may include without limitation any damage to or restoration of the Properties for which Contractor is responsible, and claims for fines and/or liquidated damages. Should the City terminate this Agreement due to a breach by Contractor, the City shall have the right to draw from the Performance Surety Bond or Letter of Credit those amounts necessary to pay any fees or other financial obligations under the Agreement and perform the towing and storage services described in this Agreement until such time as the City procures another contractor and the agreement between the City and that contractor becomes effective. The Treasurer shall have sole authority and responsibility to make demands upon the Performance Surety Bond or Letter of Credit. The Treasurer shall not allow any demands made to the Performance Surety Bond or Letter of Credit pursuant to claims arising from Section 6(c) of Appendix C to exceed a combined outstanding amount of more than four hundred thousand dollars (\$400,000.00) In addition, the Treasurer shall not allow any demands to be made to the Performance Surety Bond or Letter of Credit pursuant to claims arising from Section 6(b)

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 16 of 42 of Appendix C to exceed a combined outstanding amount of more than twice the amount of the current Base Fee established in Appendix D, Section 5, including all periodic CPI adjustments up to the date of the demand on the Performance Surety Bond or Letter of Credit. The Treasurer shall not allow any demands made to the Performance Surety Bond or Letter of Credit pursuant to claims arising from any section of the Agreement, excluding Appendices C and D, to exceed a combined outstanding amount of more than one million three hundred thousand dollars (\$1,300,000.00).

12.2.6 Depletion of Performance Surety Bond or Letter of Credit. If any portion of a Letter of Credit or Performance Surety Bond is used by City, Contractor shall provide written proof that the Performance Surety Bond or Letter of Credit has been restored to its initial value, which shall require a replacement Performance Surety Bond or Letter of Credit. Contractor's failure to do so within the time limits specified in Section 18.1.1(f) shall constitute an Event of Default as defined in Section 18 of this Agreement.

12.2.7 Dispute Resolution. In the event that a dispute arises between the City and Contractor concerning this Agreement or the use or maintenance of the Performance Surety Bond or Letter of Credit, Contractor may appeal to the Director of Transportation within fourteen (14) days of demand on the Performance Surety Bond or Letter of Credit with evidence supporting Contractor's claim for relief from the demand on the Performance Surety Bond or Letter of Credit. The Director of Transportation will respond within fourteen (14) days. Any failure of the Director of Transportation to respond within fourteen (14) days shall be deemed a rejection of Contractor's claim for relief from the demand on the Performance Surety Bond or Letter of Credit. Contractor's claim for relief from the demand on the Performance Surety Bond or Letter of Credit. Contractor's claim for relief from the demand on the Performance Surety Bond or Letter of Credit. Contractor's claim for relief from the demand on the Performance Surety Bond or Letter of Credit. Contractor's claim for relief from demands on the Letter of Credit and the Director of Transportation's response to such demand shall constitute the administrative remedy for Agreement interpretation described in Section 46 herein. Each party reserves its remedies in equity and law. No decision by the City concerning the Performance Surety Bond or Letter of Credit shall prevent Contractor from seeking restoration of the funds by appropriate legal action.

12.3 Maintenance Deposit. Upon execution of this Agreement, Contractor shall deposit with City the amount of one hundred thousand dollars (\$100,000) as a maintenance deposit. These funds may be used by City when maintenance required by this Agreement is not done in a timely manner or in accordance with the standards of this Agreement. Contractor shall be responsible for replenishing this maintenance deposit fund to maintain a balance of one hundred thousand dollars (\$100,000) within fifteen (15) days of any date that the fund falls below the minimum balance. Failure to replenish the maintenance deposit fund for more than forty-five (45) days shall be an Event of Default under this Agreement. Any interest accrued and earned on the maintenance deposit fund shall be retained by City.

12.4 Claims Fund. Contractor shall at all times maintain a Claim Fund for payment of Claims. Contractor shall maintain at least fifty-thousand dollars (\$50,000) in the Claim Fund at all times. Contractor shall be responsible for replenishing this Claims Fund to maintain a balance of fifty thousand dollars (\$50,000) within fifteen (15) days of any date that the fund falls below the minimum balance. Failure to replenish the Claims Fund for more than forty-five (45) days from the date that it falls below the minimum balance shall be an Event of Default under this Agreement.

12.5 Environmental Oversight Deposit. Upon the Effective Date of this Agreement, Contractor shall provide to the City, and shall maintain and replenish throughout the term of the property license set forth in Appendix D ("Pier 70 License") and for a period of at least ninety (90) days after termination or expiration of the Pier 70 License, an Environmental Oversight Deposit in the amount of ten thousand dollars (\$10,000), which shall be deposited in an account specified by

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 17 of 42 City. If Contractor receives a notice of violation or other regulatory order from a governmental or regulatory agency with jurisdiction over Pier 70 and or its operations (other than from the Port of San Francisco), and such notice is not cured within fourteen (14) days, the City may draw from this deposit to reimburse the City for staff costs incurred by the City while inspecting site conditions and enforcing and administering the Hazardous Materials provisions of the Pier 70 License. If Contractor receives a notice of violation or other regulatory order from a governmental or regulatory agency with jurisdiction over the site and or its operations (other than from the Port of San Francisco), and such notice is cured within fourteen (14) days, the City may draw from this deposit in an amount not to exceed \$500 to reimburse the City for staff costs incurred by the City. The City will submit an invoice to Contractor for any such costs, and Contractor will pay such invoiced amounts within thirty (30) days to replenish the Environmental Oversight Deposit. Contractor's failure to pay such costs within thirty (30) days, or to replenish the Environmental Oversight Deposit if drawn upon, will constitute an Event of Default.

13. Insurance and Performance Guarantee Requirements

All insurance policies, Performance Surety Bonds and Letters of Credit obtained pursuant to this Agreement shall be endorsed to provide that thirty (30) days prior written notice of cancellation, non-renewal or reduction in coverage or limits shall be given to SFMTA in the manner and at the addresses specified below (and during the term of Appendix D, to the Port of San Francisco in the manner and at the address specified in Appendix D).

Two copies of any Performance Surety Bond or Letter of Credit, and two copies of each original policy or policy endorsement of insurance shall be provided to SFMTA (and during the term of Appendix D, to the Port of San Francisco) upon the Effective Date of this Agreement, and complete copies of any insurance policies obtained pursuant to this Agreement shall be provided to SFMTA (and during the term of Appendix D, to the Port of San Francisco) if requested at any time.

13.1 Upon City's request, Contractor shall provide satisfactory evidence that Contractor has adequately provided for Social Security and Unemployment Compensation benefits for Contractor's Employees.

Contractor shall comply with the provisions of any insurance policy covering Contractor or the City, and with any notices, recommendations or directions issued by any insurer under such insurance policies so as not to adversely affect the insurance coverage.

14. Contractor's Representations and Warranties

Contractor hereby represents and warrants as follows:

14.1 Experience. Contractor is experienced in the operation and management of automobile towing and disposal services and hereby agrees to apply its best efforts and most efficient methods in performance of this Agreement.

14.2 Formation. Contractor is a duly formed, validly existing and in good standing limited liability company under the laws of the State of California.

14.3 Authority. Contractor has full power and authority (corporate or otherwise) to enter into this Agreement and to consummate the transactions contemplated by it. This Agreement has been duly authorized by all necessary action on the part of Contractor, and no other corporate or other action on the part of Contractor is necessary to authorize the execution and delivery of this Agreement.

14.4 Conflicts and Consents. The execution and delivery by Contractor of this Agreement and the performance by Contractor of the transactions contemplated by it will not violate AMENDED AND RESTATED AGREEMENT any federal, state or local law, rule or regulation, or conflict with or result in any breach or violation of, or constitute a default (or an event which with notice or lapse of time or both would become a default) under, or constitute an event or condition that would permit termination or acceleration of the maturity of, the Articles of Incorporation, bylaws or partnership agreement of Contractor (as applicable) or any indenture, mortgage, lease, agreement or other instrument or obligation to which Contractor is a party or by which it may be bound which would materially adversely affect the ability of Contractor to perform its obligations under this Agreement. No approval, authorization, consent or other order or action of, or filing or registration with, any person, entity or governmental authority is required for the execution and delivery by Contractor of this Agreement.

14.5 **Conflict with Orders.** The execution and delivery by Contractor or this Agreement will not conflict with any order, judgment or decree of any court, government, government agency or instrumentality, whether entered pursuant to consent to otherwise, by which Contractor may be bound or affected.

14.6 Litigation. There is no litigation, action, arbitration, grievance, administrative proceeding, suit or claim filed and pending, nor is there any investigation by a governmental agency of Contractor or any of its affiliates that, if adversely decided, could have a material adverse impact on Contractor's ability to perform its obligations under this Agreement.

15. Indemnification

Contractor shall indemnify and save harmless City and its officers, agents and employees from, and, if requested, shall defend them against any and all loss, cost, damage, injury, liability, and claims thereof for injury to or death of a person, including Employees of Contractor or loss of or damage to property, resulting directly or indirectly from Contractor's performance of this Agreement, including, but not limited to, the use of Contractor's facilities or equipment provided by City or others, regardless of the negligence of, and regardless of whether liability without fault is imposed or sought to be imposed on City, except to the extent that such indemnity is void or otherwise unenforceable under applicable law in effect on or validly retroactive to the date of this Agreement, and except where such loss, damage, injury, liability or claim is the result of the active negligence or willful misconduct of City and is not contributed to by any act of, or by any omission to perform some duty imposed by law or agreement on Contractor, its subcontractors or either's agent or employee. The foregoing indemnity shall include, without limitation, reasonable fees of attorneys, consultants and experts and related costs and City's costs of investigating any claims against the City. The foregoing indemnification does not include the limitations on Contractor's liability described in Appendix D, Sections 24.3 and 24.8.

In addition to Contractor's obligation to indemnify City, Contractor specifically acknowledges and agrees that it has an immediate and independent obligation to defend City from any claim which actually or potentially falls within this indemnification provision, even if the allegations are or may be groundless, false or fraudulent, which obligation arises at the time such claim is tendered to Contractor by City and continues at all times thereafter.

As to any intellectual property that Contractor provides to the City in the performance of this Agreement, Contractor shall indemnify and hold City harmless from all loss and liability, including attorneys' fees, court costs and all other litigation expenses for any infringement of the patent rights, copyright, trade secret or any other proprietary right or trademark, and all other intellectual property claims of any person or persons, arising as a consequence of the use by City of the intellectual property supplied by the Contractor, or any of its officers or agents.

AMENDED AND RESTATED AGREEMENT

16. Incidental and Consequential Damages

Contractor shall be responsible for incidental and consequential damages resulting in whole or in part from Contractor's acts or omissions. Nothing in this Agreement shall constitute a waiver or limitation of any rights that City may have under applicable law.

17. Liability of City

NOTWITHSTANDING ANY OTHER PROVISION OF THIS AMENDED AND RESTATED AGREEMENT, IN NO EVENT SHALL CITY BE LIABLE, REGARDLESS OF WHETHER ANY CLAIM IS BASED ON CONTRACT OR TORT, FOR ANY SPECIAL, CONSEQUENTIAL, INDIRECT OR INCIDENTAL DAMAGES, INCLUDING, BUT NOT LIMITED TO, LOST PROFITS, ARISING OUT OF OR IN CONNECTION WITH THIS AMENDED AND RESTATED AGREEMENT OR THE SERVICES PERFORMED IN CONNECTION WITH THIS AMENDED AND RESTATED AGREEMENT.

18. Default; Remedies

18.1 Each of the following shall constitute an event of default ("Event of Default") under this Amended and Restated Agreement; provided, however, that with the exception of the Events of Default listed in Section 18.1.1 below, Contractor shall have thirty (30) days after the date of written notice of the default sent by City to cure the default before an Event of Default exists:

18.1.1 The following conduct by Contractor shall constitute an Event of Default at the time it occurs without a right to cure under this Section 18, and shall be grounds for termination pursuant to Section 19:

(a) Contractor fails or refuses to perform or observe any term, covenant or condition contained in any of the following Sections of this Agreement: 5, 6, 11, 12.2.1, 13, 22, 28, 35 and 54.

(b) Substantial abandonment or discontinuance by Contractor, without the prior written consent of the City, of any or all of the services and operations required hereunder.

(c) Contractor's representation or warranty made pursuant to this Agreement which Contractor made knowing that it was not true and correct at the time when made.

(d) Failure of Contractor to replenish the maintenance deposit or Claims Fund required by Sections 12.3 and 12.4 and such failure continues for more than forty-five (45) days.

(e) Failure of Contractor to replenish the Environmental Oversight Deposit required by Section 12.5 and such failure continues for more than thirty (30) days.

(f) Failure to replace the Performance Surety Bond or Letter of Credit as required by Sections 12.2.3 and 12.2.5 within twenty (20) days of (i) City's receipt of notice of its termination or expiration, or (ii) use by City of the Performance Surety Bond or Letter of Credit, unless City's use is challenged pursuant to Section 12.2.6, in which event the twenty (20) days shall run from the date of the Director of Transportation's denial of the challenge, or fourteen (14) days from the date of the challenge, whichever is later.

18.1.2 Contractor fails or refuses to perform or observe any other term, covenant or condition contained in this Agreement.

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 20 of 42 18.1.3 Contractor (A) is generally not paying its debts as they become due, (B) files, or consents by answer or otherwise to the filing against it of, a petition for relief or reorganization or arrangement or any other petition in bankruptcy or for liquidation or to take advantage of any bankruptcy, insolvency or other debtors' relief law of any jurisdiction, (C) makes an assignment for the benefit of its creditors, (D) consents to the appointment of a custodian, receiver, trustee or other officer with similar powers of Contractor or of any substantial part of Contractor's property or (E) takes action for the purpose of any of the foregoing.

18.1.4 A court or government authority enters an order (A) appointing a custodian, receiver, trustee or other officer with similar powers with respect to Contractor or with respect to any substantial part of Contractor's property, (B) constituting an order for relief or approving a petition for relief or reorganization or arrangement or any other petition in bankruptcy or for liquidation or to take advantage of any bankruptcy, insolvency or other debtors' relief law of any jurisdiction or (C) ordering the dissolution, winding-up or liquidation of Contractor, or (D) levying of a writ of attachment or execution against any of Contractor's property.

18.1.5 Failure of Contractor to pay when due any other amount owing from Contractor to the City, including without limitation rents, taxes, fees or other charges, whether or not such amounts are related to this Agreement; provided, however, that if Contractor disputes the amount of any such obligation in good faith and is actively negotiating or litigating such dispute, Contractor's failure to pay such amount shall not constitute a default under this paragraph.

18.1.6 Failure of Contractor to abide by any of the terms or conditions of the Port MOU, set forth in Appendix C to this Agreement.

18.2 On and after any Event of Default, City shall have the right to exercise its legal and equitable remedies, including, without limitation, the right to terminate this Agreement or to seek specific performance of all or any part of this Agreement. City's right to termination for an Event of Default shall be subject to Contractor's opportunity to cure such Event of Default pursuant to the terms of Sections 18.1 and 19. In addition, City shall have the right (but no obligation) to cure (or cause to be cured) on behalf of Contractor any Event of Default; Contractor shall pay to City on demand all costs and expenses incurred by City in effecting such cure, with interest thereon from the date of incurrence at the maximum rate then permitted by law. City shall have the right to offset from any amounts due to Contractor under this Agreement or any other agreement between City and Contractor all damages, losses, costs or expenses incurred by City as a result of such Event of Default and any liquidated damages due from Contractor pursuant to the terms of this Agreement or any other agreement.

18.3 City shall have the right to offset from any amounts due to Contractor under this Agreement or any other Agreement between City and Contractor all damages, losses, costs or expenses incurred by City as a result of such Event of Default and any liquidated damages due from Contractor pursuant to the terms of this Agreement or any other Agreement.

18.4 All remedies provided for in this Agreement may be exercised individually or in combination with any other remedy available hereunder or under applicable laws, rules and regulations. The exercise of any remedy shall not preclude or in any way be deemed to waive any other remedy.

19. Termination

The City shall have the right to terminate this Agreement after written notice to Contractor upon the occurrence of any Event of Default, provided, however, that Contractor shall have a period of ten (10)

AMENDED AND RESTATED AGREEMENT

n:\ptcI\as2010\0900371\00637281.doc Page 21 of 42 days from the date of City's notice of intent to terminate to cure such default before City's termination for cause may become effective.

19.1 Termination Effective Upon Notice. Termination under this Section shall be effective immediately upon notice being given by SFMTA to Contractor and after the expiration of any applicable cure periods. Upon such termination, all rights, powers, privileges and authority granted to Contractor under this Agreement shall cease, and Contractor shall immediately thereupon vacate the Designated Facilities, except as may be permitted pursuant to Section 54 (Transition Period), below.

19.2 Termination Not Exclusive Remedy. The City's right to terminate this Agreement under this Section is not its exclusive remedy but is in addition to all other remedies provided to it by law or the provisions of this Agreement.

19.3 Duties Upon Termination. Upon termination of this Agreement, the City and Contractor shall promptly pay to the other, as soon as is determinable after the effective date of termination, all amounts due each other under the terms of this Agreement, and upon such payment neither shall have any further claim or right against the other, except as expressly provided herein. Upon the effective date of termination, Contractor shall deliver to the SFMTA the originals of all books, permits, plans, Records, licenses, contracts and other documents pertaining to Contractor's operation under this Agreement, any insurance policies, bills of sale or other documents evidencing title or rights of the City, and any and all other Records or documents pertaining to Contractor's operation under this Agreement, any insurance policies, bills of sale or other documents evidencing title or rights of the City, and any and all other Records or documents pertaining to the Designated Facilities, whether or not enumerated herein, which are requested by the City or necessary or desirable for the ownership and operation of the Designated Facilities, which are then in possession of Contractor. Contractor further agrees to do all other things reasonably necessary to cause an orderly transition of the management and operation of the services provided by Contractor under this Agreement without detriment to the rights of the City or to the continued operation of such services.

20. Provisions of Amended and Restated Agreement Surviving Termination or Expiration

This Section and the following Sections of this Agreement shall survive termination or expiration of this Amended and Restated Agreement: 1, 5, 6, 7, 9, 10, 12 (except for 12.2.5), 14, 15, 16, 17, 18.2, 18.3, 19.3, 20, 22.1, 23, 24, 25, 26, 29, 38, 39, 43, 46, 47, 48, 49, 53, 54, 55, 56 and 57; and Appendix A Sections 1.3(b), 4.1, 5.2, 12, 13.3, 15.1, 15.2, 15.7(1), 15.7(2), 15.10(7) and 15.11.

21. Conflict of Interest

Through its execution of this Agreement, Contractor acknowledges that it is familiar with the provisions of §15.103 and Appendix C 8.105 of City's Charter and § 87100 et seq. of the Government Code of the State of California, and certifies that it does not know of any facts which constitute a violation of said provisions.

22. Proprietary or Confidential Information

22.1 Contractor understands and agrees that, in the performance of the work or services under this Agreement or in contemplation thereof, Contractor may have access to private or confidential information which may be owned or controlled by City and that such information may contain proprietary or confidential details, the disclosure of which to third parties may be damaging to City. Contractor agrees that to the extent permitted by law, all information disclosed by City to Contractor shall be held in confidence and used only in performance of the Agreement. Contractor shall exercise the same standard of care to protect such information as a reasonably prudent contractor would use to protect its own proprietary data.

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 22 of 42

22.2 Contractor understands and agrees that the City's Sunshine Ordinance (Administrative Code, Chapter 67) and the California Public Records Act (Gov. Code section 6250 et seq.), apply to the Agreement and all of its appendices, that these documents are not proprietary or confidential, do not constitute trade secrets, and that the City must disclose these documents to the public, upon request, without redaction, except for employee's personal information. Contractor shall cooperate with City in the compilation, copying and production of records in its custody that are subject to requests for public records. The City is not required to take any action, or to refuse to release information where to do so would violate applicable law. During the Term, the City will endeavor to provide Contractor reasonable notice of any request for public information that seeks disclosure of confidential or proprietary information that Contractor has provided to City under the Agreement and that Contractor has identified as confidential and proprietary, with the exception of this Agreement and its appendices. Contractor may at its option then take whatever legal steps it deems appropriate to protect said information from disclosure to the public, but the City shall have no further obligation to protect said information from disclosure. However, if the Contractor takes legal action to protect said information, and if the City is required to incur legal fees and costs in such legal action, and if the Contractor does not prevail in such legal action, Contractor shall pay all legal fees and costs that the City incurs as a result of such legal action. Contractor shall clearly identify to City all information that Contractor provides to City that it considers to be proprietary, trade secret or is otherwise protected from disclosure under the California Public Records Act, the City's Sunshine Ordinance and other applicable law.

23. Notices to the Parties

Any insurance certificates or notices required under Sections 11, 12, 13, 15, 18, 19, 46 or 52 of the Agreement must be sent by first class, certified U.S. mail, postage pre-paid. All other written communications, unless otherwise indicated elsewhere in this Agreement, may be by first class U.S. mail, by email, or by fax, and shall be addressed as follows, or to such other address as designated by the parties in writing:

To Contractor:

San Francisco AutoReturn

Attention: John Wicker, President and CEO

945 Bryant Street, Suite 350

San Francisco, CA 94103	
-------------------------	--

Telephone:	415-575-2355
------------	--------------

Email:

Facsimile 415-575-2375

jwicker@autoreturn.com

To City:

San Francisco Municipal Transportation Agency Attention: Lorraine Fuqua 1 South Van Ness Avenue, 8th Floor San Francisco, CA 94103 Telephone: 415-701-4678

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 23 of 42 Facsimile:
Email:

415-701-4736 lorraine.fugua@sfmta.com

24. Ownership of Results

Any interest of Contractor or its Subcontractors, in the Records prepared by Contractor or its subcontractors in connection with services to be performed under this Agreement, shall become the property of and will be transmitted to City. However, Contractor may retain and use copies for reference and as documentation of its experience and capabilities. To the extent that the use of proprietary software or other proprietary information or intellectual property is required to access or utilize the data contained in the Records, or that Contractor holds particular work practices or methods to be proprietary, Contractor hereby grants the City a perpetual, royalty-free, nonexclusive, nontransferable, limited license, to use and reproduce said proprietary information or intellectual property, solely for City's internal purposes related to the towing, storage, and disposal of abandoned and illegally parked vehicles.

25. Works for Hire

Subject to the limited license set out in Section 24 above, any artwork, copy, posters, billboards, photographs, videotapes, audiotapes, systems designs, software, source codes, or any other original works of authorship of Contractor that are not Records, are proprietary to Contractor and shall not be considered works for hire as defined under Title 17 of the United States Code, and all copyrights in such works are the property of Contractor.

26. Audit and Inspection of Records

26.1 Records. Contractor shall maintain, in accordance with generally accepted accounting principles and business practices, all books, accounts and Records created in the performance of this Agreement in accordance with the requirements of Appendix A. Such books, accounts and Records shall be maintained throughout the Term of this Agreement at one of the Designated Facilities, or at Contractor's Headquarters Office. Records created or maintained in an electronic format shall be submitted monthly in an electronic format as specified by SFMTA, which may be an electronic database format or a static electronic format, such as PDF or an electronic format that attaches a date or time stamp to each document and record entry that cannot be erased or altered Contractor shall deliver said copy of the Records with the Monthly Management Report described in Appendix A, Section 13.

Except as otherwise specified herein, Contractor shall maintain Records related to this Agreement in a safe and secure location available for inspection and copying by City for a period of five (5) years following termination of this Amended and Restated Agreement.

26.2 City's Right to Inspect. Any duly authorized agent of City shall have the right to examine, at any time during normal business hours, all books, accounts and Records, including computer records, of the type described above.

26.3 Audit. Within sixty (60) days of the expiration or termination of the Agreement, an independent auditor approved by the SFMTA shall conduct an audit of those records pertaining to Contractor's performance during the final year of this Agreement, including a summary report of prior annual audits conducted pursuant to Appendix A, Section 12.6. Such audit shall include a review of all records involving the removal, impoundment and disposition of vehicles pursuant to

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 24 of 42 this Agreement during the previous and current audit periods. All costs of such audits shall be withdrawn from the Contractor Compliance Audit Fund, as specified in Section 12.6 of Appendix A. A certified copy of each such audit report shall be furnished promptly to SFMTA and the Controller not more than 120 days following the expiration of the Agreement.

26.4 Retention of Records. Contractor shall retain all records of subcontracts, Employee payroll and benefits, tax assessments and payments, and those records and reports described in Sections 10 and 26 of this Agreement for a period of not less than four (4) years from date of termination of this Agreement.

27. Subcontracting

Except as specifically provided in this Agreement, Contractor shall not enter into any subcontract for the performance of all or any part of this Agreement for the acquisition of towing services, security services, vehicle handling or disposal services, auctioneer or appraisal services, Customer relations services or with any PEO Contractor without the prior written consent of the City, which consent shall not be unreasonably withheld. Any attempt to enter into such a subcontract without such consent shall be null and void. Subject to the foregoing, this Agreement shall be binding upon, enforceable by, and shall inure to the benefit of any permitted subcontractor.

28. Assignment

Contractor shall not assign or otherwise transfer this Agreement or any of Contractor's rights, duties or interest under this Agreement without the prior written consent of the City. Any attempted assignment or transfer without such consent shall be null and void. Subject to the foregoing, this Agreement shall be binding upon, enforceable by, and shall inure to the benefit of the successors and permitted assigns of the Contractor.

29. Non-Waiver of Rights

The omission by either party at any time to enforce any default or right reserved to it, or to require performance of any of the terms, covenants, or provisions hereof by the other party at the time designated, shall not be a waiver of any such default or right to which the party is entitled, nor shall it in any way affect the right of the party to enforce such provisions thereafter.

30. Earned Income Credit (EIC) Forms

Administrative Code Chapter 12O requires that employers provide their employees with IRS Form W-5 (The Earned Income Credit Advance Payment Certificate) and the IRS EIC Schedule, as set forth below. Employers can locate these forms at the IRS Office, on the Internet, or anywhere that Federal Tax Forms can be found.

30.1 Contractor shall provide EIC Forms to each Eligible Employee at each of the following times: (i) within thirty (30) days following the date on which this Amended and Restated Agreement becomes effective (unless Contractor has already provided such EIC Forms at least once during the calendar year in which such effective date falls); (ii) promptly after any Eligible Employee is hired by Contractor; and (iii) annually between January 1 and January 31 of each calendar year during the Term of this Agreement.

30.2 Failure to comply with any requirement contained in Section 30.1 of this Agreement shall constitute a material breach by Contractor of the terms of this Agreement. If, within thirty (30) days after Contractor receives written notice of such a breach, Contractor fails to cure such breach or, if such breach cannot reasonably be cured within such period of thirty (30) days, Contractor fails to commence efforts to cure within such period or thereafter fails to diligently pursue such cure to

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 25 of 42 completion, the City may pursue any rights or remedies available under this Agreement or under applicable law.

30.3 Any Subcontract entered into by Contractor shall require the subcontractor to comply, as to the subcontractor's Eligible Employees, with each of the terms of this section.

30.4 Capitalized terms used in this Section and not defined in this Agreement shall have the meanings assigned to such terms in Chapter 12O of the San Francisco Administrative Code.

31. Minority/Women/Local Business Utilization; Liquidated Damages

31.1 Compliance. Contractor understands and agrees to comply fully with all applicable provisions of Chapter 12D.A ("Minority/Women/Local Business Utilization Ordinance--IV") of the San Francisco Administrative Code set forth in the RFP, Contractor's Proposal and this Agreement, and agrees to include such provisions in all subcontracts (as defined in Section 27) made in fulfillment of the Contractor's obligations under this Agreement. Said provisions are incorporated by reference and made a part of this Agreement as though fully set forth herein. Contractor's willful failure to comply with Chapter 12D.A is a material breach of contract.

31.2 Enforcement. If Contractor willfully fails to comply with any of the provisions of Chapter 12D.A set forth in this Agreement pertaining to MBE or WBE participation, Contractor shall be liable for liquidated damages in an amount equal to Contractor's net profit on this Agreement, or 10% of the total amount of this Agreement, or \$1,000, whichever is greatest. City may also impose other sanctions against Contractor authorized in Chapter 12D.A, including declaring the Contractor to be irresponsible and ineligible to contract with the City for a period of up to five years or revocation of the Contractor's MBE or WBE certification. City will determine the sanctions to be imposed, including the amount of liquidated damages, after investigation pursuant to §12D.A.16(B). By entering into this Agreement, Contractor acknowledges and agrees that any liquidated damages assessed by City shall be payable to City upon demand. Contractor further acknowledges and agrees that any liquidated damages assessed may be withheld from any monies due to Contractor on any contract with City. Contractor agrees to maintain records necessary for monitoring its compliance with this Section 31 for a period of three (3) years following termination of this contract, and shall make such records available for audit and inspection by HRC or the Controller upon request.

31.3 Subcontracting Goals. The MBE/WBE subcontracting participation goal for this contract is 12% of Gross Revenues. Contractor shall fulfill this goal throughout the duration of this Agreement. Contractor shall not participate in any back contracting to the Contractor or lower-tier subcontractors, as defined in Chapter 12D.A, for any purpose inconsistent with the provisions of Chapter 12D.A, its implementing rules and regulations, or this Section.

31.4 Subcontract Language Requirements. Contractor shall include in all subcontracts with MBEs or WBEs made in fulfillment of Contractor's obligations under this Agreement, a provision requiring Contractor to compensate any MBE or WBE subcontractor for damages for breach of contract or liquidated damages equal to 5% of the subcontract amount, whichever is greater, if Contractor fails to comply with the requirements of this Section, unless Contractor received approval from City to substitute subcontractors or to otherwise modify the commitments in the bid or proposal. Such provisions shall also state that it is enforceable in a court of competent jurisdiction. Subcontracts shall require the subcontractor to maintain records necessary for monitoring its compliance with this Section 31 for a period of three (3) years following termination of this contract and to make such records available for audit and inspection by City upon request.

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281,doc Page 26 of 42 **31.5** Payment of Subcontractors. Contractor shall pay its subcontractors within thirty (30) days after receiving an invoice or request for payment from a subcontractor, unless Contractor notifies City in writing within ten (10) working days prior to receiving payment from the City that there is a bona fide dispute between Contractor and its subcontractor, in which case Contractor may withhold the disputed amount but shall pay the undisputed amount.

32. Nondiscrimination; Penalties

32.1 Contractor Shall Not Discriminate. In the performance of this Agreement, Contractor agrees not to discriminate against any Employee, City and County employee working with such contractor or subcontractor, applicant for employment with such contractor or subcontractor, or against any person seeking accommodations, advantages, facilities, privileges, services, or membership in all business, social, or other establishments or organizations, on the basis of the fact or perception of a person's race, color, creed, religion, national origin, ancestry, age, height, weight, sex, sexual orientation, gender identity, domestic partner status, marital status, disability or Acquired Immune Deficiency Syndrome or HIV status (AIDS/HIV status), or association with members of such protected classes, or in retaliation for opposition to discrimination against such classes.

32.2 Nondiscrimination in Benefits. Contractor does not as of the date of this Agreement and will not during the Term of this Agreement, in any of its operations in San Francisco, on real property owned by San Francisco, or where work is being performed for the City elsewhere in the United States, discriminate in the provision of bereavement leave, family medical leave, health benefits, membership or membership discounts, moving expenses, pension and retirement benefits or travel benefits, as well as any benefits other than the benefits specified above, between employees with domestic partners and employees with spouses, and/or between the domestic partners and spouses of such employees, where the domestic partnership has been registered with a governmental entity pursuant to state or local law authorizing such registration, subject to the conditions set forth in §12B.2(b) of the San Francisco Administrative Code.

32.3 Subcontracts. Contractor shall incorporate by reference in all subcontracts the provisions of §§12B.2(a), 12B.2(c)-(k), and 12C.3 of the San Francisco Administrative Code (copies of which are available from Purchasing) and shall require all subcontractors to comply with such provisions. Contractor's failure to comply with the obligations in this subsection shall constitute a material breach of this Agreement.

32.4 Condition to Contract. As a condition to this Amended and Restated Agreement, Contractor shall execute the "Chapter 12B Declaration: Nondiscrimination in Contracts and Benefits" form (form HRC-12B-101) with supporting documentation and secure the approval of the form by the San Francisco Human Rights Commission.

32.5 Incorporation of Administrative Code Provisions by Reference. The provisions of Chapters 12B and 12C of the San Francisco Administrative Code are incorporated in this Section by reference and made a part of this Agreement as though fully set forth herein. Contractor shall comply fully with and be bound by all of the provisions that apply to this Agreement under such Chapters, including but not limited to the remedies provided in such Chapters. Without limiting the foregoing, Contractor understands that pursuant to §12B.2(h) of the San Francisco Administrative Code, a penalty of \$50 for each person for each calendar day during which such person was discriminated against in violation of the provisions of this Agreement may be assessed against Contractor and/or deducted from any payments due Contractor.

33. MacBride Principles—Northern Ireland

AMENDED AND RESTATED AGREEMENT

Pursuant to San Francisco Administrative Code §12F.5, the City and County of San Francisco urges companies doing business in Northern Ireland to move towards resolving employment inequities, and encourages such companies to abide by the MacBride Principles. The City and County of San Francisco urges San Francisco companies to do business with corporations that abide by the MacBride Principles. By signing below, the person executing this Agreement on behalf of Contractor acknowledges and agrees that he or she has read and understood this section.

34. Tropical Hardwood and Virgin Redwood Ban

Pursuant to § 804(b) of the San Francisco Environment Code, the City and County of San Francisco urges contractors not to import, purchase, obtain, or use for any purpose, any tropical hardwood, tropical hardwood wood product, virgin redwood or virgin redwood wood product.

35. Drug-Free Workplace Policy

Contractor acknowledges that pursuant to the Federal Drug-Free Workplace Act of 1989, the unlawful manufacture, distribution, dispensation, possession, or use of a controlled substance is prohibited on City premises. Contractor agrees that any violation of this prohibition by Contractor, its Employees, agents or assigns will be deemed a material breach of this Agreement.

36. Resource Conservation

Chapter 5 of the San Francisco Environment Code ("Resource Conservation") is incorporated herein by reference. Failure by Contractor to comply with any of the applicable requirements of Chapter 5 will be deemed a material breach of contract.

37. Compliance with Americans with Disabilities Act

Contractor acknowledges that, pursuant to the Americans with Disabilities Act (ADA), programs, services and other activities provided by a public entity to the public, whether directly or through a contractor, must be accessible to the disabled public. Contractor shall provide the services specified in this Agreement in a manner that complies with the ADA and any and all other applicable federal, state and local disability rights legislation. Contractor agrees not to discriminate against disabled persons in the provision of services, benefits or activities provided under this Agreement and further agrees that any violation of this prohibition on the part of Contractor, its employees, agents or assigns will constitute a material breach of this Agreement.

38. Sunshine Ordinance

38.1 Documents Subject to Disclosure by the City. In accordance with San Francisco Administrative Code §67.24(e), contracts, contractors' bids, responses to solicitations and all other records of communications between City and persons or firms seeking contracts, shall be open to inspection immediately after a contract has been awarded. Nothing in this provision requires the disclosure of a private person or organization's net worth or other proprietary financial data submitted for qualification for a contract or other benefit until and unless that person or organization is awarded the contract or benefit. Information provided which is covered by this paragraph will be made available to the public by the City upon request. Contractor acknowledges that this Agreement and all of its appendices are covered by this paragraph and are subject to disclosure without redaction, except for employee's personal information.

38.2 Documents Subject to Disclosure by Contractor. Contractor agrees that it shall comply with San Francisco Administrative Code § 67.29-7(c), the provisions of which are set out in full below:

AMENDED AND RESTATED AGREEMENT

In any contract, agreement or permit between the City and any outside entity that authorizes that entity to demand any funds or fees from citizens, the City shall ensure that accurate records of each transaction are maintained in a professional and businesslike manner and are available to the public as public records under the provisions of this ordinance. Failure of an entity to comply with these provisions shall be grounds for terminating the contract or for imposing a financial penalty equal to one-half of the fees derived under the agreement or permit during the period of time when the failure was in effect. Failure of any Department Head under this provision shall be a violation of this ordinance. This paragraph shall apply to any Agreement allowing an entity to tow or impound vehicles in the City and shall apply to any Agreement allowing an entity to collect any fee from any persons in any pretrial diversion program.

Contractor's compliance with Section 67.29-7(c) is a material term of the Agreement.

38.3 Public Records Requests from Public. If any member of the public communicates a public records request directly to Contractor, Contractor shall refer the requestor to SFMTA. In no event shall Contractor respond directly to a public records request or attempt to communicate a public records request to SFMTA on behalf of the requestor. SFMTA will work with Contractor, and Contractor shall cooperate with SFMTA to identify responsive records in Contractor's possession. If production of requested records involves significant staff time of Contractor, City in its sole discretion may elect to collect legally authorized fees from the requestor and credit some or all of the fees to Contractor.

39. Notification of Limitations on Contributions

Through execution of this Agreement, Contractor acknowledges that it is familiar with section 1.126 of the City's Campaign and Governmental Conduct Code, which prohibits any person who contracts with the City for the rendition of personal services, for the furnishing of any material, supplies or equipment, for the sale or lease of any land or building, or for a grant, loan or loan guarantee, from making any campaign contribution to (1) an individual holding a City elective office if the contract must be approved by the individual, a board on which that individual serves, or the board of a state agency on which an appointee of that individual serves, (2) a candidate for the office held by such individual, or (3) a committee controlled by such individual, at any time from the commencement of negotiations for the contract until the later of either the termination of negotiations for such contract or six months after the date the contract is approved. Contractor acknowledges that the foregoing restriction applies only if the contract or a combination or series of contracts approved by the same individual or board in a fiscal year have a total anticipated or actual value of \$50,000 or more. Contractor further acknowledges that the prohibition on contributions applies to each prospective party to the contract; each member of Contractor's board of directors; Contractor's chairperson, chief executive officer, chief financial officer and chief operating officer; any person with an ownership interest of more than 20 percent in Contractor; any subcontractor listed in the bid or contract; and any committee that is sponsored or controlled by Contractor. Additionally, Contractor acknowledges that Contractor must inform each of the persons described in the preceding sentence of the prohibitions contained in Section 1.126. Contractor further agrees to provide to City the names of each person, entity or committee described above.

40. Requiring Minimum Compensation for Covered Employees

Contractor agrees to comply fully with and be bound by all of the provisions of the Minimum Compensation Ordinance (MCO), as set forth in San Francisco Administrative Code Chapter 12P (Chapter 12P), including the remedies provided, and implementing guidelines and rules. The provisions of Sections 12P.5 and 12P.5.1 of Chapter 12P are incorporated herein by reference and

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 29 of 42 made a part of this Amended and Restated Agreement as though fully set forth. The text of the MCO is available on the web at www.sfgov.org/olse/mco. A partial listing of some of Contractor's obligations under the MCO is set forth in this Section. Contractor is required to comply with all the provisions of the MCO, irrespective of the listing of obligations in this Section.

The MCO requires Contractor to pay Contractor's employees a minimum hourly gross compensation wage rate and to provide minimum compensated and uncompensated time off. The minimum wage rate may change from year to year and Contractor is obligated to keep informed of the then-current requirements. Any subcontract entered into by Contractor shall require the subcontractor to comply with the requirements of the MCO and shall contain contractual obligations substantially the same as those set forth in this Section. It is Contractor's obligation to ensure that any subcontractors of any tier under this Amended and Restated Agreement comply with the requirements of the MCO. If any subcontractor under this Agreement fails to comply, City may pursue any of the remedies set forth in this Section against Contractor.

40.1 Contractor shall not take adverse action or otherwise discriminate against an employee or other person for the exercise or attempted exercise of rights under the MCO. Such actions, if taken within 90 days of the exercise or attempted exercise of such rights, will be rebuttably presumed to be retaliation prohibited by the MCO.

40.2 Contractor shall maintain employee and payroll records as required by the MCO. If Contractor fails to do so, it shall be presumed that the Contractor paid no more than the minimum wage required under State law.

40.3 The City is authorized to inspect Contractor's job sites and conduct interviews with employees and conduct audits of Contractor.

40.4 Contractor's commitment to provide the Minimum Compensation is a material element of the City's consideration for this Agreement. The City in its sole discretion shall determine whether such a breach has occurred. The City and the public will suffer actual damage that will be impractical or extremely difficult to determine if the Contractor fails to comply with these requirements. Contractor agrees that the sums set forth in Section 12P.6.1 of the MCO as liquidated damages are not a penalty, but are reasonable estimates of the loss that the City and the public will incur for Contractor's noncompliance. The procedures governing the assessment of liquidated damages shall be those set forth in Section 12P.6.2 of Chapter 12P.

40.5 Contractor understands and agrees that if it fails to comply with the requirements of the MCO, the City shall have the right to pursue any rights or remedies available under Chapter 12P (including liquidated damages), under the terms of the contract, and under applicable law. If, within 30 days after receiving written notice of a breach of this Agreement for violating the MCO, Contractor fails to cure such breach or, if such breach cannot reasonably be cured within such period of 30 days, Contractor fails to commence efforts to cure within such period, or thereafter fails diligently to pursue such cure to completion, the City shall have the right to pursue any rights or remedies available under applicable law, including those set forth in Section 12P.6(c) of Chapter 12P. Each of these remedies shall be exercisable individually or in combination with any other rights or remedies available to the City.

40.6 · Contractor represents and warrants that it is not an entity that was set up, or is being used, for the purpose of evading the intent of the MCO.

40.7 If Contractor is exempt from the MCO when this Agreement is executed because the cumulative amount of agreements with this department for the fiscal year is less than \$25,000, but . Contractor later enters into an agreement or agreements that cause contractor to exceed that amount

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 30 of 42 in a fiscal year, Contractor shall thereafter be required to comply with the MCO under this Agreement. This obligation arises on the effective date of the agreement that causes the cumulative amount of agreements between the Contractor and this department to exceed \$25,000 in the fiscal year.

41. Requiring Health Benefits for Covered Employees

Contractor agrees to comply fully with and be bound by all of the provisions of the Health Care Accountability Ordinance (HCAO), as set forth in San Francisco Administrative Code Chapter 12Q, including the remedies provided, and implementing regulations, as the same may be amended from time to time. The provisions of Chapter 12Q.5.1 of Chapter 12Q are incorporated herein by reference and made a part of this Agreement as though fully set forth. The text of the HCAO is available on the web at www.sfgov.org/olse. Capitalized terms used in this Section and not defined in this Agreement shall have the meanings assigned to such terms in Chapter 12Q.

41.1 For each Covered Employee, Contractor shall provide the appropriate health benefit set forth in Section 12Q.3 of the HCAO. If Contractor chooses to offer the health plan option, such health plan shall meet the minimum standards set forth by the San Francisco Health Commission.

41.2 Notwithstanding the above, if the Contractor is a small business as defined in Section 12Q.3(d) of the HCAO, it shall have no obligation to comply with this Section 41.1 above.

41.3 Contractor's failure to comply with the HCAO shall constitute a material breach of this Agreement. City shall notify Contractor if such a breach has occurred. If, within thirty (30) days after receiving City's written notice of a breach of this Agreement for violating the HCAO, Contractor fails to cure such breach or, if such breach cannot reasonably be cured within such period of thirty (30) days, Contractor fails to commence efforts to cure within such period, or thereafter fails diligently to pursue such cure to completion, City shall have the right to pursue the remedies set forth in 12Q.5(f)(1)-(5). Each of these remedies shall be exercisable individually or in combination with any other rights or remedies available to City.

41.4 Any Subcontract entered into by Contractor shall require the Subcontractor (unless that Subcontractor is exempt pursuant to Section 12Q.3) to comply with the requirements of the HCAO and shall contain contractual obligations substantially the same as those set forth in this Section. Contractor shall notify City's Office of Contract Administration when it enters into such a Subcontract and shall certify to the Office of Contract Administration that it has notified the Subcontractor of the obligations under the HCAO and has imposed the requirements of the HCAO on Subcontractor through the Subcontract. Each Contractor shall be responsible for its Subcontractors' compliance with this Chapter. If a Subcontractor fails to comply, the City may pursue the remedies set forth in this Section against Contractor based on the Subcontractor's failure to comply, provided that City has first provided Contractor with notice and an opportunity to obtain a cure of the violation.

41.5 Contractor shall not discharge, reduce in compensation, or otherwise discriminate against any employee for notifying City with regard to Contractor's noncompliance or anticipated noncompliance with the requirements of the HCAO, for opposing any practice proscribed by the HCAO, for participating in proceedings related to the HCAO, or for seeking to assert or enforce any rights under the HCAO by any lawful means.

41.6 Contractor represents and warrants that it is not an entity that was set up, or is being used, for the purpose of evading the intent of the HCAO.

AMENDED AND RESTATED AGREEMENT

41.7 Contractor shall maintain employee and payroll records in compliance with the California Labor Code and Industrial Welfare Commission orders, including the number of hours each employee has worked on the City Contract.

41.8 Contractor shall keep itself informed of the current requirements of the HCAO.

41.9 Contractor shall provide reports to the City in accordance with any reporting standards promulgated by the City under the HCAO, including reports on Subcontractors and Subtenants, as applicable.

41.10 Contractor shall provide City with access to records pertaining to compliance with HCAO after receiving a written request from City to do so and being provided at least ten business days to respond.

41.11 Contractor shall allow City to inspect Contractor's job sites and have access to Contractor's employees in order to monitor and determine compliance with HCAO.

41.12 City may conduct random audits of Contractor to ascertain its compliance with HCAO. Contractor agrees to cooperate with City when it conducts such audits.

41.13 If Contractor is exempt from the HCAO when this Agreement is executed because its amount is less than \$25,000 (\$50,000 for nonprofits), but Contractor later enters into an agreement or agreements that cause Contractor's aggregate amount of all agreements with City to reach \$75,000, all the agreements shall be thereafter subject to the HCAO. This obligation arises on the effective date of the agreement that causes the cumulative amount of agreements between Contractor and the City to be equal to or greater than \$75,000 in the fiscal year.

42. First Source Hiring Program

a. Incorporation of Administrative Code Provisions by Reference

The provisions of Chapter 83 of the San Francisco Administrative Code are incorporated in this Section by reference and made a part of this Agreement as though fully set forth herein. Contractor shall comply fully with, and be bound by, all of the provisions that apply to this Agreement under such Chapter, including but not limited to the remedies provided therein. Capitalized terms used in this Section and not defined in this Agreement shall have the meanings assigned to such terms in Chapter 83.

b. First Source Hiring Amended and Restated Agreement

As an essential term of, and consideration for, any contract or property contract with the City, not exempted by the FSHA, the Contractor shall enter into a first source hiring a ("Agreement") with the City, on or before the effective date of the contract or property contract. Contractors shall also enter into an Agreement with the City for any other work that it performs in the City. Such Agreement shall:

(1) Set appropriate hiring and retention goals for entry level positions. The employer shall agree to achieve these hiring and retention goals, or, if unable to achieve these goals, to establish good faith efforts as to its attempts to do so, as set forth in the Agreement. The

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 32 of 42 Agreement shall take into consideration the employer's participation in existing job training, referral and/or brokerage programs. Within the discretion of the FSHA, subject to appropriate modifications, participation in such programs maybe certified as meeting the requirements of this Chapter. Failure either to achieve the specified goal, or to establish good faith efforts will constitute noncompliance and will subject the employer to the provisions of Section 83.10 of this Chapter.

(2) Set first source interviewing, recruitment and hiring requirements, which will provide the San Francisco Workforce Development System with the first opportunity to provide qualified economically disadvantaged individuals for consideration for employment for entry level positions. Employers shall consider all applications of qualified economically disadvantaged individuals referred by the System for employment; provided however, if the employer utilizes nondiscriminatory screening criteria, the employer shall have the sole discretion to interview and/or hire individuals referred or certified by the San Francisco Workforce Development System as being qualified economically disadvantaged individuals. The duration of the first source interviewing requirement shall be determined by the FSHA and shall be set forth in each Agreement, but shall not exceed 10 days. During that period, the employer may publicize the entry level positions in accordance with the Agreement. A need for urgent or temporary hires must be evaluated, and appropriate provisions for such a situation must be made in the Agreement.

(3) Set appropriate requirements for providing notification of available entry level positions to the San Francisco Workforce Development System so that the System may train and refer an adequate pool of qualified economically disadvantaged individuals to participating employers. Notification should include such information as employment needs by occupational title, skills, and/or experience required, the hours required, wage scale and duration of employment, identification of entry level and training positions, identification of English language proficiency requirements, or absence thereof, and the projected schedule and procedures for hiring for each occupation. Employers should provide both long-term job need projections and notice before initiating the interviewing and hiring process. These notification requirements will take into consideration any need to protect the employer's proprietary information.

(4) Set appropriate record keeping and monitoring requirements. The First Source Hiring Administration shall develop easy-to-use forms and record keeping requirements for documenting compliance with the Agreement. To the greatest extent possible, these requirements shall utilize the employer's existing record keeping systems, be nonduplicative, and facilitate a coordinated flow of information and referrals.

(5) Establish guidelines for employer good faith efforts to comply with the first source hiring requirements of this Chapter. The FSHA will work with City departments to develop employer good faith effort requirements appropriate to the types of contracts and property contracts handled by each department. Employers shall appoint a liaison for dealing with the development and implementation of the employer's Agreement. In the event that the FSHA finds that the employer under a City contract or property contract has taken actions primarily for the purpose of circumventing the requirements of this Chapter, that employer shall be subject to the sanctions set forth in Section 83.10 of this Chapter.

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 33 of 42 (6) Set the term of the requirements.

(7) Set appropriate enforcement and sanctioning standards consistent with this Chapter.

(8) Set forth the City's obligations to develop training programs, job applicant referrals, technical assistance, and information systems that assist the employer in complying with this Chapter.

(9) Require the developer to include notice of the requirements of this Chapter in leases, subleases, and other occupancy contracts.

c. Hiring Decisions

Contractor shall make the final determination of whether an Economically Disadvantaged Individual referred by the System is "qualified" for the position.

d. Exceptions

Upon application by Employer, the First Source Hiring Administration may grant an exception to any or all of the requirements of Chapter 83 in any situation where it concludes that compliance with this Chapter would cause economic hardship.

e. Liquidated Damages

Contractor agrees:

(1) To be liable to the City for liquidated damages as provided in this section;

(2) To be subject to the procedures governing enforcement of breaches of contracts based on violations of contract provisions required by this Chapter as set forth in this section;

(3) That the contractor's commitment to comply with this Chapter is a material element of the City's consideration for this contract; that the failure of the contractor to comply with the contract provisions required by this Chapter will cause harm to the City and the public which is significant and substantial but extremely difficult to quantity; that the harm to the City includes not

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 34 of 42 only the financial cost of funding public assistance programs but also the insidious but impossible to quantify harm that this community and its families suffer as a result of unemployment; and that the assessment of liquidated damages of up to \$5,000 for every notice of a new hire for an entry level position improperly withheld by the contractor from the first source hiring process, as determined by the FSHA during its first investigation of a contractor, does not exceed a fair estimate of the financial and other damages that the City suffers as a result of the contractor's failure to comply with its first source referral contractual obligations.

(4) That the continued failure by a contractor to comply with its first source referral contractual obligations will cause further significant and substantial harm to the City and the public, and that a second assessment of liquidated damages of up to \$10,000 for each entry level position improperly withheld from the FSHA, from the time of the conclusion of the first investigation forward, does not exceed the financial and other damages that the City suffers as a result of the contractor's continued failure to comply with its first source referral contractual obligations;

(5) That in addition to the cost of investigating alleged violations under this Section, the computation of liquidated damages for purposes of this section is based on the following data:

A. The average length of stay on public assistance in San Francisco's County Adult Assistance Program is approximately 41 months at an average monthly grant of \$348 per month, totaling approximately \$14,379; and

B. In 2004, the retention rate of adults placed in employment programs funded under the Workforce Investment Act for at least the first six months of employment was 84.4%. Since qualified individuals under the First Source program face far fewer barriers to employment than their counterparts in programs funded by the Workforce Investment Act, it is reasonable to conclude that the average length of employment for an individual whom the First Source Program refers to an employer and who is hired in an entry level position is at least one year;

therefore, liquidated damages that total \$5,000 for first violations and \$10,000 for subsequent violations as determined by FSHA constitute a fair, reasonable, and conservative attempt to quantify the harm caused to the City by the failure of a contractor to comply with its first source referral contractual obligations.

(6) That the failure of contractors to comply with this Chapter, except property contractors, may be subject to the debarment and monetary penalties set forth in Sections 6.80 et seq. of the San Francisco Administrative Code, as well as any other remedies available under the contract or at law; and

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 35 of 42 Violation of the requirements of Chapter 83 is subject to an assessment of liquidated damages in the amount of \$5,000 for every new hire for an Entry Level Position improperly withheld from the first source hiring process. The assessment of liquidated damages and the evaluation of any defenses or mitigating factors shall be made by the FSHA.

f. Subcontracts

Any subcontract entered into by Contractor shall require the subcontractor to comply with the requirements of Chapter 83 and shall contain contractual obligations substantially the same as those set forth in this Section.

43. Prohibition on Political Activity with City Funds

In accordance with San Francisco Administrative Code Chapter 12.G, Contractor may not participate in, support, or attempt to influence any political campaign for a candidate or for a ballot measure (collectively, "Political Activity") in the performance of the services provided under this Agreement. Contractor agrees to comply with San Francisco Administrative Code Chapter 12.G and any implementing rules and regulations promulgated by the City's Controller. The terms and provisions of Chapter 12.G are incorporated herein by this reference. In the event Contractor violates the provisions of this section, the City may, in addition to any other rights or remedies available hereunder, (i) terminate this Agreement, and (ii) prohibit Contractor from bidding on or receiving any new City contract for a period of two (2) years. The Controller will not consider Contractor's use of profit as a violation of this section.

44. Preservative-treated Wood Containing Arsenic

Contractor may not purchase preservative-treated wood products containing arsenic in the performance of this Amended and Restated Agreement unless an exemption from the requirements of Chapter 13 of the San Francisco Environment Code is obtained from the Department of the Environment under Section 1304 of the Code. The Term "preservative-treated wood containing arsenic" shall mean wood treated with a preservative that contains arsenic, elemental arsenic, or an arsenic copper combination, including, but not limited to, chromated copper arsenate preservative, ammoniacal copper zinc arsenate preservative, or ammoniacal copper arsenate preservative. Contractor may purchase preservative-treated wood products on the list of environmentally preferable alternatives prepared and adopted by the Department of the Environment. This provision does not preclude Contractor from purchasing preservative-treated wood containing arsenic for saltwater immersion. The Term "saltwater immersion" shall mean a pressure-treated wood that is used for construction purposes or facilities that are partially or totally immersed in saltwater.

45. Modification of Amended and Restated Agreement

This Agreement may not be modified, nor may compliance with any of its terms be waived, except by written instrument executed and approved by Contractor and by the Director of Transportation, except where the approval of the Municipal Transportation Agency Board or the San Francisco Board of Supervisors is required by applicable law, in which case, the approval of those agencies shall also be required.

46. Administrative Remedy for Agreement Interpretation

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 36 of 42 Should any question arise as to the meaning and intent of this Agreement, the question shall, prior to any other action or resort to any other legal remedy, be referred to the Director of Transportation, who shall decide the true meaning and intent of the Agreement.

47. Agreement Made in California; Venue

The formation, interpretation and performance of this Agreement shall be governed by the laws of the State of California. Venue for all litigation relative to the formation, interpretation and performance of this Agreement shall be in San Francisco.

48. Construction

All paragraph, article, and section captions and headings contained in this Agreement are for reference and convenience only and shall not be considered in construing the scope or meaning of this Agreement. Unless this Agreement specifically provides otherwise, it is to be construed in the following manner.

48.1 Syntax. Whenever the context of this Agreement requires, the singular shall include the plural, the plural shall include the singular, the masculine shall include the feminine and the feminine shall include the masculine.

48.2 References. Unless otherwise indicated, references to Articles, Sections and subsections are to Articles, Sections and subsections in this Agreement.

49. Entire Amended and Restated Agreement

This contract sets forth the entire Agreement between the parties, and supersedes all other oral or written provisions. This contract may be modified only as provided in Section 45.

This Agreement contains the entire Agreement between the parties with respect to the subject matter of this Agreement and any prior Agreements, discussions or understandings, written or oral, are superseded by this Agreement and shall be of no force or effect. No addition or modification of any term or provision of this Agreement shall be effective unless set forth in writing and signed by the parties to this Agreement.

50. Compliance with Laws

Contractor shall keep itself fully informed of the City's Charter, codes, ordinances and regulations and of all state, and federal laws in any manner affecting the performance of this Agreement, and must at all times comply with such local codes, ordinances, and regulations and all applicable laws as they may be amended from time to time. Contractor shall comply and conform with all laws and all governmental regulations, rules and orders that may from time to time be put into effect relating to, controlling or limiting Contractor's performance under this Agreement or enjoyment of any rights or privileges granted hereby. Contractor shall secure all permits and licenses specifically required for Contractor's performance under this Agreement (copies of which shall be promptly provided to City), and shall comply with all applicable laws and regulations relating to labor employed in and relating to Contractor's performance under this Agreement. Contractor shall not use or occupy the Designated Facilities in an unlawful, noisy, improper or offensive manner and shall use its best efforts to prevent any occupancy of the Designated Facilities or use made thereof which is unlawful, noisy, improper or offensive or contrary to any law or ordinance applicable to them. Contractor shall not cause or maintain any nuisance in or about the Designated Facilities, and shall use its best efforts to prevent any person from doing so. Nor shall Contractor cause any rubbish, dirt or refuse to be placed in the streets, sidewalks or alleys adjoining the Designated Facilities or to accumulate in the Designated Facilities.

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 37 of 42

51. Services Provided by Attorneys

Any services required under this Agreement that are to be provided by a law firm or attorney must be reviewed and approved in writing in advance by the City Attorney. No invoices for services provided by law firms or attorneys, including, without limitation, as subcontractors of Contractor, will be paid unless the provider received advance written approval from the City Attorney.

52. Unavoidable Delays

Any prevention, delay or stoppage in a party's performance of any part of this Agreement due to an Unavoidable Delay shall excuse the performance of the party affected for a period of time or otherwise in a manner that bears a causal relationship and is in proportion to any such prevention, delay or stoppage.

53. Severability

Should the application of any provision of this Agreement to any particular facts or circumstances be found by a court of competent jurisdiction to be invalid or unenforceable, then (a) the validity of other provisions of this Agreement shall not be affected or impaired thereby, and (b) such provision shall be enforced to the maximum extent possible so as to effect the intent of the parties and shall be reformed without further action by the parties to the extent necessary to make such provision valid and enforceable.

54. Transition Period

54.1 Notwithstanding any other provision of this Agreement to the contrary, Contractor hereby agrees that it shall cooperate in good faith with City and with any contractor chosen by City pursuant to a competitive selection process to take over Contractor's obligations after the expiration of this Agreement, or any contractor designated by City to provide towing services following termination of this Agreement pursuant to §19 herein, for the smooth and efficient transfer of those functions.

54.2 Contractor agrees to take all actions as may be necessary or as the City may direct for the protection and preservation of any property related to this Agreement that is in the possession of Contractor and in which City has or may acquire an interest, including any environmental remediation required under Appendices C and D. Contractor acknowledges that City has a vested interest for payment of fees for vehicles that Contractor has towed and/or is storing, and Contractor shall not impair said interest and shall take all actions reasonably necessary to safeguard City's interest in said towed and stored vehicles.

54.3 Contractor shall remove any towed and/or stored vehicles from City property by the termination date and shall relocate said vehicles to a location approved by the City or (subject to City approval) authorize a successor contractor to serve as an agent of Contractor for purposes of conducting any lien sale auctions that may be required for disposal of such vehicles.

54.4 Contractor shall provide City or any successor contractor with all information necessary, in a form approved by the City, to facilitate retrieval by the registered owner of any vehicle upon which Contractor holds a lien.

54.5 Contractor agrees to assign all right, title and interest of Contractor in the 450 7^{th} Street property to the City, subject to the payment terms specified in Appendix G, as part of the transition of towing operations to a successor towing services contractor.

55. Administration of Contract

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 38 of 42
This Agreement shall be administered by and for the City by SFMTA. The Executive Director/CEO of SFMTA shall, unless otherwise stated herein, have the authority to act for or on behalf of the City in the administration of this Agreement.

56. Required Actions

Contractor agrees to execute all instruments and documents and to take all actions as may be required or necessary in order to carry out the purposes and terms of this Agreement.

57. Non-impairment of City-Owned Towing Equipment and Facilities

Nothing contained in this Agreement shall be deemed to prohibit, limit or otherwise restrict the use and operation by City of City-owned towing equipment or storage facilities, or the procurement by City of towing services for City-owned vehicles.

58. Deleted by agreement of the parties.

59. Protection of Private Information. Contractor has read and agrees to the terms set forth in San Francisco Administrative Code Sections 12M.2, "Nondisclosure of Private Information," and 12M.3, "Enforcement" of Administrative Code Chapter 12M, "Protection of Private Information," which are incorporated herein as if fully set forth. Contractor agrees that any failure of Contactor to comply with the requirements of Section 12M.2 of this Chapter shall be a material breach of the Contract. In such an event, in addition to any other remedies available to it under equity or law, the City may terminate the Contract, bring a false claim action against the Contractor pursuant to Chapter 6 or Chapter 21 of the Administrative Code, or debar the Contractor.

60. Food Service Waste Reduction Requirements. Contractor agrees to comply fully with and be bound by all of the provisions of the Food Service Waste Reduction Ordinance, as set forth in San Francisco Environment Code Chapter 16, including the remedies provided, and implementing guidelines and rules. The provisions of Chapter 16 are incorporated herein by reference and made a part of this Agreement as though fully set forth. This provision is a material term of this Agreement. By entering Agreement, Contractor agrees that if it breaches this provision, City will suffer actual damages that will be impractical or extremely difficult to determine; further, Contractor agrees that the sum of one hundred dollars (\$100) liquidated damages for the first breach, two hundred dollars (\$200) liquidated damages for the second breach in the same year, and five hundred dollars (\$500) liquidated damages for subsequent breaches in the same year is reasonable estimate of the damage that City will incur based on the violation, established in light of the circumstances existing at the time this Agreement was made. Such amount shall not be considered a penalty, but rather agreed monetary damages sustained by City because of Contractor's failure to comply with this provision.

61. Graffiti Removal. Graffiti is detrimental to the health, safety and welfare of the community in that it promotes a perception in the community that the laws protecting public and private property can be disregarded with impunity. This perception fosters a sense of disrespect of the law that results in an increase in crime; degrades the community and leads to urban blight; is detrimental to property values, business opportunities and the enjoyment of life; is inconsistent with the City's property maintenance goals and aesthetic standards; and results in additional graffiti and in other properties becoming the target of graffiti unless it is quickly removed from public and private property. Graffiti results in visual pollution and is a public nuisance. Graffiti must be abated as quickly as possible to avoid detrimental impacts on the City and County and its residents, and to prevent the further spread of graffiti. Contractor shall remove all graffiti from any real property owned or leased by Contractor in the City and County of San Francisco within forty eight (48) hours of the earlier of Contractor's (a) discovery or notification of the graffiti or (b) receipt of notification of the graffiti from the AMENDED AND RESTATED AGREEMENT

> n:\ptc1\as2010\0900371\00637281.doc Page 39 of 42

Department of Public Works. This section is not intended to require a Contractor to breach any lease or other agreement that it may have concerning its use of the real property. The term "graffiti" means any inscription, word, figure, marking or design that is affixed, marked, etched, scratched, drawn or painted on any building, structure, fixture or other improvement, whether permanent or temporary, including by way of example only and without limitation, signs, banners, billboards and fencing surrounding construction sites, whether public or private, without the consent of the owner of the property or the owner's authorized agent, and which is visible from the public right-of-way. "Graffiti" shall not include: (1) any sign or banner that is authorized by, and in compliance with, the applicable requirements of the San Francisco Public Works Code, the San Francisco Planning Code or the San Francisco Building Code; or (2) any mural or other painting or marking on the property that is protected as a work of fine art under the California Art Preservation Act (California Civil Code Sections 987 et seq.) or as a work of visual art under the Federal Visual Artists Rights Act of 1990 (17 U.S.C. §§ 101 et seq.).

Any failure of Contractor to comply with this section of this Agreement shall constitute an Event of Default of this Agreement.

62. Cooperative Drafting. This Agreement has been drafted through a cooperative effort of both parties, and both parties have had an opportunity to have the Agreement reviewed and revised by legal counsel. No party shall be considered the drafter of this Agreement, and no presumption or rule that an ambiguity shall be construed against the party drafting the clause shall apply to the interpretation or enforcement of this Agreement.

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 40 of 42

IN WITNESS WHEREOF, THE PARTIES HERETO HAVE EXECUTED THIS AMENDED AND RESTATED AGREEMENT ON THE DAY FIRST MENTIONED ABOVE. <u>CITY</u> <u>CONTRACTOR</u>

APPROVED:

NATHANIEL P. #ORD, SR. Executive Director/CEO San Francisco Municipal Transportation Agency

Approved as to Form:

DENNIS J. HERRERA City Attorney

By: Mariam Morley Deputy City Attorney

By signing this Agreement, I certify that I comply with the requirements of the Minimum Compensation Ordinance, which entitles Covered Employees to certain minimum hourly wages and compensated and uncompensated time off.

I have read and understood paragraph 33, the City's statement urging companies doing business in Northern Ireland to move towards resolving employment inequities, encouraging compliance with the MacBride Principles, and urging San Francisco companies to do business with corporations that abide by the MacBride Principles.

JOHN WICKER President and CEO TEGSCO, LLC, d.b.a. San Francisco AutoReturn 945 Bryant Street, Suite 350 San Francisco, CA 94103 Phone No.: 415-575-2355 Employer ID No.: 01-0688299

10-080 SFMTAB Resolution Number: Date:

Attest: $\underline{/c}$

Roberta Boomer, Secretary, S. F. Municipal Transportation Agency Board

Board of Supervisors

Resolution No. Dated:

Clerk of the Board

AMENDED AND RESTATED AGREEMENT

Page 41 of 42 c:\documents and settings\lfuqua\local settings\temporary internet

files\ollc2b\00637281.doc

LIST OF APPENDICES

- A. Scope of Work
- B. Operations Plan
- C. SFMTA-Port MOU
- D. Pier 70 Property License
- E. Letter of Credit & Performance Surety Bond
- F. Rate Schedule
- G. Consolidated Facility Payment Schedule

AMENDED AND RESTATED AGREEMENT

n:\ptc1\as2010\0900371\00637281.doc Page 42 of 42

SERVICE AGREEMENT AND PROPERTY USE LICENSE

FOR TOWING, STORAGE AND DISPOSAL

OF ABANDONED AND ILLEGALLY PARKED VEHICLES

by and between the

CITY AND COUNTY OF SAN FRANCISCO

AND

TEGSCO, LLC

d.b.a. SAN FRANCISCO AUTORETURN, a California Limited Liability Company

Dated: July 31, 2005

1.	DE	FINITIONS	
1.	.1	ABANDONED VEHICLE	.1
1.		ADMINISTRATIVE HOLD	2
1.		AGREEMENT.	
1.		CENTRAL DISPATCH	
1.		CITATION	$\tilde{2}$
1.		Стту	
· 1.		CLAIM	
1.		COMPLAINT	
1.	-	Controller	
	10	CITATION AND TOWING MANAGEMENT SYSTEM (CTMS)	
	11	CUSTOMER	
	12	CUSTOMER SERVICE CENTER.	$\overline{2}$
	13	Courtesy Tows	
	14	DAYS	
1	15	DEFICIENCY CLAIM	
	16	DPT.	3
	17	DELINQUENT CITATION	
1.		DESIGNATED FACILITIES	
1.		DIRECTOR OF TRANSPORTATION	
1.2		DISPATCH TOWS	
1.2		DMV	
1.2		DPH	
1.2		DPT ADMINISTRATIVE FEE	
1.2		ECAD	
1.2		EFFECTIVE DATE	
1.2		Employees	
1.2		EXPEDITED TOW	
1.2		GROSS REVENUES	3
1.2	29	LETTER OF CREDIT	4
1.3		LIEN 1 VEHICLE	
1.3		LIEN 2 VEHICLE	
1.3	32	LIEN 3 VEHICLE	4
1.3	33	LIEN CATEGORY	4
1.3	34	MANDATORY FEE	4
1.3	35	MTA	4
1.3	6	OCCURRENCE	4
1.3	37	OPERATIONS PLAN	5
1.3	8 1	PEAK HOURS	5
1.3		PEO CONTRACT	5
1.4		PEO CONTRACTOR	
1.4		POLICE HOLD VEHICLE	
1.4		PORT	
1.4	3	PORT MOU	5
1.4		PRIMARY STORAGE FACILITY	
1,4	5	Ргојест 20	5
1.4		PROPERTY, PROPERTIES	
1.4		PROPOSAL	
1.4		RECORDS	
1.4		REFERRAL FEE	
1.5		RFP	
1.5		REGIONAL SWEEPS	
1.5		ROUTINE TOWS	
1.5		SCHEDULED TOWS	
1.5	4	SCOFFLAW	6

TABLE OF CONTENTS

TOWING SERVICES MASTER AGREEMENT Page i of iv

1.	.55 SECONDARY STORAGE FACILITY	6
1.	.57 SFPD	6
1,	.58 SFPD Traffic Offender Fee	
	.59 Special Events Tows	
	.60 TERM	
	.61 TOW CAR	
	62 TOW EQUIPMENT	
	63 TOW REQUEST	6
	65 TRAFFIC CODE	0
	66 TREASURER	
	67 UNAVOIDABLE DELAYS	
	68 VEHICLE CODE	
1,	69 VIN	
	70 WAN	
1.	71 Zone Tows	7
2.	TERM OF THE AGREEMENT	7
3.	EFFECTIVE DATE OF AGREEMENT	7
4.	SERVICES CONTRACTOR AGREES TO PERFORM	
	SUBMITTING FALSE CLAIMS; MONETARY PENALTIES	
6. -	TAXES AND ASSESSMENTS	
	PAYMENT DOES NOT IMPLY ACCEPTANCE OF WORK	
8.	QUALIFIED PERSONNEL	
8.		
8.2		
8.		
8.4		
8.: 8.(
	· · · · · · · · · · · · · · · · · · ·	
9.	RESPONSIBILITY FOR EQUIPMENT	
10.	INDEPENDENT CONTRACTOR; PAYMENT OF TAXES AND OTHER EXPENSES	
. 10		
10		
10		
10	.4 PAYROLL AND TAXATION	.11
11.	REQUIRED INSURANCE	.11
12.	FINANCIAL ASSURANCES	.12
12	.1 REQUIREMENT TO PROVIDE FINANCIAL GUARANTEES	.12
12		
	12.2.1 Performance Guarantee	12
	12.2.1 Ferformance Guarantee	13
	12.2.3 Expiration or Termination of Letter of Credit	
	12.2.4 Demands Upon Letter of Credit.	
	12.2.5 Depletion of Letter of Credit	
	12.2.6 Dispute Resolution	
12	.3 MAINTENANCE DEPOSIT	.14
12		
12		
13.	INSURANCE AND PERFORMANCE GUARANTEE REQUIREMENTS	.15
	TOWING SERVICES MASTER AGREEMENT	
	Page ii of iv	

14.	CONTRACTOR'S REPRESENTATIONS AND WARRANTIES	15
14.1 14.1 14.1 14.2 14.2 14.2	 FORMATION	15 15 15 16 16
15.	INDEMNIFICATION	16
16.	INCIDENTAL AND CONSEQUENTIAL DAMAGES	16
17.	LIABILITY OF CITY	16
18.	DEFAULT; REMEDIES	17
19.	TERMINATION	
19.1 19.2 19.3	2 TERMINATION NOT EXCLUSIVE REMEDY	18 18
20.	PROVISIONS OF AGREEMENT SURVIVING TERMINATION OR EXPIRATION	
21.	CONFLICT OF INTEREST	
22.	PROPRIETARY OR CONFIDENTIAL INFORMATION	
23.	NOTICES TO THE PARTIES	
24.	OWNERSHIP OF RESULTS	
25.	WORKS FOR HIRE	-
26.	AUDIT AND INSPECTION OF RECORDS	
26.1 26.2		
26.3	AUDIT	20
26.4		
27.	SUBCONTRACTING	
28.	ASSIGNMENT	
	NON-WAIVER OF RIGHTS	
	EARNED INCOME CREDIT (EIC) FORMS	
	MINORITY/WOMEN/LOCAL BUSINESS UTILIZATION; LIQUIDATED DAMAGES.2	
31.1 31.2	COMPLIANCE	
31.3	SUBCONTRACTING GOALS	22
31.4 31.5		
	NONDISCRIMINATION; PENALTIES	
32.1	CONTRACTOR SHALL NOT DISCRIMINATE	
32.2	NONDISCRIMINATION IN BENEFITS	3
32.3 32.4	CONDITION TO CONTRACT	
	MACBRIDE PRINCIPLES-NORTHERN IRELAND	
	TROPICAL HARDWOOD AND VIRGIN REDWOOD BAN	
	DRUG-FREE WORKPLACE POLICY	
	TOWING SERVICES MASTER AGREEMENT	

36.	RESOURCE CONSERVATION	24
37.	COMPLIANCE WITH AMERICANS WITH DISABILITIES ACT	24
38.	SUNSHINE ORDINANCE	24
38. 38. 38.	2 DOCUMENTS SUBJECT TO DISCLOSURE BY CONTRACTOR.	24
39.	NOTIFICATION OF LIMITATIONS ON CONTRIBUTIONS	.24
40.	REQUIRING MINIMUM COMPENSATION FOR COVERED EMPLOYEES	.25
41.	REQUIRING HEALTH BENEFITS FOR COVERED EMPLOYEES	.27
42.	FIRST SOURCE HIRING PROGRAM	.28
42.1 42.2 42.3 42.4 42.4	 FIRST SOURCE HIRING AGREEMENT	.28 .28 .28
43.	PROHIBITION ON POLITICAL ACTIVITY WITH CITY FUNDS	
44.	PRESERVATIVE-TREATED WOOD CONTAINING ARSENIC	.29
45.	MODIFICATION OF AGREEMENT	.29
46.	ADMINISTRATIVE REMEDY FOR AGREEMENT INTERPRETATION	.29
47.	AGREEMENT MADE IN CALIFORNIA; VENUE	.29
48.	CONSTRUCTION	.29
48.1 48.2	References	.30
49.	ENTIRE AGREEMENT	.30
50.	COMPLIANCE WITH LAWS	.30
51.	SERVICES PROVIDED BY ATTORNEYS	.30
52.	UNAVOIDABLE DELAYS	.30
53.	SEVERABILITY	.30
54.	TRANSITION PERIOD	.31
55.	ADMINISTRATION OF CONTRACT	
56.	REQUIRED ACTIONS	31
57.	NON-IMPAIRMENT OF CITY-OWNED TOWING EQUIPMENT AND FACILITIES.	31

SERVICE AGREEMENT AND PROPERTY USE LICENSE FOR TOWING, STORAGE, AND DISPOSAL OF ABANDONED AND ILLEGALLY PARKED VEHICLES BY AND BETWEEN THE CITY AND COUNTY OF SAN FRANCISCO AND TEGSCO, LLC, D.B.A. SAN FRANCISCO AUTORETURN

This Service Agreement And Property Use License for Towing, Storage, and Disposal of Abandoned and Illegally Parked Vehicles ("Agreement"), dated for convenience as **July 31, 2005**, is entered into by and between the City and County of San Francisco, a municipal corporation, acting by and through its Municipal Transportation Agency, Department of Parking and Traffic, hereinafter referred to as "City", and TEGSCO, LLC, a California limited liability company doing business as San Francisco AutoReturn, hereinafter referred to as "Contractor", in the City and County of San Francisco, State of California, for the services and under the terms described herein.

Recitals

WHEREAS, The City issued a Request for Proposals for Towing, Storage and Disposal of Abandoned and Illegally Parked Vehicles on September 18, 2002 ("RFP"); and

WHEREAS, Contractor submitted a proposal dated March 10, 2003 ("Proposal") which was selected by City as the highest-ranked proposal among the proposals submitted in response to the RFP; and

WHEREAS, In the course of negotiating this Service Agreement And Property Use License For Towing, Storage and Disposal of Abandoned and Illegally Parked Vehicles the City's existing towing services contractor, The City Tow, informed the City on January 23, 2004 that it would cease to provide towing services as of March 21, 2004; and

WHEREAS, The cessation of towing services under the then-existing contract with The City Tow created emergency circumstances that required the City and Contractor, with the approval of the San Francisco Board of Supervisors, to enter into an Emergency Interim Service Agreement and Property Use License for Towing, Storage and Disposal of Abandoned and Illegally Parked Vehicles ("Emergency Interim Agreement") pending negotiation of the long-term contract pursuant to the RFP; and

WHEREAS, Contractor has been providing towing services to City pursuant to the Emergency Interim Agreement since March 22, 2004; and

WHEREAS, The parties have negotiated this long-term towing services agreement in accordance with the terms of the RFP and Proposal;

NOW, THEREFORE, based on the foregoing, and in consideration of the promises, covenants, and undertakings contained in this Agreement, City and Contractor hereby agree as follows:

AGREEMENT

1. Definitions

The following words and phrases shall have the meanings set forth below when used in this Agreement:

1.1 <u>Abandoned Vehicle</u>: A vehicle that qualifies to be towed pursuant to San Francisco Traffic Code § 159 because it is parked or left standing on a public street or highway in the City for more than 72 consecutive hours, or is deemed to be abandoned pursuant to Vehicle Code § 22669(a), or is found to be a public nuisance on private property pursuant to Vehicle Code § 22660 and Traffic Code § 234.

1.2 <u>Administrative Hold</u>: A hold placed on a vehicle impounded by DPT or SFPD pursuant to which the vehicle may not be released prior to the passage of a specified period of time, fulfillment of statutory requirements or upon written authorization by the impounding agency. The procedures and requirements for Administrative Holds are further described in Section 4.7 of Appendix A to this Agreement.

1.3 <u>Agreement</u>: This document, its Appendices, Exhibits and Attachments to Appendices, the Operations Plan as adopted in accordance with Section 14 of Appendix A, the RFP, the Proposal, and other documents attached hereto or specifically incorporated herein by reference.

1.4 <u>Central Dispatch</u>: Contractor's primary facility for taking Tow Requests and dispatching Tow Cars, as further defined in Section 11.1(d) of Appendix A.

1.5 <u>Citation</u>: A ticket for a parking violation processed by DPT.

1.6 <u>City</u>: The City and County of San Francisco, acting by and through its Department of Parking and Traffic, a division of the Municipal Transportation Agency.

1.7 <u>Claim</u>: Any claim brought against the City or Contractor or their respective agents, contractors or employees for theft of property or injury to any property or person arising out of Contractor's performance of services under this Agreement. For the purpose of Appendix A, § 5.2, "Claim" shall mean a request for compensation for personal injury, loss from or damage to towed vehicle and/or personal property, costs of recovering a vehicle that was not dropped in violation of Appendix A, Section 4.2, or the value of a vehicle that was lost in violation of Section 4.1 of Appendix A.

1.8 <u>Complaint</u>: Any problem with service provided by Contractor within the scope of this Agreement that is communicated to Contractor or City by a Customer and which is not a Claim.

1.9 <u>Controller</u>: The Controller of the City and County of San Francisco or any duly authorized agent thereof.

1.10 <u>Citation and Towing Management System (CTMS)</u>: DPT's Ticket Information Management System (TIMS) or successor system(s) used by DPT for tracking Tow Requests and Citations.

1.11 <u>Customer</u>: A member of the public who is associated with an impounded vehicle, which may include either the registered owner, the driver of the vehicle at the time that it is stopped or towed, a person who appears on behalf of the owner or the driver of the vehicle, or a purchaser of a vehicle at lien sale auction.

1.12 <u>Customer Service Center</u> The location for Customers recovering vehicles in person to pay fees and process required forms for release of a vehicle, as further described in Appendix A Section 11.1(a) and located as of the Effective Date at 850 Bryant Street, Room 145, San Francisco (Hall of Justice).

1.13 <u>Courtesy Tows</u>: Tows requested for disabled vehicles owned or used in an official capacity by either SFPD or DPT.

1.14 Days: Consecutive calendar days, including weekends and holidays, unless otherwise specified.

1.15 <u>Deficiency Claim</u>: A claim by Contractor against a registered vehicle owner equal to towing and storage charges, less any amount received from the sale of the vehicle, and which is subject to all rights and limitations set forth in California Civil Code § 3068.2 or any successor statute that creates, defines and limits Contractor's right to such claim.

TOWING SERVICES MASTER AGREEMENT

Page 2 of 33

1.16 <u>DPT</u>: Department of Parking and Traffic, a division of the Municipal Transportation Agency of the City and County of San Francisco, or any duly authorized agent thereof.

1.17 <u>Delinquent Citation</u>: A Citation that was unpaid past the original due date for payment, upon which penalties for overdue payment have accrued, and which is not scheduled for administrative review or hearing by DPT.

1.18 <u>Designated Facilities</u>: Real property and any buildings and improvements thereon used by Contractor in the performance of this Agreement, as further described in Appendix A, Section 11.1.

1.19 <u>Director of Transportation</u>: The Chief Executive Officer of the Municipal Transportation Agency of the City and County of San Francisco.

1.20 <u>Dispatch Tows</u>: Individual Tow Requests that are communicated to Contractor.

1.21 <u>DMV</u>: California Department of Motor Vehicles.

1.22 DPH: Department of Public Health of the City and County of San Francisco.

1.23 <u>DPT Administrative Fee</u>: The fee described in Section 12.1(c) of Appendix A to this Agreement, as authorized by San Francisco Traffic Code § 170.1 and Vehicle Code § 22850.5, or successor statutes.

1.24 ECAD: Electronic Computer Aided Dispatch.

1.25 Effective Date: July 31, 2005, or the date upon which all required approvals are obtained and all signatures of the parties have been affixed hereto, whichever date is later.

1.26 <u>Employees:</u> The persons directed and controlled by Contractor and/or its PEO Contractor who are providing work and services on behalf of Contractor in order to perform the work required under this Agreement, not including entities that are independent subcontractors. The classes and job descriptions of persons who are Employees of Contractor may include, but are not limited to:

(a) Cashiers, bookkeepers, and accountants.

(b) Persons who move vehicles at and between Designated Facilities (also known as "Yardmen").

- (c) Forklift and heavy equipment operators.
- (d) Clerks, secretaries, telephone operators and administrative and information technology/computer support personnel.
- (e) Tow Car operators directly employed by Contractor and/or its PEO Contractor.

(f) Guards and security personnel.

1.27 Expedited Tow: A tow needed to ensure public or officer safety and/or to eliminate a hazard.

1.28 <u>Gross Revenues</u>: Contractor's gross receipts from all fees and proceeds under this Agreement, including all revenues derived from lien sales conducted pursuant to this Agreement. "Gross Revenues" do not include DPT or SFPD Traffic Offender Fees or fees payable by vehicle owners or operators collected by Contractor on behalf of the City pursuant to the terms of this Agreement.

1.29 <u>Headquarters Office</u>: Contractor's administrative office, currently located at 945 Bryant Street, San Francisco, California.

1.30 <u>Investigative Police Hold</u>: A Police Hold imposed on an evidentiary vehicle for the purpose of criminal investigation, as further described in Appendix A, Section 4.6.

1.31 <u>Letter of Credit</u>: The letter of credit which Contractor is required to maintain to guarantee the performance of Contractor's obligations under this Agreement, as further described in Section 12.1 and 12.2 and in Appendix C, Section 6 of this Agreement. The Letter of Credit and all replacement Letters of Credit provided by Contractor during the Term of this Agreement shall be attached hereto as Appendix G and is incorporated by reference as though fully set forth herein.

1.32 Lien 1 Vehicle: A low-value vehicle, including a vehicle valued at five hundred dollars (\$500) or less, in accordance with Vehicle Code § 22670 (requiring valuation of any vehicle towed by a public agency) and § 22851.2 (regarding vehicles valued at an amount not exceeding five hundred dollars (\$500) and not towed for being abandoned) or a vehicle valued at five hundred dollars (\$500) or less pursuant to § 22851.3 (regarding vehicles towed for being abandoned) and 22851.6-22851.10 (regarding disposal procedures for low-value vehicles). If the Vehicle Code is amended subsequent to this Agreement to change the dollar amounts which trigger requirements for low-value vehicles, this Agreement shall incorporate such amendments by reference as though fully set forth herein for the purpose of defining dollar-value thresholds and legally required procedures for handling and disposal of low-value vehicles.

1.33 Lien 2 Vehicle: A medium-value vehicle valued at more than five hundred dollars (\$500) and up to and including four thousand dollars (\$4,000) in accordance with Vehicle Code § 22670 (requiring valuation of any vehicle towed by a public agency), or over five hundred dollars (\$500) and up to and including four thousand dollars (\$4,000) for the purpose of Vehicle Code § 22851.3 (regarding vehicles towed for being abandoned), and California Civil Code §§ 3067-3075 (setting forth legally required procedures for lien sales of towed vehicles). If the Vehicle Code or the California Civil Code are amended subsequent to this Agreement to change the dollar amounts which trigger requirements for medium-value vehicles, this Agreement shall incorporate such amendments by reference as though fully set forth herein for the purpose of defining dollar-value thresholds and legally required procedures for handling and disposal of medium-value vehicles.

1.34 Lien 3 Vehicle: A high-value vehicle valued at more than four thousand dollars (\$4,000), in accordance with Vehicle Code § 22670 (requiring valuation of any vehicle towed by a public agency), and California Civil Code Sections 3067-3075, setting forth required procedures for lien sales of vehicles. If the Vehicle Code or the California Civil Code are amended subsequent to this Agreement to change the dollar amounts which trigger requirements for high-value vehicles, this Agreement shall incorporate such amendments by reference as though fully set forth herein for the purpose of defining dollar-value thresholds and legally required procedures for handling and disposal of high-value vehicles.

1.35 <u>Lien Category</u>: The classification of a vehicle as a Lien 1, Lien 2 or Lien 3 vehicle in accordance with its appraised value.

1.36 <u>Mandatory Fee</u>: All fees that must be paid by Customer before Contractor may release any vehicle to the owner or purchaser of a vehicle. Mandatory Fees include: boot fees, DPT Administrative Fee, SFPD Traffic Offender Fee, towing, storage, transfer and lien fees, returned check charges and Delinquent Citation fees, as applicable.

1.37 <u>MTA</u>: The San Francisco Municipal Transportation Agency, an agency of the City and County of San Francisco.

1.38 Occurrence: Any accident or incident occurring in a single place at a single time from a single event that results in one or more claims for injury to persons or property. An accident or incident

that results in multiple claims shall be considered a single Occurrence for purposes of applying any deductible provisions of Contractor's insurance coverages.

1.39 Operations Plan: The collected written procedures manuals, lists and schedules required to be submitted by Contractor and approved by City as set forth in Appendix A, Section 14, and which are listed in Appendix B to this Agreement. When approved by City, the Operations Plan shall define the service standards for the work to be performed under this Agreement, and is hereby incorporated into this Agreement as though fully set forth herein.

1.40 Peak Hours: Monday through Friday, 7:00 a.m. to 9:00 a.m. and 3:00 p.m. to 7:00 p.m., excluding City holidays.

1.41 <u>PEO Contract</u>: Any agreement with any professional employment organization (PEO) which provides payroll, workers' compensation, work place safety, staffing and other human resources services, for the purpose of performing the services required by this Agreement. "PEO Contract" does not include contracts with subcontractors for the provision of Tow Cars, Tow Equipment and related towing services.

1.42 **PEO Contractor**: An entity with whom Contractor has entered into a PEO Contract.

1.43 <u>Police Hold</u>: A hold placed on a vehicle by the SFPD in writing which requires a vehicle to be processed in accordance with the Police Hold procedures specified in this Agreement, or a vehicle with no visible VIN which is held for inspection by the SFPD.

1.44 Port: The San Francisco Port Commission, an agency of the City and County of San Francisco.

1.45 <u>Port MOU</u>: That certain Memorandum of Understanding (Port Ref. No. MOU M-13828) by and between DPT and the Port, a copy of which is attached hereto as Appendix C and incorporated by reference as though fully set forth herein.

1.46 <u>Primary Storage Facility</u>: The facilities primarily used for short term storage of impounded vehicles currently located at 415 7th Street and 450 7th Street, San Francisco, California, as further described in Section 11.1(b) of Appendix A.

1.47 Project 20: The DPT program which allows vehicle owners and operators to perform community service in lieu of paying Citation fees, in accordance with Vehicle Code § 40215(c)(7).

1.48 Property, Properties: Real property owned by the City and licensed to Contractor for the purpose of this Agreement, collectively, the 415 7th Street, 850 Bryant Street (Hall of Justice) and Pier 70 as of the Effective Date of this Agreement, and any other properties that may be licensed to Contractor by City for the purpose of this Agreement during its Term.

1.49 Proposal: The proposal submitted by Contractor in response to the RFP, dated March 10, 2003, incorporated by reference into this Agreement as though fully set forth herein.

1.50 <u>Records</u>: The documents Contractor is required to create and maintain under this Agreement, including but not limited to: (1) complete and accurate books, accounts and documentation of financial transactions relating to all items of income received and expenses incurred in the performance of this Agreement; (2) documentation of all vehicles towed; (3) documentation of all vehicles stored; (3) documentation of all Claims; (4) all Monthly Management Reports and other reports Contractor is required to submit to City; (5) charts and diagrams of any property licensed to Contractor by City to fulfill the obligations of this Agreement; (6) other documents or reports as City may require Contractor to produce in the course of performing work under the Agreement; and (7) the Records described in Section 13 of Appendix A to this Agreement.

1.51 <u>Referral Fee</u>: The fee established in Section 12.1(a) of Appendix A to this Agreement.

1.52 <u>RFP</u>: The City and County of San Francisco's Request for Proposals for Towing, Storage and Disposal of Abandoned and Illegally Parked Vehicles, dated September 18, 2002, and all addenda thereto, incorporated by reference into this Agreement as though fully set forth herein.

1.53 <u>Regional Sweeps</u>: Pre-arranged parking or traffic enforcement operations in designated areas.

1.54 <u>Routine Tows</u>: Scheduled Tows and Dispatch Tows.

1.55 <u>Scheduled Tows</u>: Pre-arranged tow events at times and in places designated by City, including but not limited to Regional Sweeps, Peak Hour tows, Zone Tows, Special Events Tows and Abandoned Vehicle tows.

1.56 <u>Scofflaw</u>: A vehicle towed for multiple Delinquent Citations pursuant to subsection (i) of Vehicle Code § 22651.

1.57 <u>Secondary Storage Facility</u>: The facility primarily used for long term storage of impounded vehicles, located at Pier 70, San Francisco, California as of the Effective Date of this Agreement, as further described in Appendix A, Section 11.1(c).

1.58 SFPD: The Police Department of the City and County of San Francisco or any duly authorized officer or member thereof.

1.59 SFPD Traffic Offender Fee: The fee described in Section 12.1(d) of Appendix A to this Agreement, as authorized by Traffic Code §§ 170.2-A and 170.2-B and Vehicle Code § 22850.5, or successor statutes.

1.60 Special Events Tows: Tows of vehicles parked in violation of temporary parking restrictions, as authorized by Traffic Code §§ 33(c) and 130 or successor ordinances.

1.61 <u>Term</u>: The duration of this Agreement as established in Section 2 herein, and any additional period during which Contractor completes repair, remediation or other work required for the termination of the License Agreements for the Properties, as set forth in Appendices C, D and E of this Agreement, and the tasks listed in Section 54 of this Agreement for transition to a successor contractor or successor Agreement. "Term" shall include any extension of this Agreement pursuant to Section 2.

1.62 <u>Tow Car</u>: A motor vehicle which has been altered or designed and equipped for and exclusively used in the business of towing vehicles by means of a crane, hoist, tow bar, tow line, or dolly or is otherwise exclusively used to render assistance to other vehicles, and as defined in Vehicle Code § 615.

1.63 <u>Tow Equipment</u>: Tow Cars and all appurtenant computer systems, communications devices, hand tools, electric tools and towing hardware, whether or not expressly listed in this Agreement, which are necessary to perform towing services to the standards of the towing industry and as set forth in this Agreement.

1.64 <u>Tow Request</u>: A request directed to Contractor from DPT, DPH or SFPD for service by a Tow Car or Tow Equipment for the removal or relocation of a vehicle.

1.65 <u>Towed Vehicle Management System (TVMS)</u>: The electronic database system to be used by Contractor to meet record keeping, reporting and vehicle handling requirements of this Agreement, as further described in Section 10 of Appendix A.

1.66 <u>Traffic Code</u>: The Traffic Code of the City and County of San Francisco.

TOWING SERVICES MASTER AGREEMENT

Page 6 of 33

1.67 <u>Treasurer</u>: The Treasurer of the City and County of San Francisco.

1.68 Unavoidable Delays: With respect to a delay in performance, "Unavoidable Delay" shall mean any delay that is attributable to any: (a) strike, lockout or other labor or industrial disturbance (whether or not on the part of the employees of either party hereto or their contractors), civil disturbance, future order claiming jurisdiction, act of the public enemy, war, riot, sabotage, blockage, embargo, inability to secure customary materials, supplies or labor through ordinary sources by reason of regulation or order of any government or regulatory body; (b) changes in any applicable laws or the interpretation thereof; or (c) lightning, earthquake, fire, storm, hurricane, tornado, flood, washout, explosion; or (d) a governmental taking by eminent domain of all or a substantial portion of a Designated Facility; or (e) any interference with Contractor's right to access a substantial portion of a Designated Facility owned or provided for Contractor's use by City, or any other cause beyond the reasonable control of the party from whom performance is required, or any of its contractors or other representatives.

1.69 <u>Vehicle Code</u>: The Vehicle Code of the State of California.

1.70 <u>VIN</u>: The distinguishing number or other mark used for the purpose of uniquely identifying a vehicle or vehicle part, as further defined in Vehicle Code § 671.

1.71 <u>WAN</u>: Wide Area Network.

1.72 Zone Tows: Tows of vehicles parked in zones listed in Traffic Code § 71b.

GENERAL PROVISIONS

2. Term of the Agreement

The Term of this Agreement shall be from the Effective Date of July 31, 2005 through July 31, 2010. At the option and sole discretion of the City, the Term of this Agreement may be extended. Any extension of this Agreement must be in writing. In the event that this Agreement is extended beyond July 31, 2010, the City agrees to submit a request for an increase in rates to the MTA Board to reflect any increased cost to Contractor for increased rental payments for the Properties during the extended Term that are in excess of CPI adjustments to be applied during the terms of Appendices C, D and E.

3. Effective Date of Agreement

This Agreement shall become effective upon the Effective Date.

4. Services Contractor Agrees to Perform

The Contractor shall tow and store any vehicles that DPT, DPH or the SFPD orders removed from any public street or highway or from private property within the City, in accordance with the requirements of the Traffic Code and the Vehicle Code, and shall perform such other related services as are described in this Agreement, in accordance with Appendix A, "Scope of Work," and Appendix B "Operations Plan". Appendix A, Appendix B, and the Operations Plan to be adopted and amended as provided in Section 14 of Appendix A are all attached hereto and incorporated by reference as though fully set forth herein. Rates and charges to the public for services under this Agreement shall be as set forth in Appendix H, attached hereto and incorporated by reference as though fully set forth herein.

All vehicle handling and storage required by this Agreement shall be conducted at one of the Designated Facilities. As of the Effective Date of this Agreement, City hereby licenses to Contractor the use of three (3) real properties on which the Designated Facilities are currently located: Pier 70, 850 Bryant Street, Room 145, and 415 7th Street, San Francisco. The license agreements for those Properties are attached here to as Appendices D and E, respectively, and are incorporated by reference as though

fully set forth herein. The parties acknowledge that during the Term of this Agreement Contractor may, with City's prior written approval and subject to all the conditions contained in such approval, consolidate the Customer Service Center and Primary Storage Facility operations presently conducted at 415 7th Street and 850 Bryant Street to a property owned by Caltrans and leased by Contractor at 450 7th Street ("Consolidation"). Following Consolidation, should City cease to use Contractor for towing services, the terms for the City's continued use of the 450 7th Street property are attached hereto as Appendix I and are incorporated by reference as though fully set forth herein.

City hereby consents to Contractor's towing and storage of vehicles that are not towed or stored pursuant to a Tow Request with the prior written approval of City and subject to any conditions imposed in such written approval; provided however, that (i) Contractor shall at all times conduct itself in accordance with the Customer service standards of this Agreement so as not create any negative effect on Contractor's public image and reputation as the City's towing contractor, and (ii) Contractor's operations with respect to such vehicles shall not create any adverse impact on its performance of all requirements of this Agreement. City may revoke its consent at any time without cause by written notice to Contractor. The following sections of this Agreement shall apply to any towing or storage of vehicles by Contractor within the City and County of San Francisco that are not towed or stored pursuant to a Tow Request: Master Agreement Sections 8, 9, 11, 12, 13, 14, 15, 16, 17, 20, 22, 26, 27, 28, 29, 31, 32, 35, 37, 38, 40, 41, 42, 50; Appendix A Sections 1, 3, 4, 5, 6, 7, 9, 11, 12.1(b), 12.3, 12.5, 12.6, 13, 14; Appendix B, Appendices C, D and E during their respective terms, Appendix G and Appendix H.

5. Submitting False Claims; Monetary Penalties

Pursuant to San Francisco Administrative Code §21.35, any contractor, subcontractor or consultant who submits a false claim shall be liable to the City for three times the amount of damages which the City sustains because of the false claim. A contractor, subcontractor or consultant who submits a false claim shall also be liable to the City for the costs, including attorneys' fees, of a civil action brought to recover any of those penalties or damages, and may be liable to the City for a civil penalty of up to \$10,000 for each false claim. A contractor, subcontractor or consultant will be deemed to have submitted a false claim to the City if the contractor, subcontractor or consultant: (a) knowingly presents or causes to be presented to an officer or employee of the City a false claim or request for payment or approval; (b) knowingly makes, uses, or causes to be made or used a false record or statement to get a false claim paid or approved by the City; (c) conspires to defraud the City by getting a false claim allowed or paid by the City; (d) knowingly makes, uses, or causes to be made or used a false record or statement to conceal, avoid, or decrease an obligation to pay or transmit money or property to the City; or (e) is a beneficiary of an inadvertent submission of a false claim to the City within a reasonable time after discovery of the false claim.

6. Taxes and Assessments

Contractor agrees to pay prior to delinquency all taxes and assessments of any kind, including but not limited to possessory interest taxes and non-property taxes, which lawfully may be imposed on the Designated Facilities or on Contractor. Contractor is hereby granted City's consent to contest the validity or the amount of any such tax or assessment, if Contractor chooses to do so. Contractor acknowledges and understands that a possessory interest subject to property taxation may or may not be created by the Agreement and that Contractor may or may not be subject to the payment of property taxes levied on such possessory interest. Contractor further acknowledges that Contractor is familiar with San Francisco Administrative Code §§ 23.38-23.39, which require that City submit a report, which includes specified information relating to the creation, renewal, sublease, or assignment of any such possessory interest, to the County Assessor within 60 days after any such transaction. Contractor agrees to provide to City the information required by Administrative Code §§ 23.38-23.39 within thirty (30) days after any such transaction, if requested to do so by City.

7. Payment Does Not Imply Acceptance of Work

The acceptance of any payment by City and/or Contractor's deduction or offset of funds (that would otherwise be paid by City to Contractor as payment for services) from revenues generated under the Agreement shall in no way lessen the liability of Contractor to replace unsatisfactory work, equipment, or materials, although the unsatisfactory character of such work, equipment or materials may not have been apparent or detected at the time such payment was received. Work that does not conform to the requirements of this Agreement may be rejected by City and in such case Contractor must take immediate corrective action.

8. Qualified Personnel

Work under this Agreement shall be performed only by competent personnel under the supervision of and in the employment of Contractor, or through subcontracts entered into pursuant to Section 27. Contractor will comply with City's reasonable requests regarding assignment of personnel, including the removal of specified personnel upon request of City, but all personnel, including those assigned at City's request, must be supervised by Contractor. Contractor shall commit adequate resources to meet its obligations under this Agreement.

8.1 Contractor Is the Employer of the Employees

Notwithstanding any language in any PEO Contract, and notwithstanding the intent of Contractor and/or any contractor for PEO Contract services in entering into a PEO Contract or as memorialized within a PEO Contract or as a PEO Contract may be construed by any person, Contractor affirms, agrees and warrants that for all purposes under the Agreement and under relevant law applicable to the Agreement, Contractor is the employer of the Employees. Contractor affirms, agrees and warrants that for all purposes of the Agreement, specifically including but not limited to Contractor's duties under Sections 4, 5, 8, 9, 10, 14, 15, 16, 18, 21, 22, 26, 30, 32, 35, 37, 40, 41, 42, 43, 49, 50 of this Agreement, and Section 8 of Appendix A to this Agreement, Contractor is the employer of the Employees.

8.2 PEO Contract Shall Not Control or Modify the Agreement

Notwithstanding any language in any PEO Contract, and notwithstanding the intent of Contractor and/or a PEO Contractor in entering into a PEO Contract or as memorialized within a PEO Contract or as a PEO Contract may be construed by any person, Contractor affirms, agrees and warrants that all provisions of this Agreement are in full force and effect, and that a PEO Contract does not modify Contractor's obligations and duties under this Agreement. Contractor further agrees, affirms and warrants that a PEO Contract does not alter the City's rights under this Agreement against Contract does not alter the City's right to enforce all provisions of this Agreement against Contractor directly.

8.3 PEO Contract is a Subcontract to the Agreement

Notwithstanding any language in any PEO Contract, and notwithstanding the intent of Contractor and/or a PEO Contractor in entering into a PEO Contract or as memorialized within a PEO Contract or as a PEO Contract may be construed by any person, Contractor affirms, agrees and warrants that for all purposes under this Agreement and under relevant law applicable to this Agreement, any PEO Contract is a subcontract for the purpose of providing payroll, workers' compensation, work place safety and other human resources services to Contractor. Pursuant to Section 27 of this Agreement, the City hereby approves Contractor's PEO Contract with Gevity HR, which is attached hereto as Appendix F and incorporated by reference as though fully set forth herein. The parties stipulate that no agreement or legal relationship exists between the City and Gevity HR.

8.4 Contractor is Liable for All Actions of PEO Contractor

Notwithstanding any language in any PEO Contract, and notwithstanding the intent of Contractor and/or a PEO Contractor in entering into a PEO Contract or as memorialized within a PEO Contract, or as any

PEO Contract may be construed by any person, Contractor affirms, agrees and warrants that, Contractor shall be liable for all actions of its PEO Contractor insofar as said actions affect the City or Contractor's performance of its duties under the Agreement. For purposes of the Agreement, Contractor shall not assign and has not assigned any of its duties under the Agreement to its PEO Contractor.

8.5 City is not a General or Special Employer to the Employees

The parties agree no employment relationship exists between the City and the Employees. As to the Employees, the City is not a "General Employer" or "Special Employer", as those terms are defined under California law, for any purpose, including but not limited to application of California Labor Code §§ 6300 *et seq.* and California Insurance Code § 11663, or any other applicable statute. Should a court or administrative agency having jurisdiction over the issue determine that the City is a "General Employer" or "Special Employer" of the Employees, Contractor shall fully indemnify the City for all costs and liabilities arising from that finding, including but not limited to consequential and incidental damages, temporary disability indemnity, permanent disability indemnity, medical costs, penalties and fines, vocational rehabilitation costs, and attorneys fees. Contractor's duties and obligations under this section of the Agreement are material obligations guaranteed by the performance guarantee described in Section 12.

8.6 Prevailing Wages

Contractor hereby acknowledges that it has read completely and fully understands San Francisco Administrative Code Section 21.25-2 and agrees that this Agreement shall be subject to, and Contractor shall comply with, all applicable obligations and requirements imposed by that Section.

9. Responsibility for Equipment

City shall not be responsible for any damage to persons or property as a result of the use, misuse or failure of any equipment used by Contractor, or by any of its Employees, even though such equipment be furnished, rented or loaned to Contractor by City.

10. Independent Contractor; Payment of Taxes and Other Expenses

10.1 Independent Contractor. Contractor or any agent or Employee of Contractor shall be deemed at all times to be an independent contractor and is wholly responsible for the manner in which it performs the services and work requested by City under this Agreement. Contractor or any agent or Employee of Contractor shall not have employee status with City, nor be entitled to participate in any plans, arrangements, or distributions by City pertaining to or in connection with any retirement, health or other benefits that City may offer its employees. Contractor or any agent or Employee of Contractor is liable for the acts and omissions of itself, its employees and its agents. Contractor shall be responsible for all obligations and payments, whether imposed by federal, state or local law, including, but not limited to, FICA, income tax withholdings, unemployment compensation, insurance, and other similar responsibilities related to Contractor's performing services and work, or any agent or Employee of Contractor or agency relationship between City and Contractor or any agent or Employee of Contractor.

Any terms in this Agreement referring to direction from City shall be construed as providing for direction as to policy and the result of Contractor's work only, and not as to the means by which such a result is obtained. City does not retain the right to control the means or the method by which Contractor performs work under this Agreement.

10.2 Payment of Taxes and Other Expenses. Should City, in its discretion, or a relevant taxing authority such as the Internal Revenue Service or the State Employment Development Division, or both, determine that Contractor or any of its agents or Employees is an employee of the City for purposes of collection of any employment taxes, any amounts offset from revenues paid by Contractor to the City under this Agreement shall be reduced by amounts equal to both the employee and employer portions of

the tax due (and offsetting any credits for amounts already paid by Contractor which can be applied against this liability). City shall then forward those amounts to the relevant taxing authority.

Should a relevant taxing authority determine a liability for past services performed by Contractor for City, upon notification of such fact by City, Contractor shall promptly remit such amount due to the City (again, offsetting any amounts already paid by Contractor that can be applied as a credit against such liability).

A determination of employment status pursuant to the preceding two paragraphs shall be solely for the purposes of the particular tax in question, and for all other purposes of this Agreement, Contractor shall not be considered an employee of City. Notwithstanding the foregoing, should any court, arbitrator, or administrative authority determine that Contractor is an employee for any other purpose, then Contractor agrees to a reduction in City's financial liability so that City's total expenses under this Agreement are not greater than they would have been had the court, arbitrator, or administrative authority determined that Contractor was not an employee.

10.3 Contractor's Employees. Contractor shall be solely responsible for all matters relating to payment of Contractor's Employees, including compliance with Social Security, withholding and payment of any and all federal, state and local personal income taxes, disability insurance, unemployment, and any other taxes for such persons, including any related assessments or contribution required by law and all other regulations governing City Health Service System, vacation, holiday, retirement or other programs, nor, in the event that City terminates this Agreement shall Contractor have recourse of rights of appeal under City's rules and regulations which are applicable to employees.

10.4 Payroll and Taxation. Contractor shall make or cause to be made all necessary payroll deductions for disability and unemployment insurance, social security, withholding taxes and other applicable taxes, and prepare, maintain and file or cause to be filed all necessary reports with respect to such taxes or deductions, and all other necessary statements and reports pertaining to labor employed in the performance of this Agreement.

11. Required Insurance

Subject to approval by the City's Risk Manager of the insurers and policy forms, Contractor shall place and maintain throughout the Term of this Agreement, and pay the cost thereof, the following insurance policies:

11.1 Comprehensive general liability insurance with limits not less than \$5,000,000 (such limit may be provided through a primary and excess policy) each Occurrence, combined single limit for bodily injury and property damage, or in such greater amount and limits as DPT may reasonably require from time to time, including coverage for contractual liability, personal injury, broadform property damage, products and completed operations, independent contractors (excluding towing and dismantling subcontractors), and mobile equipment.

11.2 Sudden and accidental pollution insurance with limits not less than \$1,000,000 for each occurrence.

11.3 Comprehensive business/commercial automobile liability insurance with limits not less than \$1,000,000 for each Occurrence combined single limit for bodily injury and property damage, including coverage for owned, non-owned and hired automobiles. If Contractor does not own or lease company vehicles that are subject to motor vehicle registration, then only non-owned and hired coverage is required.

11.4 Garage-keeper's legal liability insurance with limits not less than \$500,000 for each Occurrence combined single limit for loss and damage to vehicles in Contractor's care, custody or control caused by fire, explosion, theft, riot, civil commotion, malicious mischief, vandalism or collision, with

TOWING SERVICES MASTER AGREEMENT

Page 11 of 33

any deductible not to exceed \$25,000 for each Occurrence. Contractor may insure or self-insure loss of non-automobile property in the care, custody, or control of the garage keeper with a limit of \$5000.

11.5 Workers' Compensation Insurance, including Employers' Liability, with limits not less than \$1,000,000 for each accident, covering all Employees employed by Contractor in the performance of this Agreement to provide statutory benefits as required by the laws of the State of California. Said policy shall be endorsed to provide that the insurer waives all rights of subrogation against the City.

11.6 Environmental impairment liability insurance with limits not less than \$1,000,000 each occurrence, covering the sudden and accidental release of hazardous materials and the resulting costs of clean up.

Except as set forth above, any deductibles in the policies listed above shall not exceed \$25,000 each occurrence. The insurance policies shall be endorsed to name as an additional insured the City and County of San Francisco and its respective departments, commissioners, officers, agents and employees.

The agreements between the Contractor and its towing and dismantling subcontractors (as applicable) shall require that reasonable insurance is maintained and that the City will be named as an additional insured on the general liability policy.

12. Financial Assurances

12.1 Requirement to Provide Financial Guarantees

Upon the Effective Date of this Agreement, Contractor shall provide, and shall maintain for the time periods specified herein, financial instruments and funds described in this Section 12 as security to ensure Contractor's performance of all terms and conditions of this Agreement and to compensate for any damage to City property and/or other actual costs to City or for reimbursement to Customers for Contractor's violation of the terms of this Agreement, as further described below.

12.2 Letter of Credit

12.2.1 Performance Guarantee. Upon the Effective Date of this Agreement, Contractor shall provide to the City, and shall maintain throughout the Term of this Agreement and for a period of at least ninety (90) days after expiration or termination of this Agreement, or until all of Contractor's obligations have been performed under this Agreement (including but not limited to investigation and remediation obligations under Appendices C, D and E), whichever date is later, a performance guarantee in the form of a confirmed, irrevocable Letter of Credit in favor of the City and County of San Francisco, a municipal corporation, acting by and through its Municipal Transportation Agency (and during the Term of Appendix D, the Port of San Francisco), in the amount of two million dollars (\$2,000,000), guarantying the faithful performance by Contractor of this Agreement and of the covenants, terms and conditions of this Agreement, including all monetary obligations set forth herein, and including liquidated damages and any dishonesty on the part of Contractor.

The City may draw upon such Letter of Credit in circumstances which include, but are not limited to:

- (a) To ensure regulatory compliance in the event that Contractor receives a notice of violation or other regulatory order from a governmental or regulatory agency with jurisdiction over Contractor's operations or the Properties used by Contractor for the performance of this Agreement and Contractor does not achieve compliance with the notice of violation or order to the satisfaction of the issuing agency within the time specified by the agency or by the City if the agency does not specify a timeframe.
- (b) To reimburse the City for any fine or other charge assessed against the City related to any notice of violation or other regulatory order issued to Contractor.

- (c) To reimburse the City for costs associated with City's environmental assessments or corrective action related to Contractor's violation of any of the requirements of Appendices D or E, which may be performed at the City's sole discretion.
- (d) To satisfy rental payment obligations for City-owned property licensed to Contractor in Appendices D or E .
- (e) To satisfy fines assessed by City against Contractor pursuant to Appendix D of the Agreement.
- (f) To compensate City for losses or damage to property caused by Contractor.

12.2.2 Letter of Credit Requirements. The Letter of Credit required by this Section 12 shall be issued on a form and issued by a financial institution acceptable to the City in its sole discretion, which financial institution shall (a) be a bank or trust company doing business and having an office in the State of California, (b) have a combined capital and surplus of at least \$25,000,000, and (c) be subject to supervision or examination by federal or state authority. If Contractor defaults with respect to any provision of this Agreement, City may, but shall not be required to, make its demand under said Letter of Credit for all or any portion thereof, to compensate City for any loss or damage which City may have incurred by reason of Contractor's default or dishonesty, including (but not limited to) any claim for fines or liquidated damages; provided, however, that City shall present its written demand to said bank for payment under said Letter of Credit only after City first shall have made its demand for payment directly to Contractor, and five (5) full days have elapsed without Contractor having made payment to City.

12.2.3 Expiration or Termination of Letter of Credit. In the event the City receives notice from the issuer of the Letter of Credit that the Letter of Credit will be terminated, not renewed or will otherwise be allowed to expire for any reason during the period from the commencement of the Term of this Agreement to ninety (90) days after the expiration or termination of this Agreement or the conclusion of all of Contractor's obligations under the Agreement, whichever occurs last, and Contractor fails to provide the City with a replacement Letter of Credit (in a form and issued by a financial institution acceptable to the City) within ten (10) days following the City's receipt of such notice, such occurrence shall be an Event of Default as defined in Section 18 of this Agreement, and, in addition to any other remedies the City may have due to such default (including the right to terminate this Agreement), the City shall be entitled to draw down the entire amount of the Letter of Credit (or any portion thereof) and hold such funds in an account with the Treasurer in the form of cash guarantying Contractor's obligations under this Agreement under the terms of this Section 12. In such event, the cash shall accrue interest to the Contractor at a rate equal to the average yield of Treasury Notes with one-year maturity, as determined by the Treasurer. In the event the Letter of Credit is converted into cash pursuant to this paragraph, upon termination of this Agreement, Contractor shall be entitled to a full refund of the cash (less any demands made thereon by the City pursuant to Section 12.2.4,) within ninety (90) days of the termination date, including interest accrued through the termination date.

12.2.4 Demands Upon Letter of Credit. City may use all or any portion of the Letter of Credit to compensate City for any loss or damage that it may have incurred by reason of Contractor's negligence or breach. Such loss or damage may include without limitation any damage to or restoration of the Properties for which Contractor is responsible, and claims for fines and/or liquidated damages. Should the City terminate this Agreement due to a breach by Contractor, the City shall have the right to draw from the Letter of Credit those amounts necessary to pay any fees or other financial obligations under the Agreement and perform the towing and storage services described in this Agreement until such time as the City procures another contractor and the agreement between the City and that contractor becomes effective. The Treasurer shall not allow any demands made to the Letter of Credit pursuant to claims arising from Section 6(c) of Appendix C to exceed a combined outstanding amount of more than four hundred thousand dollars (\$400,000.00) In addition, the Treasurer shall not allow any demands to be made to the Letter of Credit pursuant to claims arising from Section 6(b) of Appendix C to exceed a combined outstanding amount of more than four hundred thousand dollars (\$400,000.00) In addition, the Treasurer shall not allow any demands to be made to the Letter of Credit pursuant to claims arising from Section 6(b) of Appendix C to exceed a combined outstanding amount of more than twice the amount of the current Base Fee established in

Appendix D, Section 5, including all periodic CPI adjustments up to the date of the demand on the Letter of Credit. The Treasurer shall not allow any demands made to the Letter of Credit pursuant to claims arising from any section of the Agreement, excluding Appendices C and D, to exceed a combined outstanding amount of more than one million three hundred thousand dollars (\$1,300,000.00).

12.2.5 Depletion of Letter of Credit. If any portion of a Letter of Credit is used by City, Contractor shall provide written proof that the Letter of Credit has been restored to its initial value, which shall require a replacement Letter of Credit in the face amount of the required Letter of Credit. Contractor's failure to do so within the time limits specified in Section 18.1.1(f) shall constitute an Event of Default as defined in Section 18 of this Agreement.

12.2.6 Dispute Resolution. In the event that a dispute arises between the City and Contractor concerning this Agreement or the use or maintenance of the Letter of Credit, Contractor may appeal to the Director of Transportation within fourteen (14) days of demand on the Letter of Credit with evidence supporting Contractor's claim for relief from the demand on the Letter of Credit. The Director of Transportation will respond within fourteen (14) days. Any failure of the Director of Transportation to respond within fourteen (14) days shall be deemed a rejection of Contractor's claim for relief from the demands on the Letter of Credit and the Director of Transportation's response to such demand shall constitute the administrative remedy for Agreement interpretation described in Section 46 herein. Each party reserves its remedies in equity and law. No decision by the City concerning the Letter of Credit shall prevent Contractor from seeking restoration of the funds by appropriate legal action.

12.3 Maintenance Deposit: Upon execution of this Agreement, Contractor shall deposit with City the amount of one hundred thousand dollars (\$100,000) as a maintenance deposit. These funds may be used by City when maintenance required by this Agreement is not done in a timely manner or in accordance with the standards of this Agreement. Contractor shall be responsible for replenishing this maintenance deposit fund to maintain a balance of one hundred thousand dollars (\$100,000) within fifteen (15) days of any date that the fund falls below the minimum balance. Failure to replenish the maintenance deposit fund for more than forty-five (45) days shall be an Event of Default under this Agreement. Any interest accrued and earned on the maintenance deposit fund shall be retained by City.

12.4 Claims Fund. Contractor shall at all times maintain a Claim Fund for payment of Claims. Contractor shall maintain at least fifty-thousand dollars (\$50,000) in the Claim Fund at all times. Contractor shall be responsible for replenishing this Claims Fund to maintain a balance of fifty thousand dollars (\$50,000) within fifteen (15) days of any date that the fund falls below the minimum balance. Failure to replenish the Claims Fund for more than forty-five (45) days from the date that it falls below the minimum balance shall be an Event of Default under this Agreement.

Environmental Oversight Deposit. Upon the Effective Date of this Agreement, 12.5 Contractor shall provide to the City, and shall maintain and replenish throughout the term of the property license set forth in Appendix D ("Pier 70 License") and for a period of at least ninety (90) days after termination or expiration of the Pier 70 License, an Environmental Oversight Deposit in the amount of ten thousand dollars (\$10,000), which shall be deposited in an account specified by City. If Contractor receives a notice of violation or other regulatory order from a governmental or regulatory agency with jurisdiction over Pier 70 and or its operations (other than from the Port of San Francisco), and such notice is not cured within fourteen (14) days, the City may draw from this deposit to reimburse the City for staff costs incurred by the City while inspecting site conditions and enforcing and administering the Hazardous Materials provisions of the Pier 70 License. If Contractor receives a notice of violation or other regulatory order from a governmental or regulatory agency with jurisdiction over the site and or its operations (other than from the Port of San Francisco), and such notice is cured within fourteen (14) days, the City may draw from this deposit in an amount not to exceed \$500 to reimburse the City for staff costs incurred by the City. The City will submit an invoice to Contractor for any such costs, and Contractor will pay such invoiced amounts within thirty (30) days to replenish the Environmental Oversight Deposit. Contractor's failure to pay such costs within thirty (30) days, or to replenish the Environmental Oversight Deposit if drawn upon, will constitute an Event of Default.

13. Insurance and Performance Guarantee Requirements

All insurance policies and Letters of Credit obtained pursuant to this Agreement shall be endorsed to provide:

13.1 That thirty (30) days prior written notice of cancellation, non-renewal or reduction in coverage or limits shall be given to DPT in the manner and at the addresses specified below (and during the term of Appendix D, to the Port of San Francisco in the manner and at the address specified in Appendix D); and

13.2 That any insurance required by this Agreement is primary to any other insurance available to an additional insured with respect to claims arising out of this Agreement and that insurance applies separately to each insured against whom a claim is made or a suit is brought, but the inclusion of more than one insured shall not operate to increase the insurer's limits of liability.

13.3 Two copies of any Letter of Credit, and two copies of each original policy or policy endorsement of insurance shall be provided to DPT (and during the term of Appendix D, to the Port of San Francisco) upon the Effective Date of this Agreement, and complete copies of any insurance policies obtained pursuant to this Agreement shall be provided to DPT (and during the term of Appendix D, to the Port of San Francisco) if requested at any time.

13.4 Upon City's request, Contractor shall provide satisfactory evidence that Contractor has adequately provided for Social Security and Unemployment Compensation benefits for Contractor's Employees.

13.5 Contractor shall comply with the provisions of any insurance policy covering Contractor or the City, and with any notices, recommendations or directions issued by any insurer under such insurance policies so as not to adversely affect the insurance coverage.

14. Contractor's Representations and Warranties

Contractor hereby represents and warrants as follows:

14.1 Experience. Contractor is experienced in the operation and management of automobile towing and disposal services and hereby agrees to apply its best efforts and most efficient methods in performance of this Agreement.

14.2 Formation. Contractor is a duly formed, validly existing and in good standing limited liability company under the laws of the State of California.

14.3 Authority. Contractor has full power and authority (corporate or otherwise) to enter into this Agreement and to consummate the transactions contemplated by it. This Agreement has been duly authorized by all necessary action on the part of Contractor, and no other corporate or other action on the part of Contractor is necessary to authorize the execution and delivery of this Agreement.

14.4 Conflicts and Consents. The execution and delivery by Contractor of this Agreement and the performance by Contractor of the transactions contemplated by it will not violate any federal, state or local law, rule or regulation, or conflict with or result in any breach or violation of, or constitute a default (or an event which with notice or lapse of time or both would become a default) under, or constitute an event or condition that would permit termination or acceleration of the maturity of, the Articles of Incorporation, bylaws or partnership agreement of Contractor (as applicable) or any indenture, mortgage, lease, agreement or other instrument or obligation to which Contractor is a party or by which it may be bound which would materially adversely affect the ability of Contractor to perform its obligations under this Agreement. No approval, authorization, consent or other order or action of, or filing or registration with, any person, entity or governmental authority is required for the execution and delivery by Contractor of this Agreement.

TOWING SERVICES MASTER AGREEMENT

Page 15 of 33

14.5 Conflict with Orders. The execution and delivery by Contractor or this Agreement will not conflict with any order, judgment or decree of any court, government, government agency or instrumentality, whether entered pursuant to consent to otherwise, by which Contractor may be bound or affected.

14.6 Litigation. There is no litigation, action, arbitration, grievance, administrative proceeding, suit or claim filed and pending, nor is there any investigation by a governmental agency of Contractor or any of its affiliates that, if adversely decided, could have a material adverse impact on Contractor's ability to perform its obligations under this Agreement.

15. Indemnification

Contractor shall indemnify and save harmless City and its officers, agents and employees from, and, if requested, shall defend them against any and all loss, cost, damage, injury, liability, and claims thereof for injury to or death of a person, including Employees of Contractor or loss of or damage to property, resulting directly or indirectly from Contractor's performance of this Agreement, including, but not limited to, the use of Contractor's facilities or equipment provided by City or others, regardless of the negligence of, and regardless of whether liability without fault is imposed or sought to be imposed on City, except to the extent that such indemnity is void or otherwise unenforceable under applicable law in effect on or validly retroactive to the date of this Agreement, and except where such loss, damage, injury, liability or claim is the result of the active negligence or willful misconduct of City and is not contributed to by any act of, or by any omission to perform some duty imposed by law or agreement on Contractor, its subcontractors or either's agent or employee. The foregoing indemnity shall include, without limitation, reasonable fees of attorneys, consultants and experts and related costs and City's costs of investigating any claims against the City. The foregoing indemnification does not include the limitations on Contractor's liability described in Appendix D, Sections 24.3 and 24.8.

In addition to Contractor's obligation to indemnify City, Contractor specifically acknowledges and agrees that it has an immediate and independent obligation to defend City from any claim which actually or potentially falls within this indemnification provision, even if the allegations are or may be groundless, false or fraudulent, which obligation arises at the time such claim is tendered to Contractor by City and continues at all times thereafter.

As to any intellectual property that Contractor provides to the City in the performance of this Agreement, Contractor shall indemnify and hold City harmless from all loss and liability, including attorneys' fees, court costs and all other litigation expenses for any infringement of the patent rights, copyright, trade secret or any other proprietary right or trademark, and all other intellectual property claims of any person or persons, arising as a consequence of the use by City of the intellectual property supplied by the Contractor, or any of its officers or agents.

16. Incidental and Consequential Damages

Contractor shall be responsible for incidental and consequential damages resulting in whole or in part from Contractor's acts or omissions. Nothing in this Agreement shall constitute a waiver or limitation of any rights that City may have under applicable law.

17. Liability of City

NOTWITHSTANDING ANY OTHER PROVISION OF THIS AGREEMENT, IN NO EVENT SHALL CITY BE LIABLE, REGARDLESS OF WHETHER ANY CLAIM IS BASED ON CONTRACT OR TORT, FOR ANY SPECIAL, CONSEQUENTIAL, INDIRECT OR INCIDENTAL DAMAGES, INCLUDING, BUT NOT LIMITED TO, LOST PROFITS, ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT OR THE SERVICES PERFORMED IN CONNECTION WITH THIS AGREEMENT.

18. Default; Remedies

18.1 Each of the following shall constitute an event of default ("Event of Default") under this Agreement; provided, however, that with the exception of the Events of Default listed in Section 18.1.1 below, Contractor shall have thirty (30) days after the date of written notice of the default sent by City to cure the default before an Event of Default exists:

18.1.1 The following conduct by Contractor shall constitute an Event of Default at the time it occurs without a right to cure under this Section 18, and shall be grounds for termination pursuant to Section 19:

(a) Contractor fails or refuses to perform or observe any term, covenant or condition contained in any of the following Sections of this Agreement: 5, 6, 11, 12.2.1, 13, 22, 28, 35 and 54.

(b) Substantial abandonment or discontinuance by Contractor, without the prior written consent of the City, of any or all of the services and operations required hereunder.

(c) Contractor's representation or warranty made pursuant to this Agreement which Contractor made knowing that it was not true and correct at the time when made.

(d) Failure of Contractor to replenish the maintenance deposit or Claims Fund required by Sections 12.3 and 12.4 and such failure continues for more than forty-five (45) days.

(e) Failure of Contractor to replenish the Environmental Oversight Deposit required by Section 12.5 and such failure continues for more than thirty (30) days.

(f) Failure to replace the Letter of Credit as required by Sections 12.2.3 and 12.2.5 within twenty (20) days of (i) City's receipt of notice of its termination or expiration, or (ii) use by City of the Letter of Credit, unless City's use is challenged pursuant to Section 12.2.6, in which event the twenty (20) days shall run from the date of the Director of Transportation's denial of the challenge, or fourteen (14) days from the date of the challenge, whichever is later.

18.1.2 Contractor fails or refuses to perform or observe any other term, covenant or condition contained in this Agreement.

18.1.3 Contractor (A) is generally not paying its debts as they become due, (B) files, or consents by answer or otherwise to the filing against it of, a petition for relief or reorganization or arrangement or any other petition in bankruptcy or for liquidation or to take advantage of any bankruptcy, insolvency or other debtors' relief law of any jurisdiction, (C) makes an assignment for the benefit of its creditors, (D) consents to the appointment of a custodian, receiver, trustee or other officer with similar powers of Contractor or of any substantial part of Contractor's property or (E) takes action for the purpose of any of the foregoing.

18.1.4 A court or government authority enters an order (A) appointing a custodian, receiver, trustee or other officer with similar powers with respect to Contractor or with respect to any substantial part of Contractor's property, (B) constituting an order for relief or approving a petition for relief or reorganization or arrangement or any other petition in bankruptcy or for liquidation or to take advantage of any bankruptcy, insolvency or other debtors' relief law of any jurisdiction or (C) ordering the dissolution, winding-up or liquidation of Contractor, or (D) levying of a writ of attachment or execution against any of Contractor's property.

18.1.5 Failure of Contractor to pay when due any other amount owing from Contractor to the City, including without limitation rents, taxes, fees or other charges, whether or not such amounts are related to this Agreement; provided, however, that if Contractor disputes the amount of any such

TOWING SERVICES MASTER AGREEMENT

Page 17 of 33

obligation in good faith and is actively negotiating or litigating such dispute, Contractor's failure to pay such amount shall not constitute a default under this paragraph.

18.1.6 Failure of Contractor to abide by any of the terms or conditions of the Port MOU, set forth in Appendix C to this Agreement.

18.2 On and after any Event of Default, City shall have the right to exercise its legal and equitable remedies, including, without limitation, the right to terminate this Agreement or to seek specific performance of all or any part of this Agreement. City's right to termination for an Event of Default shall be subject to Contractor's opportunity to cure such Event of Default pursuant to the terms of Sections 18.1 and 19. In addition, City shall have the right (but no obligation) to cure (or cause to be cured) on behalf of Contractor any Event of Default; Contractor shall pay to City on demand all costs and expenses incurred by City in effecting such cure, with interest thereon from the date of incurrence at the maximum rate then permitted by law. City shall have the right to offset from any amounts due to Contractor under this Agreement or any other agreement between City and Contractor all damages, losses, costs or expenses incurred by City as a result of such Event of Default and any liquidated damages due from Contractor pursuant to the terms of this Agreement or any other agreement.

18.3 All remedies provided for in this Agreement may be exercised individually or in combination with any other remedy available hereunder or under applicable laws, rules and regulations. The exercise of any remedy shall not preclude or in any way be deemed to waive any other remedy.

19. Termination

The City shall have the right to terminate this Agreement after written notice to Contractor upon the occurrence of any Event of Default, provided, however, that Contractor shall have a period of ten (10) days from the date of City's notice of intent to terminate to cure such default before City's termination for cause may become effective.

19.1 Termination Effective Upon Notice. Termination under this Section shall be effective immediately upon notice being given by DPT to Contractor and after the expiration of any applicable cure periods. Upon such termination, all rights, powers, privileges and authority granted to Contractor under this Agreement shall cease, and Contractor shall immediately thereupon vacate the Designated Facilities, except as may be permitted pursuant to Section 54 (Transition Period), below.

19.2 Termination Not Exclusive Remedy. The City's right to terminate this Agreement under this Section is not its exclusive remedy but is in addition to all other remedies provided to it by law or the provisions of this Agreement.

Duties Upon Termination. Upon termination of this Agreement, the City and 19.3 Contractor shall promptly pay to the other, as soon as is determinable after the effective date of termination, all amounts due each other under the terms of this Agreement, and upon such payment neither shall have any further claim or right against the other, except as expressly provided herein. Upon the effective date of termination, Contractor shall deliver to the DPT the originals of all books, permits, plans, Records, licenses, contracts and other documents pertaining to Contractor's operation under this Agreement, any insurance policies, bills of sale or other documents evidencing title or rights of the City, and any and all other Records or documents pertaining to Contractor's operation under this Agreement, any insurance policies, bills of sale or other documents evidencing title or rights of the City, and any and all other Records or documents pertaining to the Designated Facilities, whether or not enumerated herein, which are requested by the City or necessary or desirable for the ownership and operation of the Designated Facilities, which are then in possession of Contractor. Contractor further agrees to do all other things reasonably necessary to cause an orderly transition of the management and operation of the services provided by Contractor under this Agreement without detriment to the rights of the City or to the continued operation of such services.

20. Provisions of Agreement Surviving Termination or Expiration

This Section and the following Sections of this Agreement shall survive termination or expiration of this Agreement: 1, 5, 6, 7, 9, 10, 12 (except for 12.2.5), 14, 15, 16, 17, 18.2, 18.3, 19.3, 20, 22.1, 23, 24, 25, 26, 29, 38, 39, 43, 46, 47, 48, 49, 53, 54, 55, 56 and 57; and Appendix A Sections 1.3(b), 4.1, 5.2, 12, 13.3, 15.1, 15.2, 15.7(1), 15.7(2), 15.10(7) and 15.11.

21. Conflict of Interest

Through its execution of this Agreement, Contractor acknowledges that it is familiar with the provisions of §15.103 and Appendix C 8.105 of City's Charter and § 87100 et seq. of the Government Code of the State of California, and certifies that it does not know of any facts which constitute a violation of said provisions.

22. Proprietary or Confidential Information

22.1 Contractor understands and agrees that, in the performance of the work or services under this Agreement or in contemplation thereof, Contractor may have access to private or confidential information which may be owned or controlled by City and that such information may contain proprietary or confidential details, the disclosure of which to third parties may be damaging to City. Contractor agrees that to the extent permitted by law, all information disclosed by City to Contractor shall be held in confidence and used only in performance of the Agreement. Contractor shall exercise the same standard of care to protect such information as a reasonably prudent contractor would use to protect its own proprietary data.

22.2 Contractor understands and agrees that the City's Sunshine Ordinance (Administrative Code, Chapter 67) and the California Public Records Act (Gov. Code section 6250 *et seq.*), apply to the Agreement. Contractor shall cooperate with City in the compilation, copying and production of records in its custody that are subject to requests for public records. The City is not required to take any action, or to refuse to release information where to do so would violate applicable law. During the Term, the City will endeavor to provide Contractor reasonable notice of any request for public information that seeks disclosure of confidential or proprietary information that Contractor has provided to City under the Agreement and that Contractor has identified as confidential and proprietary. Contractor may at its option then take whatever legal steps it deems appropriate to protect said information from disclosure. Contractor shall clearly identify to City all information that Contractor provides to City that it considers to be proprietary, trade secret or is otherwise protected from disclosure under the California Public Records Act, the City's Sunshine Ordinance and other applicable law.

23. Notices to the Parties

Any insurance certificates or notices required under Sections 11, 12, 13, 15, 18, 19, 46 or 52 of the Agreement must be sent by first class, certified U.S. mail, postage pre-paid. All other written communications, unless otherwise indicated elsewhere in this Agreement, may be by first class U.S. mail, by email, or by fax, and shall be addressed as follows, or to such other address as designated by the parties in writing:

To Contractor:

San Francisco AutoReturn Attention: John Wicker, President and CEO 450 7th Street San Francisco, CA 94103 Telephone: 415-626-3380 Facsimile 415-626-3381 Email: jwicker@autoreturn.com

TOWING SERVICES MASTER AGREEMENT

Page 19 of 33

To City:

City and County of San Francisco Department of Parking and Traffic Attention: Steve Bell 25 Van Ness Avenue, Suite 230 San Francisco, CA 94105 Telephone: 415-554-9825 Facsimile: 415-252-3272 Email: steve.bell@sfgoy.org

24. Ownership of Results

Any interest of Contractor or its Subcontractors, in the Records prepared by Contractor or its subcontractors in connection with services to be performed under this Agreement, shall become the property of and will be transmitted to City. However, Contractor may retain and use copies for reference and as documentation of its experience and capabilities. To the extent that the use of proprietary software or other proprietary information or intellectual property is required to access or utilize the data contained in the Records, or that Contractor holds particular work practices or methods to be proprietary, Contractor hereby grants the City a perpetual, royalty-free, nonexclusive, nontransferable, limited license, to use and reproduce said proprietary information or intellectual property, solely for City's internal purposes related to the towing, storage, and disposal of abandoned and illegally parked vehicles.

25. Works for Hire

Subject to the limited license set out in Section 24 above, any artwork, copy, posters, billboards, photographs, videotapes, audiotapes, systems designs, software, source codes, or any other original works of authorship of Contractor that are not Records, are proprietary to Contractor and shall not be considered works for hire as defined under Title 17 of the United States Code, and all copyrights in such works are the property of Contractor.

26. Audit and Inspection of Records

26.1 Records. Contractor shall maintain, in accordance with generally accepted accounting principles and business practices, all books, accounts and Records created in the performance of this Agreement in accordance with the requirements of Appendix A. Such books, accounts and Records shall be maintained throughout the Term of this Agreement at one of the Designated Facilities, or at Contractor's Headquarters Office. Records created or maintained in an electronic format shall be submitted monthly in an electronic format as specified by DPT, which may be an electronic database format or a static electronic format, such as PDF or an electronic format that attaches a date or time stamp to each document and record entry that cannot be erased or altered Contractor shall deliver said copy of the Records with the Monthly Management Report described in Appendix A, Section 13.

Except as otherwise specified herein, Contractor shall maintain Records related to this Agreement in a safe and secure location available for inspection and copying by City for a period of five (5) years following termination of this Agreement.

26.2 City's Right to Inspect. Any duly authorized agent of City shall have the right to examine, at any time during normal business hours, all books, accounts and Records, including computer records, of the type described above.

26.3 Audit. Within sixty (60) days of the expiration or termination of the Agreement, an independent auditor approved by the Controller shall conduct an audit of those records pertaining to Contractor's performance during the final year of this Agreement, including a summary report of prior annual audits conducted pursuant to Appendix A, Section 12.6. Such audit shall include a review of all records involving the removal, impoundment and disposition of vehicles pursuant to this Agreement

TOWING SERVICES MASTER AGREEMENT

Page 20 of 33

during the previous and current audit periods. All costs of such audits are to be paid by the Contractor. A certified copy of each such audit report shall be furnished promptly to DPT and the Controller not more than 120 days following the expiration of the Agreement.

26.4 Retention of Records. Contractor shall retain all records of subcontracts, Employee payroll and benefits, tax assessments and payments, and those records and reports described in Sections 10 and 26 of this Agreement for a period of not less than four (4) years from date of termination of this Agreement.

27. Subcontracting

Except as specifically provided in this Agreement, Contractor shall not enter into any subcontract for the performance of all or any part of this Agreement for the acquisition of towing services, security services, vehicle handling or disposal services, auctioneer or appraisal services, Customer relations services or with any PEO Contractor without the prior written consent of the City, which consent shall not be unreasonably withheld. Any attempt to enter into such a subcontract without such consent shall be null and void. Subject to the foregoing, this Agreement shall be binding upon, enforceable by, and shall inure to the benefit of any permitted subcontractor.

28. Assignment

Contractor shall not assign or otherwise transfer this Agreement or any of Contractor's rights, duties or interest under this Agreement without the prior written consent of the City. Any attempted assignment or transfer without such consent shall be null and void. Subject to the foregoing, this Agreement shall be binding upon, enforceable by, and shall inure to the benefit of the successors and permitted assigns of the Contractor.

29. Non-Waiver of Rights

The omission by either party at any time to enforce any default or right reserved to it, or to require performance of any of the terms, covenants, or provisions hereof by the other party at the time designated, shall not be a waiver of any such default or right to which the party is entitled, nor shall it in any way affect the right of the party to enforce such provisions thereafter.

30. Earned Income Credit (EIC) Forms

Administrative Code Chapter 12O requires that employers provide their employees with IRS Form W-5 (The Earned Income Credit Advance Payment Certificate) and the IRS EIC Schedule, as set forth below. Employers can locate these forms at the IRS Office, on the Internet, or anywhere that Federal Tax Forms can be found.

30.1 Contractor shall provide EIC Forms to each Eligible Employee at each of the following times: (i) within thirty (30) days following the date on which this Agreement becomes effective (unless Contractor has already provided such EIC Forms at least once during the calendar year in which such effective date falls); (ii) promptly after any Eligible Employee is hired by Contractor; and (iii) annually between January 1 and January 31 of each calendar year during the Term of this Agreement.

30.2 Failure to comply with any requirement contained in Section 30.1 of this Agreement shall constitute a material breach by Contractor of the terms of this Agreement. If, within thirty (30) days after Contractor receives written notice of such a breach, Contractor fails to cure such breach or, if such breach cannot reasonably be cured within such period of thirty (30) days, Contractor fails to commence efforts to cure within such period or thereafter fails to diligently pursue such cure to completion, the City may pursue any rights or remedies available under this Agreement or under applicable law,

30.3 Any Subcontract entered into by Contractor shall require the subcontractor to comply, as to the subcontractor's Eligible Employees, with each of the terms of this section.

TOWING SERVICES MASTER AGREEMENT

Page 21 of 33

30.4 Capitalized terms used in this Section and not defined in this Agreement shall have the meanings assigned to such terms in Chapter 12O of the San Francisco Administrative Code.

31. Minority/Women/Local Business Utilization; Liquidated Damages

31.1 Compliance. Contractor understands and agrees to comply fully with all applicable provisions of Chapter 12D.A ("Minority/Women/ Local Business Utilization Ordinance--IV") of the San Francisco Administrative Code set forth in the RFP, Contractor's Proposal and this Agreement, and agrees to include such provisions in all subcontracts (as defined in Section 27) made in fulfillment of the Contractor's obligations under this Agreement. Said provisions are incorporated by reference and made a part of this Agreement as though fully set forth herein. Contractor's willful failure to comply with Chapter 12D.A is a material breach of contract.

31.2 Enforcement. If Contractor willfully fails to comply with any of the provisions of Chapter 12D.A set forth in this Agreement pertaining to MBE or WBE participation, Contractor shall be liable for liquidated damages in an amount equal to Contractor's net profit on this Agreement, or 10% of the total amount of this Agreement, or \$1,000, whichever is greatest. City may also impose other sanctions against Contractor authorized in Chapter 12D.A, including declaring the Contractor to be irresponsible and ineligible to contract with the City for a period of up to five years or revocation of the Contractor's MBE or WBE certification. City will determine the sanctions to be imposed, including the amount of liquidated damages, after invéstigation pursuant to §12D.A.16(B). By entering into this Agreement, Contractor acknowledges and agrees that any liquidated damages assessed by City shall be payable to City upon demand. Contractor further acknowledges and agrees that any liquidated damages assessed may be withheld from any monies due to Contractor on any contract with City. Contractor agrees to maintain records necessary for monitoring its compliance with this Section 31 for a period of three (3) years following termination of this contract, and shall make such records available for audit and inspection by HRC or the Controller upon request.

31.3 Subcontracting Goals. The MBE/WBE subcontracting participation goal for this contract is 12% of Gross Revenues. Contractor shall fulfill this goal throughout the duration of this Agreement. Contractor shall not participate in any back contracting to the Contractor or lower-tier subcontractors, as defined in Chapter 12D.A, for any purpose inconsistent with the provisions of Chapter 12D.A, its implementing rules and regulations, or this Section.

31.4 Subcontract Language Requirements. Contractor shall include in all subcontracts with MBEs or WBEs made in fulfillment of Contractor's obligations under this Agreement, a provision requiring Contractor to compensate any MBE or WBE subcontractor for damages for breach of contract or liquidated damages equal to 5% of the subcontract amount, whichever is greater, if Contractor fails to comply with the requirements of this Section, unless Contractor received approval from City to substitute subcontractors or to otherwise modify the commitments in the bid or proposal. Such provisions shall also state that it is enforceable in a court of competent jurisdiction. Subcontracts shall require the subcontractor to maintain records necessary for monitoring its compliance with this Section 31 for a period of three (3) years following termination of this contract and to make such records available for audit and inspection by City upon request.

31.5 Payment of Subcontractors. Contractor shall pay its subcontractors within thirty (30) days after receiving an invoice or request for payment from a subcontractor, unless Contractor notifies City in writing within ten (10) working days prior to receiving payment from the City that there is a bona fide dispute between Contractor and its subcontractor, in which case Contractor may withhold the disputed amount but shall pay the undisputed amount.

32. Nondiscrimination; Penalties

32.1 Contractor Shall Not Discriminate. In the performance of this Agreement, Contractor agrees not to discriminate against any Employee, City and County employee working with such

contractor or subcontractor, applicant for employment with such contractor or subcontractor, or against any person seeking accommodations, advantages, facilities, privileges, services, or membership in all business, social, or other establishments or organizations, on the basis of the fact or perception of a person's race, color, creed, religion, national origin, ancestry, age, height, weight, sex, sexual orientation, gender identity, domestic partner status, marital status, disability or Acquired Immune Deficiency Syndrome or HIV status (AIDS/HIV status), or association with members of such protected classes, or in retaliation for opposition to discrimination against such classes.

32.2 Nondiscrimination in Benefits. Contractor does not as of the date of this Agreement and will not during the Term of this Agreement, in any of its operations in San Francisco, on real property owned by San Francisco, or where work is being performed for the City elsewhere in the United States, discriminate in the provision of bereavement leave, family medical leave, health benefits, membership or membership discounts, moving expenses, pension and retirement benefits or travel benefits, as well as any benefits other than the benefits specified above, between employees with domestic partners and employees with spouses, and/or between the domestic partners and spouses of such employees, where the domestic partnership has been registered with a governmental entity pursuant to state or local law authorizing such registration, subject to the conditions set forth in §12B.2(b) of the San Francisco Administrative Code.

32.3 Condition to Contract. As a condition to this Agreement, Contractor shall execute the "Chapter 12B Declaration: Nondiscrimination in Contracts and Benefits" form (form HRC-12B-101) with supporting documentation and secure the approval of the form by the San Francisco Human Rights Commission.

32.4 Incorporation of Administrative Code Provisions by Reference. The provisions of Chapters 12B and 12C of the San Francisco Administrative Code are incorporated in this Section by reference and made a part of this Agreement as though fully set forth herein. Contractor shall comply fully with and be bound by all of the provisions that apply to this Agreement under such Chapters, including but not limited to the remedies provided in such Chapters. Without limiting the foregoing, Contractor understands that pursuant to §12B.2(h) of the San Francisco Administrative Code, a penalty of \$50 for each person for each calendar day during which such person was discriminated against in violation of the provisions of this Agreement may be assessed against Contractor and/or deducted from any payments due Contractor.

33. MacBride Principles—Northern Ireland

Pursuant to San Francisco Administrative Code §12F.5, the City and County of San Francisco urges companies doing business in Northern Ireland to move towards resolving employment inequities, and encourages such companies to abide by the MacBride Principles. The City and County of San Francisco urges San Francisco companies to do business with corporations that abide by the MacBride Principles. By signing below, the person executing this agreement on behalf of Contractor acknowledges and agrees that he or she has read and understood this section.

34. Tropical Hardwood and Virgin Redwood Ban

Pursuant to § 804(b) of the San Francisco Environment Code, the City and County of San Francisco urges contractors not to import, purchase, obtain, or use for any purpose, any tropical hardwood, tropical hardwood wood product, virgin redwood or virgin redwood wood product.

35. Drug-Free Workplace Policy

Contractor acknowledges that pursuant to the Federal Drug-Free Workplace Act of 1989, the unlawful manufacture, distribution, dispensation, possession, or use of a controlled substance is prohibited on City premises. Contractor agrees that any violation of this prohibition by Contractor, its Employees, agents or assigns will be deemed a material breach of this Agreement.

36. Resource Conservation

Chapter 5 of the San Francisco Environment Code ("Resource Conservation") is incorporated herein by reference. Failure by Contractor to comply with any of the applicable requirements of Chapter 5 will be deemed a material breach of contract.

37. Compliance with Americans with Disabilities Act

Contractor acknowledges that the Americans with Disabilities Act ("ADA") requires that programs, services and other activities provided by a public entity to the public, whether directly or through a contractor, must be accessible to the disabled public. Contractor further acknowledges its obligation to comply with the ADA and any other federal, state or local disability rights legislation. Contractor warrants that it will fulfill that obligation, and that it will not discriminate against disabled persons in the provision of services, benefits or activities pursuant to this Agreement.

38. Sunshine Ordinance

38.1 Documents Subject to Disclosure by the City. In accordance with San Francisco Administrative Code §67.24(e), contracts, contractors' bids, responses to solicitations and all other records of communications between City and persons or firms seeking contracts, shall be open to inspection immediately after a contract has been awarded. Nothing in this provision requires the disclosure of a private person or organization's net worth or other proprietary financial data submitted for qualification for a contract or other benefit until and unless that person or organization is awarded the contract or benefit. Information provided which is covered by this paragraph will be made available to the public by the City upon request.

38.2 Documents Subject to Disclosure by Contractor. Contractor agrees that it shall comply with San Francisco Administrative Code § 67.29-7(c), the provisions of which are set out in full below:

In any contract, agreement or permit between the City and any outside entity that authorizes that entity to demand any funds or fees from citizens, the City shall ensure that accurate records of each transaction are maintained in a professional and businesslike manner and are available to the public as public records under the provisions of this ordinance. Failure of an entity to comply with these provisions shall be grounds for terminating the contract or for imposing a financial penalty equal to one-half of the fees derived under the agreement or permit during the period of time when the failure was in effect. Failure of any Department Head under this provision shall be a violation of this ordinance. This paragraph shall apply to any agreement allowing an entity to tow or impound vehicles in the City and shall apply to any agreement allowing an entity to collect any fee from any persons in any pretrial diversion program.

Contractor's compliance with Section 67.29-7(c) is a material term of the Agreement.

38.3 Public Records Requests from Public. If any member of the public communicates a public records request directly to Contractor, Contractor shall refer the requestor to DPT. In no event shall Contractor respond directly to a public records request or attempt to communicate a public records request to DPT on behalf of the requestor. DPT will work with Contractor, and Contractor shall cooperate with DPT to identify responsive records in Contractor's possession. If production of requested records involves significant staff time of Contractor, City in its sole discretion may elect to collect legally authorized fees from the requestor and credit some or all of the fees to Contractor.

39. Notification of Limitations on Contributions

This paragraph applies if this contract is in excess of \$50,000 over a 12-month period or less and is for: (1) personal services; or (2) the selling or furnishing of any material, supplies or equipment; or (3)

any combination of personal services and the selling or furnishing of any material, supplies or equipment. San Francisco Campaign and Governmental Conduct Code (the "Conduct Code") Section 3.700 et. seq., and San Francisco Ethics Commission Regulations 3.710(a)-1 - 3.730-1, prohibit the public officials who approved this contract from receiving: (1) gifts, honoraria, emoluments or pecuniary benefits of a value in excess of \$50; (2) any employment for compensation; or (3) any campaign contributions for any elective office for a period of up to six years from individuals and entities who are "public benefit recipients" of the contract. Public benefit recipients of the contract are: (1) the individual, corporation, firm, partnership, association, or other person or entity that is a party to the contract; (2) an individual or entity that has a direct 10% equity, or direct 10% participation, or direct 10% revenue interest in that party at the time the public benefit is awarded; or (3) an individual who is a trustee, director, partner or officer of the contracting party at the time the public benefit is awarded.

Contractor understands that any public official who approved this contract may not accept campaign contributions, gifts, or future employment from Contractor except as provided under the Conduct Code. Contractor agrees to notify any other individuals or entities that may be deemed "public benefit recipients" under the Conduct Code because of this contract. Upon request, Contractor agrees to furnish, before this contract is entered into, such information as any public official approving this contract may require in order to ensure such official's compliance with the Conduct Code. Upon request, the City agrees to provide, before this contract is entered into, Contractor with a list of public officials who, under the Conduct Code, approved this contract. Failure of any public official who approved this contract to abide by the Conduct Code shall not constitute a breach by either the City or Contractor of this contract. Notwithstanding anything to the contrary in this contract, neither party shall have the right to terminate the contract due to any failure by the other party to provide the information described in this paragraph.

40. Requiring Minimum Compensation for Covered Employees

Contractor agrees to comply fully with and be bound by all of the provisions of the Minimum Compensation Ordinance (MCO), as set forth in San Francisco Administrative Code Chapter 12P (Chapter 12P), including the remedies provided, and implementing guidelines and rules. The provisions of Chapter 12P are incorporated herein by reference and made a part of this Agreement as though fully set forth. The text of the MCO is available on the web at http://www.sfgov.org/oca/lwlh.htm. Capitalized terms used in this Section and not defined in this Agreement shall have the meanings assigned to such terms in Chapter 12P. Consistent with the requirements of the MCO, Contractor agrees to all of the following:

40.1 For each hour worked by a Covered Employee during a Pay Period on work funded under the City contract during the Term of this Agreement, Contractor shall provide to the Covered Employee no less than the Minimum Compensation, which includes a minimum hourly wage and compensated and uncompensated time off consistent with the requirements of the MCO. For the hourly gross compensation portion of the MCO, Contractor shall pay a minimum of \$10.77 an hour beginning January 1, 2005 and for the remainder of the term of this Agreement; provided, however, that Contractors that are Nonprofit Corporations or public entities shall pay a minimum of \$9 an hour for the term of this Agreement.

40.2 Contractor shall not discharge, reduce in compensation, or otherwise discriminate against any Employee for complaining to the City with regard to Contractor's compliance or anticipated compliance with the requirements of the MCO, for opposing any practice proscribed by the MCO, for participating in proceedings related to the MCO, or for seeking to assert or enforce any rights under the MCO by any lawful means.

40.3 Contractor understands and agrees that the failure to comply with the requirements of the MCO shall constitute a material breach by Contractor of the terms of this Agreement. The City, acting through DPT, shall determine whether such a breach has occurred.

40.4 If, within thirty (30) days after receiving written notice of a breach of this Agreement for violating the MCO, Contractor fails to cure such breach or, if such breach cannot reasonably be cured

TOWING SERVICES MASTER AGREEMENT

Page 25 of 33

within such period of thirty (30) days, Contractor fails to commence efforts to cure within such period, or thereafter fails diligently to pursue such cure to completion, the City, acting through the Contracting Department, shall have the right to pursue the following rights or remedies and any rights or remedies available under applicable law:

40.4.1 The right to charge Contractor an amount equal to the difference between the Minimum Compensation and any compensation actually provided to a Covered Employee, together with interest on such amount from the date payment was due at the maximum rate then permitted by law;

40.4.2 The right to set off all or any portion of the amount described in Subsection 40.4.1 of this Section against amounts due to Contractor under this Agreement;

40.4.3 The right to terminate this Agreement in whole or in part;

40.4.4 In the event of a breach by Contractor of the covenant referred to in Subsection 40.2 of this Section, the right to seek reinstatement of the Employee or to obtain other appropriate equitable relief; and

40.4.5 The right to bar Contractor from entering into future contracts with the City for three (3) years. Each of the rights provided in Section 40.4 shall be exercisable individually or in combination with any other rights or remedies available to the City. Any amounts realized by the City pursuant to this subsection shall be paid to the Covered Employee who failed to receive the required Minimum Compensation.

40.4.6 Contractor represents and warrants that it is not an entity that was set up, or is being used, for the purpose of evading the intent of the MCO.

40.4.7 Contractor shall keep itself informed of the current requirements of the MCO, including increases to the hourly gross compensation due Covered Employees under the MCO, and shall provide prompt written notice to all Covered Employees of any increases in compensation, as well as any written communications received by the Contractor from the City, which communications are marked to indicate that they are to be distributed to Covered Employees.

40.4.8 Contractor shall provide reports to the City in accordance with any reporting standards promulgated by the City under the MCO, including reports on subcontractors.

40.4.9 The Contractor shall provide the City with access to pertinent records after receiving a written request from the City to do so and being provided at least five (5) business days to respond.

40.4.10 The City may conduct random audits of Contractor. Random audits shall be (i) noticed in advance in writing; (ii) limited to ascertaining whether Covered Employees are paid at least the minimum compensation required by the MCO; (iii) accomplished through an examination of pertinent records at a mutually agreed upon time and location within ten (10) days of the written notice; and (iv) limited to one audit of Contractor every two (2) years for the duration of this Agreement. Nothing in this Agreement is intended to preclude the City from investigating any report of an alleged violation of the MCO.

40.4.11 Any subcontract entered into by Contractor shall require the subcontractor to comply with the requirements of the MCO and shall contain contractual obligations substantially the same as those set forth in this Section. A subcontract for the purpose of this Section 40 means an agreement between the Contractor and a third party which requires the third party to perform all or a portion of the services covered by this Agreement. Contractor shall notify the Department of Administrative Services when it enters into such a subcontract and shall certify to the Department of Administrative Services that it has notified the subcontractor of the obligations under the MCO and has imposed the requirements of the MCO on the subcontractor through the provisions of the subcontract. It

is Contractor's obligation to ensure that any subcontractors of any tier under this Agreement comply with the requirements of the MCO. If any subcontractor under this Agreement fails to comply, City may pursue any of the remedies set forth in this Section against Contractor.

40.4.12 Each Covered Employee is a third-party beneficiary with respect to the requirements of subsections 40.1 and 40.2 of this Section, and may pursue the following remedies in the event of a breach by Contractor of subsections 40.1 and 40.2, but only after the Covered Employee has provided the notice, participated in the administrative review hearing, and waited the twenty-one (21) day period required by the MCO. Contractor understands and agrees that if the Covered Employee prevails in such action, the Covered Employee may be awarded: (1) an amount equal to the difference between the Minimum Compensation and any compensation actually provided to the Covered Employee, together with interest on such amount from the date payment was due at the maximum rate then permitted by law; (2) in the event of a breach by Contractor of subsections 40.1 or 40.2, the right to seek reinstatement or to obtain other appropriate equitable relief; and (3) in the event that the Covered Employee is the prevailing party in any legal action or proceeding against Contractor arising from this Agreement, the right to obtain all costs and expenses, including reasonable attorney's fees and disbursements, incurred by the Covered Employee. Contractor also understands that the MCO provides that if Contractor prevails in any such action, Contractor may be awarded costs and expenses, including reasonable attorney's fees and disbursements, from the Covered Employee if the court determines that the Covered Employee's action was frivolous, vexatious or otherwise an act of bad faith.

41. Requiring Health Benefits for Covered Employees

Unless exempt, Contractor agrees to comply fully with and be bound by all of the provisions of the Health Care Accountability Ordinance (HCAO), as set forth in San Francisco Administrative Code Chapter 12Q, including the remedies provided, and implementing regulations, as the same may be amended from time to time. The provisions of Chapter 12Q are incorporated herein by reference and made a part of this agreement as though fully set forth. The text of the HCAO is available on the web at http://www.sfgov.org/oca/lwlh.htm. Capitalized terms used in this Section and not defined in this agreement shall have the meanings assigned to such terms in Chapter 12Q.

41.1 For each Covered Employee, Contractor shall provide the appropriate health benefit set forth in Section 12Q.3 of the HCAO. If Contractor chooses to offer the health plan option, such health plan shall meet the minimum standards set forth by the San Francisco Health Commission.

41.2 Notwithstanding the above, if the Contractor is a small business as defined in Section 12Q.3(d) of the HCAO, it shall have no obligation to comply with this Section 41.1 above.

41.3 Contractor's failure to comply with the HCAO shall constitute a material breach of this Agreement. City shall notify Contractor if such a breach has occurred. If, within thirty (30) days after receiving City's written notice of a breach of this Agreement for violating the HCAO, Contractor fails to cure such breach or, if such breach cannot reasonably be cured within such period of thirty (30) days, Contractor fails to commence efforts to cure within such period, or thereafter fails diligently to pursue such cure to completion, City shall have the right to pursue the remedies set forth in 12Q.5(f)(1)-(5). Each of these remedies shall be exercisable individually or in combination with any other rights or remedies available to City.

41.4 Contractor shall not discharge, reduce in compensation, or otherwise discriminate against any Employee for notifying City with regard to Contractor's compliance or anticipated compliance with the requirements of the HCAO, for opposing any practice proscribed by the HCAO, for participating in proceedings related to the HCAO, or for seeking to assert or enforce any rights under the HCAO by any lawful means.

41.5 Contractor represents and warrants that it is not an entity that was set up, or is being used, for the purpose of evading the intent of the HCAO.

41.6 Contractor shall keep itself informed of the current requirements of the HCAO.

41.7 Contractor shall provide reports to the City in accordance with any reporting standards promulgated by the City under the HCAO, including reports on Subcontractors and Subtenants, as applicable.

41.8 Contractor shall provide City with access to records pertaining to compliance with HCAO after receiving a written request from City to do so and being provided at least five business days to respond.

41.9 City may conduct random audits of Contractor to ascertain its compliance with HCAO. Contractor agrees to cooperate with City when it conducts such audits.

42. First Source Hiring Program

42.1 Incorporation of Administrative Code Provisions by Reference. The provisions of Chapter 83 of the San Francisco Administrative Code are incorporated in this Section by reference and made a part of this Agreement as though fully set forth herein. Contractor shall comply fully with, and be bound by, all of the provisions that apply to this Agreement under such Chapter, including but not limited to the remedies provided therein. Capitalized terms used in this Section and not defined in this. Agreement shall have the meanings assigned to such terms in Chapter 83.

42.2 First Source Hiring Agreement

42.2.1 Contractor will comply with First Source interviewing, recruitment and hiring requirements, which will provide the San Francisco Workforce Development System with the exclusive opportunity to initially provide Qualified Economically Disadvantaged Individuals for consideration for employment for Entry Level Positions. The duration of the First Source interviewing requirement shall be ten (10) days, unless business necessity requires a shorter period of time.

42.2.2 Contractor will comply with requirements for providing timely, appropriate notification of available Entry Level Positions to the San Francisco Workforce Development System so that the System may train and refer an adequate pool of Qualified Economically Disadvantaged Individuals to participating Employers.

42.2.3 Contractor agrees to use good faith efforts to comply with the First Source hiring requirements. A Contractor may establish its good faith efforts by filling: 1) its first available Entry Level Position with a job applicant referred through the First Source Program; and, 2) fifty percent (50%) of its subsequent available Entry Level Positions with job applicants referred through the San Francisco Workforce Development System. Failure to meet this target, while not imputing bad faith, may result in a review of the Contractor's employment records.

42.3 Hiring Decisions. Contractor shall make the final determination of whether an Economically Disadvantaged Individual referred by the System is "qualified" for the position.

42.4 Exceptions. Upon application by Employer, the First Source Hiring Administration may grant an exception to any or all of the requirements of Chapter 83 in any situation where it concludes that compliance with this Chapter would cause economic hardship.

42.5 Liquidated Damages. Violation of the requirements of Chapter 83 is subject to an assessment of liquidated damages in the amount of \$2,070 for every new hire for an Entry Level Position improperly withheld from the first source hiring process. The assessment of liquidated damages and the evaluation of any defenses or mitigating factors shall be made by the FSHA.

TOWING SERVICES MASTER AGREEMENT

Page 28 of 33

43. Prohibition on Political Activity with City Funds

In accordance with San Francisco Administrative Code Chapter 12.G, Contractor may not participate in, support, or attempt to influence any political campaign for a candidate or for a ballot measure (collectively, "Political Activity") in the performance of the services provided under this Agreement. Contractor agrees to comply with San Francisco Administrative Code Chapter 12.G and any implementing rules and regulations promulgated by the City's Controller. The terms and provisions of Chapter 12.G are incorporated herein by this reference. In the event Contractor violates the provisions of this section, the City may, in addition to any other rights or remedies available hereunder, (i) terminate this Agreement, and (ii) prohibit Contractor from bidding on or receiving any new City contract for a period of two (2) years.

44. Preservative-treated Wood Containing Arsenic

As of July 1, 2003, Contractor may not purchase preservative-treated wood products containing arsenic in the performance of this Agreement unless an exemption from the requirements of Chapter 13 of the San Francisco Environment Code is obtained from the Department of the Environment under Section 1304 of the Code. The Term "preservative-treated wood containing arsenic" shall mean wood treated with a preservative that contains arsenic, elemental arsenic, or an arsenic copper combination, including, but not limited to, chromated copper arsenate preservative, ammoniacal copper zinc arsenate preservative, contractor may purchase preservative-treated wood products on the list of environmentally preferable alternatives prepared and adopted by the Department of the Environment. This provision does not preclude Contractor from purchasing preservative-treated wood containing arsenic for saltwater immersion. The Term "saltwater immersion" shall mean a pressure-treated wood that is used for construction purposes or facilities that are partially or totally immersed in saltwater.

45. Modification of Agreement

This Agreement may not be modified, nor may compliance with any of its terms be waived, except by written instrument executed and approved by Contractor and by the Director of Transportation, except where the approval of the Municipal Transportation Agency Board or the San Francisco Board of Supervisors is required by applicable law, in which case, the approval of those agencies shall also be required.

46. Administrative Remedy for Agreement Interpretation

Should any question arise as to the meaning and intent of this Agreement, the question shall, prior to any other action or resort to any other legal remedy, be referred to the Director of Transportation, who shall decide the true meaning and intent of the Agreement.

47. Agreement Made in California; Venue

The formation, interpretation and performance of this Agreement shall be governed by the laws of the State of California. Venue for all litigation relative to the formation, interpretation and performance of this Agreement shall be in San Francisco.

48. Construction

All paragraph, article, and section captions and headings contained in this Agreement are for reference and convenience only and shall not be considered in construing the scope or meaning of this Agreement. Unless this Agreement specifically provides otherwise, it is to be construed in the following manner.

48.1 Syntax. Whenever the context of this Agreement requires, the singular shall include the plural, the plural shall include the singular, the masculine shall include the feminine and the feminine shall include the masculine.

48.2 References. Unless otherwise indicated, references to Articles, Sections and subsections are to Articles, Sections and subsections in this Agreement.

49. Entire Agreement

This contract sets forth the entire Agreement between the parties, and supersedes all other oral or written provisions. This contract may be modified only as provided in Section 45.

This Agreement contains the entire agreement between the parties with respect to the subject matter of this Agreement and any prior agreements, discussions or understandings, written or oral, are superseded by this Agreement and shall be of no force or effect. No addition or modification of any term or provision of this Agreement shall be effective unless set forth in writing and signed by the parties to this Agreement.

50. Compliance with Laws

Contractor shall keep itself fully informed of the City's Charter, codes, ordinances and regulations and of all state, and federal laws in any manner affecting the performance of this Agreement, and must at all times comply with such local codes, ordinances, and regulations and all applicable laws as they may be amended from time to time. Contractor shall comply and conform with all laws and all governmental regulations, rules and orders that may from time to time be put into effect relating to, controlling or limiting Contractor's performance under this Agreement or enjoyment of any rights or privileges granted hereby. Contractor shall secure all permits and licenses specifically required for Contractor's performance under this Agreement (copies of which shall be promptly provided to City), and shall comply with all applicable laws and regulations relating to labor employed in and relating to Contractor's performance under this Agreement. Contractor shall not use or occupy the Designated Facilities in an unlawful, noisy, improper or offensive manner and shall use its best efforts to prevent any occupancy of the Designated Facilities or use made thereof which is unlawful, noisy, improper or offensive or contrary to any law or ordinance applicable to them. Contractor shall not cause or maintain any nuisance in or about the Designated Facilities, and shall use its best efforts to prevent any person from doing so. Nor shall Contractor cause any rubbish, dirt or refuse to be placed in the streets, sidewalks or alleys adjoining the Designated Facilities or to accumulate in the Designated Facilities.

51. Services Provided by Attorneys

Any services required under this Agreement that are to be provided by a law firm or attorney must be reviewed and approved in writing in advance by the City Attorney. No invoices for services provided by law firms or attorneys, including, without limitation, as subcontractors of Contractor, will be paid unless the provider received advance written approval from the City Attorney.

52. Unavoidable Delays

Any prevention, delay or stoppage in a party's performance of any part of this Agreement due to an Unavoidable Delay shall excuse the performance of the party affected for a period of time or otherwise in a manner that bears a causal relationship and is in proportion to any such prevention, delay or stoppage.

53. Severability

Should the application of any provision of this Agreement to any particular facts or circumstances be found by a court of competent jurisdiction to be invalid or unenforceable, then (a) the validity of other provisions of this Agreement shall not be affected or impaired thereby, and (b) such provision shall be

enforced to the maximum extent possible so as to effect the intent of the parties and shall be reformed without further action by the parties to the extent necessary to make such provision valid and enforceable.

54. Transition Period

54.1 Notwithstanding any other provision of this Agreement to the contrary, Contractor hereby agrees that it shall cooperate in good faith with City and with any contractor chosen by City pursuant to a competitive selection process to take over Contractor's obligations after the expiration of this Agreement, or any contractor designated by City to provide towing services following termination of this Agreement pursuant to §19 herein, for the smooth and efficient transfer of those functions.

54.2 Contractor agrees to take all actions as may be necessary or as the City may direct for the protection and preservation of any property related to this Agreement that is in the possession of Contractor and in which City has or may acquire an interest, including any environmental remediation required under Appendices C, D and E. Contractor acknowledges that City has a vested interest for payment of fees for vehicles that Contractor has towed and/or is storing, and Contractor shall not impair said interest and shall take all actions reasonably necessary to safeguard City's interest in said towed and stored vehicles.

54.3 Contractor shall remove any towed and/or stored vehicles from City property by the termination date and shall relocate said vehicles to a location approved by the City or (subject to City approval) authorize a successor contractor to serve as an agent of Contractor for purposes of conducting any lien sale auctions that may be required for disposal of such vehicles.

54.4 Contractor shall provide City or any successor contractor with all information necessary, in a form approved by the City, to facilitate retrieval by the registered owner of any vehicle upon which Contractor holds a lien.

54.5 If Contractor has implemented the Consolidation (as defined in Section 4) of the Customer Service Center and Primary Storage Facility as of the termination of this Agreement, Contractor agrees to assign all right, title and interest of Contractor in the 450 7th Street property to the City, subject to the payment terms specified in Appendix I, as part of the transition of towing operations to a successor towing services contractor.

55. Administration of Contract

This Agreement shall be administered by and for the City by DPT. The Director of DPT shall, unless otherwise stated herein, have the authority to act for or on behalf of the City in the administration of this Agreement.

56. Required Actions

Contractor agrees to execute all instruments and documents and to take all actions as may be required or necessary in order to carry out the purposes and terms of this Agreement.

57. Non-impairment of City-Owned Towing Equipment and Facilities

Nothing contained in this Agreement shall be deemed to prohibit, limit or otherwise restrict the use and operation by City of City-owned towing equipment or storage facilities, or the procurement by City of towing services for City-owned vehicles.

TOWING SERVICES MASTER AGREEMENT

Page 31 of 33

IN WITNESS WHEREOF, THE PARTIES HERETO HAVE EXECUTED THIS AGREEMENT ON THE DAY FIRST MENTIONED ABOVE.

<u>CITY</u>

Recommended by:

BOND M. YEE

Acting Director Department of Parking and Traffic

APPROVED

Acting Director of Transportation Municipal Transportation Agency

Approved as to Form:

DENNIS J. HERRERA City Attorney

By Christiane Hayashi

Deputy City Attorney

MTAB Resolution Number: 05-085 Date: JUNE 7, 2005 Attest: <u>Attest:</u> Roberta Boomer Secretary, Municipal Transportation Agency Board

CONTRACTOR

By signing this Agreement, I certify that I comply with the requirements of the Minimum Compensation Ordinance, which entitles Covered Employees to certain minimum hourly wages and compensated and uncompensated time off.

I have read and understood paragraph 33, the City's statement urging companies doing business in Northern Ireland to move towards resolving employment inequities, encouraging compliance with the MacBride Principles, and urging San Francisco companies to do business with corporations that abide by the MacBride Principles.

JOHN WICKER President and CEO TEGSCO, LLC, d.b.a. San Francisco AutoReturn 450 7th Street San Francisco, CA 94103 Phone No.: 415-626-3380 Employer ID No.: 01-0688299

LIST OF APPENDICES

A. Scope of Work

B. Services and Standards (Operations Plan)

§ 2.6:

4.6(a):

5.1(e): 5.2(d):

8.2(a):

8.2(b):

8.2(c):

§ 8.3: § 8.4:

8.5:

§ 11.3: § 11.4:

Appendix A/Scope of Work Elements:

Tow Service Plan Police Department Procedures Customer Service Manual Claims/Complaint Processing Plan List of Approved Subcontractors Subcontractor Compensation Plan Subcontractor Performance Standards Employee Training Plan Policy and Procedure Manual Staffing Plan Security Plan Personal Property Plan

Appendix D/Pier 70 Premises Management Elements (see Appendix C, §14(k)).

1.	Pollution Prevention Plan
2.	Utilities Maintenance Plan
3.	Pier 70 Premises Maintenance Plan
4.	Facility plan
· 5 .	Public Auction Plan

C. 2005 DPT-Port MOU

- **D.** Pier 70 Property License
- E. 415 7th Street and 850 Bryant Street Property License
- F. **PEO Contracts**
- G. Letter of Credit
- H. Rate Schedule
- I. Consolidated Facility Payment Schedule

t Form

Introduction Form

By a Member of the Board of Supervisors or the Mayor

I hereby submit the following item for introduction (select only one):	Time stamp or meeting date		
1. For reference to Committee. (An Ordinance, Resolution, Motion, or Cha	arter Amendment)		
2. Request for next printed agenda Without Reference to Committee.			
3. Request for hearing on a subject matter at Committee.	·		
4. Request for letter beginning "Supervisor	inquires"		
5. City Attorney request.			
6. Call File No. from Committee.			
7. Budget Analyst request (attach written motion).	· · · ·		
8. Substitute Legislation File No.			
9. Reactivate File No.			
10. Question(s) submitted for Mayoral Appearance before the BOS on			
Planning Commission Description Building Inspect Note: For the Imperative Agenda (a resolution not on the printed agenda), use Sponsor(s):			
Supervisor Wiener			
Subject:	· · ·		
Extending the Agreement for Towing and Storage Services	· · · · · · · · · · · · · · · · · · ·		
The text is listed below or attached:			
Resolution amending the towing agreement with TEGSCO, dba San Francisco Aumonths, from August 1, 2015 to March 31, 2016.	toReturn, for a term of eight		
Signature of Sponsoring Supervisor:	Wiener		
For Clerk's Use Only:	, , , , , , , , , , , , , , , , , , , ,		

150689

File No. 150689

FORM SFEC-126: NOTIFICATION OF CONTRACT APPROVAL

(S.F. Campaign and Governmental Conduct Code § 1.126)		
City Elective Officer Information (Please print clearly.)		
Name of City elective officer(s):CitMembers, Board of SupervisorsCit	ty elective office(s) held: Members, Board of Supervisors	
Contractor Information (Please print clearly.)		
Name of contractor: AutoReturn		
John Wicker (CEO and Board Member); Kacy Rozelle (Board Member), (Board Member), George Hoyem (Board Member),Raymond Krouse (CF		
Contractor address: 375 Alabama Street, Suite 300, San Francisco, CA 94	4110	
Date that contract was approved: July 19, 2005A(By the SF Board of Supervisors)A	mount of contracts: \$ See comments below	
Describe the nature of the contract that was approved: Contractor provides towing, storage, lien sale and auctions services for illegally parked and abandoned vehicles on behalf of the SFMTA and the SFPD.		
Comments: This is a revenue contract; SFMTA does not pay the contractor for servic towing services; and in addition, has paid the SFMTA for administrative gross revenue for all monies collected. As of FY2013, this amount totals	(PCOs, etc.) and referral fees, and an annual 1% of	
This contract was approved by (check applicable):		
Ithe City elective officer(s) identified on this form		
a board on which the City elective officer(s) serves: <u>San Francis</u> Print Na	nco Board of Supervisors nume of Board	
the board of a state agency (Health Authority, Housing Authority Board, Parking Authority, Redevelopment Agency Commission, Re Development Authority) on which an appointee of the City elective	elocation Appeals Board, Treasure Island	
Print Name of Board		
Filer Information (Please print clearly.)	· · · · · · · · · · · · · · · · · · ·	
Name of filer: Angela Calvillo, Clerk of the Board	Contact telephone number: (415)554-5184	
Address: City Hall, Room 244, 1 Dr. Carlton B. Goodlett Pl., San Francisco, CA 94	E-mail: 4102 Board.of.Supervisors@sfgov.org	

Signature of City Elective Officer (if submitted by City elective officer)

Date Signed

Signature of Board Secretary or Clerk (if submitted by Board Secretary or Clerk)

Date Signed